# An analysis of hospital admissions for ambulatory care sensitive conditions

Jennifer Dixon

London School of Hygiene & Tropical Medicine

University of London

2001

PhD thesis



# Abstract

#### Aims

To define and identify 'ambulatory care sensitive' (ACS) and 'ambulatory care insensitive' (ACI) conditions.

To assess the quality of the NHS hospital episode statistics (HES) dataset.

To assess the reasons for variation in hospitalisation for ACS conditions across small areas. To discuss whether variations in admissions for ACS conditions could be used to indicate equity of access to primary care.

# **Main subjects**

Hospital admissions for residents of North West Thames NHS region 1991/2, 1992/3, 1993/4.

### Methods

A modified nominal group technique was used to identify ACS and ACI conditions. The quality of HES was analysed by measuring shortfalls in key fields and the reproducability of clinical coding. Factors influencing variations in hospitalisation were assessed using multivariate analysis. The results were compared to similar studies from the US and Canada.

# Results

30 ACS and 66 'weakly ACS' conditions were identified. Only 5 out of the 16 main acute providers serving the North West Thames region had adequate HES data across all three study years. In two providers, the *exact* clinical codes for main diagnosis could be reproduced in 43% and 60% respectively, and the *first 3-digits* of the ICD-9 codes in 55% and 72%. A significant positive relationship was found between admissions and socioecononomic deprivation independently of indicators for mortality and morbidity. This was weaker than that between admissions and area income found in the US and Canada. The relationship was found with an indicator of access to hospital care, but not access to primary care. The overall explanatory power of the model was weak at 10-20%.

#### Conclusions

Admissions for ACS conditions are higher in deprived areas, suggesting that access to ambulatory care may be lower. Higher rates of ACS conditions in poor areas in North America suggest lower access to timely and effective ambulatory care than in the UK NHS.

# Contents

Th <mark>anks</mark> and	acknowledgements9
Chapter 1	Introduction12
Chapter 2	Literature review17
2.1	Introduction
2.2	Equity in health care
2.3	Access to health care
2.4	Avoidable 'downstream' events as indicators of access to,
	and quality of, health services
2.5	A brief overview of the literature
2.6	Implications of the literature for the design of this study79
Chapter 3	Identifying admissions for further study
3.1	Introduction
3.2	Excluding broad categories of admissions
3.3	Identifying ACS and marker conditions
3.4	Identifying admission groups for subsequent small area analysis108
3.5	Conclusion111
Chapter 4	Data quality: completeness112
4.1	Introduction113
4.2	Methods
4.3	Results
4.4	Identifying hospitals to be excluded from further analysis128
4.5	Discussion130
4.6	Summary and conclusion

Chapter 5	Investigating the accuracy of diagnostic coding o collected hospital admissions data	f routinely 134
5.1	Introduction	135
5.2	Methods	135
5.3	Results	138
5.4	Discussion	143
5.5	Implications for the small area analysis	146
Chapter 6	Small area analysis	147
6.1	Introduction	148
6.2	Methods	148
6.3	Results	155
6.4	Discussion	189
Chapter 7	Summary of main findings, next steps and direct for the NHS	t implications 198
7.1	Context	198
7.2	Main findings	199
7.3	Next steps for analysis	201
7.4	Direct implications of the findings for the NHS	203
References.		205
Appendices		223

# Tables, figures and appendices

# page

Tables	
Table 1	Elements of some deprivation indices23
Table 2	Ambulatory care sensitive and marker conditions and
	ICD-9 codes, selected by Billings et al
Table 3	Avoidable hospital conditions and ICD-9CM codes,
	selected by Weissman et al
Table 4	Indicator conditions selected by Solberg et al
Table 5	Main conditions thought to be ambulatory care sensitive
Table 6	Resident population by ethnic group, Great Britain, 199163
Table 7	The consensus development process
Table 8	Composition of the panels
Table 9	Ambulatory care sensitive conditions
	(a) avoidable primarily through effective management
	in primary care
	(b) also with scope for prevention in ambulatory care
Table 10	Weakly ambulatory care sensitive conditions
	(a) avoidable primarily through effective management
	in primary care
	(b) also scope for prevention in ambulatory care100
Table 11	Ambulatory care insensitive
	(a) some scope for prevention or management in primary care101
	(b) little scope for prevention or management in primary care102
Table 12	Numbers of conditions in different groups104
Table 13	ACS admission groups for further analysis109
Table 14	Marker admission groups for further analysis110
Table 15	Main fields of information collected on HES data116
Table 16	The number of episodes recorded on HES data compared
	with the number on the KP70 return, by provider group,
	by year121
Table 17	The number of episodes recorded on HES data compared
	with the number on the KP70 return in Luton and Dunstable
	Hospitals, by specialty, 1991/2123

Table 18	The percentage of ACS episodes within specialties in which
	less than a 10% discrepancy was observed between HES and
	the part 1 KP70 return, 1991-1994125
Table 19	The percentage of missing or invalid data in each episode
	in selected data fields, 1991-1994127
Table 20	Hospitals meeting the relevant criteria for data quality,
	1991-1994
Table 21	Percentage agreement by external coders over the main
	diagnostic and main procedure codes, as identified by
	local coders
Table 22	The percentage of cases in each category as a percentage of all
	disagreements and as a percentage of all cases reviewed141
Table 23	Percentage agreement between external and local coders
	for asthma and diabetes142
Table 24	Percentage agreement on the main diagnosis between
	the regional coder, the local and external coders142
Table 25	Admission groups selected for further analysis, number of
	admissions in each group and percentage of all admissions,
	1991/2
Table 26(i)	The number of EDs in each hospital service area, 1991/2158
Table 26(ii)	The number of EDs in each hospital service area, 1992/3159
Table 26(iii)	The number of EDs in each hospital service area, 1993/4160
Table 27	Number of EDs in Group A, Group B and Group C EDs,
	by hospital service area, all years162
Table 28	Age-sex standardised admission rate by admission group,
	1993/4
Table 29	Comparisons of age-sex standardised admission rates for
	admission groups 1 and 16, across three study years165
Table 30	Age-sex standardised admission rates by hospital service area,
	1993/4 (ranked in order of the mean)167
Table 31	Age-sex standardised admission rates (per 1000) for admission
	group 1, by year
Table 32	Descriptive statistics for Carstairs, Standardised Mortality
	Ratio, and Standardised Illness Ratio, by hospital service area,
	by year (ranked by Carstairs score)169
Table 33	Descriptive statistics for GP (GAF2) and Hospital (HAF2) access
	factors by hospital service area, 1993/4170

Table 34	Correlations between the main independent variables172
Table 35	Correlations between Carstairs score and admission rates by
	admission group, 1993/4173
Table 36	Incremental relative risk ratios for selected variables:
	Carstairs quintile, SMR quintile and SIR quintile, 1993/4178
Table 37	Incremental relative risk ratios for 'age group' by admission
	group, data for all years combined179
Table 38	Incremental relative risk ratios, by sex, 1993/4
Table 39	Incremental relative risk ratios for GP access factor (GAF2),
	by admission group, 1993/4183
Table 40	Incremental relative risk ratios for the GP access factor (GAF2)
	for selected admission groups, 1993/4
Table 41	Incremental relative risk ratios for the hospital access
	factor (HAF2) by admission group, 1993/4186
Table 42	Incremental relative risk ratios for selected hospital service
	areas, and selected admission groups 1993/4188

# Figures

Figure 1	Dimensions of access to medical care and their indicators	29
Figure 2	The main factors that have been investigated as having influence	
	in relation to access to, and use of, care	43
Figure 3	Pathway of analysis	45
Figure 4	A simple flow diagram showing the main steps involved in	
	clinical coding	144
Figure 5	Distribution of enumeration districts by population size	156
Figure 6	Group 1 (ACS) admission rates by year	166

# Appendices

Appendix A	Hospital service areas 1993/4	223
Appendix B	Frequency distribution of values for LGAF2 and LHAF2	
	by hospital service area, 1993/4	224
Appendix C	Ratio of means of age-sex standardised admission rates	
	by admission group, by year	232
Appendix D	The incremental relative risk ratio for key variables, for	
	admission group 1(ACS), 1991/2	233

Appendix E	The value of 'pseudo r-squared' coefficients using full
	and simplified models234
Appendix F	Incremental relative risk ratios for GP (GAF2) and hospital
	(HAF2) access factors, by admission group, 1991/2, 1992/3,
	1993/4 and all three years combined235
Appendix G	Incremental relative risk ratios (IRRs) for selected hospital service
	areas, and for selected admission groups, 1993/4240

# Thanks and acknowledgements

I would like to thank and acknowledge the significant support received from many individuals during the conception, design, carrying out and writing up of this thesis.

First and foremost I would like to acknowledge the enormous debt I owe to my supervisor, Colin Sanderson, for being so generous with his time and for unfailing perseverance, patience, encouragement, excellent and timely advice and good humour throughout the project. Colin contributed significantly to the design of much of the analysis reported here, in particular those described in chapters 3, 5 and 6.

Across the Atlantic I would like to thank John Billings both for involving me in his early work in New York and his encouragement to conduct a similar study in the UK, as well as generously sharing the methods and results of his and his colleagues' work. In doing so John was the key contributor to the overall conception of the thesis.

Over several years I received exceptional and unflagging help and advice from Peter Walls. Peter helped in the analysis of the quality of the hospital episode statistics (HES) used in the main part of the study, in particular handling raw data obtained from several sources, producing basic descriptive details of the dataset and making it available in a form that could be analysed by me. Peter was also responsible, with Colin, for constructing and calculating the access variables used in the multivariate analysis described in chapter 6.

I owe a huge debt of thanks to Sheila Adam, former regional director of public health, and Sally Hargreaves, former faculty advisor in public health for North West Thames, for their encouragement to carry out a thesis, commitment to the need for academic training for public health trainees, and financial support within the regional training scheme.

Considerable thanks are also due to two key advisors. Paul Elliott took over supervision of the thesis for one year whilst Colin Sanderson was on sabbatical, and inspired in particular the extensive analysis of the adequacy of the hospital episode statistics dataset described in chapters 4 and 5. Chris Garrett, formerly of North West Thames region, now of the London NHS region, has been a source of very helpful support and advice over the years particularly over the sources of, and inconsistencies in, the main HES dataset used, the quality of data in different acute hospitals in the region, and helping to explain the complex organisational changes in the environment in the NHS over the period of study.

A few others in the former North West Thames region also offered regular support in obtaining HES and other data from the right sources, and giving advice on the reasons for apparent inconsistencies in the data. In particular I would like to thank Geoff Howard and Philip Wilcox.

Thanks are due to a number of staff at the London School of Hygiene who were involved at some point to further the project, in particular: Duncan Hunter, who helped to support the work described in chapter 3 by writing notes during the panel meetings; Mark Petticrew, who designed a computer programme to enable the external clinical coders (described in chapter 5) to re-code the sample of notes electronically; Simon Stevenson for help manipulating the raw HES dataset; Nick Black and Martin McKee for helpful comments on the manuscripts of papers that were published on the work described in chapters 3 and 5; Ben Armstrong for help with the Poisson analysis and test for dispersion; and Debbie and Jenny Stanley for unfailing and helpful administrative support.

In addition I would like to thank the GPs and consultants who, as members of the panels described in chapter 3, generously gave their time and advice towards helping to identify ambulatory care sensitive and ambulatory care insensitive conditions. In the study described in chapter 5, Meg Evans at the North West Thames region gave much support and help both in refining the design of the study and being the 'expert regional coder' described. In the same study the four external coders (from CHKS) carried out the tasks I set for them very efficiently and cheerfully, completing the study in good time, for which I was very grateful. The two lead clinical coders at both study hospitals – Luton & Dunstable and Lister – offered me their full support and co-operation, as did the information officers, the medical records clerks and the chief executives of the hospitals. Mike O'Connor and Sheila Adam at North West Thames both helped to secure funding for this study.

I am very grateful to various people at the King's Fund. In particular I would like to thank: Julia Neuberger for her support and for allowing some working time to be used to work on the thesis, as did Ken Judge and Nick Mays in my early days at the Fund; John Howard, Billy Butlin and Patrick Drury in the IT department, for the loan of a Fund computer and other equipment at home, and for efficient and timely IT support to allow me to complete the thesis; and Mike Damiani for help to construct the map shown in appendix A.

There are others who have also helped along the way whose support I was extremely grateful for, who I hope will allow me to mention their contribution anonymously.

Finally perhaps the very largest debt of gratitude is owed to Stephen Burke and John Vorhaus for offering patient and selfless support to allow me to work on the thesis at home, often at very unsociable times. The work in this thesis would not have been possible without their help.

#### Chapter 1 Introduction

This study is essentially an inquiry into the equity of an aspect of access to health care in the UK National Health Service (NHS).

John Billings and colleagues in the US studied hospitalisation rates of people living in lowincome compared to high-income areas in New York City (1). The study focused on admission rates for three groups of conditions: so-called 'ambulatory care sensitive' (ACS) conditions; 'markers'; and 'discretionary surgeries'.

ACS conditions included those for which an admission to hospital was thought to be preventable by 'timely and effective' ambulatory care (such as diabetes and asthma). 'Ambulatory care' included pre-hospital care which was not provided in an inpatient setting, such as primary, outpatient, community health and accident and emergency care.

The study found that residents aged under 65 years living in low-income areas were hospitalised for ACS conditions over five times more than the rate of those in the same age group living in high- income areas, and almost 70% of the variation was associated with the average income of the residents of the area. Because of a lack of area-based data on prevalence of disease (one indicator of 'need'), Billings' *et al* were unable to take this factor fully into account in the analysis. However survey data showed that the prevalence of two of the commonest ACS conditions (asthma and diabetes) was approximately three times higher in low- relative to high-income groups - much lower than the variation in ACS rates observed.

In contrast there was no significant difference in admission rates between areas for the second group of conditions analysed - 'marker' conditions (those for which admission was thought to be relatively ambulatory care-*insensitive* and for which admission was thought to be urgent and 'mandatory' (such as fractured femur and appendicitis)). Admission rates for a third group of conditions - high cost 'discretionary' surgical procedures (such as joint replacements, coronary artery bypass surgery) - were found to be approximately half in residents of low-income areas compared to the high income areas, with area income explaining just over 20% of the variation.

Billings *et al* suggested that residents of low-income areas were less likely to have adequate health insurance cover. As a result, they would have less access to 'timely and effective' ambulatory care, and were likely to be admitted to hospital more often for conditions that could be managed in an ambulatory setting. Hospitalisation rates for ACS conditions were therefore used in the study as an outcome measure for equity of access to 'timely and effective' ambulatory care. Inadequacies in health insurance cover for residents of lowincome areas could also lead to poor access to high cost 'discretionary' surgeries, and hence the lower admission rates observed for this population group. In contrast, Billings argued, access barriers to mandatory and emergency care, required for patients with marker conditions, were less evident in this population.

In those aged under 65 years examined by Billings *et al*, variation in admission rates for ACS conditions between low and high income areas was greatest for those aged 25-64 (those least likely to have adequate health insurance cover) and lowest for the under fives (a higher proportion of which had access to the state funded health insurance programme, Medicaid, compared to other age groups). Billings *et al* repeated the analysis using admissions data for the age group 65 years and older, almost all of whom qualified for Medicare (federally funded health insurance). Again there was an inverse relationship between income and admission rates for ACS conditions, but the variation was three-fold rather than five-fold seen in the under 65 years group. Again admission rates for marker conditions were similar between low- and high-income areas. Billings *et al* suggested that differences in ACS admission rates across age groups reflected the relative access to insurance coverage (and, in turn, ambulatory care).

Billings' findings were perhaps not surprising given the inequities of the US health care system, the barriers to accessing health care for those on low income, and the relative lack of investment in primary care. Similar analyses were subsequently carried out in Canada (2) and Spain (3) - countries with more egalitarian health systems compared to the US. Much less variation across small areas between socioeconomic groups was found. These findings raised the question of whether variations in hospitalisation for ACS conditions could be used as an indicator of equity of access to ambulatory care in a country.

The UK NHS scores highly on equity criteria compared to most other countries (4). Over 98% of the population are registered with a general practitioner (GP) who usually works in

a group practice. Care is largely free at the point of use, with approximately 20% of the population subject to user charges (mainly for prescription drugs), and the NHS provides near comprehensive coverage of health benefits. Nevertheless inequities in access to ambulatory, particularly primary care, remain, particularly in cities, and especially in London. Given this, would the pattern of admission rates for ACS conditions be similar to that observed in the US, or more like that observed in Canada or Spain? If there were variations in admission rates for ACS conditions between residents of socio-economically deprived compared to less deprived areas, how much of this could be due to lack of access to 'timely and effective' ambulatory, particularly primary, care provided by the NHS? If so, could the relationship between the level of access to primary care and admission rates for ACS conditions be quantified in some way?

These questions were the starting point of the work presented in this thesis. The overall aim was to identify patterns of admission rates (by small area) to NHS hospitals for ACS and marker conditions, and to investigate reasons for the variations. Several hypotheses were to be examined:

- that rates of hospitalisation for ACS conditions would vary across areas more than would be expected by chance;
- that hospitalisation rates for marker conditions would vary across areas randomly;
- that higher hospitalisation rates for ACS conditions would be found in more socioeconomically deprived areas even after adjusting for health care 'need';
- that higher hospitalisation rates for ACS conditions would be significantly associated with lower access to primary care facilities.

To investigate these hypotheses, an ecological cross-sectional study was proposed in which hospital admission rates for ACS and marker conditions were to be analysed by small geographical areas. It was intended that the analysis would be confined to hospitalisations of residents living within the former North West Thames region over a period of three years 1991/2, 1992/3 and 1993/4. Analysis of Billings' third group of conditions - discretionary surgeries - was not proposed because a significant proportion of admissions for these conditions in the UK occur in the private sector for which data are not routinely available.

But before the analysis could begin, three preparatory pieces of work were required. First it

was necessary to identify and select groups of admissions which were to be analysed further. In particular there was a need to identify a list of conditions which UK clinicians agreed were 'ambulatory care sensitive' and 'markers': given the likely differences in clinical practice style between the US and UK it was thought inappropriate to rely upon the list of ACS and marker conditions derived by US clinicians. Since admission rates might vary simply due to data error it was necessary to assess the quality of routinely collected admissions data. Thus the second piece of work was an investigation of the *completeness* of routine data on hospital admissions in North West Thames region. The third was an investigation of the *quality of diagnostic coding* of routine admissions data, because admissions for specific conditions were to be analysed.

The thesis is therefore set out as follows:

- chapter 2 contains a review of related published work, and defines the scope of the study;
- chapter 3 describes how groups of admissions were identified and selected for further analysis, in particular how 'avoidable' and 'marker' conditions were identified. In this chapter there has been a substantial contribution to the analysis by Dr Colin Sanderson, and the chapter draws on a paper published in the Journal of Health Services Research and Policy in October 2000 (5). Jennifer Dixon was largely responsible for : the conception and design of the study; conducting the study; carrying out the initial analysis, the initial writing up of the paper; and all of sections 3.4 and 3.5 of the chapter. Colin Sanderson was largely responsible for further analysis and final drafting of the paper, in particular the results and conclusions;
- chapter 4 describes the completeness of data on hospital admissions within North West
   Thames region and the criteria for selecting, for further study, the hospitals with the
   best quality data;
- chapter 5 describes the investigation into the quality of diagnostic coding of hospital admissions data. This chapter draws on a paper published in the Journal of Public Health Medicine in 1998 a paper contributed to by a number of individuals (6). In the chapter Jennifer Dixon was responsible for the conception and design of the study,

obtaining funding, managing the fieldwork (conducted by clinical coders), analysis of the results and writing up. The other authors made such contributions as providing advice on the sample size, advising on the study design, designing a computer programme to enable coders to enter their data directly onto a laptop computer, and commenting on drafts of the paper;

- chapter 6 describes the small area analysis. Colin Sanderson contributed to the overall design of the study, in particular in simplifying the regression model used, helping to decide on the formula to be used to calculate the access factors, and suggesting which of the many results should be presented. Peter Walls helped mainly to calculate the access factors, as well as providing the dataset in a form that could be analysed by me;
- chapter 7 summarises the main findings, suggests next steps for analysis, and outlines the direct implications of the study for the NHS.

Because some of the chapters are long, to help the reader a schematic outline is shown at the beginning of each chapter.

# 2.1 Introduction

The study aims to investigate equity of access to ambulatory care in the UK NHS by area, using hospitalisation rates for ambulatory care sensitive (ACS) conditions as an indicator of access to ambulatory care.

The first aim of the literature review was to clarify the concept of equity in health care to be used in the study. The second was to examine the concept of access. The third was to examine how researchers had used hospital admissions as a negative indicator of access to 'upstream' care - timely and effective ambulatory care. The fourth was to summarise briefly the results of some of the published studies that have investigated variations in access to, or use of, health care, specifically to identify the main factors associated with the variations. The fifth was to consider methodological issues when conducting small area analyses of hospitalisation rates. The sixth was to outline the implications of the literature for the design of the proposed study.

This chapter contains some long sections. To help the reader, at the beginning of each section a box summarises an outline of the contents.

# 2.2 Equity in health care

Section	n outline		
2.2.1	Conc	epts of equity	
2.2.2	Conc	epts of need	
2.2.3	Measuring need		
	(# )	self-reported illness	
	(b)	socioeconomic deprivation	
	(c )	mortality	

## 2.2.1 Concepts of equity

Equity (justice or fairness) is a broad issue which has inspired extensive research and discussion amongst many, including philosophers, economists, and politicians. The more specific concept of equity in health and health care has also been widely discussed. Mooney (7) offers several definitions of equity in health care which include:

- equal resources per capita;
- equal inputs per capita;
- equal health per capita;
- equal access for equal need;
- equal use for equal need;
- equal treatment for equal need;
- equal outcomes (health) for equal need.

It is not the objective here to provide a background to the debate, to discuss the pros, cons and inconsistencies of each of the equity principles outlined above – there is an extensive literature on the subject\*. Disagreement remains over which principle is best to use, although the definition most often used in research to assess equity in health care is 'equal access for equal need'. Here I simply seek to clarify this definition, which will be used in this thesis.

Two dimensions of equity are usually discussed in the literature – 'horizontal' equity (the equal treatment of equals) and 'vertical' equity (the unequal treatment of unequals) (7). Neither dimension is as straightforward a concept as it first appears (8). For example, measuring the extent of horizontal equity in health care presents difficulties because of the problems in assessing which individuals have the same needs. Vertical equity - unequal access for unequal need, for example different levels of use of health care for different levels of illness - has presented the added problem for researchers involving judgements about exactly how much differential access is appropriate for differing levels of 'need'. This judgement is necessarily subjective and often arbitrary. Instead researchers have tended to investigate whether or not higher levels of 'need' are associated with higher levels of access, rather than being concerned with how much higher.

These difficulties are compounded because both 'need' and 'access' are complex concepts and are difficult to measure directly. Need is considered below; access is discussed in the following section.

#### 2.2.2 Concepts of need

There has been much debate as to what 'need' for health care actually is. Bradshaw (9) defined need as including:

- 'felt need' where an individual or group perceives a need;
- 'expressed need' where an individual or group turns 'felt need' into action (ie through a demand);
- 'normative need' 'that which the expert or professional, administrator or social scientist defines as need in any given situation';
- 'comparative need' where an individual or group with specific characteristics is not receiving the same services as others with the same characteristics.

In a comprehensive and useful review of the literature on access to health care, Goddard and Smith suggest that an individual's 'need' for health care can include two related but separate issues - the individual's *level of illness*, and their *capacity to benefit* from treatment (10). The authors go on to pose further questions:

- to what extent should non-clinical contributions to need be considered? (such as an individual's social circumstances or their preferences);
- how is the relevant concept of health status to be measured? (for example can 'selfreported' health status be relied upon?);
- at what stage should need be measured? (for example at the point of demand or at the point of treatment?).

Goddard and Smith note that most empirical studies 'have paid scant attention' to the concept of need and tend to make crude assumptions, for example that levels of need are

<sup>\*</sup> For a useful introduction and discussion see Le Grand 1991 (32), Mooney (25) and Culyer (28).

#### either:

- (a) the same in the population groups studied; or
- (b) assessed on the basis of 'self-reported' health status, thus ignoring the potential variations between population groups in the way that health status is interpreted and reported in surveys. Furthermore a self-reported level of health gives no indication about the individual's capacity to benefit; or
- (c) assessed on the basis of a bio-medical measure, thus ignoring the potential systematic variation in the measurement, and the influence on need of unmeasured factors (such as socioeconomic circumstances); or
- (d) indicated by the characteristics of the population of the area in which the individual (under study) lives rather than their own circumstances; or
- (e) indicated by the results of some other study, which may have been based on a measure reflecting utilisation (leading to the potential for circularity in argument).

In practice, researchers who have attempted to adjust for need have tended to use a definition relating to the ill health of the individual, since information on this is more likely to be available than a measure of an individual's capacity to benefit from treatment. In such studies, depending on the data used in analysis, either assumptions (b) or (d) tend to be made, again mainly for practical reasons (discussed further below).

#### 2.2.3 Measuring need

As noted above, the need for health care in an individual can be related to the level of ill health and/or the capacity to benefit from treatment. In small area analyses, of the kind to be conducted in this thesis, aggregate datasets are used to investigate variations in hospitalisation rates. What data pertaining to need might be available to aid such analyses?

The main dataset available on hospitalisation in the UK NHS is the 'Hospital Episode Statistics' (HES) dataset. When an individual is hospitalised as an inpatient, a number of items of information are collected routinely. However information on the *level of ill health* or severity of illness is not directly recorded. For example, for each admission the main diagnosis (and up to 6 other diagnoses) are recorded, as are the main (usually surgical)

procedures carried out. The recording of co-morbidities may indicate the level of ill health of the patient, but in the UK NHS co-morbidities are often not recorded as secondary diagnoses on HES data (a finding reported in chapter 5).

However, a large number of factors are known to influence the level of health in an individual (11). The HES dataset includes some information on these, such as age, sex, and more recently some limited information on ethnicity, but not on others, such as socioeconomic status or a functional measure of health status. Hospital admission rates can therefore be stratified or adjusted for age and sex but data are not good enough to do this for other potential need factors such as health status, socioeconomic circumstances or ethnicity.

On the other hand, routinely collected data on hospital admission do include the postcode of residence of the person admitted. Hospitalisation rates can therefore be calculated by small area and the characteristics of the area (using data from other sources) used to indicate relative need. The main types of information available at small area level include:

- (a) self-reported health status (a near 100% sample of the population, using data from the 1991 Census);
- (b) socioeconomic deprivation (a near 100% sample of the population, from the Census);
- (c) mortality rates (a near 100% sample, from the Office for National Statistics).

Data of all three kinds are usually available to researchers aggregated on an area-level basis, rather than by individual. Clearly the problems of ecological fallacy arise when ascribing rates of admission for individuals to what can be called *area-aggregated individual-level* data, since the individuals hospitalised may not be typical of residents in their area (12).

Each of the three kinds of data is discussed briefly below.

#### (a) Self-reported illness

Large scale national population surveys such as the General Household Survey and the Health Survey for England (HSE) do contain questions on self-reported health status, and in the HSE these questions are combined with measurements of risk factors such as blood pressure and cholesterol levels. However these surveys include only a sample of the population, and the data are not available by small area.

Data on the prevalence of self-reported illness across a sample of the population large enough to yield adequate data for each 'small area' (enumeration district or electoral ward) in England, can be obtained from only one source at present, the Census. In 1991, for the first time the Census (of all persons in all households in England) contained the following question on self-reported limiting long-term illness : 'does the person have any long-term illness, health problem or handicap which limits his/her daily activities or the work he/she can do?'(13). The advantage of this measure is that it is the only item of data on the health status of the population that is routinely available at small area level. The measure typically used in small area analysis is the age-sex standardised rate of limiting long term illness by area, known as the standardised illness ratio (SIR). The main drawbacks of the measure are (i) that self-reported health status is confounded by the expectations of the reporter, (ii) and that it is not specific to any particular disease or condition, and so gives very little information about the potential to benefit from health care.

## (b) Socioeconomic deprivation

Because information at small area level on limiting long term illness from the Census has only been available since the mid 1990's, and because of the potential biases arising from self-reporting of health status, researchers have often used small area-based measures of socioeconomic deprivation as a proxy measure for 'need' for health care.

Socioeconomic deprivation is partly an 'individual' concept and partly a 'community' concept. Factors relating to an *individual* can contribute to socioeconomic status, for example type of employment, level of educational attainment, type of housing lived in, access to a car, or age - data on these variables are available from the 1991 census, aggregated by small area. '*Community*' factors include material deprivation, quality of local housing, availability of key facilities, and general levels of unemployment – information on which are also available from the census. Both 'individual' and 'community' socioeconomic factors have been shown to influence health, although there is debate as how and their relative importance (11)(14).

There is no single definition of a 'deprived area'. In the past researchers have used a number of indicators, usually obtained from the census, to assess area deprivation. The indicators are based on 'objective' socioeconomic criteria, for example level of unemployment, housing tenure, proportion of lone-parent households, which can be used singly or combined to form a composite measure, for example the Townsend (15), the Carstairs (16), the Jarman (17) and the Dept. of Environment (18), indices. The first three were developed specifically for use in the context of health care services, in particular for resource allocation and planning purposes.

The table below shows the elements of the main composite indicators used in the UK.

Element	Index			
	Carstairs	Jarman (UPA)	Townsend	Department of the Environment
Unemployment	x	x	x	x
No car	x		x	
Low Social Class	x			
- Unskilled		x		
Overcrowding	x	x	x	x
Not owner-occupied			x	
Lacking amenities				x
Single parent		x		x
Aged under 5		x		
Lone pensioners		x		x
1-year immigrants		x		
Ethnic minorities		x		x
Vacant dwellings				x
Level and access (old)				x
Level and access (<5)				x
Permanent sickness				x
Large households				x

# Table 1

#### Elements of some deprivation indices

Source: adapted from Morris R, Carstairs V. Which deprivation? A comparison of selected deprivation indexes. Journal of Public Health Medicine 1991;13: 318-26

The table shows that a variety of variables have been included in each respective indicator.

The Carstairs and Townsend indicators are most similar with three out of four variables in common.

Two key questions are, how were the variables that comprise each index selected, and how is each variable 'weighted' relative to the others and why?

For the three indicators that were developed specifically for use in planning health services, the variables were initially selected according to whether they seemed likely to be associated with material disadvantage and ill health. Of these, for the Carstairs and Townsend indices, the variables finally selected were those most strongly correlated with the measures of ill health (levels of mortality or morbidity). For the Jarman index, the variables selected were those thought to be most associated with higher workload for general practitioners.

On the weighting of each variable within each index, in each index slightly different statistical methods are used. In the Carstairs score, the 4 variables are standardised but unweighted. In the Townsend score, the 4 variables are also standardised and unweighted, but the natural logarithm of two of the variables (unemployment and overcrowding) is used. For the Jarman score, the 8 variables are standardised, then transformed into their natural logarithm, and then each given a weighting determined from a national survey of general practitioners (which aimed to identify the relative pressure on GP's workloads from each of the variables) (17).

Which index, if any, is most suitable for the investigation proposed here? There could be two main criteria for suitability: first that the deprivation indicator includes factors that affect the 'need' for health care over and above the factors represented by SIR and SMR; second, that the deprivation indicator is a measure of social exclusion which may affect access to health care.

For these two criteria, most research has concentrated on how far different deprivation indicators are associated with 'need' for health care. Usually the research has investigated the level of correlation between deprivation measures and ill health (rather than capacity to benefit from care). For example, there is evidence from one recent study that the level of health in a population (health status as measured using the SF-36 questionnaire) is more

highly correlated with the Townsend index than the Jarman index (the Carstairs index was not investigated), and particularly highly correlated with two socioeconomic variables – housing tenure and car ownership (19). Other work has measured the correlation of a range of 'health' variables (including SMRs, whether classified as 'permanently sick') with each index and found that the correlation coefficients were strongest for those variables highly associated with the Carstairs index (no car, unemployment, overcrowding, and social class measures (20). The authors concluded that, of the 5 most readily available indicators, the Carstairs and Townsend indices were 'best buys' to use when investigating variations in health. Certainly all the indicators shown above have been shown to be highly correlated (20)(21), and all four have been commonly used in small area analyses in the UK. It is not clear from the available research whether the same levels of socioeconomic deprivation in different locations are correlated with the same level of 'illness' in those locations.

Unfortunately less research has been undertaken to investigate how far various measures of area deprivation really do measure the extent of social exclusion from health care.

# (c) Mortality

Information on the cause of, and age and address at, death is available from death certificates that are completed for every death in the UK. The data are collated by the Office for National Statistics and are available at small area level, typically in the form of age and sex adjusted rates, using indirect standardisation to produce the standardised mortality ratio or SMR.

The advantages of using the SMR as an indicator of need are that it is available at small area level, and mortality for specific conditions can be identified, although typically in small area analyses, 'all cause' SMR has been used. The main drawback is that the relationship of mortality to illness levels is uncertain, as is the relationship with potential to benefit from health care. Both of these relationships may be tenuous for some conditions (eg chronic conditions, many of which may be 'ambulatory care sensitive').

Researchers have noted these drawbacks, in particular that two of the three indicators described above do not give useful information about the 'need' for care for specific

conditions, such as ACS conditions. A few small area analysts have carried out community surveys to estimate the prevalence of specific diseases under investigation. For example, in a study of variations in hospitalisations for 5 ACS conditions in California Bindman *et al* (22) estimated the prevalence of each of these conditions in the geographical area under study using information from a telephone survey of residents. The results were validated against data obtained from a national prevalence survey. In a multivariate analysis, prevalence of illness was the second most influential variable influencing hospitalisation after access to care as perceived by residents.

In Sheffield in the UK, Payne and Saul analysed hospitalisation rates by electoral ward for coronary angiography, angioplasty or coronary artery bypass graft, according to 'need', as indicated by the prevalence of angina per head of population as ascertained by a community survey (23). They found that rates in the most deprived areas were half those in the least deprived.

Chaturvedi and colleagues used GP consultation rates for those conditions as a proxy indicator of 'need' (24). They found a marked positive association between GP consultation and hospitalisation rates for selected conditions (varicose veins) but not for others (including hernia, gallstones, and hip operations for osteo-arthritis). However, the level of GP consultations was also likely to be influenced by factors other than need such as supply and accessibility, and the study was not able to take into account other factors such as the social class, or deprivation of area of residence, of patients.

In conclusion, all three commonly used measures of 'need' discussed above are problematic in some way. In particular, none addresses the issue of capacity to benefit from health care. Nevertheless, these three are the main measures available at small area level. And since it is highly debatable as to which is the better indicator of need, often in small area analyses in the UK, all three types have been used.

#### 2.3 Access to health care

Section	n outline	•	
2.3.1	Acces	s and use	
2.3.2	Some ways of thinking about access		
	(2)	Aday and Anderson's model	
	(b)	multiple dimensions	
	(c )	multiple levels	
	(d)	'perceived' and 'in-system' access and outcomes	

#### 2.3.1 Access and use

'Access' is sometimes taken to mean *opportunity to use* health care (25). But some researchers and policy-makers often use the term access to mean *utilisation* of care: access is not measured directly but inferred from scrutiny of utilisation patterns. In the literature there has been debate over whether in principle it is best to investigate equity of the 'opportunity to use' health care or 'actual use' (26)(27)(28)(29). Mooney *et al* favour measuring the 'opportunity to use health services' or the costs to the individual of utilisation (25). They argue that to measure utilisation alone ignores the preferences and choices of individuals seeking treatment – a point reinforced by Le Grand (30)(31)(32). Culyer *et al* favour measuring actual utilisation arguing that if the poor have the same opportunity to use care as the non-poor but have a lower take-up of services, then lack of use rather than opportunity would be the major concern of policy-makers (28). They argue that utilisation is an important indicator of 'realised' access and that it is no accident that use and access have been confused for so long.

Various definitions of access that have been put forward also favour taking the utilisation of health care as the measure:

'The proof of access is in the use of services, not simply the presence of a facility' (33)

'The timely use of personal health services to achieve the best possible outcomes' (34)

In practice, most researchers investigating access using quantitative methods have had to rely on utilisation data, partly because the data are easily available. So while 'equal access for equal need' is the more favoured definition of equity in health care, in research 'equal use for equal need' has been used more.

2.3.2 Some ways of thinking about access

# (a) Aday and Andersen's Model

Lu Ann Aday and Ronald Andersen and colleagues in the 1970's (35)(36) developed a model to help identify and classify simple and measurable indicators which might reflect the access to, and use of, health care (see figure 1 below).

The model reflects the particular definition of access that at least one of the authors espouse:

'Those dimensions which describe the potential or actual entry of a given population group to the health care delivery system' (36).

The model can essentially be divided into two parts: one (on the left of the figure) containing measures that reflect the potential or probable access of an individual to health care; the other (on the right) containing measures of 'realised' or actual access.

To the left, factors reflecting and influencing potential access are divided into characteristics of *individuals*, of the *community or area* and of the *health care delivery system* (availability). Individual and community factors are grouped according to whether they *predispose* an individual to seeking care, *enable* an individual to seek care, or reflect the *need* of the individual to receive care. As shown, some factors may be influential both at the individual and community level (36). To the right of the figure, features of realised access include not only levels of utilisation of care but also, for example, of satisfaction. Andersen and Aday's early model has been used and developed further by other researchers (37)(38).

# Figure 1 Dimensions of access to medical care and their indicators\*

Poten	tial access	Realised acces	55
Individual	System (area/community)	Objective	Subjective (satisfaction)
Predisposing	Availability	Use	Convenience
Aged 6 or less	Drs/population	Preventive exam	Travel time
Aged 65 or over	Beds/population	Dr visits	Waiting time
Sex Race	Dentists/population	Hospital admissions Dental visits	Visit cost
Education	Area/community Characteristics	Use relative to need	Provider behaviour
Cushkus.	D. B. and a	C	Courtesy of staff
Financing Income	% aged 65 or above	Use/ disability Dental want	Quality of care
Health insurance	Enabling		
Visit cost	% below poverty Region		
Organisation Particular provider Specialty of provider	Rural residence Central city residence		
Travel time	Need		
Prior appointment Waiting time Time with Dr	Infant mortality		
Need			
Perceived health			
Dental symptoms Disability days**			

\* Adapted from Figure 1 in Anderson RM, McCutcheon A, Aday LA, Chiu GY, Bell R. Exploring the dimensions of access to medical care. Health Services Research 1983;18(1):49-74.

\*\* The number of days of self-reported disability

As Andersen notes, equity of access to care is implied if the predominant factors influencing utilisation are *need* for care or individual demographic factors such as age (reflecting need), and inequity is implied if *predisposing* or *enabling* factors determine use rather than need (38). Thus a key task of research into access is to try to investigate the respective influence of various predisposing or enabling factors and 'need' for health care.

#### (b) Multiple dimensions

Aday and Andersen's model implies that access to health care is a *multi-dimensional* concept. The dimensions of factors that enable and predispose individuals or communities to access health care include:

- financial (for example income or health insurance status);

- geographical (for example by distance to health care facilities, available transport);

- educational (for example access influenced by educational attainment, social class, understanding of illness and treatment, previous experience);

- cultural (for example race or language health beliefs);

- gender;
- generational (for example age);

- technological (for example the ability to use communications technology to access health care).

These factors have also been noted by other writers on access, although they have been categorised slightly differently (for example Penchansky & Thomas (39)). Many of the factors listed above are known to have overlapping effects (for example income, area of residence and ethnicity) and they may also influence utilisation of health care in ways other than through access, for example by influencing the need for care. The effect of each factor on utilisation is likely to be difficult to identify separately. The direction of any causal relationship between factors may also be complex (for example a high supply of health service facilities in a community may result in higher levels of use, or vice versa) and cannot be inferred from cross-sectional studies.

# (c) Multiple levels

The model also illustrates that access is a *multi-level* concept - that factors influencing access can operate at different levels, such as poverty influencing the individual directly (such as lack of access to a car) and through the community (such as poor public transport).

# (d) 'Perceived' and 'in-system' access, and outcomes

It is worth noting that there are at least three dimensions of access (and their measurable indicators) that are not made explicit in the Anderson and Aday model.

First, perceived barriers to care may have greater effects than actual barriers (22)(40). Second, it is necessary to distinguish between access to care once the patient is actually in the health care system or 'in-system access' (for example, access to care within the GP surgery, such as to a prescription or referral to an outpatient clinic) and access to the first point of contact or entry. 'Care' in this sense includes clinical care and 'interpersonal' care – the interaction of health care professionals and their users and carers or 'the management of the social and psychological interaction between client and practitioner' (41). Third, and particularly relevant to this thesis, is the concept of access to timely, effective and appropriate care (care that produces better clinical outcomes for that person) rather than just to any form of care. This implies the need to take outcomes and the quality of care into consideration, not just access to, or use of, care.

As noted in chapter 1, in this thesis, hospitalisation rates for ACS conditions will be used as a negative 'downstream' indicator of access to upstream 'timely and effective' ambulatory care. Hospitalisation will thus be an indirect indicator of the quality of ambulatory care, the implication being that some hospital admissions are likely to be avoidable if good quality ambulatory care 'upstream' had been accessible. Section 2.4 below explores these concepts further. 2.4 Avoidable 'downstream' events as indicators of access to, and quality of, health services

# Section outline

- 2.4.1 The concept of avoidable events
- 2.4.2 Identifying potentially avoidable admissions

## 2.4.1 The concept of avoidable events

The concept of access to health services builds on the idea that the *quality* of care provided is important (for example the timeliness with which it is provided, and how effective it is in improving health outcomes) (34).

The pathway of an individual through a health care system may be complicated, making it quite difficult to assess which type of 'upstream' care influenced which type of 'downstream' avoidable and negative events. However in the literature there seem to be two main areas of focus:

- (a) the quality of *ambulatory care* (upstream) and its effect on *hospital admission* (downstream potentially avoidable event);
- (b) the quality of ambulatory and/or inpatient(hospital) care (upstream) and its effect on subsequent mortality (downstream potentially avoidable event).

Billings' work, outlined in the introduction (1), used hospital utilisation for 'ACS' conditions as a proxy *negative* indicator of access to, and quality of, ambulatory care (point (a)). Others have conducted similar investigations (see below), but work with this focus has been relatively uncommon. A few have challenged the idea that good quality primary care might result in fewer 'downstream' admissions. For example a recent multi-centre randomised controlled trial in the US suggested that improved primary care was associated with *increased* hospitalisation (42). However this study was criticised because care given to

intervention group was not primary care but intensive specialist care, patients were severely ill with end stage chronic disease, and continuity of care was absent (43)(44).

There has been much more analysis of the effect of 'upstream' care on mortality (point (b)). For example, in the UK, as in many other countries, individual 'sentinel' cases or events are routinely investigated to determine whether care was lacking in cases of peri-operative and maternal deaths (45)(46)(47).

Less common and more controversial has been the analysis of variations in *rates* of events to indicate deficiencies in quality or access. This approach was used by Rutstein and colleagues in the 1970's, who, in a team called the Working Group on Preventable and Manageable Diseases, developed a method of measuring quality of care that counted cases of unnecessary disease and disability and unnecessary or untimely deaths (48)(49). The working group identified three groups of conditions:

- single index (or sentinel) cases (where even one case of disease disability or death would indicate a problem);

- 'rates-linked' conditions (where critical increases in rates of disease or death could serve as indices of the quality of care; and

- medico-social conditions (such as homicides, alcoholism and drug dependence - categories "requiring further definition").

For each group a list of diagnoses and conditions were defined. The working group suggested that a failure of lifestyles, medical care, public health and social environmental factors (similar to Aday and Andersen's 'individual', 'community' and 'health delivery system' factors noted earlier) could contribute to critical increases, for example, in the rates-linked conditions and would serve as a 'warning signal' about the quality of care.

The conditions drawn up by the Working Group created a lot of interest in the early to mid-1980's. Most of the resulting investigation focused on avoidable deaths rather than admissions, and 'rates-linked' conditions rather than single index cases in which a failure of quality of medical care was often difficult to prove (50).

In the US researchers continued to analyse 'avoidable' mortality rates and hospitalisation

rates using Rutstein's 'rates-linked' conditions, although the analyses were frequently conducted across social and ethnic groups rather than geographical areas. Higher rates of avoidable hospitalisation were found among disadvantaged groups (51)(52).

In the UK, in 1983 Charlton *et al* found considerable variations in standardised mortality ratios (SMRs) for a subset of conditions on Rutstein's list across geographical areas across England and Wales, after adjusting for socioeconomic factors (53). Other studies found similar variations even after adjusting for morbidity (as measured by hospitalisation or disease registration)(54). Maps showing avoidable death 'hot spots' in the UK were produced and a link between the level of the SMRs and the quality of health care was suggested (55). Some argued that this link had not been proven because of the probable long lag time between medical care and death, and because of the multitude of other factors which could affect mortality apart from the quality of health care received (56). Others had previously found that socio-demographic factors were far more influential than health service characteristics on the mortality rates observed (57).

But since many of Rutstein's 'rates-linked' conditions were relatively uncommon (so that large geographical areas were often required so as to include enough events to demonstrate statistically significant differences between areas) and because of the difficulty of linking cause (the quality of 'upstream' medical care) to effect (the rates of 'downstream' events observed), interest in avoidable mortality in the UK waned in the late 1980's. A review of the literature on avoidable mortality concluded that in-depth studies at individual level were more likely to produce information about factors relating to the quality of health services than studies of aggregate data (58), although other authors continue to argue the merits of the latter (59).

Readmission rates to hospital (within 28 days of discharge) have also been examined in a number of studies as a 'downstream' outcome measure of the quality of 'upstream' hospital, and to some extent ambulatory, care after discharge from hospital. But in the UK, one indepth analysis of a sample of readmissions showed that less than half were in fact avoidable (60), and the author suggested that if readmissions were used as an indicator, only those that were avoidable should be counted (61). Others have come to similar conclusions (62)(63). This suggests that if rates of avoidable events are to be used as an indicator, then there must be very careful definition and identification of what is an avoidable event, in advance of any

analysis.

# 2.4.2 Identifying potentially avoidable admissions

Crucial to the work outlined above has been the identification of a valid list of conditions for which hospitalisation is likely to indicate a problem of the quality of, or access to, 'upstream' ambulatory care. In studying rates of admissions or individual 'sentinel' events, researchers have usually identified in advance a list of conditions for which admission or death could indicate avoidable morbidity or mortality.

For example. Rutstein and colleagues selected conditions on the assumption that 'if everything had gone well, they would have been prevented or managed, to avoid unnecessary disease, disability or death' (48). The Working Group collaborated with several major medical institutions in the US, such as the Centers for Disease Control and the Veterans Administration, and consulted experts widely.

More recently the few groups of researchers investigating rates of hospitalisation for ACS conditions have developed their own lists of conditions. Billings *et al* analysed hospitalisation rates for three groups of conditions (1). The first group were 'ambulatory care sensitive' (or ACS) conditions:

'diagnoses for which timely and effective ambulatory care can to help reduce the risk of hospitalization either by preventing the onset of the illness or condition, controlling an acute episodic illness or condition or managing a chronic disease or condition'

The second were 'marker' conditions:

'diagnoses for which the provision of timely and effective outpatient (ambulatory) care is likely to have little impact on the need for hospital admission and where there is substantial agreement amongst practitioners on the clinical criteria for admission'.

The third were 'referral sensitive' or 'discretionary' surgeries:

'high cost/high technology surgical procedures for which impediments to access or referral to specialty care may reduce the chances of having the surgery'

An advisory panel of physicians (six internists and paediatricians) drew up a list of ICD-9 CM codes (International Classification of Diseases – 9<sup>th</sup> revision, Clinically Modified codes) for each group using a modified Delphi technique, specific details of which have not been published. The 'ambulatory care sensitive' and 'marker' conditions selected are set out in table 2 below, having converted the ICD-9CM codes to ICD-9 codes used in the UK. Where the panel thought that conditions were ACS or marker under specific circumstances, for example in specific age groups, this is indicated in the table.

# Table 2 Ambulatory care sensitive (ACS) and marker conditions and ICD-9 codes\*, selected by Billings et al

ACS Conditions	ICD-9 codes
Congenital syphilis	90, 90.1, 90.2, 90.3, 90.4, 90.5, 90.6, 90.7, 90.9**
Whooping cough	33, 33.1, 33.8, 33.9
Rheumatic fever	390
(without heart involvement)	
Rheumatic fever	391,391.1, 391.2, 391.8, 391.9
(with heart involvement)	
Tetanus	37
Acute poliomyelitis	45, 45.1, 45.2, 45.9
Bacterial meningitis	320***
Epilepsy	345, 345.1,345.2,345.3,345.4,345.5,345.6,345.7,345.8,345.9
Convulsions	780.3
Suppurative and unspecified	382, 382.1, 382.3, 382.4, 382.9****
otitis media	
Acute pharyngitis	462
Acute tonsillitis	463
Acute respiratory infections	465, 465.8, 465.9
Chronic pharyngitis	472.1
Pulmonary tuberculosis (TB)	11,11.1,11.2,11.3,11.4,11.5,11.6,11.7,11.8,11.9
Other respiratory TB	12,12.1,12.2,12.3,12.8
TB of meninges and central nervous	13,13.1,13.8,13.9
system	
TB on intestines, peritoneum or mesenteric glands	14
TB of bones and joints	15,15.1,15.2,15.7,15.8,15.9
TB of genitourinary system	16,16.1,16.2,16.3,16.4,16.9
TB of other organs	17,17.1,17.2,17.3,17.4,17.5,17.6,17.7,17.8
Miliary TB	18,18.8,18.9
Chronic bronchitis	491,491.1,491.2,491.8,491.9
Emphysema	492
Bronchiectasis	494
Chronic obstructive airways disease	496
Acute bronchitis and bronchiolitis	466,466.0,466.1^
Pneumococcal pneumonia	481
#### Table 2 Continued

ACS Conditions	ICD-9 codes
Other bacterial pneumonia	482.2,482.3,482.9^^
Pneumonia due to other specified	483
organism	
Bronchopneumonia	485
(organism unspecified)	
Pneumonia (organism unspecified)	486
Asthma	493,493.1.493.9
Heart failure (left)	428,428.1,428.9
Hypertensive heart disease	402,402.1,402.9
Acute oedema of lung	518.4
Hypertension	401,401.9
Other forms of ischaemic heart disease	411^^^
Angina pectoris	413^^^
Cellulitis (finger or toe)	681,681.0,681.1,681.9^^^^
Other cellulitis or abscess	682,682.1,682.2,682.3,682.4,682.5,682.6,682.7,682.8,682.9^
Acute lymphadenitis	683
Other local infections of skin or subcutaneous tissue	686,686.1, 686.8, 686.9
Diabetes mellitus	250,250,1,250,2,250,7,250,9
Hypoglycaemia	251.2
Other non-infective gastroenteritis and colitis	558
Infections of kidney	590,590,1,590,2,590,3,590,8,590,9
Other disorders of urethra or	599.599.9
Volume depletion	176.5
Volume depletion	270.5
Kwashioskos	260
Nutritional managemus	260
Vitamin D deficiency	201
Lack of expected normal physiological development	1 783.4**
Pelvic inflammatory disease	614,614.1,614.2,614.3,614.4,614.5,614.6,614.7,614.8,614.9
Marker Conditions	ICD-9CM codes
Appendicitis	540.540.0.540.1.540.9~, 541~, 542~
Acute myocardial infarction	410
Intestinal obstruction	560.560.8.560.9
Fractured neck of femur	820 820 0 820 1 820 2 820 3 820 8 820 9

\* ICD-9 codes International Classification of Diseases, Ninth Revision.

\*\* all codes as applied to those aged under 1 year

\*\*\* in those aged 1-5 years

••••• all codes excluding cases with procedures D15, D15.1 or D15.2 ^all codes only with a secondary diagnosis of 492,494,496

^^ all codes excluding those in patients aged under 2 months and those with a sceondary diagnosis of 282.6 ^^^ excluding cases with a surgical procedure

^^^^ excluding cases with a surgical procedure except S47, S47.1-S47.9

~ in those aged under 5 years

--- all codes only with surgical procedures H01.1, H01.2, H01.3, H01.8, H01.9, H02.9

---- only in cases with an inpatient length of stay >5 days or who died

----- in those aged <1 or >45 years

Weissman *et al* in New Hampshire used three main criteria for selecting conditions as an indicator of whether admission to hospital could represent shortcomings of ambulatory care treatment (64). The criteria were: clinical consensus about avoidability of admission in persons aged under 65 years of age; importance of the clinical problem; and whether or not the conditions were clearly coded on hospital discharge data. The resulting list of 12 conditions, mostly a subset of conditions selected by Billings *et al*, is shown in table 3 below.

# Table 3 Avoidable hospital admissions and ICD-9CM \* codes, selected by Weissman et al

Avoidable hospital conditions	ICD-9CM* codes
Ruptured appendix	540.0, 540.1
Asthma	493
Cellulitis	681, 682
Congestive heart failure	428, 402.01, 402.11, 402.91
Diabetes	250.1, 250.2, 250.3, 251.0
Gangrene	785.4
Hypokalaemia	276.8
Immunizable conditions	032, 033, 037, 072, 045, 055
Malignant hypertension	401.0, 402.0, 403.0, 404.0, 405.0, 437.2
Pneumonia	481, 482, 483, 485, 486
Pyelonephritis	590.0, 590.1, 590.8
Perforated or bleeding ulcer	531.0, 531.2, 531.4, 531.6, 532.0, 532.2, 532.4, 532.6,
L.	533.0, 533.1, 533.2, 533.4, 533.5, 533.6

ICD-9 CM indicates International Classification of Diseases, Ninth Revision, Clinical Modification. ICD-9 CM codes were
used in the USA in the 1990's, whereas ICD-9 codes (un-clinically modified) were used in this period the UK.

Bindman and colleagues in California selected five ACS conditions 'using the opinions of clinicians': asthma, chronic obstructive pulmonary disease, congestive heart failure, diabetes, and hypertension (22). Again many of these were a subset of conditions identified by Billings (1).

Solberg and colleagues in Minnesota developed a list of 15 conditions according to four criteria (65):

- clarity of case definition and coding;

- case frequency high enough (at least 1 admission per 10 000) to warrant the development of review criteria;

- broad coverage of age, gender and care types;

- perceived likelihood that the hospital admission would have occurred either at an earlier stage of the disease or not at all if the quality of ambulatory care had been good.

The indicator conditions were selected by representatives from three health maintenance organisations 'by group consensus'. More specific details were not published, neither were the specific codes. The conditions are shown in table 4.

## Table 4 Indicator conditions selected by Solberg et al (65)

#### Avoidable conditions

Diabetic acidosis Ruptured appendicitis Gangrene of extremity Hypokalaemia Pulmonary embolism / infarction Cellulitis Peptic ulcer with perforation, bleeding, or obstruction Transient ischaemic attack or cerebrovascular accident under aged 65 (resulting from hypertensive crisis) Primary breast cancer - radical surgery Drug toxicity (drug overdose/toxicity, bleeding resulting from excess anticoagulant therapy) Endometrial cancer (as an extension of cervical cancer) Asthma Premature birth Ectopic pregnancy Toxaemia of pregnancy

Other US researchers investigating rates of preventable hospitalisation have used the lists of conditions (or a subset) developed by the researchers already named (3)(66)(67)(68).

No such list of 'preventable hospitalisations' has been developed systematically in the UK. However there has been some recent work in which admission rates for a few conditions thought to be manageable in primary care have been analysed, for example asthma, diabetes and epilepsy (69)(70). Similarly the National Performance Assessment Framework (71), a national set of performance indicators for the NHS published in 1999 and 2000, contains one based on admission rates for diabetes and asthma, purporting to indicate the quality of management of these chronic diseases in primary care.

The methods used by all of the researchers noted above to identify a list of conditions have not been published in detail, but have been based in general on developing consensus among experts, usually clinicians. The studies do not appear to have attempted to:

- gain consensus explicitly on the *degree* of likely avoidability or preventability of admissions for certain conditions (although Billings *et al* did specify the circumstances under which the conditions were likely to be ambulatory care sensitive, as summarised in table 2);

- distinguish between the likely effect of ambulatory care in *preventing the onset of the* condition, or *preventing admission for an existing condition*;

- distinguish between the relative contribution of different types of ambulatory care, such as primary, outpatient or accident and emergency care.

But despite the different criteria and methods used to identify the conditions, there is a fair amount of overlap between those identified by each group of researchers. The table below shows the main ambulatory care sensitive conditions that commonly feature in all lists.

## Table 5 Main conditions commonly thought to be ambulatory care sensitive

#### Condition

Asthma Diabetes Pneumonia Cellulitis Congestive heart failure Chronic obstructive airways disease

## 2.5 A brief overview of the literature on access to, and use of, health services

## Section outline

2.5.1	Broad	types	of in	nvesti	gation
-------	-------	-------	-------	--------	--------

### 2.5.2 Reflections on the literature

- (a) Multivariate analyses using cross sectional data
- (b) Smaller scale and qualitative studies

## 2.5.3 Main findings

- (a) Studies in the US
- (b) Studies in the UK
- (c) Contrasting the findings for medical and surgical admissions

## 2.5.4 Factors found to influence access to and use of health care

- (a) Age and sex
- (b) Indicators of need
  - (i) Socioeconomic deprivation
- (c) Ethnicity
- (d) Patient lifestyle
- (e) Perceived access
- (f) 'Health System factors
  - (i) The hospital itself
  - (ii) Levels of supply and access to supply

Hospital care

Ambulatory care

- (iii) Financing and organisation of care Direct financial incentives Organisation of care
- (iii) Physician practice style
- 2.5.5 Summary

#### 2.5.1 Broad types of investigation

The literature on access to health care spans several disciplines, including health services research, medical sociology, policy analysis, health economics, epidemiology, geography, and operational research. In many studies the original aim was not to investigate access at all but some other issue such as how to control health care costs, or the doctor-patient relationship. Nevertheless the results shed some light on the question of access to, and use of, care. There is a very large amount of literature in this area and most studies have been conducted in North America. This section simply gives an overview of the UK and North American literature, examines the broad types of investigation that have been carried out, and identifies the main factors found to influence access to and use of health care.

Broadly, the research can be divided into two main types: large scale multivariate studies, typically either ecological analyses using aggregate cross-sectional data (72)(73); or smaller scale observational and qualitative studies (74). The focus on this review is mainly on the former, because of the intended analysis in this thesis.

Much of the research has aimed to identify the factors that influence access to and use of care. Figure 2 summarises the main factors that have been investigated; the list shown is not exhaustive. The figure looks similar to the Andersen/Aday model (figure 1) shown earlier, except that only utilisation of care is listed as a measure of realised access, since utilisation (hospitalisation) is the indicator of interest in this thesis.

## Figure 2 The main factors that have been investigated as having influence in relation to access to, and use of, care

	Explanatory and utilisation variables	Utilisation variables	
Column ( a)	Column ( b)	Column ( c)	
Area A			
'Individual' factors	Levels of utilisation	Levels of utilisation	
(a) self-reported health status	- primary care	- inpatient care	
(b) whether classified as 'permanently sick'	- community health services		
(c) age/sex	- accident and emergency care		
(d) income, occupation	- outpatient care		
(e) access to a car			
(f) housing tenure			
(g) education			
(n) ethnicity			
(I) HIESTYle			
j) possession of health insurance			
Area-lavel aggregated data of 'individual' fa (especially (a) to (h)) *Community' factors (k) physical environment (eg pollution) (l) availability of public transport	ctors		
Area-lavel aggregated data of 'individual' fa (especially (a) to (h)) "Community' factors (k) physical environment (eg pollution) (l) availability of public transport	ctors		
Area-layel aggregated data of 'individual' fa (especially (a) to (h)) "Community' factors (k) physical environment (eg pollution) (l) availability of public transport "Health System' factors	ct0/3		
Area-level aggregated data of 'individual' fa (especially (a) to (h)) 'Community' factors (k) physical environment (eg pollution) (l) availability of public transport 'Health System' factors (m) supply of facilities (eg staff, beds)	ct0/3		
Area-lavel aggregated data of 'individual' fa (especially (a) to (h)) "Community' factors (k) physical environment (eg pollution) (l) availability of public transport "Health System' factors (m) supply of facilities (eg staff, beds) (n) distance to facilities	ctors		
Area-level aggregated data of 'individual' fa (especially (a) to (h)) "Community' factors (k) physical environment (eg pollution) (l) availability of public transport "Health System' factors (m) supply of facilities (eg staff, beds) (n) distance to facilities (o) quality of facilities	ctors		
Area-level aggregated data of 'individual' fa (especially (a) to (h)) *Community' factors (k) physical environment (eg pollution) (l) availability of public transport *Health System' factors (m) supply of facilities (eg staff, beds) (n) distance to facilities (o) quality of facilities (p) practice style of clinical staff	«««»		
Area-level aggregated data of 'individual' fa (especially (a) to (h)) *Community' factors (k) physical environment (eg pollution) (l) availability of public transport *Health System' factors (m) supply of facilities (eg staff, beds) (n) distance to facilities (o) quality of facilities (p) practice style of clinical staff (q) funding of health service facilities	«««»		

Columns (b) and (c) in figure 2 list some of the factors used as indicators of realised access. Essentially these are measures of utilisation (mainly numbers or rates of contacts) in the ambulatory ((column (b)) or inpatient ((column (c)) setting. These factors are frequently used as *dependent* variables in small area analyses.

Column (a) groups the main types of variables that have been investigated to see whether they are associated with utilisation. They are typically used as *independent* variables in small area analyses. As in Andersen and Aday's model these are divided into 'individual', 'community' and 'health system' factors.

The factors listed under the heading 'individual' include all individual-level data that are available routinely, typically from population surveys such as the General Household Survey (75) and the 4<sup>th</sup> National Survey of Ethnic Minorities (76). As noted above, routinely collected NHS data on hospital admission also include some 'individual' variables such as age and sex.

'Area-aggregated individual-level' data include 'individual' factors that have been aggregated by area (typically obtained from the census). 'Community'factors include those that may affect need or access to care over and above individual factors, such as levels of unemployment or access to public transport. 'Health system' variables are available from routine sources and are often expressed by institution (eg numbers of staff) or by area (staff per head of catchment, or area, population). The factors that have been investigated include the supply of facilities and their distance from users, the quality of care provided, how facilities were organised or funded, and the 'clinical practice style' of physicians. These factors have been measured for the varying types of ambulatory as well as inpatient care.

The two main types of investigation using the variables summarised above have been : a) population surveys, which have been able to link *self-reported* utilisation of health care with self-reported data on income and health status in individuals, allowing 'individual use' to be analysed by 'individual need'; and b) ecological studies using health service routine data on the *actual* utilisation of care by individuals. The latter tend to be able to link *individual* use with few individual-level variables (such as age and sex) but more often have used area-aggregated individual-level data and 'community' factors as proxy indicators of need (such as socioeconomic deprivation).

Most of these ecological studies and surveys have measured the influence of variables in column (a) on utilisation variables in column (c), and much less commonly on those in column (b). There has also been a lot of work examining the association between variables within each column. More of this has involved variables within column (a) (for example the association between area-level morbidity (as a measure of need) and the supply of health care facilities) than within column (b) (for example the association between use of primary care and use of care in accident and emergency (A&E) departments).

Investigation of the effect of variables in column (b) on column (c) - such as the effect of utilisation of primary care, or A&E care, on subsequent utilisation of hospital care - has been unusual. This may be because most of the studies outlined have been conducted in the US and UK where there has been a relative lack of routine data on utilisation in ambulatory care. The US literature typically contains studies investigating associations across one link, such as between 'individual' or 'community' factors and hospitalisation, rather than across two links, for example the influence of 'individual' or 'community' factors on utilisation of ambulatory care and the influence of use of ambulatory care and 'individual' or 'community' factors on subsequent utilisation of hospital inpatient care (see figure 3 below). This type of investigation may increase with the growing emphasis in both the US, and the UK, on managed care, the recognition that family physicians can play a significant 'gate-keeping' role, and as information systems allow integration of data between health care facilities in the acute and ambulatory care sectors.

#### Figure 3 Pathway of analysis



The sections below offer broad reflections from the literature first using multivariate analyses of cross –sectional data, and then using smaller scale and qualitative methods of investigation.

#### 2.5.2 Broad reflections on the literature

#### (a) Multivariate analyses using cross-sectional data

Two broad strands of work apparent from the literature are illustrated in figure 2 as Area A and B. Area A includes multivariate analyses that have mainly used data from population surveys. Area B includes multivariate analyses that have mainly used routinely collected health service data on utilisation.

Studies in Area A have tended to use *population survey data* to link information in individuals, for example on self-reported utilisation of health care and self-reported variables listed under 'individual' in column (a). The concern of many researchers in this area has been equity of access to care, and typically rates of utilisation between population groups of interest (such as by race, income, or insurance status) have been compared after some adjustment for need (usually age, sex, self-reported health status and ethnicity) and other factors. Because the geographical location (rather than broad type) of the health services utilised is usually not available from survey data, health system variables such as supply of facilities, distance to, quality or organisation of services available to users have generally not been taken into consideration even though these are known to have important effects on utilisation (77)(72).

Studies in Area B generally include ecological studies using information that has been routinely collected by health services, typically hospital discharge data. Typically utilisation rates of hospital (rather than ambulatory) care have been compared between geographical areas or between specific population groups (such as ethnic or, in the US, income group). In these studies 'health system' variables can be taken into account because the location of health care used and the residence of the patient are likely to be known. The 'health system' factors investigated have related more to hospital facilities than ambulatory care, for example the supply of beds, the numbers of doctors per capita working in the area, or (in the

US) the method of funding the facility (for example fee-for-service or prospective payment). There has been much less investigation of the effect of different forms of organisation, or utilisation, of ambulatory care on hospitalisation rates. Generally studies in Area B, in contrast to those in Area A, have used less satisfactory data on 'need' with which to explain variations in utilisation, because researchers have mostly had to rely on 'area-aggregated individual-level' measures.

The concern of many US researchers conducting small area studies (included in Area B) has been on the *appropriateness* of variations in utilisation in view of the higher health care costs in high rate areas, and therefore on *efficiency* over and above equity. In contrast, in the UK small area analysis of hospitalisation rates has mainly been used to investigate the geographical *equity* of allocation of resources within the National Health Service, for example (78).

It was only in the 1990's that increasing numbers of US researchers who were interested in equity and access, rather than efficiency, began to use small area analysis in their enquiries. This may have been prompted by several factors. First was the growing realisation that there were specific conditions for which ambulatory care may have greater potential to reduce the need for hospitalisation. Second was the identification of these conditions - known variously as ACS conditions (1)(22) or 'avoidable hospital conditions' (AHCs) (64) (as outlined earlier). Third was the observation that high hospitalisation rates for these conditions occurred in patients living in low income areas with a lack (in terms of quality and quantity) of primary care. Finally, mapping hospitalisation rates for a group of these conditions by small area allowed presumed difficulties in access to ambulatory care to be pinpointed geographically, thus facilitating the targeting and evaluation of policies to improve access. But as yet only a few studies of this kind have been conducted.

## (b) Smaller scale and qualitative studies

Qualitative studies in Area A (figure 2) include those found in the medical sociology literature. Researchers have tried to identify factors, particularly in individual patients and doctors, that influence utilisation of care. Much of this research has tended to be small-scale and qualitative because of the number and complexity of factors involved. For example

studies have attempted to examine the effects of individual factors such as health beliefs, illness behaviour, lifestyles, culture, or social support, on patients' decisions to seek care and their actual use of health care (79). Others have examined 'in-system access' - for example the effect of the doctor-patient interaction on access of patients to subsequent hospital care (80)(81).

These studies have demonstrated the complex array of factors that influence use of health care, including how the health system is organised and financed, and psychosocial factors relating to doctors and patients (74). For example Newton *et al*, in a UK study based on interviews with 15 general practitioners, found that no less than 23 factors - both clinical and non-clinical - influenced the treatment received by patients. Non-clinical factors related to personal characteristics of the referring doctor (such as toleration of risk, legal implications), of the patient (such as expectations, social circumstances and ability to assert their views and feelings), the doctor's relationship with the consultant being referred to and the doctor's level of clinical experience (81). It has not been possible to take many of these factors into account in larger scale surveys or ecological research that rely on routinely collected data.

Smaller scale studies in Area B also include prospective or retrospective studies of individual admissions that have attempted to identify factors that have led to hospitalisation. These studies have more often investigated the *appropriateness* of admission for each patient (for example whether a lack of an appropriate and alternative location of care resulted in the admission) rather than the *avoidability* (whether the condition could have been prevented to an extent that admission was not needed), although the two terms are often confused in the literature. Again a complex number of reasons for admission has been described which simply cannot be taken into account in large ecological studies.

There have been a few other types of investigations which do not fit into the pattern described, for example those which have combined actual utilisation data with survey data. For example, in the UK, Carr Hill *et al* combined data from the 4<sup>th</sup> National Morbidity Survey in General Practice (providing some individual-level data) with information from the 1991 census (providing area-aggregated individual-level data) (82). The 4<sup>th</sup> National Survey covered 502 493 patients who consulted a GP from September 1991 to August 1992 in one of 60 sample practices in England and Wales – equivalent to a 1% sample of the

population. For each face to face consultation, information was recorded which included the reasons for the consultation, as well as key 'individual' factors such as age, sex and socioeconomic circumstances. This way patients' utilisation of GP care and referral to outpatient care ('individual use') could be linked with factors denoting 'individual need'.

The factors found to influence hospitalisation in all these studies may have a different level of influence in different countries, depending on the specific mix of types of reimbursement, incentives, organisation of health care, the health seeking behaviour and expectations and culture of patients or the health care professionals. This is important to note when considering the implications of results from other health systems for the UK.

## 2.5.3 Main findings

Most of this section will consider studies investigating the influence on hospital utilisation of the different factors listed in columns (a) and (b) in Figure 2. The aim will be to identify the factors that have consistently been found to be influential, and which will thus be important to investigate in this thesis. There will be a particular focus on the findings of small area studies (included in Area B in figure 2) that have investigated admission rates for ACS and marker conditions, although some observational and qualitative studies are also included. Because of the importance of 'need' in relation to hospitalisation, attention will be given to how this has been taken into account in the analyses. Factors in column (a) in figure 2 which have been found to influence the use of ambulatory care (column (b) in figure 2) will also be briefly reviewed, because access to ambulatory care is also likely to influence hospitalisation and the factors influencing access to ambulatory care may be different from those influencing hospital care.

#### (a) Studies in the US

In the US there has been a great deal of investigation into access to care using aggregate survey data (Area A in figure 2), such as from the National Health Interview Survey.



Poverty, lack of health insurance and non-white ethnicity have long been associated with an overall lower use of all types of health services after adjusting for differences in health status (83)(84)(85). For example, uninsured children with treatable conditions such as asthma have been shown to be less likely to receive care and be at risk of substantial avoidable morbidity (86). The introduction of publicly funded insurance (the Medicaid and Medicare programme) and other public initiatives have been shown to have had a major impact in reducing financial and other barriers to accessing care for the poor over the last twenty years in the US, effectively equalising the use of ambulatory and hospital care between recipients and the privately insured after controlling for need (as indicated by self-reported health status) (83). However recent studies still report inequities in utilisation by income and race (87)(88)(89).

As noted earlier, in general the small area analysis literature from North America (Area B in figure 2) has focused on questions of the cost-effectiveness (efficiency) of variations rather than questions of equity of use of health care. In an extensive review of the North American literature on small area analysis in 1987, Paul Shaheen showed that there had been more studies of admissions for surgical procedures than medical conditions, and that studies had generally not been able to take 'need' into account. 'Supply' of health services was most consistently shown as being the major factor influencing variations in rates of surgical procedures, whereas 'community' and 'individual' factors, such as poverty and health insurance status, appear to exert a stronger influence on the variation in medical admissions (72). But since many of the studies reviewed did not adjust for need, the effect of supply may be over-emphasised. In most small area studies, a large amount of variation remained unexplained, leading some to believe that factors that were not easily quantifiable - such as health beliefs and illness behaviour (74), patient lifestyle choices or the differing practice style of clinicians - may be influential (90).

There has been much less work of this kind in the US to examine variations in utilisation of ambulatory care. One reason is that while survey data are available (the results are referred to above), health service data on utilisation of ambulatory care are less routinely available (91).

#### (b) Studies in the UK

In line with the objective of the NHS to provide equal access for equal need, there has also been extensive investigation into variations in use of health care based on survey data (Area A in figure 2). Using data from the General Household Survey (GHS), higher rates of inpatient and outpatient care and GP consultations have been found among more socioeconomically deprived groups and most ethnic groups. Because the prevalence of illness is also higher in these groups, many studies have sought to adjust for these differences, although the most appropriate method for doing so is still debated (92)(93)(94)(95)(96)(97). Using GHS data, O'Donnell and Propper showed that use of ambulatory and inpatient care in the UK was more influenced by self-reported health status than any other factor. Their conclusion - that there was no evidence to suggest systematic inequity in use of inpatient care between social or ethnic groups (92) - has been underlined more recently by other similar work (98).

There has also been a considerable amount of analysis of rates of hospital admissions over the past 20 years using routine NHS utilisation data (Area B in figure 2). The most extensive was by researchers at the University of York, in developing a needs-based resource allocation formula for NHS funding (99). One of their studies involved an analysis of the determinants of use of hospital resources between 4 985 small areas (aggregates of electoral wards, or so-called 'synthetic' wards) covering all of England. The key variables tested included indicators of 'need' (mortality, limiting long-standing illness, permanent sickness and low birth weight) and indicators of supply or 'health system' variables (accessibility of NHS hospitals, accessibility of GP services, provision of nursing and residential homes, and accessibility of private hospitals). Of these, the study found that the factors most positively associated with acute episodes of inpatient admission (surgical as well as medical, emergency as well as elective) were all-cause standardised mortality ratio (for those aged under 75 years), limiting long-standing illness (for those under 75 years), the proportion of persons of pensionable age living alone (all 'need' variables), and the levels of access to GP services and to private hospitals ('supply' or 'health system' variables). How these 'access' factors were constructed is briefly discussed later, and in the methods section in chapter 6.

In a review of the literature on variations in hospitalisation rates, Sanders *et al* noted that in the UK surgical admissions had been investigated far more than medical admissions, and again the supply of hospital facilities (usually beds) had a consistently positive influence on the rates (73). However a significant amount of variation remained unexplained, and, as so often in literature reviews on this topic, there was an inadequate discussion as to the extent that 'need' factors were taken into account in the analysis. Conclusions are very difficult to draw from such reviews.

Some researchers have analysed variations in hospitalisation not by area of residence, but by general practice of registration. In one such study, involving 209,136 emergency and elective admissions in London in 1996, Reid found that admission rates between practices (for all, emergency, and elective admissions, respectively) varied twofold and that the differences in admission rates between practices were influenced more by socioeconomic factors (measured at area-level) (these explained 45% of the difference observed) than by health system factors (100). The latter factors included '*GP practice*' factors - a range of features of the general practices including whether a system of child surveillance was provided, or whether minor surgery was offered - which explained approximately 10% of the variation, and 'hospital factors' - the specific admitting hospital - which also explained approximately 10%.

As in the US, in the UK there has been much less analysis of variations in use rates of ambulatory care using routine health service data. In England & Wales there is no routinely collected minimum dataset for primary, outpatient or A&E care. The most extensive dataset is the National Morbidity Survey of General Practice (referred to above); otherwise studies have used information from selected individual practices.

Such as it is, the literature shows that GP consultation rates are influenced by factors similar to those for hospital utilisation. Generally there is a positive association with socioeconomic deprivation, morbidity, ethnic minority status and age ('need' factors), and a positive association with the accessibility to health service facilities (as measured by inverse distance) (24)(101)(102)(103)(104)(105)(106)(107)(108)(109)(110). Again these studies

have not been able to adjust adequately for need for health care.

One study that did attempt to adjust for need, referred to earlier, combined data from the 4<sup>th</sup> National Morbidity Survey in General Practice (providing some individual-level data) with information from the 1991 census (providing area-aggregated individual-level data) (82). Characteristics of individual patients (such as age, employment status and whether classed as permanently sick) were much more powerful predictors of consultation rates than area-aggregated data on 'need' from the census, or 'health system' factors (distance from the centre of the enumeration district of residence to the centre of the enumeration district of the general practice). Studies combining data in this way are very rare.

The broad findings suggest slight differences between the US and UK literature in that the UK work generally points to a stronger influence of 'need' factors over and above 'health system' factors on utilisation rates, whereas the reverse is apparently true in the US. However the picture is confused because much of the investigation in the US has focused on variations in admission rates for surgical procedures, whereas in the UK the work has been dominated by researchers at the University of York who have examined variations in admission rates for medical and surgical conditions together. Furthermore, in both countries, studies in which 'need' has been explicitly taken into account in some way in the analysis are relatively uncommon. Since 'need' may be confounded with 'health system' factors, such as supply of hospital beds, it is difficult to tease out their respective influence on the variations.

(c) Contrasting the findings for medical and surgical admissions

One advantage of using routine utilisation data rather than survey data is that analysis of admission rates for medical and surgical conditions, and a range of specific clinical conditions is more likely to be possible. This is important because the extent, and direction, of factors influencing hospitalisation may be different according to the types of conditions studied.

As previously noted, small area studies in general show that hospitalisation rates vary more

for medical conditions than surgical (72)(73)(111) and that across a variety of locations studied there has been a pattern of conditions showing high or low variation. Low variation conditions include appendicitis and myocardial infarction (73) and are typically conditions requiring mandatory and immediate treatment (such as those included in the list of marker conditions identified by Billings (1)). High variation conditions include many medical conditions (subsequently identified by other researchers as ACS conditions) such as asthma and diabetes as well as some surgical, typically discretionary, procedures such as hysterectomy (72)(73). Furthermore there appears to be consistency in the pattern of admission rates across locations over time. For surgical procedures this pattern has been dubbed a 'surgical signature' (112), leading to a hypothesis that unexplained variation might be due to the distinctive clinical practice styles of different physicians.

For specific clinical conditions, in the US, small area studies have found that hospitalisation rates for lower income and certain ethnic groups (black and hispanic) are *lower* for some conditions (for example for so-called 'discretionary' procedures for coronary heart disease such as coronary artery bypass grafting, hip/joint replacement (34)(113)(114)), and *higher* for others (medical admissions such as diabetes and its complications (1)(115)(116)) and avoidable injury (117)). Observational studies have also demonstrated higher rates of hospitalisation of lower-income people for asthma (118)(119) and diabetes (120), many of which were classified by Billings and others as ACS. While these studies have not tended to take formally into account differences in underlying levels of disease between income groups, it has been noted in a few studies that variations in admission rates are many times higher than variations in prevalence of the condition (identified from survey data) across income groups (1) (121).

In contrast much less variation by socioeconomic group has been observed for conditions requiring mandatory and immediate treatment such as fractured femur and appendicitis (1)(118).

These findings are confirmed by small area studies of hospitalisation for a range of conditions defined as ambulatory care sensitive (ACS) and 'markers'. In particular in the US a strongly *negative* association has been found between hospitalisation for ACS

conditions and income, having health insurance cover, and non-white ethnicity respectively (1)(22)(64)(67)(122) (123). Researchers have concluded that this may be a result of significant access barriers to 'upstream' care - ambulatory care - for low-income and non-white populations in the US. For example, Billings *et al* found that the overall average ratio of standardised admission rates for ACS conditions (in those aged under 65) between high and low income areas was 1: 4.65, the greatest differences (ratio of 1:7 or 1:6) related to admissions for congestive heart failure, hypertension and asthma, and even the lowest ratio was 1:3 (admission for the diagnosis of angina) (34).

In the UK, while an overall positive relationship between hospitalisation and socioeconomic deprivation has been found (73)(99), a negative association has been found for some discretionary procedures (such as coronary artery bypass grafting, hernia, gallstones and dilatation and curettage), although most of the relevant studies have been unable to adjust adequately for other need factors (24)(104)(124).

Variations in hospitalisation rates for a range of mainly medical conditions (identified by others as ACS) have been only recently investigated as such in the UK. Giuffrida and colleagues analysed hospitalisation rates (inpatients and day cases) for several conditions - asthma, diabetes and epilepsy (classified by others as ACS) – across 'small' areas covered by the former NHS Family Health Services Authorities (approximately 300, 000 population) in England, for the period 1989/90 to 1994/5 (69). They found, in a univariate analysis, that there was a significant (p < 0.01) strong correlation with socioeconomic deprivation (correlation coefficients between 0.69-0.78), and a similarly significant (p < 0.01) strong correlation with supply factors (supply of hospital and primary care facilities) (0.90-0.98). Because of overlapping (confounding) effects between these two variables, in a multivariate analysis the correlation coefficients for each dropped to approximately 0.50. Furthermore they found that socioeconomic factors had a similar sized impact on hospitalisation rates to age and sex.

2.5.4 The factors that have been found to influence access to, and use of, health care.

In the paragraphs below, rather than discuss an exhaustive list of factors that have been shown to influence access and use, only findings for the main factors are summarised.

(a) age and sex

Most studies in the UK and US investigating variations in hospitalisation have compared rates that have been standardised for age and sex, as both those factors have long been shown to have an important influence on the use of inpatient, outpatient, and primary care. For example in both countries hospitalisation rates increase markedly for patients over age 60, and females generally use more health care than males (125). The advantage of standardising for age and sex is that it may be easier then to identify other factors influencing variations in use of health care. The disadvantage is that effects of interest specific to different age groups may be hidden.

For an example of the influence of age, as noted in the introduction, Billings found large differences in hospitalisation rates for a range of ACS conditions between residents of high and low income areas for people in age groups 25-54 years, but much smaller differences in those over 65 years and under 18 years (1)(2). While the rates were not adjusted formally for morbidity, the researchers suggested that these findings may reflect differences in health insurance coverage, and therefore access to care, since children and the elderly are more likely to be covered by public health insurance.

Different age groups may also experience differences in the incidence, prevalence and severity of disease (as has been shown for asthma and wheezing illness (126)(127)), or differences in socioeconomic deprivation (128) or preferences for different types of care.

With regard to gender effects, a number of studies suggest that access to certain types of care by women may be lower than for men, and conclude that this is evidence of gender bias in the behaviour of health care professionals. But in a review of the international literature on the use of *specialist* health care by gender, Raine found that of five major care

areas – coronary artery disease, renal transplantation, human immunodeficiency virus and acquired immune deficiency syndrome, mental illness, and other (mainly invasive) procedures – the evidence for gender inequity was mixed and weak (129). Most of the research has focused on coronary artery disease, and no clear gender inequity in investigation and treatment was apparent. However, similar investigation into gender bias in hospitalisations for ambulatory care sensitive conditions has not been conducted.

Nevertheless, the literature suggests that age and sex are important factors influencing use of care, and there may be factors specific to different age and gender groups that influence access to care.

## (b) Indicators of need

Three types of 'need' indicator were considered earlier, and all three have been found to be positively associated with hospitalisations for medical and surgical conditions. In the paragraphs below, the focus is on the evidence of the impact of one of the three indicators, socioeconomic deprivation, on access to and use of care. Socioeconomic deprivation is examined more closely here because of the central hypothesis in this thesis that higher hospitalisation rates for ACS conditions would be found in more socioeconomically deprived areas even after adjusting for other 'need' factors.

### (i) Socioeconomic deprivation

#### Quantitative studies using aggregated datasets

Concerning *inpatient care*, in the UK many researchers using survey data have found that, after adjusting for self-reported health status, either there is no consistent evidence of inequity in the utilisation of NHS inpatient care between people in different socioeconomic circumstances, or, if anything, there is a 'pro-poor' bias (92)(130).

It was also noted earlier that researchers conducting small area analyses have found that socioeconomic deprivation is strongly and positively correlated with hospitalisation rates

(72)(99), although researchers have not been able to adjust adequately for other need factors such as prevalence or severity of disease.

In the US, discussed earlier, the positive influence of socioeconomic deprivation on hospitalisation rates can be reversed for specific types of admissions such as 'discretionary' surgeries like coronary artery bypass grafting. In the UK there has also been a focus in investigating hospitalisations by small area for specific conditions and procedures, although such work has tended to be small-scale. Care for patients with coronary artery disease has been the single most investigated topic. In a survey of 180 GP practices in Nottinghamshire between 1993 and 1997, Hippisley-Cox found that rates of hospital admission by practice for coronary angiography and revascularisation techniques were lower, and the length of time waiting for admission higher, in practices serving more deprived populations, with presumably greater need (131). In Cheshire, admission rates for cardiothoracic surgery (and for hip and knee replacements) were again found to be lower in less affluent areas (132).

As noted earlier, in Sheffield admission rates for coronary artery revascularisation procedures were lower in more deprived areas after adjusting for need – half the number of revascularisations per head of population with angina (23). In contrast, in Northern and Yorkshire NHS region higher rates of admission for coronary artery bypass grafting (CABG) and percutaneous transluminal arterioplasty (PCTA) were found in more deprived areas, but the authors concluded that the rates were not as high as the underlying standardised mortality ratio (the need indicator selected) for coronary heart disease would suggest it should be (133). Kee *et al* analysed 24 179 admissions of patients with a diagnosis of ischaemic heart disease. Rates of cardiac catheterisation and angiography were 85.5 per 100 000 in non-deprived males compared to 123 per 100 000 in deprived males. However after adjusting for potential confounding factors (such as age), the difference was not found to be statistically significant (134).

As referred to above, Giuffrida and colleagues analysed hospitalisation rates for asthma, diabetes and epilepsy finding that socioeconomic factors had a similar sized impact on hospitalisation rates to age and sex (69).

Concerning **hospital outpatient care**, analysis of General Household Survey data in the UK again suggests, after adjustment for self-reported health status, no systematic inequity in the use of outpatient care by people of different socioeconomic status (92). Small area analysis of use rates for outpatient care using actual utilisation data has generally not been performed in the UK because of a lack of adequate data. With regard to other forms of ambulatory care, such as district nursing and health visiting, and hospital casualty departments, the significant positive influence of socioeconomic deprivation and morbidity has also been demonstrated in the UK using small area analysis (135)(136).

Concerning **primary care**, while the UK NHS offers primary care which is free at the point of use, there is still a cost to the patient (for example travel costs, time off work) which may have a greater deterrent effect in socioeconomically deprived groups. Data from the General Household Survey suggests that people who are socioeconomically deprived tend to consult more often than those who are not, but again (and this is still debated) these differences disappear once there has been adjustment for need in the analysis (92).

Research in the UK on variations in GP consultation rates has tended to focus on variations between population groups (typically categorised by deprivation) within or between general practices rather than small geographical areas. The evidence is more straightforward for some forms of primary preventive care - care for which, in theory, the need in the targeted populations is equal, whether deprived or not. For example, data on the uptake of screening for cervical cancer and childhood immunisations are available by practice, and rates are consistently lower among socioeconomically deprived populations (137)(138)(139)(140)(141).

As noted earlier, research using the MSGP-4 dataset shows a positive relationship between GP consultation rates and socioeconomic deprivation (82). More deprived groups may be consulting for more serious conditions, as suggested by Saxena, who showed that while consultation rates for children were higher in more deprived socioeconomic groups (without adjusting in any way for other need variables), these consultations were more likely to be for intermediate and more serious categories of illness, and less likely for preventive care than less deprived groups (109).

#### More focused studies

Many observational and qualitative studies have examined in more detail the influence of socioeconomic deprivation on use of health services (38)(142). In the US poverty is a barrier to patients seeking care and a lack of insurance (or having public insurance) deters providers from offering care (143)(144). As a result the low-income and uninsured tend not to have a regular source of care, receive less preventive care, delay seeking care, use less ambulatory care (uninsured persons versus insured), experience poorer health outcomes and use proportionately more A&E and inpatient care for certain conditions (143) (145)(146)(147)(148). However, doctors may lower the clinical threshold of admission for patients in response to non-clinical factors, such as poverty, perceived non- compliance, and lack of social support (149)(150). However vivid these findings are, it has generally not been possible to take them into account in small area analyses using area-aggregated data because information on these factors is not collected routinely either through the census or in health services (74)(142).

In the UK there has also been a wealth of small scale studies analysing the effect of socioeconomic factors on access to, and use of, health care. These studies show persistent barriers to access resulting from socioeconomic deprivation (151)(152) which can result in untimely and inadequate treatment. For example, the supply and quality of primary care in deprived areas in inner cities, particularly London, has long been known to be worse than in other areas (153)(154). The National Survey of NHS Patients (General Practice) in 1998 (155) showed that a significantly higher percentage of patients in more socioeconomically deprived health authorities put off a visit to the GP because of inconvenient hours. Another study showed that patients from deprived areas were over twice as likely to perceive problems in accessing daytime GP services (156).

There is a large literature on the effect of socioeconomic status of patients on the GP-patient interaction, which may affect access to subsequent inpatient care. In the UK, the more socioeconomically deprived patients have been shown to be given shorter consultations, less advice, more prescriptions, and have a lower likelihood of receiving, or being referred for, a diagnostic test (80), although there is mixed evidence on follow-up visits and

subsequent inpatient hospitalisation. Scott suggested that there were three mechanisms by which socioeconomic status could influence GP consultations independent of health status: social distance between GP and patient; health knowledge and beliefs of the patient; and the extent that professional power of the physician can be moderated by patients. A study of 500 consecutive patients in 4 practices in Bedfordshire and Hertfordshire (areas included in the analysis described in chapter 6), reported that general practitioners listened less, examined less, and gave less advice to patients in Social Classes V and IV (manual occupational groups) compared to those in Social Class I and II (professional and managerial occupational groups) (157). Patients in more socioeconomically deprived groups have been shown to experience less continuity of primary care, are less compliant with treatment, and have a poorer quality interaction with professionals (158).

Socioeconomic status may also be associated with different *preferences* of the patient for care, which in turn influence whether or not a patient will be referred (or will refer him/herself) to hospital, and be subsequently hospitalised. For example, Coulter *et al* showed that patients with menstrual disorders who had lower educational achievement tended to prefer surgical treatment (as opposed to more conservative treatment) and that the likelihood of GP referral was related to patient preferences for surgery (159). Also in the UK, Armstrong showed that perceived pressure from patients on the GP to refer led to increased referrals (160). Such pressure is likely be associated with better informed and articulate patients with higher socioeconomic status. It may be that more socioeconomically deprived patients prefer to go to a hospital casualty department for care (given difficulties in accessing primary care), and once there clinicians have been shown to be more inclined to admit socio-economically deprived patients than others with similar illnesses (161).

The influence of socioeconomic deprivation on *GP referral to hospital*, and the way it is mediated, is obviously complex. There are such a large number of factors influencing the decision to refer (81) that teasing out the independent effect of deprivation of the patient is difficult, in particular from the effect of morbidity. Furthermore many studies examining variations in GP referrals to hospital have examined *all* referrals rather than referrals by specialty or by condition. It is possible that socioeconomic deprivation may have a stronger impact on the decision to refer for some conditions than others, analagous to the hypothesis

in this thesis that deprivation might have more effect on hospitalisation for ACS than marker conditions. For example, a cross sectional survey of new referrals in Nottinghamshire in 1993 found a strong positive (r-squared = 32%) and significant (p<0.0001) association between socioeconomic deprivation (as measured using Jarman scores (17)) and medical referrals, but a much smaller effect on surgical referrals (r-squared - 2.3%, p<0.04) (131).

In summary, analysis of information from large-scale surveys suggests that, after adjustment for self-reported health status, inequity in the use of health care between socioeconomic groups has not been consistently demonstrated in the UK NHS. However small area analysis of variations in use for specific conditions (mainly investigations and treatment for coronary heart disease) suggests otherwise. Qualitative research also points to significant barriers to entering the health system, and 'in-system' access, due to a number of complex reasons relating to the individuals seeking care, the communities they live in, and the health system and professionals providing care.

## (c) Ethnicity

The Department of Health in England defines an ethnic group as 'a group of people who share characteristics such as language, history, culture, upbringing, religion, nationality, geographical and ancestral origins and place' (162). There is much debate as to what ethnicity actually is, and clearly the definition of an ethnic group, for an individual or for a whole group, may be inconsistent or may change over time. In Britain, a question in the 1991 Census asked respondents to record their ethnic group as being either: white; Black-Caribbean; Black African; Black-Other; Indian; Pakistani; Bangladeshi;Chinese; or Other. On these definitions, the results are shown in the table below.

Ethnic Group	Proportion of total population (%)	Proportion born in Britain (%)
White	94.5	95
Black – Caribbean	0.9	54
Black- African	0.4	36
Black - other	0.3	84
Indian	1.5	42
Pakistani	0.9	50
Bangladeshi	0.3	37
Chinese	0.3	28
Other – Asian	0.4	} 44
Other – non Asian	0.5	}

### Table 6 Resident Population by Ethnic Group, Great Britain, 1991

Source: 1991 Census, Local Base Statistics, OPCS Crown Copyright

Ethnicity may important to consider in the context of this thesis. A significant proportion of the population in the area covered (the former North West Thames NHS region) is made up of non-white ethnic groups, and ethnicity may have a key influence on access to, and use of, hospital care. Also the prevalence and severity of illness for ambulatory care sensitive conditions, such as diabetes, may be higher.

However the influence of ethnic group on use of health care is heavily confounded by socioeconomic status and the other 'need' factors discussed above, and most studies have not adequately separated out these effects.

There has been extensive investigation in the US as to the impact of race on variations in hospitalisation rates. Analysis using survey data (eg from the National Health Interview Survey) suggests that non-white ethnicity is associated with an overall lower use of all types of health services (ambulatory and inpatient) after adjusting for differences in health status. Analysis using routinely collected health service data suggests a negative association between non-white ethnicity and hospitalisation for discretionary procedures (for example investigations and treatment for coronary heart disease), and a positive association with ACS conditions.

In the UK most analyses have used data from population surveys, in particular the General Household Survey (GHS) or the 4<sup>th</sup> National Survey of Ethnic Minorities (SEM), from which self-reported information on ethnicity, income and health, with utilisation of health care can be linked. Despite the fact that acute hospitals have been required by the Department of Health to record ethnic group routinely since 1995, data on ethnicity on all types of routine NHS data are too poor to use in small area analyses.

Concerning *inpatient care*, using the GHS or SEM, several researchers have found no consistent difference in the rates of inpatient hospital utilisation between ethnic groups after adjusting for self-reported health status (76)(163)(164). The main exception was the Chinese population, in whom lower rates of utilisation of all health services have been consistently found.

Small area analysis studies have largely had to rely on area-aggregated data from the census. In the extensive analysis by the University of York, referred to earlier (99), researchers analysed the effect on acute hospital admissions (medical and surgical) of a variable measuring the proportion of the population <u>not</u> in black ethnic groups. In a model which included a range of 'individual', 'community' and 'health system' factors (including 'need factors relating to socioeconomic deprivation and SMR), the results suggested that areas with higher proportions of black residents had *lower* use of inpatient care than expected. The study team interpreted this finding as reflecting supply not need, suggesting that electoral wards with large ethnic minority populations tend to be located close to acute hospitals. A more likely explanation may be that black residents may be under-using inpatient care relative to their needs. There have been no small area studies of ACS admissions in the UK that have examined the influence of ethnicity.

Most smaller scale observational studies analysing use rates between different ethnic groups

have not taken into account important confounding factors other than age and sex, for example socioeconomic deprivation or underlying prevalence of disease. It is therefore extremely difficult to assess the impact of 'ethnicity' *per se* on variations in hospitalisation rates. A recent review of the literature on ethnicity and use of NHS care for patients with coronary heart disease concluded that the evidence was so weak that it was impossible to draw conclusions about the equity of use between ethnic groups (165). But the few studies of this type that have adjusted for 'need' factors in some way suggest that there is inequity in access to care. For example, Shaukat *et al* showed that the time between onset of symptoms and being seen in outpatients for patients with suspected coronary artery disease was at least twice as long for Indian patients, having adjusted for the severity of disease subsequently found (166). Lear *et al* showed that following myocardial infarction, fewer Asians than Europeans were referred for diagnostic tests(167).

Concerning *outpatient care*, Smaje & Le Grand found that people in minority ethnic groups in the age group 0-44 years were *less* likely to report an outpatient attendance (164), although the picture was reversed for older members of the Black-Caribbean and Black-African population, who reported higher rates.

Concerning *primary care*, in contrast consultation rates in general practice have been found to be higher among ethnic populations, after adjusting for self reported health status, especially amongst the Asian and Caribbean population (164). The exception was the Chinese, young Pakistani females, and Africans, in whom lower rates have been reported.

Analysis of the MSGP-4 dataset appears to confirm the findings from survey data, with GP consultation rates higher in Caribbeans and Asians, and lower in the Chinese and African population after socioeconomic factors have been taken into account (82). Other surveys, that have not taken health status into account, suggest significant barriers to care for people from different ethnic groups. For example, in the National Patients Survey of General Practice in 1998, non-white people were less likely to report that they understood their GP, that they were given enough information and that their opinions had not been taken seriously by the GP, and were more likely to wait longer before being seen in outpatients (155).

Studies examining uptake of preventive care in primary care, for example screening for breast and cervical cancer, suggest that lower uptake rates are associated more with socioeconomic deprivation than ethnicity (10). In a review of a small number of studies on childhood immunisation, Smaje suggested that uptake has generally been found to be *higher* among minority ethnic groups, especially South Asian people compared to the white population (168).

To conclude, it is unclear from studies using aggregated survey data, such as the GHS or SEM, whether there is systematic inequity in use of NHS hospital care for ethnic populations. However, smaller scale quantitative studies on utilisation for specific conditions (typically coronary heart disease) suggest otherwise. There has been no relevant work analysing the impact on ethnicity on variations in hospitalisation for ACS conditions.

### (d) Patient behaviour and lifestyle

The influence of patient lifestyle on hospitalisation has been examined mostly in studies of individual admissions rather than studies using aggregate data, and has been found to be important. For example drug or alcohol misuse has been found to contribute to 17% of admissions which could have been avoided, and lack of patient compliance with treatment has been estimated to lead to between 4% and 20% of admissions (169)(65)(170) and 11% of readmissions (171). Compliance has been shown to be influenced by a number of factors including, age, social class, the level of patient knowledge and beliefs about the illness and treatment, and the level of environmental support (172).

The influence of lifestyle-related factors on hospitalisation for ACS-type conditions has been estimated in ecological studies, by investigating the proportion of admissions in which substance misuse has been recorded on hospital discharge data. Billings showed that the proportion was almost 12% in New York City and there was a strong positive correlation between substance misuse and socioeconomic deprivation (1). The extent to which health care-seeking behaviour could be a factor explaining variations in hospitalisation rates for ACS-type conditions has also been included in one multivariate analysis using data from a specially designed community survey, but was found to have no significant influence (22). No similar work has been conducted in the UK.

## (e) Perceived access

Bindman *et al* in California sought to measure the *perceived* access to care of populations living in the geographical area under scrutiny (22). Residents of these areas were asked to rate, on a 5-point scale, their difficulty in receiving medical care when needed. Perceived access to ambulatory care was the single most important variable influencing variations in rates between zip code clusters - explaining over 50% of the variation. In a multivariate model, perceived access and prevalence were independently significant predictors of hospitalisation. The authors concluded that hospitalisation rates for ACS conditions were a suitable indicator of perceived access to ambulatory care; however this could be a circular argument.

In the UK, while ad hoc surveys show how various population groups perceive their access to care (155), few researchers, if any, have linked this information meaningfully with measurement of actual use of inpatient care in a way similar to the work in the US by Bindman *et al* described above.

### (f) 'Health System' factors

The influence of four factors upon hospitalisation and ambulatory care is discussed briefly here: the hospital itself; supply of health services; organisation of care; and physician practice style.

#### (i) The hospital itself

Many studies have examined the impact of specific health system variables (discussed below) on variation in hospital utilisation rates across small areas. But few have examined the variation in small areas within and between hospital 'catchment' areas to identify the strength or otherwise of such a 'hospital effect'. Hospital catchment areas are sometimes referred to in the literature as 'hospital market areas' or 'hospital service areas', and have various definitions but generally mean areas in which the residents go mainly for their care to the particular provider in question. One important study, by Tedeschi and colleagues in the US, found that hospital utilisation rates in small areas (zip codes) within a hospital market area were more similar to each other than rates in small areas in other hospital market areas (173). After adjusting for income and age, the hospital market area explained 35% of the variance in surgical use rates and 39% in medical admission rates. The factors causing the apparently strong hospital effect were not discussed, although could clearly include a combination of those discussed below, as well as, for example, differences in the diagnostic coding of routine data between providers (see chapter 5). Tedeschi and colleagues concluded that the 'hospital effect' was important to take into account in further small area studies. How researchers have defined and identified 'hospital service areas' is discussed below.

#### (ii) Levels of supply and access to supply

Many cross-sectional ecological studies in US and UK have compared hospitalisation rates between hospital catchment areas because the supply of health system facilities can more easily be calculated than in geographical areas defined for other reasons (123)(174). Supply has usually been measured in terms of facilities per capita in an catchment area, such as the number of hospital beds or number of physicians. More sophisticated work has taken into account not only 'supply' but also distance from a facility (and neighbouring facilities) and the size of the population 'competing' for access to those facilities; in the UK these factors have been constructed as one variable which has been termed 'GP access' factor or 'hospital access factor' (99)(175), discussed below.

## Hospital care

The supply of hospital facilities (such as beds, or doctors) in an area has consistently been found to have had an important positive influence on hospitalisation, and the association has been found more consistently for surgical procedures than medical admissions (72)(73)(99) (176)(177)(178)(179)(180)(181). For example, in a large cross sectional ecological study across 3 selected NHS regions in England and 4 NHS Health Boards in Scotland, the reasons for variation in rates between these areas (consisting of populations of approximately 250,000) for coronary revascularisation procedures in NHS and private providers were investigated(177). The researchers concluded that the variations were unrelated to need (measured as SMR per district for coronary heart disease) but were mostly due to the supply of hospital facilities available to carry out the procedures: the higher use rates observed for deprived communities were related to the proximity of residence to the provider carrying out the procedures.

There is a well documented inverse relationship between utilisation and distance to hospital facilities (99)(182), and with the size of the population 'competing' for the hospital resources. Researchers at the University of York have calculated a 'hospital access factor' for each small area in England, based on a formula that combined supply (the number of hospital beds in 5 specialties), distance to facilities from each small area, and the size of the populations competing for that supply in neighbouring small areas (99). In the multivariate analysis, the 'hospital access factor' was positively associated with hospitalisation rates, although the effect was weaker than for most of the 'needs' variables investigated.

#### Ambulatory care

The supply of ambulatory care facilities has received much less investigation, and most of the research has focused on primary care. In the US a number of studies have shown that the supply of primary care physicians or facilities, or outpatient facilities had a negative effect on overall hospitalisation rates (183)(184). Similarly in Canada, an overall negative association between the supply and access to primary care (general practice) facilities and hospitalisation has been found (185). In three studies of ACS-type conditions, two also found a negative relationship between the supply of primary care physicians and admission rates (66)(123). However one of these found that, in a multivariate analysis, the negative association became weak and the positive association of supply of hospital beds did not reach statistical significance (123). The third study found a negligible effect between ACS admission rates and either the overall supply of physicians, or the proportions of specialists available (186). However this study was conducted in the Medicare population who, as a whole, enjoy significantly better access to health benefits than the rest of the US population.

In the UK, a recent large study by Jarman and colleagues, using data on 8 million discharges in England from 1991/2 to 1994/5, found a negative correlation between emergency admissions and the supply of general practitioners (r=-0.35, p<0.001)(187). But while very few large ecological analyses have examined the effect of supply of ambulatory care on hospital utilisation, observational studies have shown that inappropriate admissions can result from a lack of supply of ambulatory care facilities (188)(189).

In the study referred to earlier, researchers at the University of York investigated the association between hospitalisation rates and the supply of, and distance from, general practitioners and the size of the population competing for those facilities (99). This way the researchers attempted to measure access to general practitioner care, not just supply. The construction of the variable was similar to that used to construct the 'hospital access factor', discussed further in chapter 6. The York team put forward two alternative hypotheses as to the effect of the 'GP access factor':

- that there would be a *positive* relation between access to general practitioners and hospitalisation rates because, for example, GPs may be more likely to refer patients to hospital; or
- that there would be a *negative* relation since GP or other primary care may substitute for hospital care.

The team found an overall *positive* relationship after adjusting for need using standardised mortality ratios, standardised illness ratio and measures of socioeconomic deprivation.

However, they did not distinguish in their analysis between admission rates for surgical and medical conditions.

Other small area studies have demonstrated the effect of 'health system' factors on admission rates for ACS-type conditions. For example a study of admission rates for asthma (an ambulatory care sensitive condition) by general practice in a deprived part of East London found that a 'health system' factor (whether a practice had one or more GPs) was the strongest factor influencing admission rates (190): rates in practices with one GP were 1.7 times higher than those in practices with 3 GPs or more, and those in practices with two GPs were 1.3 times more. However this, like other studies, made no attempt to adjust adequately for need, making conclusions difficult to draw.

Not surprisingly the *supply* of ambulatory care has been found to be positively associated with *use* of ambulatory care (66) (91) and longer distance to ambulatory care facilities and longer waiting times have been found to deter use (110)(141)(191)(192)(193)(194)(195). Distance has been shown to have a greater deterrent effect on use of general practice care by women (192), the elderly and those in lower social classes even within same health status groups, although equal rates of referral and hospitalisation have been found after initial GP consultation (193).

Other work in the UK has investigated the relation between supply of primary care and need - that is, the relationship between variables within column (a) in figure 2. Since 1948 there have been strict controls over where general practices (the focus of much primary care activity) can be located, and the number of patients who can be registered with each practice. Despite efforts over the last 30 years, the so-called 'inverse care law' persists, that is to say that there are fewer GPs and attached practice staff and worse primary care facilities in socially deprived areas, in particular inner cities and in particular in London (107)(196)(197)(198). A recent national survey showed that the population in London, in particular people from black and ethnic minority groups, felt that they had greater difficulty accessing care from their GP than residents from other parts of England (155). The direct relation between lack of supply of primary care and use of ambulatory or hospitalisation rates in these areas has yet to be demonstrated.

## (iii) Financing and organisation of care

#### Direct financial incentives

In the US, different arrangements and incentives for payment and organisation of care have been shown to have important effects on rates of health care utilisation.

Concerning inpatient care, significantly lower rates of hospitalisation have been observed among patients enrolled in managed care organisations, compared to those with similar risks in traditional fee-for-service insurance plans (199). 'In-system' access to discretionary procedures by hospitalised patients has been shown to vary considerably by insurance status, after adjusting for need (200). Other 'system' factors influencing hospitalisation rates include the introduction of new technologies (201), audit or surveillance of clinical practice (202), and the introduction of financial incentives for providers such as prospective payment (199)(203)(204) or disincentives to patients such as co-payments (205)(206)(207). Fewer studies have analysed the outcomes of care as a result of different methods of financing and organisation of care, although one large study (the Medical Outcomes Study) indicated that elderly and poor chronically ill patients had worse physical health outcomes in health maintenance organisations than in fee-for-service systems of care (208).

In the UK there have been fewer ecological studies of the effects of altering financial incentives to providers or users upon hospitalisation rates, because until the 1991 NHS reforms there were few direct financial incentives operating on patients or providers which might influence the utilisation of inpatient care. For acute hospitals after 1991, which were paid on a block contract basis (or cost and volume), the incentives to increase or reduce the level of hospitalisations were weak compared to the US (209).

For general practitioners, in 1991 the GP fundholding scheme was introduced whereby GP practices could be given a budget with which to buy prescription drugs and a limited range of hospital care (outpatient care and mainly elective surgery). The incentives for reducing expenditure on levels of hospital and outpatient care were weak, as was the impact of such
incentives (210). In 1996 fundholding practices were given responsibility for purchasing medical admissions as well, although again the effect on admission rates was modest (209). Between 1991 and 1994 (the period covered by the analysis in this thesis) four main financial incentives operated for general practitioners relating to the way they were paid: *capitation payments* (which may have encouraged GPs to take on more patients up to a limit); *per-item fees* (for example for giving certain vaccinations); *target payments* (flat rate fees paid after a predetermined level of service has been reached) and *sessional fees* (for providing certain clinics eg promoting 'wellness' or managing chronic diseases such as asthma and diabetes). Of these, the incentives of the capitation payment (more payment for more patients) and sessional fees are likely to have most bearing on the care of patients with ACS conditions. On capitation payments, there is no strong evidence to suggest that GPs systematically increased their registered list of patients to attract higher income. Sessional fees have been shown to stimulate the provision of specific clinics (eg health promotion) in inverse proportion to the need of the local population (211), particularly in London (107).

### Organisation of care

There has been increasing recognition that the way that *ambulatory* care is organised (affecting access and the quality of care provided) can have an important influence on the risk of subsequent admission to hospital. For example, in the US, case studies that have investigated factors in ambulatory care which may have led to admission in an individual patient have identified: sub-optimal clinical management (including inadequate follow up in ambulatory care, delayed diagnosis, failures of communication of results and poor planning); adverse drug reactions (65)(169)(170)(212); and a lack of health care facilities available in the community (189) (171)(213)(214)(215).

Despite efforts on both sides of the Atlantic, there is still no agreement as to which indicators might best measure the quality of primary care (69)(216)(217), and a comprehensive set is not yet available in the UK at least. Instead researchers have examined specific ad hoc features relating to various aspects of the organisation of primary care.

For example, whether populations have a regular source of care has been investigated as a factor. It is known from observational studies that barriers to primary care (real or

perceived) result in greater use of a hospital emergency department, and therefore an increased risk of admission in the US (145)(146)(218)(219)(220) and UK (221)(222)(223)(224)(225) (226)(227). In the US, as noted above, people without a regular source of care tend to receive less ambulatory care (228), or preventive care (148); and lack of a regular source of care is associated with poverty, ethnic minority status, and greater use of A&E (and thus possibly greater likelihood of admission).

It is not clear how far having a regular 'facility' to go to, or a regular physician, affects the direct risk of subsequent hospitalisation. In the UK, almost all the population is registered with a general practitioner, but while patients may go to one practice for primary care, they may not receive care from the same clinician. The reasons for this partly relate to 'health system' factors (for example GPs in group practice may decide not to work with a personal list of patients (229)), and partly due to preferences of patients (158). However the link between the provision of continuity of care with a particular clinician, quality of care, access to and the use of ambulatory or inpatient hospital care remains unclear, despite the arguments for the beneficial effect on patient care (230)(231)(232).

More often observational or cohort studies have been used to investigate the effects of different forms of treatment provided in, or the organisation of, ambulatory care for specific conditions (labelled by others as ACS) using hospitalisation as an outcome measure. The aspects of quality of care that have been investigated include: levels of appropriate prescribing, organisation of care (for example sharing care with specialists, availability of a dietition for diabetic patients, having a practice-based protocol for care); and whether there has been clinical audit of care. Different forms of treatment or organisation of ambulatory care have been shown to reduce the need for admission in patients with asthma and diabetes (45)(233)(234)(235)(236)(237)(238)(239)(240)(241)(242)(243)(244)(245). For example Griffiths and colleagues in east London (239) found that higher admission rates for asthma by practice were associated most strongly with small size of practice and a high rates of night visiting, over and above differences in prescribing practice. However the prevalence of asthma (a measure of need) across practices was not taken into account in the analysis. Similarly a study of 318 patients with diabetes (another ACS condition) across 12 practices in Nottinghamshire found that more effective control of the disease was observed in

practices holding special clinics, with GPs having a special interest in diabetes, and access to dietetic support and better equipment (246). A recent systematic review of the international literature on the effect of organisation on asthma care, found a lack of good quality research in the area. However studies that used the following outcome measures attendances at accident and emergency departments, admissions to hospital, and readmissions – were found to result in a lower level of activity in these areas after a range of interventions in ambulatory care, for example using nurses to educate patients about asthma and to train patients more in the management of their disease and closer monitoring in primary care (247).

## (iv) Physician practice-style

The large amount of variation in hospitalisation rates found in small area studies which remained unexplained after taking account of many of the factors discussed above led to the hypothesis that this may be due to differences in physician practice-style affecting the threshold of admission, particularly for treatments over which physicians have more scope for discretion in their decision-making (90)(112)(248)(249). The influence of physician practice-style was also suggested by Wennberg's observation that variations in surgical admission rates across locations appeared to be stable over time, leading some to suggest that surgeons working in those locations left a 'surgical signature' because of their particular practice-style. Attention then focused on measuring variations in use of care between physicians, rather than between geographical areas or population groups as discussed above. Initially few of these studies took casemix into account (250), but those that did showed that a large amount of variation between physicians remained unexplained by health or sociodemographic characteristics of patients or supply of local health care facilities. This has been particularly well illustrated in Canada (251)(252).

The majority of studies in this area have focused on analysing reasons for differences in rates of surgical procedures. Regarding medical admissions - which are more likely to be for ACS conditions - Connell *et al* examined patterns of hospitalisation for patients with one ACS-type condition (diabetes) and found that in higher rate areas, patients were

admitted with less severe disease (253). This finding suggested that higher rate areas may be served by hospitals where physicians (referring or admitting) have a lower threshold for admission, rather than where patients necessarily have poor access to primary care.

But this finding has been contradicted in a few of the studies analysing hospitalisation for specifically ACS conditions. Two studies of avoidable hospitalisation across small areas in California surveyed physicians to ascertain their *propensity to admit* patients (22)(123). The physicians were asked to comment on hypothetical clinical vignettes to investigate how far clinical criteria (such as severity of disease) and also social criteria (such as being homeless, known to be non-compliant with treatment, being uninsured, and having no regular doctor) would influence their decision to admit a patient. No association was found between hospitalisation rates for ACS conditions and the propensity of physicians to admit. The team concluded that attempts to reduce hospitalisation may be better focused on improving access to care and reducing social inequalities rather than attempts to change physician practice-style. Two other studies of ACS admissions assessed the clinical severity of patients on admission (measured using clinical codes recorded on routinely collected discharge data) to assess the 'clinical threshold' of admission of the admitting physicians. But using this necessarily crude method, severity of disease or physician practice-style were not found to have a significant effect (1)(22).

The gender of the doctor may have important influence on practice style and subsequent hospitalisation although the evidence is mixed. For example although Coulter showed no evidence of an effect on the treatment of menorrhagia by GPs (254), others have shown that female GPs tended to refer women less often to hospital with gynaecological complaints and had higher uptake rates of cervical cancer screening among their patients (255).

The age and experience of physicians (referring or admitting) may also be important. For example, in Australia higher hospitalisation rates for diabetes were associated with patients from general practitioners of younger age with smaller caseloads of diabetic patients (181), suggesting that the level of clinical experience, or toleration of risk, of the referring doctor may be a significant factor.

How the referring or admitting physician reacts to predisposing or enabling factors (referred to earlier) in patients is also likely to be relevant. For example in the US, insurance status of the patient has been shown to influence clinical decision-making by ambulatory care physicians; fewer services are selected for uninsured patients than insured (256). In the UK, children from socially deprived backgrounds were more likely to be admitted from accident and emergency departments than those with similar illnesses from non-deprived families (161).

In general the *referral behaviour of general practitioners* in the UK and its impact on subsequent hospitalisation has not been well researched. One study found that while there was a positive relationship between the level of GP referral rates to hospital and surgical admissions, the relationship between referrals and medical admissions (which are more likely to include ACS conditions) was not significant (257).

## 2.5.5 Summary

Large scale population surveys, and ecological studies using routine hospital data, have both demonstrated the significant influence on hospitalisation rates of 'individual' 'community' and 'health system' factors.

Many small area studies have been unable to take account of need adequately in the analyses owing to the paucity of data on the levels of ill health, particularly for specific diseases, and are subject to problems of ecologic fallacy. However, 'need' factors (such as socioeconomic deprivation, limiting long term illness, and mortality and in the US income and race) have been shown to exert powerful effects on hospitalisation. Of particular note is that socioeconomic deprivation appears to have different effects according to the conditions for which individuals are hospitalised – for example deprived populations tend to experience more hospitalisation for ACS conditions, and less for discretionary procedures, such as for coronary heart disease.

'Health system' factors (especially the supply of hospital facilities) have been shown to be

related to hospitalisation rates most consistently, particularly for surgical admissions. In the US, the payer of health care has also significant influence. However, since most studies investigating the effect of 'health system' factors have not made an adjustment for need in the analysis, the effect of supply variables is likely to be over-estimated. This may be so even if there is an adjustment for need, as need and supply are commonly counfounded. For example, more socioeconomically deprived, ethnic and older, populations tend to live in inner cities in areas where the supply of inpatient facilities is likely to be higher.

The influence of supply of, or access to, ambulatory care on the subsequent use of hospital care has been investigated much less - most studies have been carried out in the US. There, several studies have found a negative association between supply of primary care physicians and hospitalisation rates, especially for ACS conditions. In the UK, a positive relation has been found for all admissions (not distinguishing between medical and surgical) (99).

There have been fewer ecological studies analysing the impact of varying factors upon use of ambulatory care because fewer data are available for analysis. Primary care has been investigated more than other forms of ambulatory care. Multivariate analyses of variations in use of primary care suggest that 'individual' factors have more explanatory power than 'community' or 'system' factors.

More qualitative studies have demonstrated the complex array of factors that may influence both whether an individual seeks care, and how the individual is treated within the health care system including access to further health care. These factors relate to structural aspects of the health care system (such as physical location of facilities, and the requirement of copayments), and characteristics of physicians, patients as well as the condition treated. This complexity is likely to mean that what can be gained from the results of ecological studies is limited, although the results of these studies may raise useful questions for further indepth analyses.

# 2.6 Implications of literature for the design of this study

Section	outline	
2.6.1	Genera	al comments
2.6.2	Specifi	c methodological issues
(a)	Why a	small area analysis?
(b)	Size of	small area
(c)	Identify	ying random variation
(d)	Туре о	f analysis
(e)	Selectio	on of variables
	(i)	Dependent variable
		Identifying ACS and marker conditions
		Accuracy of coding
	(ii)	Independent variables
		Need
		Age and sex
		Health system factors
		Ethnicity
		Other factors

#### 2.6.1 General comments

One important message from this review is that access to and use of health care is influenced by a range of factors, many of which are interrelated in a complex way. In particular, access and use are influenced by the 'need' for care, and it is important to take account of 'need' in any investigation of the equity of access to ambulatory care.

The literature also shows that, while a great deal of research has been conducted, there are significant gaps. In particular there has been no published study in the UK in which variations in admission rates for a range of ACS conditions have been analysed.

In the proposed investigation, equity of access to ambulatory care will be investigated. The definition of equity to be used will be equity of access for equal need, and hospitalisation

rates for ACS conditions will be used as a negative indicator of access to timely and effective ambulatory care. The form of ambulatory care focused on in this study will be that provided in primary care rather than in other settings (for example hospital outpatient or accident and emergency departments, or minor injuries clinics). The reason for this decision was partly to keep the investigation relatively simple in the first instance, and partly because for the large majority of people in the UK the first point of contact for health care is their general practice, probably with a GP. Variations in hospitalisation rates for ACS conditions will be compared to variations observed for 'marker' (non-discretionary) conditions - as examined in the US by Billings.

Of all the possible factors which might influence variations in hospitalisation rates, it was thought important to analyse the impact of eight of those discussed above: standardised illness ratio; standardised mortality ratio; socioeconomic deprivation; age; sex; access/supply of hospital facilities; access/supply of primary care facilities; and 'hospital effect'. It was thought that this would be a useful start, and more than enough work for one thesis. Chapter 6 describes the investigation.

To gain further understanding of why admission rates for ACS conditions were high in particular locations, qualitative studies, rather than large scale multivariate analyses using routine data, would also be required but were beyond the scope of this thesis.

## 2.6.2 Specific methodological issues

#### (a) Why a small area analysis?

A small area analysis approach has been favoured by researchers investigating variations in admission rates for ACS and marker conditions. The term 'small area analysis' essentially indicates that the factor of interest (admissions) has been calculated and compared across small geographical areas, and reasons for variation between small areas investigated. In most studies of this kind, there have been particular reasons for choosing an analysis by area, rather than, for example, by population group. These include the following: - spatial distance from a person's home to health care facilities may be an important factor to take into consideration in the analysis. These data are not available for individuals, but are available aggregated by small area;

- routine data are available to document the characteristics of the area (eg from the census) and include in a multivariate analysis;

- identifying high or low admission rates by area on a map may be a useful way to interest policy-makers in further investigation, or in the design of interventions, focused on areas of interest; and

- information on 'health system' characteristics (eg numbers of hospital beds) and 'need' (such as mortality rates) are available by area, and not specific to individuals.

All of these were relevant to the investigation in this thesis, so a small area approach will be used.

The main disadvantages of small area analyses are that:

- attribution of the characteristics of an area to an individual (in this case one who is admitted), which may not be appropriate (the problem of 'ecologic fallacy');
- there is little information on health status available at small -area level, making it difficult to make an adequate adjustment for need in the analysis;
- the use of aggregated datasets necessarily means a broad-brush enquiry. More focussed and detailed studies will be needed to understand a fuller picture of, for example, the relationship between admission for an ambulatory care sensitive condition, socioeconomic deprivation and access to ambulatory care.

## (b) Size of small area

In the UK and US, routine hospital admissions data are 'geo-coded' - in the UK a patient's postcode in recorded, in the US a zip code - allowing admission rates by small area to be calculated. In the US 'small areas' have included individual zip codes (roughly 30, 000 population) (1), zip code clusters (22), cities (176) or States (72). In the UK they have generally included enumeration districts (pop size approximately 500), electoral wards (pop size approximately 10,000), health authorities (size from 500,000) and regions (approximately 3-5 million).

There appears to be no optimim size of small area for analyses: choice of size depends upon the aims and objectives of the study, and the number of events of interest per area. In the case of hospital admissions, choosing small areas with a *large* population size may be advantageous because more admissions are captured and less of the variation between area will be due to chance. But if area-based information is used to document the characteristics of the area, then small areas covering *small* populations may be better because the likelihood of ecologic fallacy is reduced (12). Some researchers have suggested that whatever the size of population covered by a small area, a more important factor to consider for the analysis is the *variation* in population size between small areas, because a larger variation can make it more difficult to detect the extent of random versus non-random variation in events between these areas (258). Others have suggested that, at the very least, researchers should document the variation in population size between areas (72).

In this study census enumeration districts (EDs) (rather than the larger electoral wards) were chosen as small areas partly to reduce the risk of ecologic fallacy, and partly to identify hospital service areas (see below and chapter 6) with more refinement. The disadvantage of choosing EDs rather than larger areas was that there would be more random variation in hospitalisation rates between them.

## (c) Identifying random variation

A number of papers investigating the best statistical test for random variation among small areas have been published. Paula Diehr and colleagues (259) used simulation models to test the robustness of a variety of statistical tests. They favoured the chi-squared model over others such as the extremal quotient and the systematic component of variation used by other researchers. Others have noted the importance of testing for spatial autocorrelation – which tests for the possibility that scores between neighbouring small areas may not be independent of spatial location (260).

## (d) Type of analysis

In most small area analyses, researchers have gone further than simply documenting variations between small areas. Usually some type of multivariate regression analysis is performed (72) (73), in which the correlation of a number of independent variables with the dependent variable (hospitalisations) is investigated to identify which independent variables have most influence.

There are various types of regression analyses, used under different conditions. If the number of events per small area is small and they are random in nature, as is the case with hospitalisations, then a Poisson regression is normally used (261). This type of regression will be used in this thesis.

- (e) Selection of variables
- Dependent variable hospital admissions for specific groups of conditions by small area

# Identifying ACS and marker conditions

The dependent variables to be used will be admissions for ACS and marker conditions. A

crucial first step in the study will be to identify such conditions in a way that has face validity in the UK. The literature suggests that consensus development may be a suitable method for this. The US studies discussed above did not appear to consider the *degree* of likely avoidability of admissions for certain conditions, or the extent to which ambulatory care might have an effect on either *preventing the onset of the condition*, or *preventing admission for an existing condition*. Furthermore most of the US studies discussed were not specific about the type of ambulatory care which could have most effect in reducing the need for hospitalisation. It might be important to fill these gaps; chapter 3 gives more details.

#### Accuracy of coding

Clearly any small area analysis is vulnerable to the quality of the data used. It is striking that this fact is largely ignored in the literature. It was thought important in this thesis to analyse the quality of routine data on hospitalisation in the HES dataset, and in particular the diagnostic coding. Chapters 4 and 5 describe this analysis.

#### (ii) Independent variables

#### Need

A crucial step in the analysis will be to take some account of 'need' - in particular the prevalence of the specific ACS and 'marker' conditions under study. As noted above, researchers have had great difficulty in allowing for this owing to the lack of geographically based information on prevalence of disease or health status. However reference has been made to the small area variations work at the University of York, in which a combination of area-based standardised mortality ratio (SMR), and standardised illness ratio (SIR) were calculated for small areas (electoral wards), using data from the 1991 census. This offers a promising way to account for ill health, however non-specific, since other data on health status are not available by small area. Of course this does not take into account the potential to benefit from health care.

Another 'need' variable to take account of in the analysis is socioeconomic deprivation. As

shown above, research suggests that the relationship between deprivation and hospitalisation rates is different for different types of conditions: medical (and in the US, ACS) conditions; 'discretionary' (elective) surgeries; and 'marker' conditions such as fractured neck of femur. As indicated in the introduction, one of the questions to be addressed in this thesis is whether there is such a strong association between socioeconomic deprivation and hospitalisation rates for ACS and marker conditions in the UK NHS. If so, this may indicate that access to ambulatory care (in particular primary care) is a problem in the NHS (as suggested in the US), or not (as suggested in Canada). From the literature discussed above, it was decided to use the Carstairs index in the analysis.

#### Age and sex

Clearly it is important, for the reasons outlined earlier, to take into account the effects of age and sex in analyses of hospitalisations.

### 'Health system' factors

From the literature, of all the potential 'health system' factors it seems most important to take account of access to primary care facilities (in particular access to care by general practitioners), and access to inpatient facilities. As discussed, the main literature on constructing such variables is that produced by researchers at the University of York; it was decided to construct the variables in a similar way.

After Tedeschi's work (173), the other 'health system' variable it might be important to take account of is the 'hospital effect'. This involves identifying hospital 'catchment', 'market' or 'service' areas. The term 'hospital service area' (HSA) will be used in this thesis. Two main methods have been to define a HSA. In the first, the small area is 'assigned' to a hospital's HSA if there are more admissions from the area to that hospital than to any other. This is termed in the literature the 'plurality rule' (262). *All* admissions occurring in residents of that small area are then assumed to be subject to the same hospital effect as for Hospital A, even though in fact a proportion of admissions may have occurred in other hospitals.

In the second, a small area is assigned to a hospital's service area if more than a specified

*proportion* of all admissions in residents occur at that hospital - say 50%. Again, all admissions occurring in residents of that small area are then assumed to be subject to the same hospital effect as for Hospital A.

In this analysis, both approaches were used. At first EDs were assigned to a hospital service area according to the plurality rule, then EDs were only included in the hospital service area if a specified proportion of admissions has occurred in the hospital. A provisional cut-off proportion was defined that was low enough not to exclude too many EDs from the analysis and yet high enough to avoid blurring any individual 'hospital effect' (see chapter 6).

A further decision was taken at this stage to limit the small area analysis to census enumeration districts only in HSAs serving the largest acute hospitals within the North West Thames region. Larger hospitals would be more likely to, it was assumed, a) admit patients with a full range of ACS and marker conditions (because the hospitals would have the required full range of specialities), b) have staff with greater expertise in diagnostic coding of routine activity data, and c) have a larger number of EDs contained within their HSA.

Finally, because patterns of admissions change over time, HSAs were to be calculated for each year for which admissions data were available at the beginning of the study – 1991/92, 1992/93, 1993/94.

#### Ethnicity

The UK literature contained mixed messages on the impact of ethnicity on variations in hospital admission rates. Because of this, and the inadequacy of coding of ethnicity on routine NHS admissions data, it was decided to exclude this variable from the analysis.

## Other factors

Finally, other factors shown in the discussion above to have impact on hospitalisation rates included perceived access, and clinician practice style. These, and other factors not generally investigated, such as utilisation of ambulatory care, were thought to be of secondary importance in this thesis, to be investigated using multivariate analyses if the

variations between small areas in the UK were shown to be large, and if a significant amount of variation remained after multivariate analysis using the eight main factors above.

# Chapter 3 Identifying admissions for further study

Chap	ter outlig	LC .
3.1	Introd	luction
3.2	Exclu	ding broad categories of admissions
3.3	Identi	fying ACS and marker conditions
	3.3.1	Methods
		(a ) The approach to consensus development
		(b) Developing consensus
		(c) Selecting conditions to be reviewed
		(d) Selection of panellists
		(e) Data analysis
	3.3.2	Results
		(a) Attendance
		(b) Items excluded or modified during panel meetings
		(c) 'Ambulatory care sensitive' and 'ambulatory care insensitive'
		conditions
	3.3.3	Discussion
		(a) Consensus process
		(b) Validity
		(c) Substantive results
3.4	Identi	fying admission groups for subsequent small area analysis
	3.4.1	UK-defined ACS and marker admission groups
	3.4.2	Other admission groups
3.5	Conclu	usion

#### 3.1 Introduction

The next step in the project was to identify and select the admissions that would be studied further in the small area analysis described in chapter 6. This chapter is divided into three sections. First there is a brief discussion about several broad categories of admissions that were excluded at the start from further analysis. This is followed by a description of how ACS and marker conditions were identified. The third section shows how and why other groups of admissions were identified to be included in the small area analysis.

## 3.2 Excluding broad categories of admissions

From the outset it was decided to exclude from further analysis three broad categories of admissions. The first was admissions in persons aged 75 years or older. One reason was because an aim of the study was to be able to compare findings with those of Billings' *et al* (1), as noted in the introduction, whose study examined admission rates in persons aged under 65 years. 75 years rather than 65 years was chosen in order to ensure that there were a reasonable number of events (admissions) per small area for the analysis. Another reason for excluding these older patients was because identifying the diagnosis that caused the admission (recorded on routine hospital activity data) was likely to more difficult in an age group with a greater prevalence of multiple morbidities.

The second group excluded were admissions in well babies born in hospital since these admissions were unlikely to be influenced by access to, or quality of, primary care. The third group were admissions in certain specialties (see below) - obstetrics, psychiatry, mental handicap, dental medicine - because these had also been excluded in the Billings' study. Admissions for neoplasms were also excluded for the same reason.

## 3.3 Identifying ACS and marker conditions

As noted in chapter 2, the US literature shows that lists of conditions for which ambulatory care could prevent a hospital admission have typically been drawn up using a literature review and the considered judgement of a few experts. The methods by which the

judgement was arrived at have not been fully described, although it is clear that in at least one case a method to develop a consensus was used (1). It was thought that a systematic method of arriving at a consensus should be used in this study.

The US studies discussed above did not appear to attempt to gain consensus on the *degree* of likely avoidability of admissions for certain conditions, or the extent to which ambulatory care might have an effect on either *preventing the onset of the condition*, or *preventing admission for an existing condition*. This thesis aimed to fill these gaps.

### 3.3.1 Methods

(a) The approach to consensus development

There are three generally accepted formal methods to develop a consensus opinion:

- a consensus development conference (263)(264), in which a 'jury' hears and considers expert evidence before establishing a consensus view. This was rejected as too costly for our purpose and inappropriate because of the numbers of clinical conditions to be considered;
- a *Delphi exercise* (265), in which a panel of around 10 experts complete a series of postal questionnaires, interspersed by anonymised postal feedback on the results of each successive round. This was also ruled out on the grounds that the advantages of meetings for explanation and clarification of the task and 'live' debate between panellists would outweigh the disadvantages of cost and loss of anonymity;
- the nominal group technique (NGT) (266)(267), like Delphi, involves several rounds, but experts meet at least once to discuss their differences and usually complete questionnaires at their meetings.

The nominal group technique (NGT) was accepted as the most appropriate method for this study. However because it was anticipated that the number of conditions for 'experts' to review would be too large to be completed in a meeting, it was decided to allow experts to

complete the questionnaire outside of the meeting and post back the results. This combination of meetings and questionnaire rounds is closest to what has been described as a modified NGT (267).

### (b) Developing consensus

Three panels of participants were convened. Each panel was provided with a different list of clinical conditions, and for each condition members were asked to consider three questions:

- Q1 Can *onset* of the condition be prevented by timely and effective ambulatory care (eg by appropriate screening or risk factor management)?
- Q2 (Given onset) can *admission* be averted by timely and effective ambulatory care (eg by remedy, stabilisation or preventing complications)?
- Q3 Once admission is indicated, should it take place within 48 hours?

'Ambulatory care' was defined as care of an acute, episodic or chronic illness or condition in any setting other than in a hospital inpatient setting. Care included screening and risk factor management as well as treatment. Care was 'timely and effective' if it stopped the condition deteriorating to the point at which admission was needed, and was 'practical and reasonable' in the current NHS. Panellists were asked to respond on the basis of their own experience of patients admitted to hospital for the condition.

The first two questions were designed to help identify ambulatory care sensitive (ACS) conditions and ambulatory care insensitive (ACI) conditions, the third to identify 'marker' conditions. For each question, each panellist assigned each condition to one of the following six categories: 1 (0-4% of the time); 2 (5-29%); 3 (30-49%); 4 (50-69%); 5 (70-94%) and 6 (95-100%). There was space on the questionnaires for comments, and participants were asked not to respond for conditions outside their normal area of expertise. No evidence from the literature was supplied.

For panel A in the first round, only questions 2 and 3 were posed. It was agreed at the first meeting that question 1 should be added to all subsequent questionnaires for all panels, to avoid confusion between effective primary prevention and effective care of existing disease.

Each panel met on two occasions and completed three rounds of the same questionnaire. The schedule is shown in table 7. The meetings were facilitated, ensuring that all had an opportunity to speak, all were drawn into the discussion, and all conditions were discussed. An observer recorded key points and decisions.

Table 7 The consensus dev	evelopment process
---------------------------	--------------------

Schedule	Task		
week 1	1st panel meeting	-	introduce project distribute and discuss questionnaire
			discuss each condition on list panellists take questionnaire away with them
week 4	1st round questionnaire	e returned,	completed
week 8	2nd round questionnair with group response (n	re sent out: nodes and	ranges) and addressee's own response in round 1
week 12	2nd round questionnair	e returned	completed
wcck 16	3rd round questionnaire with group response (m	e sent out: nodes & ra	nges) and addressee's own response in round 2
week 17	2nd panel meeting	-	discuss results
wcek 20	3rd round questionnair	e returned	
week 22	Group responses (medi	ans and in	terquartile ranges) analysed

## (c) Selecting conditions to be reviewed

The numbers of conditions reviewed had to be kept to manageable proportions. It was decided not to consider diagnoses likely to occur in admission categories already excluded (see above) - ICD-9 codes in several ICD-9 chapters:V (mental disorders); XI (complications of pregnancy, childbirth, and puerperium); XV (conditions arising in the perinatal period); and codes 520-525 (diseases and conditions of teeth). ICD-9 codes in Chapter II (neoplasms) were not expected to yield many ACS conditions, and so these were largely excluded. The remaining conditions were identified from a frequency distribution of

all ICD-9 diagnosis codes (268) that appeared on hospital discharge data for the 3.5 million residents of North West Thames NHS Region during one year – 1990/91.

Conditions represented by ICD-9 codes were selected for review if they had appeared more than 400 times in discharge records, (not necessarily as the main diagnosis) during 1990/91. Some codes were grouped where this seemed appropriate (for example, cellulitis of the back or buttock formed one group; cellulitis at other sites formed another). Codes indicating evidently non-specific symptoms and signs were excluded (for example, 799.9: 'unknown cause of morbidity or mortality'). In a secondary list, codes were selected if they appeared between 100 and 400 times and it seemed likely that the onset or admission could be affected by ambulatory care. The choice of these was made by a public health physician (JD). In all, there were 166 conditions (45 with less than 400 discharges) in the initial list.

### (d) Selection of panellists

Clinicians who were advisors in each clinical specialty to the then North West Thames Regional Health Authority were asked to suggest possible panel members. Of 27 general practitioners (GPs) contacted, 17 agreed to participate. Of the 31 consultants contacted, 17 agreed. The 34 participating clinicians were grouped into 3 panels. Panels A and B reviewed conditions normally treated by clinicians working in the specialties of general medicine, paediatrics or general surgery. Panel C reviewed conditions in other specialties such as gynaecology, orthopaedics and ear, nose and throat (ENT). Consultants were allocated to groups according to their specialty, and the GPs according to their own choice. The composition of each panel is shown in table 8.

Panel	No. of GPs	No. of Consultants	Specialties represented
	5	5	General Surgery
			General Medicine
			Rheumatology
			Cardiology
			Renal medicine
			General practice
в	5	5	Gynaecology
			Orthopaedics
			Ophthalmology
			ENT
			General practice
С	7	7	General Medicine
			General Surgery
			Paediatrics
			Respiratory Medicine
			Cardiology
			General practice

#### Table 8 Composition of panels

#### (c) Data analysis

To each question, panellists indicated their response by marking one of the 6 categories noted above. Median percentages and interquartile ranges were calculated for the responses to each question and each condition, using an interpolation method for grouped data (269) which assumes that values are spread evenly across categories. Thus the minimum possible median score for each question was 2.5% and the maximum possible was 97.5%. Scores for panellists who only completed one round were excluded. For panellists who did not complete the final round, second round scores were taken as final. The extent to which scores for a condition converged between rounds 1 and 3 was examined using only results for participants who had completed scores in all three rounds. For each panel, the closeness of agreement was indicated by the median interquartile range over the conditions considered. Consensus was characterised as 'poor' if, after excluding at most one outlier panellist per condition, the remainder of the responses covered more than three adjacent

categories.

Conditions were defined as ambulatory care sensitive (ACS) if 70% or more admissions were judged avoidable by better prevention or if 70% or more were judged avoidable by better management in ambulatory care. Conditions were defined as 'weakly ACS' if not ACS and either 50-69% of admissions were judged avoidable by better prevention or 50-69% were judged avoidable by better management in ambulatory care. The rest were defined as ACI (ambulatory care insensitive). Each of these three groups was subdivided into 'urgent' and 'others' according to whether 70% or more admissions were judged to be normally required within 48 hours or not (question 3).

'Potential markers' are the subset of ACI conditions for which less than 30% of admissions could be avoided by prevention of onset and less than 30% avoided by appropriate management. Actual 'marker conditions' are, in turn, a subset of these for which 95% or more admissions were judged necessary in less than 48 hours. The layout of table 10 helps to clarify these definitions.

#### 3.3.2 Results

#### (a) Attendance

Data were discarded for one panellist who only completed one round of the questionnaire. Two members of panel C did not complete the first questionnaire and one from each panel did not complete the third questionnaire.

#### (b) Items excluded or modified during panel meetings

There were 12 conditions that the panels were unable to classify, either because they were too unspecific (eg 'other diseases of nasal cavity and sinuses' - code 478.1), or because they were considered unlikely ever to be the main cause of an admission and therefore difficult to classify (eg 'coronary atherosclerosis', 414.0). The four-digit ICD-9 codes for diabetes did not distinguish between insulin dependent and non-insulin dependent diabetes, and these

were considered separately in the second and third rounds of the questionnaire. Thus the initial 166 conditions became 155. One condition, ophthalmic complications of diabetes (250.4), was inadvertently reviewed by two panels. The final scores were based on average response rates over all questions of 99% for GPs and 97%, 84% and 91% for consultants in panels A, B and C, respectively. Almost all of the non-responses by consultants arose because they felt that the question lay outside their area of expertise.

(c) 'Ambulatory care sensitive' and 'ambulatory care insensitive' conditions

Tables 9, 10 and 11 give the results for the individual conditions. As well as identifying conditions that are ACS (table 9), weakly ACS (table 10) and ACI (table 11), the table also separates those for which admission is 'usually urgent' and 'usually not urgent', and those for which prevention is more or less likely to play a role. 'Markers' are shown in bold. Results for conditions with fewer than 400 admissions in 1990/91 (labelled in the table as 'also') are indicated by listing their ICD-9 codes in the appropriate areas of the tables.

# Table 9 Ambulatory care sensitive conditions (responses to Q1= or >70% or Q2=or >70%)

		ADMISSI (Q3 = or i	ION USU >70%)	ALLY UR	GENT					ADMISSI	ON NO (Q3 = (	FUSUALLY or <70%)	,		•
		Q1		02		03				Q1		02		Q3	
ICD-0	Condition	Median	IQR	Median	IOR	Median	IQR	ICD-9	Condition	Median	IQR	Median	IQR	Median	IQR
Code								Code							
														•	
345.9	Epilepsy unspec	3.0	3.0	78.3	25.8	97.0	3.0	78.1	Viral warts	4.2	15.4	97.2	2.8	2.5	2.5
493.0	Asihma extrinsic	3.3	3.3	76.3	20.6	96.7	3.3	706.2	Sebaceous cyst	2.8	2.8	97.2	2.8	2.8	2.8
493.9	Asthma unspec	3.3	3.3	74.2	24.7	96.7	3.3	727.4	Ganglion and cyst of	2.5	2.5	77.1	19.4	2.5	2.5
813.4	Fracture of radius & ulna	3.6	7.4	70.0	29.2	96.9	3.1		Synovium, tendon, bursa						
	(lower, closed)							622.0, 622.1,	Dysplasia cervix, mucous	5	15.0	72.5	22.5	2.5	2.5
79.9	Viral infection unspec	3.6	9.5	74.2	22.1	90.0	19.4	622.7	polyp, erosion of cervix						
250.0	Diabetes (insulin dependent)	3.0	3.0	73.6	24.3	70.0	25.0		Dyspepsia and other	25.0	27.5	76.3	29.4	13.3	19.6
								536.8	Stomach function disorders						
Agree?	463, 599.0							also	373.2, 374.0, 616.0						

(a) avoidable primarily through effective management in ambulatory care (Q1<30%)

Key: 'agree?' indicates poor consensus (see text); 'also' indicates fewer than 400 admissions in the North West Thames dataset; bold type indicates marker conditions; IQR, interquartile range; nec, not elsewhere classified; unspec. unspecified; w, with; w/o, without; ACS, ambulatory care sensitive.

		ADMISSI >70%)  Q1	on usual	LY URGEI  02	NT (Q3	03				ADMISSIO URGENT ( Q1	N NOT Q3 <70	USUALLY 6)  Q2		Q3	
ICD-0	Condition	Median	IQR	Median	IQR	Median	IQR	ICD-9	Condition	Median	IQR	Median	IQR	Median	IQR
code								code							
251.2	Hypophycaemia unspec	76.3	22.3	79.4	16.8	97.3	2.7	703.0	Ingrowing toenail	60.0	18.6	96.4	9.4	3.0	3.0
682.0	Cellulitis of face, etc	50.0	26.7	74.2	22.1	79.4	15.6	564.0	Constipation	70.0	28.1	80.0	25.8	55.7	17.9
250.0	Diabetes (non-insulin dependent)	32.0	27.1	80.0	25.8	70.0	28.1	280	Iron deficiency anaemia	60.0	12.5	74.2	22.1	37.1	12.9
								707.1	Lower limb ulcer (except decubitus)	60.0	16.7	74.2	22.1	20.6	28.4
								531.9	Gastric ulcer unspec w/o	46.0	21.5	70.0	25.0	30.0	22.5
								535.5	Gastritis and duodenitis unspec	46.0	24.0	70.0	22.5	30.0	25.0
agree?	435							agree? also	401.9, 530.1, 532.9 532.7						

Table 9Ambulatory care sensitive conditions (Q1 = or >70% or Q2 > 70%) and (Q1>30%)(b) Also with scope for prevention in ambulatory care (Q1 > 30%)

Kay: 'ngree?' indicates poor consensus (see text); 'also' indicates fewer than 400 admissions in the North West Thames dataset; bold type indicates marker conditions; IQR, interquartile range; nec, not elsewhere classified; unspec, unspecified; w, with; w/o, without; ACS, ambulatory care sensitive.

		ADMISS URGEN  Q1	ion US T (Q3 =	UALLY  pr >70%)  Q2	)	03				ADMISSIC URGENT	ON NOT (Q3 <70'	USUALLŸ %)  Q2		Q3	
ICD-8 code	Condition	Median	IQR	Median	KQR	Median	KQR	ICD-9 code	Condition	Median	1QR	Median	IQR	Median	KQR
345.1	Generalised curvulsive epilepsy	3.0	3.0	61.1	13.3	97.0	3.0	626.2	Excessive or frequent menstruation	4.0	11.3	58.6	11.4	3.3	3.3
464.4	Croup	2.5	2.5	60.0	16.7	97.5	2.5	626.8	Dysfunctional or functional uterine	4.0	11.3	60.0	10.0	3.3	3.3
481	Pneumococcal pneumonia	9.2	18.7	58.6	17.1	96.7	3.3		Haemonhage, NOS						
485	Bronchopneumonia	16.3	16.3	56.3	19.4	95.4	19.9	715.9	Osteoarthritis (unspec if generalised or	4.5	17.3	54.0	22.0	3.2	3.2
465.9	Upper respiratory tract infection	17.5	18.8	60.0	15.0	96.7	3.3		localised)						
	Unapec							354.0	Carpai tunnel syndrome	17.5	22.5	60.0	12.9	2.5	2.5
		1						332	Parkinson's disease	3.1	3.1	65.7	27.8	37.8	12.2
								340	Multiple sclerosis	2.8	2.8	55.7	16.2	41.4	14.3
								553.3	Diaphragmmatic hemia	15.4	18.8	62.9	12.9	11.3	32.2
								556	Idiopathic proctocolitis	2.7	2.7	59.1	10.9	44.3	19.3
								714.0	Rheumatoid arthritis	2.8	2.8	64.3	17.0	35.7	19.1
				1				780.6	Pyrexia of unknown origin	2.5	2.5	57.1	21.4	51.4	21.7
								574.2	Calculus of gall bladder without	4.1	11.7	57.8	14.4	58.8	16.3
									cholecystitis			1			
								558	Other non-infective gastroenteritis/colitis	23.8	29.4	62.5	12.5	56.7	20.0
				1				562.1	Diverticular disease of colon	17.5	18.1	61.8	11.8	44.4	20.3
								585	Chronic renal failure	27.5	25.4	58.3	15.0	58.8	11.3
agree7	428.9							agree?	569.3, 553.3, 722.1, 715.3						
								also	365.9, 378.9, 603.9, 626.6, 714.3, 471.0, 625.0, 625.3						

Table 10Weakly ambulatory care sensitive conditions (not ACS and; Q1 >50% or < 70%, or Q2 >50% or < 70%)</th>(a) avoidable primarily through effective management in ambulatory care (Q1  $\leq 30\%$ )

Key: agree? indicates poor consensus (see text); "also" indicates fewer than 400 admissions in the North West Thames dataset; bold type indicates marker conditions; IQR, interquartile range; nec, not elsewhere classified; unspec, inspecified; w, with; w/o, without; ACS, ambulatory care sensitive.

		ADMISSI	ON USU/ (Q3 =or?	ALLY 70%)						ADMISSI <70%)	ON N	OT USUALL	Y URG	ENT (Q3	
		en 🛛		02		Q3				Q1		QZ		Q3	
ICD-0	Condition	Nedian	IQR	Nedian	IQR	Nedian	IQR	ICD-8	Condition	Median	<b>IQR</b>	Median	IQR	Median	IQR
code								code							
9.1	Calitis, enteritis, etc, of intectious origin	50.0	22.5	61.1	11.1	77.5	23.8	250.4 496.0	Diabetes with ophthalmic manifestations Chungic ainways obstruction pec	55.7 56.3	16.2	60.0 60.0	10.0	57.5 68.6	12.5
250.1	Diabetes with ketoacidosis	60.8	10.8	58.2	11.8	97.0	3.0	428.0	Concestive heart failure	33.8	22.8	59.0	13.0	64.4	19.0
428.1	Left heart failure	53.8	29.4	59.2	10.8	95.0	21.3	285.9	Anaemia unspecified	55.0	25.0	61.1	11.1	43.3	20.0
780.3	Convulsions	51.4	29.0	65.7	22.3	96.4	9.4	535.6	Duodenitis	45.0	25.0	64.3	17.0	30.0	25.0
413	Angina pectoris	45.0	25.0	53.3	21.7	96.9	3.1	614.9	Pelvic inflammatory disease female, unspec	40.0	12.5	60.0	12.5	41.1	11.1
427.3	Airial fibrillation and flutter	35.7	25.4	60.0	10.0	80.7	17.9	565.0	Anal fissure	57.5	12.5	58.9	11.1	25.8	24.6
								454.9	Varicose veins of lower extremities w/o ulcer or inflammation	40.0	16.7	60.0	16.7	2.8	2.8
								727.0	Synovitis and tenosynovitis	35.0	21.9	65.0	21.9	3.2	3.2
agree?	788.2, 959.4							agree?	8.8, 443.9, 474.0, 618.0, 381.0, 466.0						
also	25.1							also	9.3, 625.6, 473.9, 455.0, 455.2, 455.6, 384.2, 530.2, 535.4, 682.5						

Table 10Weakly ambulatory care sensitive conditions (not ACS and; Q1 >50% or <70%, or Q2 Q1 >50% or <70%)</th>(b) also scope for prevention in ambulatory care (Q1 >30%)

Key: "agree?' indicates poor consensus (see text); "also indicates fewer than 400 admissions in the North West Thames dataset; bold type indicates marker conditions; IQR, interquartile range; nec. not elsewhere classified; unspec, unspecified; w, with; w/o, without; ACS, ambulatory care sensitive

# Table 11 Ambulatory care insensitive conditions (Q1<50% and Q2<50%)</th>

		ADMISSI URGENT Q1	ON USI (Q3 = d	JALLY x>70%)  Q2		03				ADMISSIO 70%)  Q1	N NO	USUALLI	URG	ENT (Q3 Q3	
ICD-0	Condition	Median	IQR	Nedian	<b>IQR</b>	Median	<b>IQR</b>	ICD-9	Condition	Median	IQR	Median	IQR	Median	IQR
Code								code							
451.1	Presides and thromoophiebies of	40.0	12.5	41.4	14.3	95.8	32.9	218.0	Uterine fibroid	2.5	2.5	32.5	24.7	3.2	3.2
	deep vessels							474.1	Hypertrophy of tonsils and	2.9	2.9	45.0	21.7	2.9	2.9
578.9	Haemorrhage of gastrointestinal	42.2	14.4	37.0	13.0	96.8	3.3		adenoids						
	tract unspec							715.1	Osteoarthritis (localised	8.1	19.4	44.0	22.5	3.2	3.2
592.1	Calculus of urelier	10.4	18.7	38.8	16.3	96.4	12.6		and primary)						
410	Acute myocardial infarction	36.7	35.8	4.2	12.3	97.5	2.5	628.9	Infertility, female of unspec	26.9	22.4	36.7	13.3	2.5	2.5
415.1	Pulmonary embolism	42.5	23.8	2.8	2.8	97.5	2.5		origin						
577.0	Acute pancrealitis	40.0	16.7	3.1	3.1	97.5	2.5	381.2	Chronic mucoid otitis media	37.1	12.9	42.9	12.9	2.8	2.8
466.1	Acute branchialitis	3.0	3.0	45.0	28.0	96.7	3.3	381.3	Chronic non-supporative	37.5	12.5	44.3	16.2	3.1	3.1
574.0	Calculus of gall bladder with	3.6	9.4	36.3	20.6	97.0	3.0		otitis media (other/unspec)						
	acure cholecystilis							611.7	Signs and symptoms in breast	3.6	9.5	33.3	21.0	38.6	14.3
566	Abcess of anal and rectal regions	3.3	3.3	35.6	22.2	81.3	16.3	592.0	Calculus of kidney	17.5	18.1	43.8	19.4	40.8	10.8
850	Concussion	33.3	37.5	35.0	30.0	96.9	3.1	530.3	Stricture & stenosis of	46.0	24.0	36.7	21.3	45.0	27.5
									oesophagus						
agree?	550.1, 436, 977.9, 574.1							agree?	424.1, 598.9, 600, 605, 602						
also	282.6, 532.4, 616.3							aiso	511.9, 555.9, 381.4, 626.9, 715.0, 728.6						

(a) Some scope for prevention or management in ambulatory care (Q1>30% or Q2>30%)

Key: agree? indicates poor consensus (see text); also indicates fewer than 400 admissions in the North West Thames dataset; bold type indicates marker conditions; IQR, interquartile range; nec, not elsewhere classified; unspecified; w, with, w o, without; ACS, ambulatory care sensitive.

# Table 11 Ambulatory care insensitive conditions (Q1<50% and Q2<50%)</th>

		ADMISS URGEN	ION USI [ (Q3 =0	UALLY r>70%)						ADMISSION (Q3 <70%)	I NOT U	SUALLY UR	GENT		
		<b>Q1</b>		62		Q3				Q1		Q2		(D3	
ICD-0	Condition	Median	IOR	Nedian	IQR	Nedian	<b>IQR</b>	ICD-9	Condition	Median	IQR	Median	IQR	Median	IQR
code								code							
512	Pneomothorax	2.5	2.5	2.7	2.7	97.0	3.0	366.1	Senile cataract	2.5	2.5	2.5	2.5	2.8	2.8
540.9/1	Acute appendicitis w/o	2.7	2.7	2.7	2.7	97.3	2.7	366.9	Cataract unspec	3.1	3.1	3.1	3.1	2.5	2.5
	perionitis or unqualified							470	Deflected nasal septum	2.5	2.5	3.6	9.5	2.5	2.5
623.2	Fracture of tibia and fibula, shaft,	2.5	2.5	8.1	19.4	97.2	2.8	550.9	Inguinal hemia w/o	3.2	3.2	4.2	15.4	2.8	2.8
	closed								obstruction/gangrene						
560.9	Intestinal obstruction unspec	3.6	12.6	15.4	24.9	96.4	9.4	627.1	Postmenopausal bleeding	2.5	2.5	4.0	15.5	3.8	12.5
\$20.8	Fractured femur unspec, closed	15.4	18.8	2.5	2.5	97.5	2.5	752.5	Undescended testicle	2.5	2.5	3.3	3.3	3.0	3.0
965.4	Poisoning by paracetamol.	27.5	31.3	3.2	3.2	97.5	2.5	620.2	Ovarian cyst	2.8	2.8	4.5	13.7	21.7	22.1
	acetanilide, phenacetin							203.0	Multiple myeloma	2.5	2.5	3.3	3.3	40.9	10.9
578.0	Haematemesis	28.2	24.4	23.8	26.8	97.5	2.5	599.7	Haematuria	3.0	3.0	17.5	15.0	22.9	22.9
813.0	Fracture of radius & ulna (upper,	3.1	3.1	3.6	9.5	82.5	31.3								
	unspec, closed)														
Agree?	854														
Also	821.0, 574.5, 426.9, 560.8														

# (a) Little scope for prevention or management in ambulatory care (Q1<30% or Q2<30%)

Key agree? indicates poor consensus (see text), 'also' indicates fewer than 400 admissions in the North West Thames dataset; bold type indicates marker conditions; IQR, interquartile range; nec, not elsewhere classified; unspec, unspecified; w, with, w/o, without; ACS, ambulatory care sensitive.

Table 12 gives a cross-classification of the final results for question 1 against question 2, indicating where the panels felt that the opportunities for avoiding admission lay. For 30 (19.3%) of the conditions, at least 70% of admissions were thought to have been avoidable through more timely and effective management, compared to only 2 (1.3%) avoidable through prevention of onset (and for both of these, most admissions were also avoidable through good management).

The final grouping of conditions, and the criteria used to group them, were sent to all panellists for comment. Of the 34 panellists contacted, nine responded. All indicated that they were satisfied with the final groupings except for an ENT surgeon, who was concerned about 'non-suppurative otitis media' (ICD-9 381.3).

Q2: % of admissions avoidable		Q1: % of adm onset of the	nissions avoidable condition	by preventing the	)		
by better management		0-4%	5-29%	30-49%	50-69%	70% +	TOTAL
0-4%	All >95% urgent				0	0	23
5-29%	All >95% urgent				0	0	8
30-49%	All	to the second			2	0	30
50-69%	IIA	C_19	13	17	15	0	64
70 <del>-94</del> %	Ail	11	2	4	6	2	25
95% +	All	3	0	1	1	0	5
TOTAL		68	25	36	24	2	155

# Table 12 Numbers of conditions in different groups

ACI= Ambulatory care insensitive, ACS = Ambulatory care sensitive

#### 3.3.3 Discussion

#### (a) Consensus process

In the USA, lists of conditions for which ambulatory care could prevent a hospital admission have typically been drawn up using a literature review and the considered judgement of a few experts (1)(22)(64). This study used a more transparent method of arriving at a consensus, using a modified nominal group technique.

There were some difficulties with the task. The panels were asked to respond on the basis of their own experience of patients admitted to hospital for each condition. Initially some of the hospital consultants had difficulty thinking in terms of their overall experience rather than specific cases. Also, many factors other than the condition itself can influence the preventability of any particular admission (such as age, co-morbidity and social/informal support). By the second meeting, panellists were clearer about the task, although for some the difficulty was not entirely resolved. The meetings helped panellists to focus and summarise their thinking, and in general they thought that the process had been effective in stimulating debate and producing a list of conditions likely to be amenable to ambulatory care.

One difference between this and earlier studies of ACS conditions is the distinction made between effective prevention of onset and effective management of existing disease. This has obvious advantages, but it does lead to some ambiguity when deriving consolidated lists of ACS, or weakly ACS, conditions from two more specific questions. For example if 50% of cases are preventable and 50% treatable on an ambulatory basis, the overall percentage of avoidable admissions could in theory lie anywhere between 50% and 100%. However in this exercise only 2 conditions were considered to be ACS on grounds of preventability alone.

A more general problem in arriving at consensus statements is how to convert a distribution of values into two or more categories. The definitions of ambulatory care sensitive conditions used in this study involved arbitrary cut points (70% and 50% avoidable), and an arbitrary, if undemanding, definition of 'poor' consensus. For conditions with medians in the mid-range, even quite tight distributions can lead to substantial interquartile ranges, but only one condition in the study (transient cerebral ischaemia) had an interquartile range of more than 50%. For all but a handful of others, the ranges were less than 30%.

#### (b) Validity

What can be said about the validity of the findings? Firstly, the outcome of consensus studies depends on the expertise in the panel and the quality of the evidence (270)(271) they use. This study involved consultants and GPs nominated by peers with an interest in clinical policy, and then self-selected on grounds of their own interest and availability. This was done because a secondary aim was to develop a list which made sense to local clinicians and to raise discussion locally. However, clinicians who participated came from a range of hospitals and practices - urban, suburban and rural, teaching and non-teaching - and most specialties were represented. In this limited sense those participating were reasonably representative - a pattern noted in consensus studies (272).

Second, there were very marked differences between panels in median scores for the different questions and lists of conditions, suggesting that at very least the process had some power to discriminate. Third, although the panellists were not provided with a literature review, the results were supported by the limited evidence available from published empirical studies on the effect of ambulatory care on hospitalisation rates, particularly for asthma (234)(235)(236) and diabetes (242)(273)(274). Fourth, there were many similarities between the results and those of American studies on this subject (see table 5, chapter 2) despite differences in practice between the UK and US and the different methods used to select and group conditions.

If the study were to be repeated, the single greatest improvement would be to have a substantial overlap between the lists considered by each panel, to allow a proper assessment of the reproducability of the consensus development process. The results for the condition mistakenly reviewed by two panels is at least consistent with a reproducible process, but are hardly conclusive evidence of it. Recent studies of a broadly similar consensus process have given kappas ranging from 0.51 (275)(276) to 0.83 (274).

#### (c) Substantive results

A number of general themes can be identified. The clear view of the panels was that the scope for avoiding admission through better ambulatory care is very substantial. Although ambulatory care was defined in the way outlined above, in the discussions it was clear that most panellists were concerned with the impact of primary care. Furthermore the panellists thought that timely and effective management of existing disease offered far more scope to reduce avoidable admissions than prevention of onset.

In general, GPs and consultants were in good agreement. The GPs were slightly the more positive about the potential for avoiding admissions than the consultants. Possibly, GPs were the more sanguine about the potential of ambulatory care. Alternatively, in spite of instructions to consider the full range of patients currently admitted with each condition, GPs may have been thinking about the patients registered in their practice they have experienced, while consultants may have been thinking about the patients they cared for in hospital. However the small differences involved should not be over-interpreted.

Clearly, the results of this study cannot be regarded as definitive. At best, they may provide a fair reflection of the clinical consensus (or lack of it) at a given point in time, and thus a possible agenda or shortlist for reviews in which clinical consensus is checked against research results, condition by condition. The risk of short-listing without a literature review in this way is that there may be false negatives or positives: conditions that the literature would support as ACS, but which were not shortlisted by the panel on the basis of their own knowledge, or vice versa.

Meanwhile technical and social developments can be expected to alter thresholds for hospital admission one way or another. The value of the approach in improving equity and effectiveness remains to be seen, but if it finds favour more widely in the NHS it will be necessary to rerun the exercise. However, it may be possible in future to produce revised lists on the basis of a much more limited process, given the very limited changes made after the first round.

Other studies have used much shorter lists of conditions such as asthma and diabetes – admissions for these two conditions have been used recently by the NHS Executive as an indicator of the quality of chronic care management, within a wider set of indicators in national performance assessment framework (155). The advantage of the approach in this study, by contrast, is that it involved far more events, and so as a potential indicator of quality of care could be used for smaller areas or shorter periods of time. Also, if the longer list of conditions is used as a basis for routine monitoring of NHS performance, the scope for 'gaming' (ie focusing management attention on indicator conditions, perhaps at the expense of others) is much reduced.

It has been suggested that improvements in primary care may *increase* hospital utilisation (42). Although such a contention remains controversial (44), it is not inconsistent with the hypothesis that good primary care can cost-effectively reduce admission rates for certain types of condition while increasing them for others. Good primary care will improve appropriate access to the NHS and may well increase admission rates for some non-ACS conditions, if cost-effective treatment demands secondary care. The benefits of better access to timely and effective ambulatory, in particular primary, care may lie not necessarily in cost savings for secondary care, but in improved efficiency in the system as a whole.

### 3.4 Identifying admission groups for subsequent small area analysis

3.4.1 UK-defined 'ACS' and 'marker' admission groups

The study described above identified two groups of ACS conditions (ACS and weakly ACS) and two groups of marker conditions (potential markers and markers). From these a total of 16 groups of admissions for ACS or marker conditions (termed 'admission groups') were defined for potential further study in the small area analysis, based upon:

- whether an ACS or marker condition had been coded on discharge data;

- the extent to which the admission was thought by the panels to be required within 48 hours;

- whether the ICD-9 diagnostic code for the ACS or marker condition was coded in first place in the admission (ie as the main diagnosis), or whether it appeared as a secondary diagnosis;

- whether or not the admission was a readmission (an admission within 28 days of the discharge date of a previous admission).

How the 16 admission groups were defined is shown in table 13 (for ACS conditions) and table 14 (for marker conditions) below.
Table 13

ACS admission groups for further analysis

Admission group	Strong ACS^^	Weak ACS^^	Urgency*	Position of diagnostic code in first episode**	Admissions^ or readmissions
1	Yes	Yes	0-100% (Any urgency)	Main	Admissions
2	Ycs	Yes	0-100% (Any urgency)	Main or secondary	Admissions
3	Ycs	Yes	0-100% (Any urgency)	Main	Readmissions
4	Yes	Yes	0-100% (Any urgency)	Main or secondary	Readmissions
5	Yes	Yes	70% plus (Urgent)	Main	Admissions
6	Yes	Yes	70% plus (Urgent)	Main or secondary	Admissions
7	Yes	Yes	70% plus (Urgent)	Main	Readmissions
8	Yes	Yes	70% plus (Urgent)	Main or secondary	Readmissions
9	Yes	No	0-100% (Any urgency)	Main	Admissions
10	Yes	No	0-100% (Any urgency)	Main	Readmissions
11	Yes	No	70% plus (Urgent)	Main	Admissions
12	Yes	Na	70% plus (Urgent)	Main	Readmissions
13	Yes	No	90% plus (Very urgent)	Main	Admissions
14	Yes	Yes	90% plus (Very urgent)	Main	Readmissions

 Indicates the overall response in the final round of questionnaires to question 3: 'Once admission is indicated, should it take place within 48 hours?' Responses in the range 0-100% are referred to as 'any urgency', responses of 70% and over are referred to as 'urgent', responses of 90% or more are referred to as 'very urgent'.

\*\* Indicates the position of the ICD-9 diagnostic code recorded on routine hospital episode statistics data. Main' denotes first recorded diagnosis, 'secondary' denotes a secondary diagnosis. A hospital admission for one patient may consist of more than one finished consultant 'episode' (see chapter 4).

^ Admissions here refer to all admissions including readmissions.

^^ 'Strong' and 'weak' ACS conditions are defined as shown in tables 9 and 10.

Table 14

Marker admission groups for further analysis

Admission group	Markers	Potential markers	Urgency*	Position of diagnostic code in first episode**	Admissions^ or readmissions
15	Yes	Yes	90% plus (Very urgent)	Main	Admissions
16	Yes	No	90% plus (Very urgent)	Main	Admissions

Indicates the overall response in the final round of questionnaires to question 3: 'Once admission is indicated, should it take place within
 48 hours?' Responses in the range 0-100% are referred to as 'any urgency' responses of 70% and over are referred to as 'urgent',
 responses of 90% or more are referred to as 'very urgent'.

•• Indicates the position of the ICD-9 diagnostic code recorded on routine hospital episode statistics data. 'Main' denotes first recorded diagnosis, 'secondary' denotes a secondary diagnosis. A hospital admission for one patient may consist of more than one finished consultant 'episode' (see chapter 4).

^ Admissions here refer to all admissions including readmissions.

^^ 'Marker' and 'potential marker' conditions are defined as shown in table 11.

### 4.2 Other admission groups

A further seven admission groups were defined as follows:

- Group 17 all admissions in surgical specialties;
- Group 18 all admissions in medical specialties;
- Group 19 emergency admissions in surgical specialties;
- Group 20 emergency admissions in medical specialties;
- Group 21 admissions for conditions selected by Billings *et al* in the US as 'ambulatory care sensitive' (recorded as 'main' diagnosis in the first episode);
- Group 22 admissions for conditions selected by Billings *et al* in the US as 'ambulatory care sensitive' (recorded as 'main' or 'secondary' diagnosis in the first episode);

- Group 23 - admissions for conditions selected by Billings et al as 'markers'.

The rationale for selecting groups 17 to 20 was first, to see if there were significant differences between medical and surgical admissions, for which the rates of admissions across EDs would be higher and subject to less random variation than admissions for ACS conditions and markers (in general medical admissions were thought more likely to be ambulatory care sensitive than surgical, as suggested from the findings shown in tables 9-11 and the literature review (see chapter 2)). Second, it was thought useful to investigate whether or not there were differences between all admissions and emergency admissions, in particular to assess the effect on these of access to general

practitioner and hospital facilities. The rationale for selecting groups 21 to 23 was to enable some comparison with the findings made by Billings *et al* (1).

### 3.5 Conclusion

In this chapter the broad categories of admissions to be excluded from the study were identified and the reasons outlined. A list of ACS and marker conditions was identified using a consensus development method. A total of twenty three 'admission groups' of conditions were identified, potentially to be included in the small area analysis described in chapter 6.

Chap	ter outline	
4.1	Introdu	ction
4.2	Method	s
	4.2.1	Checking for duplicate episodes
	4.2.2	Completeness of the total count of activity
	4.2.3	Completeness of specific data fields
4.3	Results	
	4.3.1 Du	iplicate episodes
	4.3.2 To	tal count of activity
	4.3.3 Co	empleteness of specific data fields
4.4	Identify	ing hospitals to be excluded from further analysis
4.5	Discussi	on
4.6	Summa	ry and conclusion

### Chapter 4 Data quality: completeness

Chapi	er outlind	e
4.1	Introd	uction
4.2	Metho	ds
	4.2.1	Checking for duplicate episodes
	4.2.2	Completeness of the total count of activity
	4.2.3	Completeness of specific data fields
4.3	Results	
	4.3.1 D	Puplicate episodes
	4.3.2 T	otal count of activity
	4.3.3 C	Completeness of specific data fields
4.4	Identif	ying hospitals to be excluded from further analysis
4.5	Discuss	sion
4.6	Summ	ary and conclusion

### 4.1 Introduction

Routinely collected NHS information on inpatient and daycase admissions in NHS hospitals was to be used in the small area analysis. A common set of data is collected in all acute NHS hospitals in England for every patient admitted. These data are available from individual providers, health authorities, regional health authorities or the Department of Health, in what is known as the 'hospital episode statistics' (HES) dataset.

The aim of the study described in this chapter was to investigate the completeness of this dataset, and specifically to identify hospitals in which data on inpatient activity were relatively incomplete in order to exclude them from further analysis. It was thought important to spend time analysing the quality of the main dataset used because there has been widespread criticism of its adequacy for research purposes.

### 4.2 Methods

Hospital episode statistics (HES) on all inpatient and daycase episodes occurring in acute hospitals within the former North West Thames region for the years 1991/92, 1992/93 and 1993/4 were obtained from the regional Information Department. The region covered an area whose boundary began at the river Thames in central and West London, and ended in Bedfordshire to the North West of the city – an area covering approximately 3.5 million residents.

Some terminology should be clarified here. In HES, inpatient and daycase activity is recorded as 'finished consultant episodes' (FCEs or episodes) rather than discharges or admissions in individual patients. An *episode* is the length of time a patient spends under the care of one particular specialist while an inpatient in hospital. Episodes are recorded by specialty - the specialty is based on the clinical qualifications of the relevant consultant. The period spent in hospital by a patient (an admission) is called a *'provider spell'*. A provider spell (or admission) consists of one or more episodes of care. Information is collected by financial year. The data included inpatient and daycase episodes which had been coded by 'provider code' and 'provider site', by year, and by specialty. A *provider code* identifies either a single provider unit, or, if the provider unit is linked with one or more smaller

hospitals, the provider code refers to the group of hospitals, such as is the case for many NHS Trusts. *Provider 'site'* identifies individual hospitals. Across the three years a total of 1 925 780 episodes had been recorded, 0.39% of which had no provider coded recorded. An *NHS Trust* is a community or acute facility which had been granted 'trust status' by the Department of Health, allowing it in theory more freedoms and less direct line management from health authorities than other NHS facilities. In 1991 there were a few Trusts in the region under study, but by 1994 most acute and community facilities were NHS Trusts. Primary care facilities were not termed NHS Trusts in the period 1991-94.

Only the large acute general or teaching hospitals (those providing care on-site in most major specialties in 1991-94) in the region were selected for further study. Hospitals which were originally designated as 'special health authorities' between 1991 and 1994, such as the Royal Marsden, the Brompton and Hammersmith Hospitals (all in London), were excluded. This was partly because they were specialist hospitals providing care to patients in one or two specialties and unlikely to admit a patients with a broad range of ACS conditions, and partly because their inpatient activity dataset was known to be relatively incomplete since they only entered the NHS internal market in 1993-94 and had fewer incentives than existing NHS Trusts to improve the completeness of data recording.

For those episodes with a provider code recorded, those providers with the highest number of recorded admissions were selected. Two main difficulties emerged in identifying these providers. First, during the period 1991-4 provider codes and site codes changed for most hospitals, and the names of several hospitals also changed as hospitals became NHS Trusts. Sets of old codes and names and the corresponding new codes and names were obtained from the regional Information Department.

Second, some of the episodes had been assigned a provider or site code which was not exactly the same as that issued by North West Thames regional health authority, although it was similar. For example instead of the correct codes being 'EQM' (provider group code) and '01' (site code), 'EQM01' '1' had been recorded. To ensure the selection of all episodes occurring in each provider, all provider and site codes recorded on the dataset were listed and grouped together where they obviously represented the same hospital or provider group. This was checked with the regional Information Department. This way a list of codes were produced which could be used to identify all episodes occurring in the provider group and

hospitals, the provider code refers to the group of hospitals, such as is the case for many NHS Trusts. *Provider 'site'* identifies individual hospitals. Across the three years a total of 1 925 780 episodes had been recorded, 0.39% of which had no provider coded recorded. An *NHS Trust* is a community or acute facility which had been granted 'trust status' by the Department of Health, allowing it in theory more freedoms and less direct line management from health authorities than other NHS facilities. In 1991 there were a few Trusts in the region under study, but by 1994 most acute and community facilities were NHS Trusts. Primary care facilities were not termed NHS Trusts in the period 1991-94.

Only the large acute general or teaching hospitals (those providing care on-site in most major specialties in 1991-94) in the region were selected for further study. Hospitals which were originally designated as 'special health authorities' between 1991 and 1994, such as the Royal Marsden, the Brompton and Hammersmith Hospitals (all in London), were excluded. This was partly because they were specialist hospitals providing care to patients in one or two specialties and unlikely to admit a patients with a broad range of ACS conditions, and partly because their inpatient activity dataset was known to be relatively incomplete since they only entered the NHS internal market in 1993-94 and had fewer incentives than existing NHS Trusts to improve the completeness of data recording.

For those episodes with a provider code recorded, those providers with the highest number of recorded admissions were selected. Two main difficulties emerged in identifying these providers. First, during the period 1991-4 provider codes and site codes changed for most hospitals, and the names of several hospitals also changed as hospitals became NHS Trusts. Sets of old codes and names and the corresponding new codes and names were obtained from the regional Information Department.

Second, some of the episodes had been assigned a provider or site code which was not exactly the same as that issued by North West Thames regional health authority, although it was similar. For example instead of the correct codes being 'EQM' (provider group code) and '01' (site code), 'EQM01' '1' had been recorded. To ensure the selection of all episodes occurring in each provider, all provider and site codes recorded on the dataset were listed and grouped together where they obviously represented the same hospital or provider group. This was checked with the regional Information Department. This way a list of codes were produced which could be used to identify all episodes occurring in the provider group and

site. For some individual hospitals six different codes were recorded.

Nevertheless, 18 acute hospitals with the most activity were selected in this way. Of these, two (St Albans City and Hemel Hempstead) had effectively merged as an NHS Trust and were operating as one hospital on two sites (with one A&E department) and were treated as one hospital. Another two, Barnet and Edgware General Hospital had also formed into one NHS Trust (the Wellhouse NHS Trust) although were still operating as two separate hospitals (each with its own A&E department). Because of difficulty analysing the completeness of their data by hospital rather than by Trust (see below) in this chapter Edgware and Barnet were analysed as one hospital. This left 16 'hospitals' for analysis.

The data for each hospital were investigated in three ways: the number of *duplicate episodes* were identified and excluded; completeness of the *total count* of episodes recorded was identified; and completeness of data collected in key fields *within each episode* was assessed. The results were used to identify the hospitals with inadequate data, to be excluded from further analysis.

4.2.1 Checking for duplicate episodes

A number of fields of information are collected on HES data, including those in table 15 below.

Episodes having the same data in the following fields within each hospital were identified as duplicate:

- patient ID number (unique to each hospital);
- date of birth (or age);
- spell number;
- episode number;
- spell start date;
- episode start date.

The duplicates were then removed from the HES dataset.

name	description
ADMIDATE	admission date
ADMIMETH	admission method
EPISTART	episode start date
EPIEND	episode end date
EPIDUR	episode duration
EPIORDER	consultant episode number
DISDATE	date of discharge
DISDEST	discharge destination
DOB	date of birth
SPELDUR	spell duration
STARTAGE	age
SEX	sex
ETHNOS	ethnicity
EDCODE	ED code
DIAG01	diagnostic code 1
DIAG02	diagnostic code 2
DIAG03	diagnostic code 3
DIAG04	diagnostic code 4
DIAG05	diagnostic code 5
DIAG06	diagnostic code 6
DIAG07	diagnostic code 7
OPER01	procedure code 1
OPER02	procedure code 2
OPER03	procedure code 3
OPER04	procedure code 4
MAINSPEF	consultant specialty function code
PATID	patient identification number
PRCODE	provider code
SITERET	site code
PROVSPNO	provider spell number
RESHA	HA of residence

### Table 15 Main fields of information collected on HES data

### 4.2.2 Completeness of the total count of activity

To check the total count of activity recorded, the number of episodes recorded on the NWT HES dataset for each of the 16 study hospitals was compared to the corresponding number of episodes recorded on the 'KP70 return' for each of the three study years.

The 'KP70 return' is a manually completed return made by each NHS provider to the Department of Health that summarises the total count of inpatient and daycase activity annually. The KP70 has three parts:

- part 1 shows the total number of episodes carried out in each specialty;
- part 2 shows the total number of episodes for mothers using maternity facilities;
- part 3 shows the total number of episodes of well babies born in hospital.

The relevant part 1 KP70 returns were obtained on diskette from the Department of Health, the part 3 returns were obtained directly from NWT providers.

But before comparisons could be made two difficulties emerged. First, the aim was to assess the completeness of recorded activity in the 16 largest acute *hospitals* in the region. While data were available by individual hospital on the NWT HES database, data were available only for *NHS Trusts* (which might consist of a number of hospitals) on the KP70 returns. During the study period 1991-1994 most hospitals had grouped with smaller ones to form NHS Trusts, each typically comprising one large acute hospital and several small peripheral specialist hospitals or clinics, such as drug rehabilitation units or facilities for care of the elderly, or those with mental illness or mental handicap. Therefore it was necessary to find out which hospitals had become NHS Trusts during 1991-92 to 1993-4, and identify the hospitals which were grouped within each Trust, and their codes. The information was obtained from the NWT Information Department.

However grouping the hospitals into NHS Trusts was not straightforward. This was because many of the peripheral hospitals had changed names and codes over the period, or had closed or merged with another hospital between 1991-94. The picture was particularly confused in London: two teaching hospitals (Charing Cross Hospitals and St Mary's Hospitals) were grouped with many other smaller hospitals, some of which were functioning almost as stand-alone specialist hospitals because they retained their own A&E department. Furthermore, while the hospitals had joined to form NHS Trusts at the beginning of one of the study years, hospitals had closed or merged at different times within the study year. The situation was particularly confused at Charing Cross Hospital, and its subsequent NHS Trust (Riverside NHS Trust): the main hospital site was not operating as a general hospital providing most of the common specialties, because the organisational

arrangements of this NHS Trust were so dynamic (which would make calculating the 'hospital access factor' (to be used in the small area analysis) difficult) it was decided to exclude Riverside NHS Trust entirely from further analysis.

It was not possible to check the completeness of data in the 16 study hospitals without including data from their attached peripheral hospitals. This was potentially a problem because it was thought likely that the completeness of coding would be lower in the peripheral hospitals. It was not possible to solve this problem, but it was alleviated by two factors. Many of the peripheral hospitals served patients admitted into psychiatric care or for mental handicap - the specialties which were excluded from this study and in which very few episodes for ACS conditions were recorded. Also many of the peripheral hospitals had closed between 1991-1994 leaving the main acute hospital as the sole hospital in the provider group.

The second problem related to the fact that the small area analysis was to exclude admissions into the specialties of psychiatry, obstetrics (and related specialties), dental medicine, and admissions in two groups of patients - patients aged 75 years and over and well babies born in hospital (see chapter 3). Ideally comparisons of the total count of episodes recorded on the NWT HES dataset and the KP70 return would exclude these admissions as well. It was possible to exclude from the KP70 return the counts for whole specialties - psychiatry, obstetrics, dental medicine and related specialties – but not for patients aged 75 and older.

Excluding episodes in well babies born in hospital also presented a problem. These episodes were identified on the NWT HES database (most were coded under paediatrics, having a date of birth the same as the admission date) and excluded, but they could not be identified separately in the part 1 KP70 return. However they could be identified from the part 3 KP70 return. These were then subtracted from episodes recorded in paediatrics on the part 1 KP70 return, before comparisons were made between HES and the part 1 KP70 figures for paediatrics.

4.2.3 Completeness of specific data fields (within each episode)

Episodes from each of the 16 study hospitals could be identified from the NWT HES database using the relevant provider codes and site codes.

The main data fields of relevance to this project are:

- date of birth, age and sex (required in later analyses to exclude patients aged 75 and over, to exclude babies born in hospital and to standardise the admission rates by age and sex);
- postcode (to assign the admission to a small area in the subsequent analysis);
- main diagnosis (to identify an admission for ACS and marker conditions).

Thus for each hospital, the following fields were analysed for episodes in all specialties (except obstetrics, psychiatry and related specialties, and dental medicine) and in all ages:

### - age

The proportion of episodes with no age (or date of birth) recorded were identified. The proportion of episodes with an age over 110 years recorded were also identified and assumed to be invalid.

### - sex

The proportion of episodes with no sex recorded was identified.

The following fields were analysed for episodes in patients aged 0-74 years (excluding well babies) in all specialties except obstetrics, psychiatry, dental medicine and related specialties;

### - postcode

The proportion with either a missing postcode or an invalid postcode were identified. To check for invalid codes, postcodes recorded in the relevant episodes were checked against the NHS Users Postcode Directory.

### - main diagnosis

The proportion with either a missing ICD-9 code or a 'dustbin' ICD-9 code (799.9 or 799) in the field of main diagnosis was identified. The diagnosis code 799 or 799.9 is often entered

either when the coding clerk cannot assign clinical codes to the patient discharge, for example because of missing casenotes or if the diagnoses are unclear from the casenotes.

Hospitals were then graded according to the completeness of the total count of activity and the completeness of data fields within each episode. The datasets were examined using Oracle and Excel computer software.

### 4.3 Results

### 4.3.1 Duplicate episodes

The HES dataset contained a total of 1,935,780 episodes recorded between 1991/2 and 1993/4. Of these, 7549 (0.39%) had no provider code and were excluded. In the remainder, 29,938 (1.5%) duplicate episodes were identified - over 20,000 occurring in one of the study hospitals in one year - and removed. 1, 898,293 episodes remained.

### 4.3.2 The total count of activity

Of the 16 study hospitals, 6 were NHS Trusts with a number of attached hospitals. The number of episodes recorded in each of the 16 study hospitals (plus the episodes recorded in peripheral hospitals in the same NHS Trust) were identified and compared to the KP70 return. The results are shown in table 16 and refer to patients aged under *and* 75 years and over. Data for patients in all specialties and all ages are shown, and the name of the main acute hospital, rather than of the NHS Trust, is given. The number of episodes on the HES dataset divided by the number on the KP70 return is shown in the column HES/KP70 as a percentage.

Provider group	Year	Episodes	KP70	HES/KP70 %
Ashford Hospitals	91	20341	18159	112.0
	92	19690	21470	91.7
	93	20015	•	•
Bedford Hospitals	91	30102	29764	101.1
	92	30929	32118	96.3
	93	33838	35108	96.4
Central Middlesex Hospitals	91	28036	27998	100.1
	92	26560	28832	92.1
	93	28907	26155	110.5
Ealing Hospital	91	24669	25189	97.9
	92	22076	26145	84.4
	93	25896	27627	93.7
Hillingdon Hospitals	91	28584	29999	95.3
	92	29674	29288	101.3
	93	32373	32365	100.0
Lister Hospitals	91	40416	34733	116.4
	92	44825	35736	125.4
	93	53126	38614	137.6
Luton & Dunstable Hospitals	91	30206	41970	72.0
	92	31487	43251	72.8
	93	43969	45594	96.4
Mount Vernon Hospitals	91	20616	19416	106.2
	92	22681	22625	100.2
	93	24659	24636	100.1
Northwick Park Hospitals	91	40292	38094	105.8
	92	39728	39402	100.8
	93	38916	38760	100.4
QE2 Hospitals**	91	28903	30738	94.0
	92	30457	30157	101.0
	93	32565	32268	100.9
Charing Cross Hospitals**	91	62457	59470	105.0
	92	73472	64741	113.5
	93	59736	59393	100.6
St Albans City and	91	29722	28564	104.1
Hemel Hempstead Hospitals*	92	30745	30851	99.7
	93	33571	31800	105.6

## Table 16The number of episodes recorded on HES data compared to the<br/>number on the KP70 return, by provider group by year

### Table 16 Continued

Provider group	Year	Episodes	K <b>P7</b> 0	HES/KP70 %
St Mary's Hospitals**	91	62208	55903	111.3
	92	53536	55238	96.9
	93	67763	57033	118.8
West Middlesex Hospitals	91	29869	30155	99.1
in est initialitebert frospitalo	92	30750	30598	100.5
	93	31373	29935	104.8
Watford Hospitals**	91	31254	31718	98.5
	92	32691	32347	101.1
	93	32995	36387	90.7
Edgware & Barnet Hospitals**	91	49388	52304	94.4
	92	51657	49957	103.4
	93	51471	50310	102.3
otal		1739234		

:...

No KP 70 data were available for Ashford hospital in 1992/3 and 1993/4

Indicates where more than one hospital shared the same NHS Trust provider code.

The table shows that a total of 1,739,234 episodes were recorded in the 16 main acute hospitals in NWT - over 92% of the total recorded in the region. There were discrepancies between HES and KP70 returns across most hospitals and across years, although for many hospitals the differences were less than 5%. However discrepancies between HES and the KP70 returns were sometimes larger in individual specialties. Typically, but not always, the number of episodes recorded on HES was lower than recorded on the KP70 returns. Space does not permit a table showing comparisons by specialty in all hospitals and in all three study years. However a fairly typical example is shown in table 17.

Table 17

The number of episodes recorded on HES data compared to the number on the KP70 return in Luton and Dunstable Hospitals, by specialty, 1991-92

	No episodes on	No episodes	HES/KP70
Specialty	NWT dataset	on KP70 retu	:n (%)
General surgery	3414	3446	99.1
Urology	1793	1804	99.4
Trauma and orthopaedics	2398	2422	99.0
Ear, nose throat	1756	1722	99.1
Ophthalmology	1536	1550	99.1
Plastic surgery	3	0	-
Paediatric surgery	278	282	98.6
Accident and emergency	410	421	97.4
Anaesthetics	328	331	99.1
General medicine	6309	6372	99.0
Dermatology	35	36	97.2
Medical oncology	138	139	<b>99.2</b>
Neurology	47	47	100
Paediatrics	3127	7988	39.1
Geriatric medicine	3158	3193	98.9
Gynaecology	3615	3648	99.1
General pathology	1	0	-
Haematology	83	85	97.6

Over all specialties, 72% of episodes on the HES dataset were recorded in the KP70 return in 1991-92. But as seen in table 17 most of the discrepancy occurred in paediatrics, and other excluded specialties (not shown here) such as psychiatry, obstetrics, dental medicine and related specialties. This is an important observation because ACS and marker conditions were mainly recorded in specialties in which the discrepancy was relatively small.

Of the specialties in which episodes for ACS and marker conditions occurred, the discrepancies were particularly wide (>10%) in some years in 7 providers in paediatrics (code 420), as shown in the example above. A problem was that it was not possible to be

confident that the number of episodes in well babies had been excluded from both the HES data and the part 1 KP70 return because the number on the part 3 KP70 return often bore no relation to the number identified on the HES data. In some providers, there were wide discrepancies in paediatrics in all three years. On discussion with information officers in these providers, the likely explanation was that the episodes of well babies born in hospital were often recorded on separate maternity information systems and inadequately transferred to HES, and vice versa. Depending upon the source material used to compile the part 3 KP70 return, the return could easily over- or underestimate the number of episodes for these babies. In other providers there was a marked discrepancy in one particular year, and the explanation given was that there had been a problem transferring information from the maternity information system to the main hospital information system typically because the hospital IT system had been upgraded or replaced in that particular year. The consequence relevant to this study is that it is therefore not possible to make reliable comparisons of the total count of activity in paediatrics between HES and the KP70 return for these 7 providers.

Finally, in four providers there were also major discrepancies between HES and the KP70 in the specialty of gynaecology. Again, discussion with the information officers in each provider revealed that the likely explanation for this was that episodes in this specialty were often recorded on the maternity information system (when gynaecological conditions were treated in maternity patients) and incompletely transferred to the hospital's main HES database. Thus for similar reasons to paediatrics, it was not possible to obtain an accurate picture of the completeness of the activity recorded in gynaecology.

To be more confident about the completeness of the count of activity in specialties in which ACS episodes were recorded, the number of episodes with a main diagnosis of an ACS condition (in patients aged under 75 years and excluding well babies) was identified from the NWT HES database by hospital and specialty. Then the proportion of the ACS episodes occurring in specialties with less than a 10% discrepancy between the HES data and the KP70 return was identified. The results are shown in table 18 below. In the table, the seven providers in which there were marked discrepancies with paediatrics in one or more of the study years are identified. For these providers the figures in brackets in the third column are the findings excluding paediatrics in the years in which activity on HES and KP70 showed the greatest discrepancy. A similar calculation was not made by excluding gynaecology

because the proportion of ACS episodes occurring in the specialty was very small.

Table 18

The percentage of ACS episodes within specialties in which less than a 10% discrepancy was observed between HES and the part 1 KP70 return, in 16 acute providers in NWT, 1991-94\*

Provider group	Year	% of ACS episodes	
		including paediatrics	excluding paediatrics
Ashford^	912	99.3	
	923	87.6	100
	934	KP70 data not available	
Bedford	912	90.7	
	923	99.6	
	934	100	
Cental Middlesex	912	100	
	923	99.7	
	934	53.3	
Charing Cross	912	96.0	
	923	92.6	
	934	82.0	
Ealing^	912	84.4	100
	923	78.1	99.9
	934	84.5	99.4
Edgware and Barnet^	912	99.9	
	923	69.8	83.5
	934	99.9	
Hemel Hempstead	912	83.6	99.2
& St Albans^	923	66.4	87.2
	934	86.4	95.5
Hillingdon^	912	99.1	100
	923	99.4	
	934	99.9	
Lister	912	99.5	
	923	99.2	
	934	99.5	
Luton & Dunstable^	912	83.9	100
	923	84.5	100
	934	86.2	100
Mt Vernon	912	93.8	
	923	84.9	
	934	100	

#### Table 18 Continued

Provider group	Year	% of ACS episodes including paediatrics	excluding paediatrics
Northwick Park	912	98.7	
	923	99.6	
	934	99.9	
QE2^	912	89.5	100
	<b>92</b> 3	100	
	934	100	
St Mary's	912	78.4	
	923	55	
	934	92.9	
West Middlesex	912	99.9	
	923	100	
	934	99.8	
Watford	912	100	
	923	99.1	
	934	98.7	

Less than 100% indicates that there were fewer episodes recorded on HES than the KP70.
 Providers in which there were marked discrepancies between HES and KP70 (>10%) in paediatrics.

The table shows that, in most hospitals, over 90% of ACS episodes were recorded in specialties with less than a 10% discrepancy between HES and the KP70 return.

#### Completeness of specific data fields 4.3.3

Table 19 shows the percentage of missing or invalid data in each episode found in the fields of date of birth, age, sex, postcode, and main diagnosis for each of the 16 main hospitals.

Table 19

The proportion of missing or invalid data in each episode found in the data fields: date of birth; age; sex; postcode; and main diagnosis, by provider, 1991-94.

Ashford         912         0         0         0         0         0         1         1         1           Bedford         912         0.09         0	Hospital	Year	Date of Birth^	Age^ (%)	Sex^ (%)	Postcode* Main* (%)	diagnosis
923         0         0         0         0         1         1.1           Bedford         912         0.08         0         0         0.5         3.5           233         0.04         0         0         0.5         3.5           234         0.03         0         0         2.5         3.1.4           923         0.14         21         0         0.9         2.7           934         0.13         25         0         1.3         18.7           Charing Cross         912         0.13         0         0         0.7         7.6           Ealing         912         0.09         0         0         1.2         9         9           923         0         0         0         0         0         7.6         1.73           Ealing         912         0.22         0         0         0.12         9         1.73           Edgware and         912         0.22         0         0         0.1         2.62           Herrel Hompstead         912         0.21         0.1         0         0         2.62           Herrel Hompstead         912         0.	Ashford	912	0	0	0	0.4	3.5
934         0.09         0         0         0.9         0.71           Bedford         912         0.08         0         0         0.6         11.3           923         0.03         0         0.5         3.5         0.5           Cental Middlesex         912         0.15         7         0         2.5         31.4           923         0.13         25         0         1.3         18.7           Charing Cross         912         0.03         0         0         1.4         14.5           923         0.03         0         0         1.4         14.5         9.7         7.6           Ealing         912         0.03         0         0         1.2         9.7         7.6           Barnet         923         0.22         0         0         0.2         1.73           Edgware and         912         0.22         0         0         0         2.2           Barnet         923         0.10         0         0         2.4         2.62           Hemel Hempstead         912         0.00         0         0         1.1         0.9           924         0.		923	0	0	0	1	1.1
Bedford         912         0.08         0         0         0.6         11.3           923         0.04         0         0         0.5         3.5           293         0.04         0         0         0.5         3.5           Cental Middlesex         912         0.15         7         0         2.5         31.4           923         0.14         21         0         0.9         27           924         0.13         25         0         1.3         1.6           923         0.03         0         0         1.4         14.5           923         0.03         0         0         1.4         14.5           923         0         1         0         2.3         6.1           923         0         0         0         1.2         9           923         0         0         0         1.2         2.9           923         0.22         0         0         0.3         0.2           Ealing         912         0.22         0         0         1.2         2.9           454 (bans         923         0.11         0         0		934	0.09	0	0	0.9	0.71
923         0.04         0         0         0.5         3.5           Cental Middleaex         912         0.15         7         0         2.5         31.4           923         0.14         21         0         0.9         27           934         0.13         25         0         1.3         18.7           Charing Cross         912         0.13         0         0         0.8         11.8           923         0.03         0         0         1.4         14.5           923         0.03         0         0         1.2         9           923         0         0         0         1.2         9           923         0         0         0         1.2         9           923         0.22         0         0         0.2         1.73           Edgware and         912         0.22         0         0         1.2         2.9           924         0.22         0         0         1.2         2.9         2.4           K St Aban         923         0.01         0         0         1.2         2.9           4 St Abans         923	Bedford	912	0.08	0	0	0.6	11.3
934         0.03         0         0         2.8         6.5           Cental Middlesex         912         0.15         7         0         2.5         31.4           934         0.13         25         0         1.3         18.7           Charing Cross         912         0.13         0         0         0.8         11.8           923         0.03         0         0         1.4         14.5           934         0.03         0         0         1.2         9           923         0.03         0         0         1.2         9           923         0         1         0         2.3         6.1           934         0         0         0         1.2         9           923         0.22         0         0         0.3         0.2           Barnet         923         0.23         0         0         0         6           934         0.22         0         0         1.1         2.62           Hereet Herrstead         912         0.01         0         0         1.1         0.9           94         0.06         0         0.1		923	0.04	0	0	0.5	3.5
Cental Middlesex         912         0.15         7         0         2.5         31.4           934         0.13         25         0         1.3         18.7           Charing Cross         912         0.13         0         0         0.8         11.8           923         0.03         0         0         1.4         14.5         9           934         0.03         0         0         1.4         14.5         9           923         0.03         0         0         1.2         9         9           923         0         1         0         2.3         6.1         1.73           Edgware and         912         0.22         0         0         0.1         2.62           Hemel Hempstead         912         0.01         0         0         1.2         2.9           & St Albans         923         0.01         0         0         1.2         2.9           #st Albans         923         0.01         0         0         1.1         0.9           Hemel Hempstead         912         0.01         0         0.19         1.3         2.19           Lister		934	0.03	0	0	2.8	6.5
923         0.14         21         0         0.9         27           934         0.13         25         0         1.3         18.7           Charing Cross         912         0.13         0         0         0.8         11.8           923         0.03         0         0         1.4         14.5           934         0.03         0         0         7.6           Ealing         912         0.09         0         0         1.2         9           934         0         0         0         1.2         1.73         1.73           Edgware and         912         0.22         0         0         0.3         0.2           Barnet         923         0.01         0         0         1.2         1.73           Edgware and         912         0.22         0         0         0.1         2.62           Hemel Hempstead         912         0.01         0         0         1.2         2.9           & \$1 Abana         923         0.11         0         0.02         0.9         1.1           Go 0         0         0         0         0.11         0.9	Cental Middlesex	912	0.15	7	0	2.5	31.4
934         0.13         25         0         1.3         18.7           Charing Cross         912         0.13         0         0         0.8         11.8           923         0.03         0         0         1.4         14.5           934         0.03         0         0         1.4         14.5           923         0         1         0         2.3         6.1           923         0         0         0         1.2         9           934         0         0         0         1.2         1.73           Edgware and         912         0.22         0         0         0.3         0.2           934         0.22         0         0         0.1         2.62         0           Hemel Hempstead         912         0.01         0         0         1.1         0.9           45 K Albana         923         0.11         0         0.27.6         1.1         0.9           934         0.23         0.11         0         0.22         0.9         1.1           912         0.09         0         0         1.1         0.9           106		923	0.14	21	0	0.9	27
Charing Cross         912         0.13         0         0         0.8         11.8           934         0.03         0         0         1.4         14.5           934         0.03         0         0         0.7         7.6           Ealing         912         0.09         0         0         1.2         9           923         0         1         0         2.3         6.1           934         0         0         0         1.2         1.73           Edgware and         912         0.22         0         0         0.3         0.2           Barnet         923         0.21         0         0         0         1.2         1.73           Edgware and         912         0.01         0         0         0.1         2.62           Hemel Hempstead         912         0.01         0         0         1.1         0.9           4.51 Albans         923         0.11         0         0.02         0.9         1.1           934         0.78         0.278         0.8         1.09         1.1           11ingdon         912         0.06         0.19         0.1		934	0.13	25	0	1.3	18.7
923         0.03         0         0         1.4         14.3           934         0.03         0         0         0.7         7.6           Ealing         912         0.09         0         0         1.2         9           934         0         0         0         1.2         1.73           Edgware and 934         912         0.22         0         0         0.3         0.2           Barnet         923         0.23         0         0         0.1         2.62           Hernel Herrspitead         912         0.01         0         0         1.2         2.9           & St Albans         923         0.01         0         0         1.1         0.9           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.06         0         0.1         0.9           934         0.06         0         0         2.6         1.1           934         0.06         0         0.1.9         0.6         1.1           Dunstable         912         2.11         0         0         0         0.1.8         7.91 <td>Charing Cross</td> <td>912</td> <td>0.13</td> <td>0</td> <td>0</td> <td>0.8</td> <td>11.8</td>	Charing Cross	912	0.13	0	0	0.8	11.8
934         0.03         0         0         0.7         7.6           Ealing         912         0.09         0         0         1.2         9           934         0         0         0         1.2         1.73           Edgware and Barnet         912         0.22         0         0         0.3         0.2           Barnet         923         0.23         0         0         0         1.2         1.73           Edgware and 934         0.22         0         0         0.3         0.2         0.24           Barnet         923         0.21         0.0         0         0.1         2.62           Hemel Hempstead         912         0.01         0         0         1.2         2.9           & St Albans         923         0.01         0         0.8         0.01           934         2.78         0         2.11         0.9         1.1           923         0.11         0         0.02         0.9         1.1           923         0.11         0         0         0         0         0           P12         0         0         0         0		923	0.03	0	0	1.4	14.5
Ealing         912         0.09         0         0         1.2         9           Edgware and Barnet         912         0.22         0         0         0.3         0.2           Edgware and Barnet         912         0.22         0         0         0.3         0.2           Barnet         923         0.23         0         0         0         1.2         1.73           Edgware and 934         0.22         0         0         0.3         0.2         6.1           Barnet         923         0.23         0         0         0.1         2.62           Hemel Hempstead         912         0.01         0         0         0.8         0.01           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0         1.1         0.9           923         0.11         0         0.02         0.9         1.1         0.9         1.1           912         0.10         0         0         0         0         1.1         0.23           Lister         912         2.11         0         0         1.8		934	0.03	0	0	0.7	7.6
923         0         1         0         2.3         6.1           934         0         0         0         1.2         1.73           Edgware and Barnet         912         0.22         0         0         0.3         0.2           Hernel Hempstead         912         0.01         0         0         0.1         2.62           Hernel Hempstead         912         0.01         0         0         1.2         2.9           & St Albana         923         0.01         0         0         1.2         2.9           & St Albana         923         0.01         0         0         1.1         0.9           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0         1.1         0.9           934         0.06         0         0.9         1.1         0.9         1.1         0.2           Lister         912         2.11         0         0         1.9         0.6           Dunatable         923         0.04         0         0         0.3         2.1           Mt Vernon         912 <td>Ealing</td> <td>912</td> <td>0.09</td> <td>0</td> <td>0</td> <td>1.2</td> <td>9</td>	Ealing	912	0.09	0	0	1.2	9
9.34         0         0         0         1.2         1.73           Edgware and Barnet         912         0.22         0         0         0.3         0.2           Barnet         923         0.23         0         0         0         0         0           Hernel Hempstead         912         0.01         0         0         0.1         2.62           Hernel Hempstead         912         0.01         0         0         0.8         0.01           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0         1.1         0.9           934         0.06         0         0.19         1.3         2.19           Lister         912         0         0         0         0         1.02           934         0         0         0         0         1.02         1.1         0.23           Lister         912         2.11         0         0         1.9         0.6         1.8         7.91           Mt Vernon         912         0.16         0         0.7         0.3         9.3         0.2		923	0	1	0	2.3	6.1
Edgware and Barnet         912         0.22         0         0         0.3         0.2           Barnet         923         0.23         0         0         0         0         6           Barnet         923         0.23         0         0         0         0.1         2.62           Hermel Hempstead         912         0.01         0         0         1.2         2.9           & St Albans         923         0.01         0         0         0.8         0.01           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0         1.1         0.9           934         0.06         0         0.19         1.3         2.19           Lister         912         0         0         0         0         2.6           934         0         0         0         0.92         0.6         2.1           Lister         912         2.11         0         0         1.9         0.6           Dunetable         923         0.04         0         0         0.7         0.3           934		934	0	0	0	1.2	1.73
Barnet         923         0.23         0         0         0         6           Hernel Hempstead         912         0.01         0         0         0.1         2.62           Hernel Hempstead         912         0.01         0         0         1.2         2.9           & St Albans         923         0.01         0         0         0.8         0.01           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0.11         0.9           934         0.06         0         0.19         1.3         2.19           Lister         912         0         0         0         0         2.6           923         0         0         0         0         0.1         0.23           Luton &         912         2.11         0         0         1.9         0.6           Dunstable         923         1.79         0         0         0         0         1.8         7.91           Mt Vernon         912         0.16         0         0         0.2         0.1         0.2         0.1 <tr< td=""><td>Edgware and</td><td>912</td><td>0.22</td><td>0</td><td>0</td><td>0.3</td><td>0.2</td></tr<>	Edgware and	912	0.22	0	0	0.3	0.2
934         0.22         0         0         0.1         2.62           Hemel Hempstead & St Albans         912         0.01         0         0         1.2         2.9           d. St Albans         923         0.01         0         0         0.8         0.01           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0         1.1         0.9           934         0.06         0         0.11         0.9         0.02         0.9         1.1           934         0.06         0         0.19         1.3         2.19           Lister         912         0         0         0         0         0.23           Luton &         912         2.11         0         0         1.9         0.6           Dunstable         923         1.79         0         0         3.6         2.1           Mt Vermon         912         0.16         0         0         0.7         0.3           934         0.02         0         0         0.4         4.9         9.2           934         0.02         <	Barnet	923	0.23	0	0	0	6
Hemel Hempstead & St Albans       912       0.01       0       0       1.2       2.9         & St Albans       923       0.01       0       0       0.8       0.01         934       2.78       0       2.78       0.8       1.09         Hillingdon       912       0.09       0       0       1.1       0.9         923       0.11       0       0.02       0.9       1.1       0.9         934       0.06       0       0.19       1.3       2.19         Lister       912       0       0       0       0       0.1       0.23         934       0       0       0       0       0.1       0.23       0.1         Lister       912       2.11       0       0       0.1       0.23         Dunstable       923       1.79       0       3.6       2.1         Dunstable       923       0.04       0       0       0.6       0.3         Dunstable       923       0.04       0       0.6       0.3       0.3         934       0.25       0       0       0.4       0.2       0.1         QE2       912 <td></td> <td>934</td> <td>0.22</td> <td>0</td> <td>0</td> <td>0.1</td> <td>2.62</td>		934	0.22	0	0	0.1	2.62
& St Albans         923         0.01         0         0         0.8         0.01           934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0         1.1         0.9           934         0.06         0         0.11         0.02         0.9         1.1           934         0.06         0         0.19         1.3         2.19           Lister         912         0         0         0         0         2.6           934         0         0         0         0         0.19         1.3         2.19           Lister         912         0         0         0         0         0.1         0.23           Luton &         912         0.1         0         0         1.9         0.6           Dunstable         923         1.79         0         0         0.16         0         0.7         0.3           923         0.04         0         0         0.6         0.3         0.2           Mt Vernon         912         0         0         0         0.2         0.1         0.2	Hemel Hempstead	912	0.01	0	0	1.2	2.9
934         2.78         0         2.78         0.8         1.09           Hillingdon         912         0.09         0         0.11         0.9         0.9         1.1         0.9           934         0.06         0         0.11         0         0.02         0.9         1.1         0.9           Lister         912         0         0         0         0         0         0.11         0.23           Lister         912         0         0         0         0         0.1         0.23           Luton &         912         2.11         0         0         1.9         0.6           Dunstable         923         1.79         0         0         3.6         2.1           934         0         0         0         1.8         7.91         0.16         0         0.6         0.3           Dunstable         923         0.04         0         0         0.6         0.3         0.2         0.1           Wetrwon         912         0.16         0         0         0.5         1.1         0.2           QE2         912         0.01         0         0.02         0.4 <td>&amp; St Albans</td> <td>923</td> <td>0.01</td> <td>0</td> <td>0</td> <td>0.8</td> <td>0.01</td>	& St Albans	923	0.01	0	0	0.8	0.01
Hillingdon       912       0.09       0       0       1.1       0.9         934       0.06       0       0.19       1.3       2.19         Lister       912       0       0       0       0       0.22       0.9       1.1         934       0.06       0       0.19       1.3       2.19         Lister       912       0       0       0       0       0.1       0.23         Luton &       912       2.11       0       0       0.1       0.23         Luton &       912       2.11       0       0       1.9       0.6         Dunsable       923       1.79       0       0       3.6       2.1         934       0       0       0       0.7       0.3       0.6         Dunsable       923       0.16       0       0       0.7       0.3         934       0.25       0       0       0.2       0.1         Mt Vernon       912       0.16       0       0       0.2       0.1         923       0.02       0       0       0.2       0.1       0.2         QE2       912       0.01<		934	2.78	0	2.78	0.8	1.09
923         0.11         0         0.02         0.9         1.1           934         0.06         0         0.19         1.3         2.19           Lister         912         0         0         0         0         0         2.19           Lister         912         0         0         0         0         0         0         0.1         0.23           Lister         912         2.11         0         0         1.9         0.6         0.1         0.23           Luton &         912         2.11         0         0         1.9         0.6           Dunstable         923         1.79         0         0         3.6         2.1           Mt Vernon         912         0.16         0         0         0.7         0.3           934         0.25         0         0         0.4         0.9           Northwick Park         912         0.01         0         0.2         0.1           QE2         912         0.01         0         0.2         0.1           QE2         912         0.01         0         0.5         1.1           923         0.01	Hillingdon	912	0.09	0	0	1.1	0.9
934         0.06         0         0.19         1.3         2.19           Lister         912         0         0         0         0         0         0.19         2.6           923         0         0         0         0         0         0.19         0.1         0.23           Luton &         934         0         0         0         1.9         0.6         0.23           Luton &         912         2.11         0         0         1.9         0.6           Dunstable         923         1.79         0         0         3.6         2.1           Mt Vernon         912         0.16         0         0         0.7         0.3           923         0.04         0         0         0.2         0.1           Northwick Park         912         0         0         0.2         0.1           923         0.01         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         0.2           QE2         912         0.01         0         0         0.3         0.7           934         0.01		923	0.11	0	0.02	0.9	1.1
Lister         912         0<		934	0.06	0	0.19	1.3	2.19
923         0         0         0         0         0         0         0.1         0.23           Luton &         912         2.11         0         0         1.9         0.6           Dunstable         923         1.79         0         0         3.6         2.1           Mt Vernon         912         0.16         0         0         0.7         0.3           923         0.04         0         0         0.6         0.3           923         0.04         0         0         0.2         0.1           Mt Vernon         912         0.16         0         0.7         0.3           923         0.04         0         0         0.6         0.3           934         0.25         0         0         0.2         0.1           Northwick Park         912         0         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         0.2           QE2         912         0.01         0         0         0.7           934         0.02         0         0         0.1         1.46 <td< td=""><td>Lister</td><td>912</td><td>0</td><td>0</td><td>0</td><td>0</td><td>2.6</td></td<>	Lister	912	0	0	0	0	2.6
934         0         0         0         0.1         0.23           Luton &         912         2.11         0         0         1.9         0.6           Dunstable         923         1.79         0         0         3.6         2.1           934         0         0         0         1.8         7.91           Mt Vernon         912         0.16         0         0         0.7         0.3           923         0.04         0         0         0.6         0.3           923         0.04         0         0         0.6         0.3           934         0.25         0         0         0.2         0.1           Northwick Park         912         0         0         0.02         0.4         4.9           923         0.01         0         0.02         0.4         0.2           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.1         1.46           Si Mary's         912         0.07         0         0         5.3         18.2           923 <td< td=""><td></td><td>923</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0.1</td></td<>		923	0	0	0	0	0.1
Luton & Dunstable         912 923         2.11 1.79         0         0         1.9         0.6           Dunstable         923         1.79         0         0         3.6         2.1           Mt Vernon         912         0.16         0         0         0.7         0.3           934         0.25         0         0         0.6         0.3           923         0.04         0         0         0.6         0.3           923         0.04         0         0         0.6         0.3           934         0.25         0         0         0.2         0.1           Northwick Park         912         0         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         0.2         0.2           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.1         1.46           Si Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2		934	0	0	0	0.1	0.23
Dunstable         923         1.79         0         0         3.6         2.1           934         0         0         0         0         1.8         7.91           Mt Vernon         912         0.16         0         0         0.7         0.3           923         0.04         0         0         0.66         0.3           934         0.25         0         0         0.66         0.3           Northwick Park         912         0         0         0         0.4         0.25           QE2         912         0         0         0         0.4         0.5         934           Northwick Park         912         0         0         0         1.4         0.5           923         0.01         0         0.02         0.4         0.2           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.3         0.7           934         0.01         0         0         0.3         0.7           934         0.08         0         0         2.1         19.2      <	Luton &	912	2.11	0	0	1.9	0.6
934         0         0         0         1.8         7.91           Mt Vernon         912         0.16         0         0         0.7         0.3           923         0.04         0         0         0.6         0.3           934         0.25         0         0         0.6         0.3           934         0.25         0         0         0.2         0.1           Northwick Park         912         0         0         0         0.4         4.9           934         0.02         0         0.19         0.4         4.9           934         0.02         0         0.19         0.4         4.9           934         0.02         0         0.19         0.4         4.9           923         0         0         0         0.2         0.2           QE2         912         0.01         0         0         0.1         1.16           St Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0	Dunstable	923	1.79	0	0	3.6	2.1
Mt Vernon         912         0.16         0         0.7         0.3           923         0.04         0         0         0.6         0.3           934         0.25         0         0         0.6         0.3           Northwick Park         912         0         0         0         1.4         0.5           923         0.01         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         4.9           934         0.02         0         0.19         0.4         4.9           923         0         0         0         0.5         1.1           923         0         0         0         0.3         0.7           934         0.01         0         0         0.3         0.7           934         0.01         0         0         0.1         1.46           St Mary's         912         0.07         0         0         5.3         18.2           934         0         0         0         2.1         19.2         2.7           West Middlesex         912         0.01         0		934	0	0	0	1.8	7.91
923         0.04         0         0         0.6         0.3           934         0.25         0         0         0.2         0.1           Northwick Park         912         0         0         0         1.4         0.5           923         0.01         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         4.9           934         0.02         0         0.19         0.4         0.2           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.3         0.7           934         0.01         0         0         0.1         1.46           St Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         21.7           934         0.02         0         0	Mt Vemon	912	0.16	0	0	0.7	0.3
934         0.25         0         0         0.2         0.1           Northwick Park         912         0         0         0         1.4         0.5           923         0.01         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         4.9           934         0.02         0         0.19         0.4         0.2           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.3         0.7         934         0.01         0         0.1         1.46           St Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         21.7           934         0.02         0         0         1.11         19.5		923	0.04	0	0	0.6	0.3
Northwick Park         912         0         0         1.4         0.5           923         0.01         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         4.9           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.3         0.7           923         0.01         0         0         0.3         0.7           934         0.01         0         0         0.1         1.46           St Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0         21.7           923         0         3.0         0         0         21.7         934         0.02         0         0.1         1.11		934	0.25	0	0	0.2	0.1
923         0.01         0         0.02         0.4         4.9           934         0.02         0         0.19         0.4         0.2           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.3         0.7           923         0.01         0         0         0.3         0.7           934         0.01         0         0         0.3         0.7           934         0.01         0         0         0.3         0.7           934         0.01         0         0         0.1         1.46           St Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0.1         19.5           923         0         3.0         0         0         21.7         934         0.02         0         0         1.11	Northwick Park	912	0	0	0	1.4	0.5
934         0.02         0         0.19         0.4         0.2           QE2         912         0.01         0         0         0.5         1.1           923         0         0         0         0.3         0.7           934         0.01         0         0         0.3         0.7           934         0.01         0         0         0.1         1.46           Si Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0         21.7           934         0         3.0         0         0         21.7         34         0.02         0         0.1         1.11		923	0.01	0	0.02	0.4	4.9
QE2         912         0.01         0         0.5         1.1           923         0         0         0         0.3         0.7           934         0.01         0         0         0.1         1.46           Si Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0.1         19.5           923         0         3.0         0         0.1         19.5           923         0.01         0         0.1         19.5           923         0         3.0         0         21.7           934         0.02         0         0.1         1.11		934	0.02	0	0.19	0.4	0.2
923         0         0         0         0.3         0.7           934         0.01         0         0         0.1         1.46           Si Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0.1         19.5           923         0         3.0         0         0.1         19.5           923         0.02         0         0.01         1.11	QE2	912	0.01	0	0	0.5	1.1
934         0.01         0         0.1         1.46           St Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0.1         19.5           923         0         3.0         0         0.1         19.5         2.7           934         0.01         0         0         0.1         19.5         2.7		923	0	0	0	0.3	0.7
St Mary's         912         0.07         0         0         5.3         18.2           923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0.1         19.5           923         0         3.0         0         0.1         19.5         19.5           924         0.02         0         0.1         1.11         10.1         11.1		934	0.01	0	0	0.1	1.46
923         0.08         0         0         2.1         19.2           934         0         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0.1         19.5           923         0         3.0         0         0.1         19.5         21.7           934         0.02         0         0         0.1         1.11	St Mary's	912	0.07	0	٥	5.3	18.2
934         0         0         4.5         2.7           West Middlesex         912         0.01         0         0         0.1         19.5           923         0         3.0         0         0.1         19.5         2.7           934         0.02         0         0.1         19.5         1.11		923	0.08	0	0	2.1	19.2
West Middlesex         912         0.01         0         0         0.1         19.5           923         0         3.0         0         0         21.7           934         0.02         0         0         0.1         1.11		934	0	0	0	4.5	2.7
923         0         3.0         0         0         21.7           934         0.02         0         0         0.1         1.11	West Middlesex	912	0.01	- ō	ō	0.1	19.5
934 0.02 0 0 0.1 1.11		923	0	3.0	0	0	21.7
		934	0.02	0	0	0.1	1.11

#### **Table 19** Continued

Hospital	Year	Date of Birth^	Age^ (%)	Sex^ (%)	Postcode* (%)	Main* diagnosis (%)	
Watford	912 973	0.05	0	0	0.4	10.3	
	934	0.02	ŏ	Ő	0.3	0	

^ results refer to episodes in all ages and all specialties.

 results exclude episodes occurring in the following: patients aged 75 years or older; well babies born in hospital; psychiatry, obstetrics, dental medicine and related specialties.

The table shows that in most hospitals, data recording the fields of date of birth or age, sex and postcode was fairly complete, and valid (using the rudimentary tests for validity as described in the methods). The completeness of postcoding was lowest in St Mary's Hospitals. However there were significant gaps in the recording of main diagnosis: six hospitals had greater than 10% of episodes uncoded for main diagnosis in at least one year, and one hospital (Central Middlesex) had more than 30% uncoded.

### 4.4 Identifying hospitals to be excluded from further analysis

From the results shown above, incomplete data recording appeared to be most problematic in the following areas: the total count of activity in specialties in which ACS conditions had been recorded; and the fields with each episode for main diagnosis and postcode.

Some criteria were developed for excluding hospitals from the small area analysis. It was decided that hospitals should be excluded from the small area analysis if there was a discrepancy of 10% or more between HES and KP70 in specialties in which 90% of ACS conditions were recorded (see table 18). For providers in which a wide discrepancy in paediatrics had been noted in particular years, the criterion applied to all specialties except paediatrics in the relevant year. It was further decided that hospitals in which 5% or more episodes had missing or invalid data in the fields of postcode, main diagnosis, age or date of birth and sex (see table 19) should be excluded. The hospitals meeting the relevant criteria or not in each year are shown in the table below – the bold type indicates those meeting the criteria.

#### Provider group Year HES/KP70\* Key episode^ Included <=10%? fields <=5%? in small area (Y/N) (Y/N) analysis? Ashford 912 923 Y Y Y Y Y Y 934 Ŷ Ŷ Ŷ Bedford N Y 912 Y N Y 923 Ŷ Y 934 Ň Ň Cental Middlesex 912 Y N N N Ŷ 923 N N 934 Ň N NNN Charing Cross 912 Y Y N 923 NN 93 Ň Ealing 912 Y N Ň 923 N N Y Y 934 Ŷ Y Edgware and Barnet 91 Ŷ Y Y 923 Ñ Y NY N Y 934 Hemel Hempstead 712 & St Albans 923 N Y Y N Y Ŷ 934 Hillingdon 912 Y Y Y 923 934 Y Y ¥ ¥ Lister 912 923 Y Y Y Y 934 Y Y Luton & Dunstable 912 Y Y Y Y N 923 Y ¥ 934 Ŷ Ν Mt Vernon 912 923 N Y Y N Y 934 Y Northwick Park 912 Y Y Y Y Y 923 Y Y ¥ 934 QE2 912 Y C 923 ¥ Y Y Y 934 Ŷ Ŷ NNY St Mary's 912 N N Y NNY 923 934 West Middlesex 912 N N 923 Y Y N Y N Y

934

#### Table 20 Hospitals meeting the relevant criteria for data quality, 1991-4

### Table 20 Continued

Provider group	Year	HES/KP70* <=10%? (Y/N)	Key episode^ fields <=5%? (Y/N)	Included in small area analysis?	
Watford	912	Y	N	N	
	923	Y	N	N	
	934	Y	Y	Y	

•this column refers to where 90% or more ACS episodes occurred in specialties with less than a 10% discrepancy between the HES data and the KP70 return

^ this column refers to where less than 5% of episodes examined had incomplete data in the fields of date of birth or age, sex, postcode, and main diagnosis.

Using these criteria 28 'hospital-years' of data out of a total of 48 remained, as shown, for he small area analysis.

### 4.5 Discussion

The main aim of the analysis presented here was to gauge the completeness of some of the data recorded on the NWT HES dataset in the 16 largest acute providers, and to exclude from subsequent investigation providers with the most incomplete data.

The study has several limitations with regard to the small area analysis. First, for simplicity, episodes rather than admissions were analysed - admission rates are to be analysed in the small area analysis. However over 90% of admissions in the NWT dataset consisted of only one episode.

Second, the data also included episodes occurring in *non-NWT* residents within the study hospitals (data which would not be used in the later analysis). Episodes in non-NWT patients may have been atypical, for example in patients who could have been transferred into NWT providers because of being more complex cases. However, as discussed with the NWTRHA Information Department, they are more likely to represent patients living across the boundary in neighbouring regions and to be typical cases. In either case it is not clear whether the completeness of data recording would be systematically different in these cases. Third, the analysis did not include episodes of NWT residents occurring in other hospitals within or outside the region, unless they were grouped with the 16 hospitals within NHS Trusts.

Fourth, part of the analysis compared the completeness of activity recorded on HES with that recorded on the KP70 return. The KP70 figures are not a 'gold standard' measure but are the only available standard against which to assess completeness of recorded activity. However since they are compiled from the hospital's own information systems - the same as HES - a question arises of which source of data is the most reliable estimate of activity, and whether discrepancies really indicate a problem of completeness. There may be several 'legitimate' reasons for differences. For example KP70 returns include unfinished episodes (for example patients still in hospital) as well as finished episodes, whereas HES data only includes finished episodes. HES data for each financial year is often submitted later than the KP70 return, allowing more fully coded episodes to be recorded on HES. Where the KP70 figures were lower than the HES data, this could be because: the KP70s were returned before all HES data for the year were completely coded; of errors in manual returns; and because of ambiguity in the criteria for collecting data for KP70. Where the KP70 was higher, this could be because of incomplete HES data. Despite the knowledge that KP70 was not a 'gold standard' estimate - in this study it was decided that discrepancies with HES, in either direction, were likely to indicate problems with the completeness of data recording.

Finally the checks on validity of the data recorded within each episode were somewhat crude. For example episodes with missing data, or 'dustbin' ICD-9 codes (799, 799.9), only were identified for the field of main diagnosis. More sophisticated analysis would have used algorithms to assess the validity of the main diagnosis, for example checking whether gynaecological conditions occurred in men.

Nevertheless there were several useful results. Despite the large amount of resources used to upgrade information systems in providers during 1991-94, it is perhaps surprising that so many duplicate episodes were identified. The information officers in the hospital in which 20 000 episodes were duplicated in one year suggested this had occurred because of moving over to a new hospital information system during the year in question.

The study highlighted that there may be significant under-recording of activity on HES data in maternity specialties, gynaecology and paediatrics.

Perhaps a more worrying finding was the high proportion of episodes without a main diagnosis recorded (or having a 'dustbin' ICD-9 code) in some hospitals, in particular the Central Middlesex Hospital. This may reflect the low priority clinical coding is afforded in many hospitals, possibly because the income of providers in 1991-94 was less dependent upon the accuracy of diagnostic coding than on the accuracy of coding by specialty.

The criteria chosen to exclude some of the hospitals from further analysis were to a large extent arbitrary. They represented a compromise between a desire to include only hospitals with almost totally complete data, and wanting to keep a substantial number of hospitals for analysis. Nevertheless, out of a possible 48 hospital-years of data, 20 were excluded using these criteria.

One issue to be resolved was whether to include in the small area analysis the ACS conditions which were likely to be recorded in the specialities of paediatrics and gynaecology, in view of the large discrepancies between the HES data and KP70 in these specialties. For paediatrics, the problem appeared to be because episodes in well babies born in hospital were inconsistently recorded on HES data. Furthermore the number of ACS conditions identified in chapter 3 which could potentially be recorded in paediatrics, and the proportion of ACS conditions actually found to have been recorded on HES data, were sizeable. For these reasons it was decided not to exclude episodes in paediatrics from the subsequent analysis. For gynaecology, fewer gynaecological conditions had been found to be ACS, and, as noted in chapter 3, several of the panellists were unhappy with classifying these conditions because they were often symptoms which could apply to several conditions. (for example 'irregular menstrual bleeding' or 'excessive menses'). The proportion of ACS episodes recorded in gynaecology was found to be small, although this could have been because of incomplete transfer of information from the hospital's maternity information system. For these reasons it was decided to exclude all episodes recorded in gynaecology from subsequent analysis.

### 4.6 Summary and conclusion

Complicated though the investigation was in this chapter, it uncovered findings that were important to take note of in the small area analysis.

The total count of activity in the 16 study hospitals, and the coding of age or date of birth, sex and postcode were relatively completely recorded on HES data. A larger proportion of episodes had missing or invalid coding for main diagnosis. Hospitals were excluded for further study if there was a discrepancy of 10% or more between HES and KP70 in specialties in which 90% of ACS conditions were recorded, and in which 5% or greater episodes had missing or invalid data in the fields of postcode, main diagnosis, age or date of birth and sex.

20 out of a total 48 possible hospital-years of data were excluded using these criteria. Finally, it was decided that episodes or admissions recorded in the specialty of gynaecology were to be excluded from further analysis.

The investigation in this chapter largely focused on the completeness of data recorded. An evaluation of the quality of diagnostic coding of admissions on HES data is described in the next chapter.

### Chapter 5

r 5 Investigating the accuracy of diagnostic coding of routinely collected hospital admissions data

### **Chapter outline** 5.1 Introduction 5.2 Methods 5.2.1 Selection of hospitals 5.2.2 Selection of case notes 5.2.3 Recoding 5.2.4 Analysis 5.3 Results 5.3.1 Retrieval 5.3.2 **Diagnostic codes** 5.3.3 **Procedure codes** 5.3.4 **Review by regional coder** 5.4 Discussion 5.5 Implications for the small area analysis

### 5.1 Introduction

Having accurate diagnostic information for hospital activity in the NHS is crucial to this study. While the volume of recorded inpatient and day case activity is regularly analysed (277), little is known about the accuracy of clinical coding in routine data, despite the substantial cost to the NHS of collecting them.

The few studies in this area have estimated the 'accuracy' of clinical coding to be between 60-90% (278)(279)(280)(281)(282)(283). Strictly, most of these studies have been concerned with *reproducibility* of the coding process rather than with accuracy of the diagnostic codes, the question at issue being whether different coders, given the same set of medical records, register the same set of codes, rather than involving independent clinical review of patients or test results (284). But interpretation of these studies is difficult because experienced clinical coders have generally not been used to review codes (285)(286) and the reviewers used were often not blind to the original codes recorded. Furthermore, the studies were of data collected before the study period of 1991-94 when the incentives for collecting accurate data were different and hospital information systems less sophisticated.

This study set out to address these methodological issues by using experienced clinical coders who were unaware of the original codes recorded, and by examining inpatient data collected after 1991.

### 5.2 Methods

### 5.2.1 Selection of hospitals

Funding was available to conduct the study in only two hospitals in the NHS region under study - the former North West Thames region. The hospitals chosen - Lister and Luton & Dunstable Hospital were large acute NHS Trusts typical of those serving non-metropolitan areas and which had been selected in chapter 4 for inclusion in the small area analysis.

### 5.2.2 Selection of case notes

Routine hospital episode statistics were obtained from the hospitals for patients discharged between April 1991 and January 1993, the most recent data at the time of the study. 'Finished consultant episodes' of care were grouped to represent hospital admissions. However Luton & Dunstable Hospital only supplied clinical codes relating to the first episode in each admission, and so these only were used for both hospitals. Hospital admissions in several broad admission groups (referred to in chapter 3 and 4) were also excluded.

From the remaining first episodes (37,192 at Lister Hospital, 39,679 at Luton & Dunstable Hospital) five random samples were selected for each hospital:

- a 'general' group containing a random selection of any diagnoses (target: n=385);

- two 'disease-specific' groups representing ACS conditions: asthma (target: n = 165), diabetes (target: n = 165);

- two 'disease specific' groups representing 'marker' conditions: fractured femur (target: n = 85) and appendicitis with appendicectomy (target: n = 85).

The 'general' group was chosen in order to give a general picture as to the reproducibility of clinical coding across a range of diagnoses - including ACS and marker conditions. At the time the study started, the full range of ACS and marker conditions had not been fully identified from the study described in chapter 3. Disease-specific groups of admissions for one or two ACS and marker conditions were chosen to allow a more in depth analysis of coding of two common ACS and marker conditions. The total sample size was 885 for each hospital. In each group, half the sample was drawn from 1991/2 and half from 1992/3 (the most recent data available at the time of the study). In each of the four disease-specific groups, half the sample was drawn with the diagnosis coded as the *main* diagnosis (in coding regulations this is the diagnosis which was the reason why the patient was admitted) and half as a *secondary* diagnosis within the first finished consultant episode. In the 'general' group' approximately 2% of the sample drawn in each hospital had no clinical codes assigned. In these cases the local coders were asked to code the admissions as usual.

### 5.2.3 Recoding

Four clinical coders from a private company - CASPE Healthcare Knowledge Systems (CHKS) - were employed as the external coders. All were experienced and had worked for several years in the NHS as coders, and three had trained other NHS coders.

For each episode, these 'external' coders could record up to 7 diagnoses (using ICD-9 codes) (268), and up to 4 procedures (using OPCS-4 codes (287)). The external coders used the same source material (case notes) as the local coders, and both used the national rules used in the NHS for clinical coding (288)(289). To help coders identify each episode in the casenotes, information was provided showing the specialty, start and end date of each finished consultant episode of care.

Where local and external coders disagreed, the casenotes were reviewed by the most senior coding manager at North West Thames Regional Health Authority. All coders were unaware of codes recorded by others.

### 5.2.4 Analysis

Agreement between external and local coders was identified at two levels: *exact* agreement over the main diagnosis (or main procedure); and *approximate* agreement (based on the first 3-digits of the ICD-9 code for diagnosis, and the letter and first 2-digits of the OPCS-4 code for procedure).

Kappa statistics were calculated for the 'general' groups, but not for the disease-specific groups because these did not provide a full 'square' design: for example, there were no data on the numbers of cases which were not coded as asthma by the local coders, but would have been coded as such by the external coders. Percentage agreement was calculated for all groups, with 95% confidence intervals for levels of approximate agreement. The sequencing of codes was also compared.

#### 5.3 Results

#### 5.3.1 Retrieval

Of the 885 discharges chosen in each hospital, the case notes of 804 were found in Lister Hospital (90.8%) and 803 (90.7%) in Luton & Dunstable Hospital.

#### 5.3.2 **Diagnostic codes**

1

The level of agreement (expressed as the percentage agreement between the external and local coders over the codes for main diagnosis) is shown in columns (a) and (b) in table 21.

Fable 21	Percentage agreement by external coders over the main diagnosis and
	main procedure codes as identified by local coders

		Local main code same as external code		Local main code same as external secondary code		Local main code not recorded by externals	
Group		Exact*	Approximate*	Exact*	Approximate**	Exact*	Approximete**
	Hospital	(a)	(b)	(a)	(b)	(a)	(b)
General	A	43	55 (50.5-61.2)	5	7	52	38
	в	60	72 (66.8-76.4)	5	8	35	20
Asthma^		35	86 (75.3-92.3)	3	4	62	10
	в	53	91 (81.1-95.9)	4	6	43	3
Diabetes*		6	70 (58.3-86.2)	13	25	81	5
	В	9	75 (63.8-88.4)	13	22	78	3
Appendicitis^		51	78 (72.7-90.9)	3	6	46	16
	в	65	80 (68.8-88.4)	1	1	34	19
Frectured femur*	•	47	84 (72.7-90.9)	4	4	49	12
	в	30	89 (73.6-96.5)	0	0	70	11
Main procedure		52	73 (65.9-78.3)	7	8	41	19
	В	69	82 (75.3-86.4)	4	4	27	14

The 95 per cent confidence intervals are shown in parentheses

• For diagnoses: identical ICD-9 four-digit codes; for procedures: identical OPCS codes

\*\* For diagnoses: first three digits of ICD codes identical; for procedures: the first two characters of the OPCS codes identical. ^ Result shown in these disease- specific groups are for where the disease was locally coded as the main diagnosis.

Column (a) shows that, in the 'general' group, the *exact* agreement over the code for main diagnosis was 43% at Lister Hospital and 60% at Luton & Dunstable Hospital. Agreement was slightly higher for fractured femur and appendicitis but the possibility of chance agreement was greatest in these groups because fewer categories were involved. Exact agreement was very low for diabetes. Column (b) shows that *approximate* agreement was higher for all groups, as expected, and very good for asthma (agreement of 86% and 91% in the two hospitals) and fractured femur. Kappa values for the 'general' groups were 0.54 for Lister Hospital, and 0.72 for Luton & Dunstable Hospital. These values are very close to the percentage agreement because, given the large number of categories used and reasonably even spread of observations across them, the probability of chance agreement was small. This improvement in reproducibility was striking for diabetes; the local coders in both hospitals differed from the external coders in their use of the fourth digit of the ICD-9 code.

Procedure codes were investigated for first episodes in the 'general' groups. The figures in the last two rows of table 21 refer to records in which both sets of coders had recorded at least one procedure in the first episode. Disagreement about whether a particular procedure was the main or a secondary one was relatively uncommon.

A second analysis was done including all records in the 'general' groups. In this analysis, if neither coder recorded a particular procedure, this counted as an agreement; if one did and the other did not, this was a disagreement. For Lister Hospital, there was exact agreement for 58% and approximate agreement for 70% (kappa = 0.66). For Luton & Dunstable Hospital, the corresponding figures were 76% and 83% (kappa = 0.80). The adjustment involved in the kappa statistic related almost entirely to the possibility of chance agreements that there had been no procedure.

To identify how far disagreements in diagnostic codes were due to differences in *sequencing* of codes, we investigated whether the main diagnosis recorded by the local coders had been recorded by the external coders as a secondary diagnosis instead. The results are shown in table 21 in columns (c) and (d). For the 'general' group from Lister Hospital, the locally-coded exact main diagnosis was identified as a secondary diagnosis by the external coders in 5% of cases, but in 52% it appeared nowhere on the externally coded

list of diagnoses. The results were similar for the other disease groups at Lister Hospital except for diabetes, where the proportion recorded as a secondary diagnosis by the external coders was higher. The results for Luton & Dunstable Hospital were also similar. In general, therefore, disagreements in coding did not appear to be due to sequencing differences. The external coders tended to code more secondary diagnoses or comorbidities than the local coders in both hospitals, especially for chronic conditions like diabetes.

Where there was not even approximate agreement over the main diagnosis, the disagreements fell into the following five broad categories:

- where one group of coders had recorded a symptom, and the other had recorded a diagnosis; related to the symptom; eg abdominal pain (ICD-9 code 789.0) and acute appendicitis (540.9);
- (2) where both coders had recorded codes for very similar if not identical conditions, often where one code was more precise than the other; eg diabetes (250.0) and diabetes in pregnancy (648.0); acute appendix (540.9) and appendix unqualified (541); pneumonia (486) and pneumococcal pneumonia (481);
- (3) where there was obvious disagreement between coders over similar conditions; eg acute appendicitis (540.9) and mesenteric lymphadenitis (289.2), chronic tonsillitis (474.0) and acute tonsillitis (463); emphysema (492) and asthma (493.9);
- (4) where coders had recorded codes for conditions which were not similar but obviously related; eg diabetes (250.0) and cellulitis (682.7), phlebitis of the deep vessels of the lower extremities (415.1) and pulmonary embolus (451.1);
- (5) where coders had recorded completely different conditions; eg bronchopneumonia (485) and fracture of the vault of the skull (800.1);
- (X) where external coders were not able to reach a decision and had not recorded a main diagnosis.

The proportion of disagreements in each category at each hospital is shown in table 22; the disagreements were spread evenly across all categories in both hospitals, but the proportion of cases in category X was lower at Luton & Dunstable Hospital.

Table 22

The proportion of cases in each category\* as a percentage of all disagreements (where external and local coders did not agree over the first three digits of the ICD-9 code for main diagnosis) and as a percentage of all cases reviewed

Category	Hospital A		Hospital B		
	% of disagreements (n= 295)	% of all cases reviewed (n=804)	% of disagreements (n=213)	% of all cases reviewed (n=803)	
1	13.6	5.0	19.1	5.1	
2	17.6	6.5	11.2	3.0	
3	23.4	8.6	29.3	7.7	
4	10.5	3.8	15.8	4.2	
5	18.3	6.7	19.5	5.2	
x	16.6	6.1	5.1	1.2	
Total	100	36.7	100	26.4	

The results are shown for all disease groups combined.

\* For a description of categories, see text.

For all disease groups together, the level of exact agreement over the main diagnosis increased between 1991/2 and 1992/3 from 38% to 42% at Lister Hospital and from 68% to 79% at Luton & Dunstable Hospital. The increase was statistically significant for Luton & Dunstable Hospital only (p<0.05).

Table 23 shows how the external coders coded asthma and diabetes where local coders had recorded these as secondary diagnoses. In most cases both sets of coders agreed that the diabetes and asthma were secondary diagnoses. However, for asthma, the external coders thought that in 17% (Lister Hospital) and 18% (Luton & Dunstable Hospital) of cases, this was the main, rather than the secondary diagnosis. Also the external coders had not coded asthma at all in 9% (in Lister Hospital) and 14% (Luton & Dunstable Hospital) of discharges in this group.

### Table 23

Percentage agreement between external coders and local coders over asthma and diabetes when these were coded by local coders as secondary diagnoses; for disagreements, the percentage of cases where the external coders had recorded asthma or diabetes as a main diagnosis or not at all

	Agreement	Disagreement		
	Secondary diagnosis (3-digit ICD-9 code) %)	ain diagnosis (3-digit ICD-9 code) (%)	Not coded (3-digit ICD-9 code) (%)	
Hospital A				
Disease group				
Asthma	74 (62.4-83.0)	17 (9.7-27.7)	9 (3.9-18.3)	70
Diabetes	88 (78.1-94.0)	7 (2.7-15.9)	5 (1.56-13.3)	76
Hospital B				
Disease group				
Asthma	68 (55.9-78.3)	18 (0.3-29.9)	14 (7.3-24.7)	72
Diabetes	92 (83.4-96.6)	4 (1.1-11.4)	4 (1.1-11.4)	83

The fractured femur and appendicitis (with appendicetomy) groups are omitted because of small numbers. 95% confidence intervals are shown in parentheses.

### 5.3.3 Review by the regional coder

Table 24 shows that where there was not even approximate agreement over main diagnosis, the third coder disagreed with both local and external coders in a high proportion of cases, but more so with the local than with the external coders. The differences were statistically significant (p<0.05) at Lister Hospital but not at Luton & Dunstable Hospital.

# Table 24 The percentage agreement on the main diagnosis (on the first three digits of the ICD-9 code) between the regional coder and the local coders and external coders

	Agreement of regional coder with:				
	Local coders	External coders	Neither	n	
Hospital A	17(13.0-21.9)	30(24.9-35.6)	53 (47.1-58.8)	295	
Hospital B	27(21.3-33.6)	38 (31.4-44.9)	35 (28.7-41.9)	213	

The 95% confidence intervals are shown in parentheses.
### 5.4 Discussion

The *accuracy* of clinical codes depends upon how closely they reflect the clinical condition of the relevant patient, (path A in figure 4). However this study involved comparison of two sets of codes independently abstracted from the casenotes (path B in figure 4), and is thus a study of the reproducability of the coding process.

The results will not allay the widespread suspicions of the full clinical codes on the hospital episode statistics (HES) dataset. However, agreement about the first three characters in diagnostic and procedure codes was relatively good, particularly for certain conditions such as asthma. This, and the significant increase in agreement between years in Luton & Dunstable Hospital, is encouraging. The pattern of results was similar in both hospitals and reflects the results of earlier studies noted earlier.

Also table 22 indicates that for diagnoses, the differing codes chosen were for related conditions in a substantial proportion of cases. The disagreements in categories 5 and X are the most worrying. There may have been legitimate reasons for disagreement. For example the external coders, who coded 3-26 months after discharge, may have had access to information added to the notes after the local coding had been done, while the local coders may have had the benefit of querying the diagnoses 'on the spot' with clinicians.

The 'general' groups are nearest to being a presentation of the overall population of hospital discharges. Agreement in this group was lowest possibly because of rare, complex and ill-defined cases that were difficult to code. Exact agreement was higher for the more acute conditions selected - asthma, appendicitis and fractured femur. Diabetes seemed to present particular problems in the way the fourth digit of the ICD-9 code was used by different coders.

Not examined in the study was the extent to which the main diagnosis, as coded on hospital episode statistics data, really did represent the cause of admission. Also unexamined was the extent to which the main diagnosis was a code representing 'symptoms and signs' which may have been due to an ambulatory care sensitive or marker condition.





What do the results imply about the quality of coding in the hospitals studied? Interpretation depends partly upon the quality of the external coders. All had coded for more than four years, and three had trained NHS regional and hospital coding managers in courses run by the Department of Health. In the past, CHKS coders have been found to be reliable (285), and in this study the proportion of disagreements over main diagnosis and procedure was consistent across all four coders. However they cannot be regarded as providing a gold standard. The very experienced regional 'third' coder agreed more often with the external than with the local tearns, but often disagreed with both.

Also it is not possible with a study of this kind to assess the extent to which coding differences result from poor quality information in case notes or coding errors. The clinical information in the casenotes was frequently ambiguous, missing, illegible, or poorly organised (problems that have been noted elsewhere (290)). This was especially so in Lister Hospital, where there was more disagreement between all three groups of coders (table 24) and more disagreements in category X. While diagnosis is often complex and genuinely uncertain, it is the clinician's responsibility to document clearly codable conditions (291) and it is not for coders to try to work out what the diagnoses are. But even if clinical information is unambiguous, there may be intrinsic uncertainty in the process of clinical coding - in assigning and sequencing codes - and some level of disagreement between coders seems inevitable.

Could similar results be expected elsewhere? Neither hospital had unusual features, the pattern of results in each was broadly similar, and the results mirror those found in other studies noted above. As such the results presented here may well be generalisable, although there may well be greater variations between other acute hospitals in the region which were not included in this study. However some admissions excluded from the sample such as those in older people may be relatively difficult to code and so the results may underestimate the level of agreement across *all* admissions in acute hospitals. The case notes of approximately 9% of the sample could not be found, and they too could have been more difficult to code, although they were spread across all disease groups, specialties and ages and there was no evidence to suggest they were different to the admissions reviewed. Also only first episodes in the admissions were reviewed – secondary episodes may have

been more difficult or easier to code. But since typically more than 90% of admissions on HES data contain just one episode, the effect of secondary episodes on the results is likely to have been very small.

### 5.5 Implications for the small area analysis described in this thesis

As noted above, this study did not investigate directly the *accuracy* but the *reproducibility* of clinical coding between two groups of coders. Furthermore, the reproducibility of coding was investigated in only two of the potential 16 main acute hospitals in NWT region whose data were to be used in the subsequent small area analysis.

The results were therefore only partially illuminating as to the accuracy of clinical coding of hospital episode statistics for the potential 23 admission groups, and the data in the other 14 main acute hospitals, to be used in the small area analysis. It was not possible, or appropriate, to use the findings from this study directly, for example in a sensitivity analysis in the small area analysis. The findings can at best provide a greater understanding as to the potential reasons for observed variations in admissions for specific conditions due to clinical coding.

# Chapter 6

# Small area analysis

6.1	Introdu	ction
6.2	Method	8
	6.2.1	Identifying small areas
	6.22	Building the dataset
		(a) Dependent variable
		(i) Admissions per ED
		(ii) calculating age-sex standardised admission rates
		(iii) Palsson regression
		(b) Independent variables
		(i) SIR and SMR
		(ii) Socioeconomic deprivation
		(iii) age group and sex
		(Iv) Hospital and GP access factors
		(v) Hospital service area
6.3	Results	
	6.3.1	Selection of data for analysis
		(a) Number and population of EDs
		(b) Assignment of EDs to hospital service areas
		(c) Selection of HSAs for further analysis
	6.3.2	Univariate analyses and comparisons by HSA
		(a) Admission rates
		(b) Carstairs score
		(c) SMR and SIR
		(d) GP and hospital access factors
	6.3.3	Bivariate analysia
		(a) Correlation between main independent variables
		(b) Correlation between admission rates and socioeconomic deprivation
	6.3.4 M	ultivariate analysis using Poisson regression
		(a) developing the model
		(b) running the Polsson regression model
		(i) SMR, SIR and Carstairs index
		(ii) age groups
		(iii) sex
		(iv) GP and hospital access factors
		(v) hospital service area
6.4	Discussi	a <b>.</b>
	6.4.1	Methodological lasues

### 6.1 Introduction

This chapter describes the main analysis of the thesis - a small area analysis of admission rates for ACS and marker conditions. As outlined in chapters 1 and 2, there were three broad aims of the analysis:

- to investigate the relationship between admissions for ACS and marker conditions and socio-economic deprivation;
- to investigate the relationship between socio-economic deprivation and access to ambulatory care (in this case primary care);
- to investigate, in a multivariate analysis, the influence of several other factors –
   morbidity, access to hospital and general practice facilities and the 'hospital effect' –
   on admissions for ACS and marker conditions (as identified in chapter 3).

The geographical area under study was the former North West Thames NHS region - an area covering approximately 3.5 million people - and admissions data for 1991/2, 1992/3 and 1993/4 were used.

In this chapter the methods and main results are described and discussed.

### 6.2 Methods

### 6.2.1 Identifying small areas

Data identifying the geographical boundaries for the North West Thames region were obtained from the regional Information Department. Geographical boundaries for the enumeration districts (EDs) from the 1991 census were obtained from the Office for National Statistics. Approximately 70 EDs straddled the boundaries of the NHS region and other regions and were excluded from the analysis, leaving 8094 for analysis.

### 6.2.2 Building the dataset

To provide a basis for the analyses, a dataset was constructed containing admissions by enumeration district (ED) for various groups of diagnoses (the dependent variable) and eight main independent variables: standardised mortality ratio (SMR); standardised illness ratio (SIR); socio-economic deprivation (Carstairs index); age group; sex; access to hospital facilities; access to general practitioner services; and 'hospital service area'. The dataset was prepared by assigning a score to each ED for each variable as described below.

### (a) The dependent variable

### (i) Admissions per ED

As explained in chapter 4, routine hospital data in the UK NHS (hospital episode statistics or HES data) contain information on activity recorded as 'finished consultant episodes' (FCEs) rather than admissions. All FCEs for NWT residents during the study period (1991/2, 1992/3 and 1993/4) were obtained. FCEs occurring in providers within the NWT region were obtained from the regional Information Department (as described in chapter 4). FCEs occurring in providers outside the region were obtained from: North East Thames (NET) Regional Health Authority (covering admissions in providers in NET region); South West Thames (SWT) Regional Health Authority (admissions in SWT region); and Mersey Regional Health Authority (admissions in all other providers in England by NWT residents). These data were merged into one dataset. Data from private hospitals were not sought, partly because it was likely that the numbers of admissions for ACS and marker conditions would be very small (most admissions to private providers are for elective surgery (292), data would be difficult to obtain, and the quality of data (particularly postcode and diagnostic code) were unknown.

Duplicate episodes were removed (see chapter 4), as were FCEs for patients aged 75 and older, in well babies born in hospital and in the specialties of obstetrics, psychiatry, dental medicine, gynaecology and related specialties.

The remaining FCEs were grouped to represent admissions in individual patients. This was done for each acute provider listed in the HES dataset by matching the following data fields: patient identification number; age/date of birth; FCE start date; FCE end date; and FCE number. Across the whole dataset, the ratio of admissions to FCEs was 1:1.08.

In chapter 3, a total of 23 potential groups of conditions were identified for further analysis. It was thought that some of these should be included in the small area analysis because there might be important differences in the pattern of admission rates for strong and weak ACS conditions and markers, with different levels of urgency of admission. To reduce the admission groups to a more manageable number, groups 2, 4, 6, 8 and 22 were excluded. This was because ACS conditions may be recorded on HES as a main *or secondary* diagnosis, but as noted in chapter 5, the condition causing admission is most likely to be coded as a *main* diagnosis. A total of 18 admission groups remained for further analysis as summarised in the table below.

Admission	Description	Admissions	% of	
group			all admissions	
1	Strong and weak ACS conditions, any urgency, all	87669	25.92	
3	Strong and weak ACS conditions, any urgency, readmissions	25278	7.47	
5	Strong and weak ACS conditions, urgent, all admissions	31424	9.29	
7	Strong and weak ACS conditions, urgent, readmissions	7719	2.28	
9	Strong ACS conditions, any urgency, all admissions	30408	8.99	
10	Strong ACS conditions, any urgancy, readmissions	14738	4.36	
11	Strong ACS conditions, urgent, all admissions	16341	4.83	
12	Strong ACS conditions, urgent, readmissions	3685	1.09	
13	Strong ACS conditions, very urgent, all admissions	20252	5.99	
14	Strong ACS conditions, very urgent, readmissions	10256	3.04	
15	Strong and weak markers, very urgent, all admissions	14577	4.31	
16	Strong markers, very urgent, all admissions	13566	4.01	
17	Admissions in surgical specialties	174736	51.66	
18	Admissions in medical specialties	167161	48.42	
19	Emergency admissions in surgical specialties	14690	4.34	
20	Emergency admissions in medical specialties	33769	9.98	
21*	Admissions for ACS conditions defined by Billings	3994	1.18	
23*	Admissions for marker conditions defined by Billings	6176	1.83	

# Table 25 Admission groups selected for further analysis, number of admissions in each group and percentage of all admissions, 1991/2

 The list of ACS and marker conditions drawn up by Billings' et al comprised ICD-9CM (Clinically modified) codes. UK HES data contains ICD-9 codes, which lack the fifth digit found for some ICD-9 CM codes (which add precision to a diagnostic code) All ICD-9CM codes with the fifth digit were dropped from Billings' et al groupings – this affected only a small number of conditions. As shown in the table, ACS admission group 1 comprised almost 26% of all admissions per study year, and admission group 16 (markers) approximately 4%.

Using the postcode recorded on the HES dataset, each admission was allocated to an enumeration district using a postcode-enumeration district 'look-up' table obtained from the Office for National Statistics.

(ii) calculating age-sex standardised admission rates

Age-sex standardised admission *rates* per ED were calculated for each of the 18 admission groups for the preliminary univariate and bivariate analyses.

Data on population size in the seven age bands by sex for all EDs in NWT region were obtained from the Office for National Statistics, using data from the 1991 census. The data were adjusted to allow for the estimated undercounting in some age groups using adjustment factors supplied by ONS (293). In the most extensive previous small area analysis in England (conducted by researchers at the University of York), the impact on the results of using direct standardisation rather than indirect were found to be negligible (99). The direct method was used here (by applying local rates to the England & Wales population in 1991). All the rates were expressed as admissions per 1000 population.

### (iii) Poisson regression

To examine the influence of the independent variables on variations in admissions for the 18 admission groups, a multivariate Poisson regression analysis was performed. A Poisson regression rather than other forms of regression was conducted mainly because the numbers of events (admissions) in the EDs were small.

In this analysis, the *number* of admissions for each age and sex group was the dependent variable. This was related to the population number in each age and sex group for each ED and the independent variables described below.

### (b) Independent variables

### (i) Standardised mortality ratio (SMR) and standardised illness ratio (SIR)

Data on mortality (for persons under 75 years all causes) by ED for five years 1988-1992 (the most recent years available at the time of the study) were obtained from ONS. The SMR was calculated for each ED using indirect standardisation. Indirect standardisation was used because this method is typically used in small area studies for these variables (York), having lower variance in scores per ED compared to the direct method. Mortality data for five years, rather than one year, were used with the aim of reducing random variation in the SMR for ED populations.

The standardised illness ratio (SIR) was calculated using responses to the question on limited longstanding illness from the 1991 census (Scrivener 96). Age-sex standardised SIRs were calculated for each ED again using the indirect method.

### (ii) Socio-economic deprivation

Data on socio-economic deprivation by ED were obtained from the ONS, derived from the 1991 census. A composite index of socio-economic deprivation - the Carstairs index - was calculated for each ED. The Carstairs index includes variables reflecting unemployment, proportion of households with no car, proportion of the population in a low socioeconomic class, and overcrowding (20). Carstairs scores generally range from -3 to +2, higher scores (ie the less negative) indicating higher levels of socioeconomic deprivation.

### (iii) Age group, and sex

In the Poisson regression analysis, the numbers of people in each age and sex group per ED are treated as a particular kind of independent variable. In the dataset, a row of data is used for each age and sex group for each ED, with age and sex group indicated by single codes. For example using seven age groups (see below) and two sex groups, there were a total of 14 rows of data per ED showing the number of admissions by age and sex group.

### Seven age groups were identified as follows:

Ages (years)	Age group
Under 1 year	1
1-4	2
5-14	3
15-24	4
25-44	5
45-64	6
65-74	7

Two sex groups were:

Sex	Group
Female	1
Male	2

### (iv) Hospital service area

Hospital service areas (HSAs) (as defined in chapter 2) were identified for 17 main acute hospitals in the North West Thames region - those examined in chapter 4 but with Edgware and Barnet separated. As explained in chapter 4, in the analysis Charing Cross Hospital (Riverside NHS Trust) was excluded, although the number of EDs in this HSA is shown in the tables below.

(v) Hospital and GP access factors

A formula was used to calculate a score per ED that estimated the access of residents of the ED to GP and hospital facilities respectively. The formula was very similar to that developed by the University of York as part of a project to refine the allocation formula for NHS resources across England, as discussed in chapter 2 (99)

The formula used was:

$$HA(ED_i) = \sum_{h=1}^{3} \left[ Beds_h / (d_{ih} + d_{offset})^2 / \sum_{i=1}^{n_h} (population_i / (d_{ih} + d_{offset})^2) \right]$$

Where:

HA(ED <sub>i</sub> )	-	Hospital access factor for ED i
$\sum_{h=1}^{3}$	-	sum over the 3 hospitals h taking the most admissions from this ED
Beds h	-	the number of acute beds in hospital h
d in	-	distance (km) from centroid of postcode of hospital h to centroid of ED i
doffset	-	distance (= $5$ km) to be added to $d_{ih}$ to avoid very high access near provider
$\sum_{i=1}^{n_{k}}$	-	sum over the $n_h$ EDs in the service area of hospital h
population ,	-	population of ED,

The model was based on the number of beds in hospitals serving the ED divided by the square of the distance from the ED to each hospital, taking into account the populations in other EDs who may 'compete' for those hospital beds. The square of the distance was used because this is typically used in calculations of this sort, and it was used in the formula developed by the University of York as referred to above. The offset' distance of 5km means that 5km was added to each distance to avoid implausibly high access factors when the distances involved were very small, for example in EDs very close to the hospitals. The figure of 5km was necessarily arbitrary – for example 10 or 20km could also have been used. However the shorter cut-off distance was used because of the density of the population and hospitals in London.

The resulting hospital access factor score (termed HAF2) was calculated for each ED. The higher the score, the higher the presumed access.

The GP access factor was calculated in a very similar fashion, except GP practices were used instead of hospitals, and the supply of full-time-equivalent (FTE) GPs instead of beds. The formula used was:

$$GA(ED_i) = \sum_{g=1}^{n_i} \left[ FTE_g / (d_{ig} + d_{offset})^2 / \sum_{i=1}^{n_i} (population_i / (d_{ig} + d_{offset})^2) \right]$$

GA(ED <sub>i</sub> )	-	GP access factor for ED i
$\sum_{g=1}^{n_{i}}$	-	sum over the n, GP practices g in the ward containing ED,
FTE	-	the number of FTE GPs in practice g
d <sub>ig</sub>	-	distance (km) from centroid of postcode of practice $\mathbf{g}$ to centroid of $\mathbf{ED}_i$
doffeet	-	distance (= 5km) to be added to $d_{ig}$ to avoid very high access near provider
$\sum_{i=1}^{n_j}$	-	sum over the n, EDs in the ward containing ED i.
population,	-	population of ED,

Again this formula is similar to that developed by the University of York in the work referenced above. However it was not clear from the University of York work which 'offset' distance was used. 5km was chosen here.

Also, because few studies have developed a formula in such a way to measure access to hospital or general practice or primary care, it was not clear whether to take the square of the distance, or some other power. We opted *a prioi* to use the square of the distance in the analysis (the scores were termed GAF2 or HAF2), but also did some exploratory analysis using GAF (distance unsquared), GAF4 (distance to the power of 4), GAF6 (power of 6) and HAF4 (power of 4).

### 6.3 Results

- 6.3.1 Selection of data for analysis
- (a) Number and population of EDs

A total of 8094 EDs were located wholly within the boundaries of the former NWT region in each of the study years. 340 EDs had no recorded population. The distribution of

population in the remaining 7754 is shown below - the mean population size was 429.



### Figure 5 Distribution of EDs by population size

### (b) Assignment of EDs to hospital service areas

For each ED, the total number of admissions of NWT residents in each study year was identified, by hospital. All 7754 (populated) EDs in NWT region were then assigned to a particular hospital on the basis that the *largest proportion* of residents of the ED had occurred in that particular hospital (the 'plurality' rule) (262). The number of EDs assigned in this way to each of the 17 main acute hospitals in the former NWT region are shown below in the table below (column (a)).

Column (a) in table 26 (i) shows that of the total 7754 EDs with a population, in 6956 (89.7%) the largest proportion of admissions from the ED occurred in one of the 17 main acute hospitals within the region. In the remaining 794 EDs (approximately 10%) the largest proportion had occurred in a hospital outside of the region. This proportion was similar across the three study years.

As a next step, EDs with *less than 40%* of admissions occurring in the 17 main hospitals were identified (shown in column 'b') and excluded, as were EDs with a population too small (<100) to calculate a Carstairs score (Column 'c'). The number and proportion of EDs remaining made up each of the 17 hospital service areas (HSAs) as shown in columns 'e' and 'f'. On average less than 7% of EDs were excluded because of fewer than 40% of admissions occurring in the HSA (column 'f'). This was judged to be an acceptable proportion. However in several hospital service areas - Barnet, Ealing, Edgware, Central Middlesex – over 10% EDs were excluded using the 40% rule. These hospitals are located in a similar area – west and north west of London – and residents in local EDs may have tended to have been referred to inner London hospitals for admission.

The geographical location of EDs in each of the 17 HSAs in 1993/4 is shown in appendix A. The dots in the map show the geographical centroid of the ED rather than the boundaries, and dots of the same colour relate to the same HSA. The number and pattern of EDs within the HSAs remained fairly constant over the three study years because the flow of patients to the 17 study hospitals remained largely unchanged (data not shown).

		A	В	C	D	6	1
					b+c	a - (b+c)	((a - (b+c))/a)*100
Acute	Name of	EDs assigned	EDs with <40% of	Remaining EDs	Total	EDs included in H.S.A	% of EDs in 'a'
Hospital	H.S.A	under plurality	admissions to this	with small population	EDs exluded	for further study	included in the H.S.A
1		rule	hospital				
Ashford	ASHF	256	8	3	11	245	95.7
Bernet	BARN	347	55	3	58	289	63.3
Bedford	BEDF	425	9	3	12	413	97.2
Central Middlesex	CENT	344	60	1	61	283	82.3
Ealing	EALI	384	57	1	58	326	84.9
Edgware	EDGW	323	37	6	43	280	86.7
Hemai Hempstead	HEME	438	9	7	16	422	96.3
Hillingdon	HILL	289	3	4	7	282	97.6
Linker	LIST	443	10	4	14	429	96.8
Luion	LUTO	589	4	3	7	582	98.8
Mit Vernon	MTVE	196		3	11	185	94.4
Northwick Park	NINC	403	14	1	15	388	96.3
Queen Elizabeth I	OE2	326	6	2		318	97.5
Charing Cross*	RIVE	877	31	15	46	831	94.8
St Mary's	STM2	717	23	12	35	682	95.1
Watiord	WATE	336	5	1	6	330	98.2
West Middlesex	WMID	263	12	4	16	247	93.9
TOTAL		6956	351	73	424	6532	93.9

#### The number of EDs in each hospital service area, 1991/2 Table 26 (i)

Bold type indicates the hospitals in which data were judged (in chapter 4) likely to be better quality.
 Underlining indicates the hospitals for which better quality data were available in 1991/2, 1992/3 and 1993/4, ie for all three study years.
 Charing Cross (Riverside NHS Trust) was removed from all further analysis – see chapter 4.

		<b>a</b>	8	c	d	•	f
					b+c	a - (b+c)	((a - (b+c))/a)*100
Acute Hospital	Name of H.S.A	EDs assigned under plurality rule	EDs with <40% of admissions to this hospital	Remaining EDs with small population	Total EDs exluded	EDs included in H.S.A for further study	% of EDs in 'a' included in the H.S.A
Ashiord	ASHE	245	1	1	2	243	99.2
Barnet	BARN	322	68	6	74	248	77.0
Bedlord	BEDF	422	5	3	8	414	98.1
Central Middlesex	CENT	321	64	1	65	256	79.8
Ealing	EALI	371	33	1	34	337	90.8
Edgware	EDGW	314	38	4	42	272	86.6
Hamel Hempstead	HEME	445	19	6	25	420	94.4
Hillingdon	HILL	278	4	2	6	272	97.8
Linter	LIST	465	6	2	8	457	98.3
Luton	LUTO	579	2	1	3	576	99.5
Mit Vernon	MTVE	208	19	1	20	188	90.4
Northwick Park	NMIC	408	8	2	10	398	97.5
Queen Elizabeth II	0.52	315	5	1	6	309	98.1
Charing Cross*	RIVE	914	32	16	48	866	94.7
St Mary's	STM2	701	17	7	24	677	96.6
Watlord	WATE	336	10	3	13	323	96.1
West Middlesex	WMID	251	9	4	13	238	94.8
TOTAL		6895	340	61	401	6494	94.2

# Table 26 (ii) The number of EDs in each hospital service area, 1992/3

**Bold type indicates the hospitals in which data were judged (in chapter 4) likely to be better quality.** <u>Underfining indicates the hospitals for which better quality data were available in 1991/2, 1992/3 and 1993/4, ie for all three study years.</u> • Charing Cross (Riverside NHS Trust) was removed from all further analysis – see chapter 4.

		a	В	C	d	•	f
					b+c	a - (b+c)	((a - (b+c))/a)*100
Acute	Name of	EDs assigned	EDs with <40% of	Remaining EDs	Total	EDs included in H.S.A	% of EDs in 'a'
hospital	H.S.A	under plurality	admissions to this	with small population	EDs exluded	for further study	included in the H.S.A
		rule	Hospital				
Ashford	ASHF	254	3	1	4	250	98.4
Barnet	BARN	318	55	7	62	256	80.5
Bedford	BEDF	437	9	3	12	425	97.3
Central Middlesex	CENT	326	56	1	57	269	82.5
Ealing	EALI	351	36	1	37	314	89.5
Edgware	EDGW	301	28	1	29	272	90.4
Hemel Hempstead	HEME	457	15	4	19	438	95.8
Hillingdon	HILL	292	2	2	4	268	98.6
Lister	LIST	485	4	3	7	478	98.6
Luton	LUTO	536	5	2	7	529	98.7
Nit Vernon	MITVE	203		2	10	193	95.1
Northulck Park	NWIC	412	12	1 1	13	399	96.8
Queen Elizabeth II	QE2	343	10	2	12	331	96.5
Charing Cross*	RIVE	847	40	9	49	798	94.2
St Mary's	STM2	746	18	14	32	714	95.7
Wationd	WATE	336	8	2	10	326	97.0
West Middlesex	DIMW	254	4	2	6	248	97.6
TOTAL		6898	313	57	370	6528	94.6

# Table 26 (iii) The number of EDs in each hospital service area, 1993/4

Indid type indicates the hospitals in which data were judged (in chapter 4) likely to be better quality.
 <u>Underlining</u> indicates the hospitals for which better quality data were available in 1991/2, 1992/3 and 1993/4, ie for all three study years.
 • Charing Cross (Riverside NHS Trust) was removed from all further analysis – see chapter 4.

### (c) Selection of HSAs to be included in further analysis (after applying the 40% rule)

Table 26 (i), column 'e', shows that of the original dataset of 7754 EDs per study year, 6532 were included in 1991/2, 6494 in 1992/3 and 6528 in 1993/4. The number of EDs was reduced further for two reasons. First, EDs in HSAs of hospitals with data judged likely to be of poorer quality (see chapter 4 table 20) were removed, and the hospitals shown in bold type in table 26 are those for which data were judged likely to be of better quality. The number of EDs in these HSAs is shown in table 27 as **Group A** EDs. Group A EDs comprised 38.8% of the original total of 7754 in 1991/2, 37.1% in 1992/3 and 55.7% in 1993/4.

Second, further EDs were removed because there were difficulties in obtaining data to calculate the 'GP access factors' from two health authorities: Bedfordshire and Hounslow & Spelthorne. The lack of data on the number and location of GPs in these two health authorities meant that GP access factors could not be calculated in most EDs within these areas. This affected EDs in many HSAs but particularly those serving Bedford Hospital, Luton & Dunstable Hospital, Ashford Hospital and Lister Hospital, Stevenage. Because relatively few EDs remained in Luton and Ashford HSAs, these EDs were also removed and excluded from the multivariate analysis. The remaining EDs are shown in table 27 as Group B EDs. Group B EDs comprised 28.4% of the original total of 7754 in 1991/2, 21.8% in 1992/3 and 52.5% in 1993/4.

Finally, for the purpose of examining trends, from Group B EDs, only HSAs for which better quality data were available in all three study years were selected (those in table 26 indicated in bold type and underlined, excluding Luton and Ashford HSAs). The number of EDs in these four HSAs (Hillingdon, Lister, Northwick Park and Queen Elizabeth II) is shown in table 27 as **Group C** EDs. Group C EDs comprised 17.3% of the original total of 7754 in 1991/2, 17.4% in 1992/3 and 18.1% in 1993/4.

	Number of EDs 1991/2				Number of I 1992/3	EDa			Number of EDs 1993/4			
Hospital service area (H.S.A)	From Table 25	Group A	Group B	B Group C	From Table 26		Group B	Group C	From Table 26	Group A	Group B	Group C
Ashford	245	245	[74]		243	243	[71]		250	250	[75]	
Barnet	289	289	280		248				256	256	254	
Bedford	413				414	414	[0]		425			
Central Middlesex	283				256				269			
Ealing	326				337	337	336		314	314	312	
Edgware	280				272				272	272	272	
Hemel Hempstead	422	422	398		420				438	438	419	
Hillingdon	282	282	282	282	272	272	272	272	288	288	287	287
Lister	429	429	363	363	457	457	380	380	478	478	390	390
Luton & Dunstable	582	582	[39]		576	576	[34]		529			
Mount Vernon	185	185	182		188				193	193	190	
Northwick Partk	388	388	387	387	398	398	397	397	399	399	399	399
Queen Elizabeth II	318	318	316	316	309	309	307	307	331	331	329	329
Charing Cross*	831				866				798			
St Mary's	682				677				714	714	684	
Watford	330				323				326	326	294	
West Middlesex	247				236				248	248	248	
TOTAL	6532	3140	2208	1348	4390	3006	1692	1356	6528	4507	4078	1405
% of all EDs in NWT	80.7	38.8	28.4	17.3	54.2	37.1	21.8	17.4	80.7	55.7	52.5	18.1

#### Table 27 Number of EDs in Group A, Group B and Group C by hospital service area (HSA), all years

Group A EDs : EDs in HSAs in which admissions data were thought to be adequate (from chapter 4) Key:

Group B EDs: A subset of Group A EDs for which it was possible to calculate access factor scores

Croup C ED: A subset of Group B EDs for which data were judged to be adequate across all three study years
 Charing Cross (Riverside NHS Trust) was removed from all further analysis – see chapter 4.

In summary **Group A** EDs were those in HSAs in which the quality of admissions data were judged to be adequate (in chapter 4). **Group B** EDs were a subset of Group A for which it was possible to calculate access factor scores. **Group C** EDs were a subset of Group B for which admissions data were judged adequate for all three study years, allowing an analysis across time.

In different analyses below, different groups of EDs are used. Group A EDs are generally used to describe the frequency distributions of variables and for some bivariate analyses. Group B and C EDs are used for bivariate analyses using access factors and for the multivariate analysis.

### 6.3.2 Univariate analyses and comparisons by HSA

### (a) Age-sex standardised admission rates across EDs

Using Group A EDs, the distribution of the age-sex standardised rates for all 18 admission groups is shown in table 28. Only data for one year, 1993/4 are shown.

The data show that, for the ACS admission groups (1-14), the mean rates were highest in group 1, at 31.09 per 1000 admissions, and high in groups 5 and 9. Rates were much lower in marker admission groups (15 and 16) at between 4-5 per 1000. Rates in Billings' ACS (group 21) and marker (group 23) groups were very low at 1-2 per 1000.

The rate in group 19 (emergency surgical) was roughly 8% of that in group 17 (all surgical admissions), whereas the rate in group 20 (emergency medical admissions) was 18% of that in group 18 (all medical admissions). The proportion of readmissions tended to be higher in admission groups representing strong ACS conditions. The pattern seen in the table mirrors that shown in table 25, although in table 25 data on all EDs were used, while in table 28 data from a subset of those in table 25 were used.

To understand better the change in rates over time, the rates in all Group C EDs across the region were calculated, for group 1 and group 16 only (for simplicity). The results are shown in table 29.

# Table 28 Age-sex standardised admission rates by admission group, 1993/4

Admission	Description	Rate per 1000						
Group		Mean	Median	Standard deviation	n			
1	Strong and weak ACS conditions, any urgency, all	31.10	28.67	16.89	4450			
3	Strong and weak ACS conditions, any urgency, readmissions	7.43	5.41	7.92	4450			
5	Strong and weak ACS conditions, urgent, all admissions	11.57	9.82	9.31	4450			
7	Strong and weak ACS conditions, urgent, readmissions	3.12	1.63	4.84	4450			
9	Strong ACS conditions, any urgency, all admissions	11.52	9.72	9.24	4450			
10	Strong ACS conditions, any urgency, readmissions	7.11	4.06	10.10	4450			
11	Strong ACS conditions, urgent, all admissions	5.92	4.50	6.24	4450			
12	Strong ACS conditions, urgent, readmissions	1.37	0.00	3.13	4450			
13	Strong ACS conditions, very urgent, all admissions	7.13	5.61	6.65	4450			
14	Strong ACS conditions, very urgent, readmissions	3.58	2.20	4.71	4450			
15	Strong and weak markers, very urgent, all admissions	4.94	4.02	4.69	4450			
16	Strong markers, very urgent, all admissions	4.57	3.67	4.49	4450			
17	Admissions in surgical specialties	61.75	58.69	26.44	4450			
18	Admissions in medical specialties	73.02	55.42	82.77	4450			
19	Emergency admissions in surgical specialties	5.31	4.11	5.53	4450			
20	Emergency admissions in medical specialties	13.77	11.25	12.64	4450			
21	Admissions for ACS conditions defined by Billings	1.56	0.00	2.54	4450			
23	Admissions for marker conditions defined by Billings	2.06	1.44	2.73	4450			

Group A EDs

# Table 29 Comparisons of age-sex standardised admission rates across 3 study years, for admission groups 1 and 16.

		1991/2			1992/3		1993/4			
Admission	Brief description	Mean	Standard deviation	n	Mean	Standard deviation	n	Mean	Standard deviation	n
1	Strong and weak ACS conditions, any urgency, all	28.09	14.04	1348	28.82	13.86	1356	30.74	15.09	1405
16	Strong markers, very urgent, all admissions	4.61	4.28	1348	4.64	4.28	1356	4.72	4.38	1405

Group C EDs

The results show that there was an increase in the mean rates in 1993/4 in group 1 ACS admissions - from 28.09 in 1991/2 to 30.74 in 1993/4. The rates for marker conditions were stable across the three years.

These results for admission group 1 (ACS conditions) were plotted on an error bar graph, showing the mean and 95% confidence intervals of the values (figure 6 – note truncated axis).

### **Figure 6**



Group 1 (ACS) admission rates

The figure shows that the increase between 1993/4 and the earlier years was statistically significant (p < 0.05).

The standardised admission rates were then examined by HSA, as shown in the next table. For simplicity, again the results for only two admission groups are shown ((group 1 (ACS) and group 16 (markers)) for 1993/4.

### Table 30

### Age-sex standardised admission rates by HSA, 1993/4 (ranked by descending order of the mean)

	Admission group 1 (ACS)								
Hospital service Area	Mean	Standard Deviation	n=						
West Middlesex	34.79	16.85	248						
Ealing	34.59	15.98	313						
Northwick Park	33.72	15.71	399						
Watford	33.35	15.81	318						
St Mary's Paddington	33.26	23.86	687						
Hillingdon	33.03	13.88	285						
Edgware	31.98	17.85	272						
Mt Vernon	30.26	13.55	193						
Barnet	30.10	15.39	256						
Hemel Hempstead	28.12	14.21	421						
Lister	28.01	13.73	477						
Queen Elizabeth II	27.11	13.33	331						
Ashford	24.60	12.8	250						
TOTAL	31.09	16.89	4450						

Group A EDs

	Admission group 16 (marker)								
Hospital service Area	Mean	Standard Deviation	n=						
Lister	5.39	4.88	477						
Ealing	5.08	4.67	313						
West Middlesex	5.04	4.62	248						
Barnet	4.93	4.75	256						
Watford	4.88	4.14	318						
Queen Elizabeth II	4.82	4.50	331						
Edgware	4.76	4.03	272						
Hemel Hempstead	4.48	4.08	421						
Hillingdon	4.39	4.00	285						
Mt Vernon	4.25	3.59	193						
Northwick Park	4.00	3.36	399						
St Mary's Paddington	3.92	5.54	687						
Ashford	3.85	3.51	250						
TOTAL	4.57	4.48	4450						

Group A Eds

The results show that there were marked differences between HSAs for admission group 1, for example the mean rate in West Middlesex was 34.79 compared to Ashford at 24.6 per 1000. The differences were smaller, although proportionately as great, between HSAs for group 16 (marker) admissions.

Table 31 below shows that the rates increased in three of the four HSAs in 1993/4, with the rise being most noticeable in Northwick Park.

	1991/2							
Hospital service area	Mean	Standard Deviation	n					
Hillingdon	32.07	16.04	282					
Lister	26.89	13.73	363					
Northwick Park	28.37	12.87	387					
Queen Elizabeth II	25.61	13.11	316					
Total	28.09	14.04	1348					

### Table 31 Age-sex standardised admission rates (per 1000) for admission group 1, by year

	1992/3							
Hospital service area	Mean	Standard Deviation	N					
Hillingdon	33.93	14.37	272					
Lister	26.87	13.39	380					
Northwick Park	29.31	13.26	397					
Queen Elizabeth II	26.07	13.52	307					
Total	28.82	13.86	1356					

	1993/4								
Hospital service area	Mean	Standard Deviation	N						
Hillingdon	33.04	13.85	287						
Lister	29.07	15.86	390						
Northwick Park	33.72	15.71	399						
Queen Elizabeth II	27.11	13.33	329						
Total	30.74	15.09	1405						

Group C EDs

### (b) Carstairs score

Using Group A EDs, the mean and standard deviation of scores per HSA per year were calculated. Data from 1993/4 are presented (table 32), but since data from two HSAs (Luton and Bedford) were not adequate in this year (see table 26), data for 1992/3 were used.

Standard Deviation 1.33 1.55	1993/4 Mean 0.95	Standard Deviation	1993/4 Mean	Standard Deviation	
Standard Deviation 1.33 1.55	Mean 0.95	Standard Deviation	Mean	Standard Deviation	
1.33 1.55	0.95	0.40			
1.55		0.40	0.87	0.23	193
	0.98	0.40	0.89	0.24	438
1.50	1.01	0.50	0.84	0.24	331
1.65	1.03	0.44	0.91	0.21	256
2.08	1.06	0.88	0.86	0.24	326
1.68	1.02	0.44	0.88	0.22	250
2.47	1.03	0.45	0.91	0.25	414
1.75	1.01	0.52	0.85	0.23	478
1.86	0.93	0.40	0.88	0.21	399
2.29	0.92	0.42	0.89	0.27	272
1.76	1.02	0.45	0.86	0.22	288
3.07	1.01	0.44	0.86	0.24	576
1.88	0.98	0.44	0.86	0.22	248
3.04	1.00	0.54	0.90	0.24	314
2.83	0.98	0.71	0.76	0.28	714
	4       1.68         1       2.47         1       1.75         9       1.86         3       2.29         3       1.76         3       3.07         2       1.88         9       3.04         4       2.83	1.68         1.02           1         2.47         1.03           1         1.75         1.01           9         1.86         0.93           3         2.29         0.92           3         1.76         1.02           3         3.07         1.01           2         1.88         0.98           9         3.04         1.00	1.68         1.02         0.44           1         2.47         1.03         0.45           1         1.75         1.01         0.52           9         1.86         0.93         0.40           3         2.29         0.92         0.42           3         1.76         1.02         0.45           3         3.07         1.01         0.44           2         1.88         0.98         0.44           9         3.04         1.00         0.54	4         1.68         1.02         0.44         0.88           1         2.47         1.03         0.45         0.91           1         1.75         1.01         0.52         0.85           9         1.86         0.93         0.40         0.88           3         2.29         0.92         0.42         0.89           3         1.76         1.02         0.45         0.86           3         3.07         1.01         0.44         0.86           2         1.88         0.98         0.44         0.86           3         3.07         1.01         0.44         0.86           3         3.04         1.00         0.54         0.90           4         2.83         0.98         0.71         0.76	4         1.68         1.02         0.44         0.88         0.22           1         2.47         1.03         0.45         0.91         0.25           1         1.75         1.01         0.52         0.85         0.23           9         1.86         0.93         0.40         0.88         0.21           3         2.29         0.92         0.42         0.89         0.27           3         1.76         1.02         0.45         0.86         0.22           3         3.07         1.01         0.44         0.86         0.24           2         1.88         0.98         0.44         0.86         0.22           3         3.07         1.01         0.44         0.86         0.22           3         3.07         1.01         0.44         0.86         0.22           3         3.04         1.00         0.54         0.90         0.24           2         3.04         1.00         0.54         0.90         0.24           3         2.83         0.98         0.71         0.76         0.28

### Table 32 Descriptive statistics for Carstairs, SMR and SIR scores, by HSA 1993/4 (ranked by Carstairs score)

\* Data refer to 1992/3

The table shows that the HSAs with higher mean Carstairs scores (ie greater deprivation) were St Mary's, Ealing, West Middlesex, Luton and Hillingdon. The HSAs with the lowest scores were: Mt Vernon, Hemel Hempstead and QE2. Across most of the region, the mean scores were mostly negative, indicating lower levels of socio-economic deprivation than the rest of England & Wales.

(c) Standardised mortality ratio (SMR) and standardised illness ratio (SIR)

Table 32 also shows information on the SMR and SIR scores by HSA. The table shows that the mean SMR in most HSAs was close to 1, indicating SMRs similar to that in England & Wales. Lower mean values (ie lower rates compared to England & Wales) were found in Northwick Park (0.93) and Edgware (0.92).

For the SIR, all the mean scores were below 1, indicating lower rates of people reporting

limiting longstanding illness than in England & Wales as a whole. St Mary's HSA contained EDs with the lowest mean score, which was surprising considering that it is located in a deprived inner city area (Paddington). Undercounting in the 1991 census, which affected deprived inner city areas the most (294) might have excluded residents who were most likely to report illness, although most of the undercounted were estimated to be young males who are generally a healthy group.

### (d) GP and hospital access factors

Table 33 below shows descriptive statistics for the two access factors, by hospital service area for 1993/4.

	GP Access Fact		
Hospital service area (HSA)	Mean	Standard Deviation	n
St Mary's Paddington	0.546	0.517	684
Ealing	0.382	0.446	312
Edgware	0.329	0.467	272
Northwick Park	0.270	0.364	399
West Middlesex	0.242	0.384	248
Barnet	0.179	0.355	254
Hillingdon	0.161	0.241	287
Mt Vernon	0.133	0.205	190
Watford	0.122	0.208	294
Queen Elizabeth II	0.075	0.240	329
Hemel Hempstead	0.074	0.153	419
Lister	0.052	0.196	390

### Table 33 Descriptive statistics for GP (GAF2) and hospital access factors (HAF2), by hospital service area, 1993/4 (ranked in order of the mean)

### Table 33 Continued

	Hospital Access		
Hospital service area (HSA)	Mean	Standard Deviation	N
Lister	0.086	0.074	390
Hillingdon	0.060	0.048	287
St Mary's Paddington	0.032	0.148	684
West Middlesex	0.021	0.107	248
Watford	0.014	0.071	294
Edgware	0.014	0.059	272
Mt Vernon	0.013	0.079	190
Barnet	0.009	0.058	254
Hemel Hempstead	0.007	0.038	419
Queen Elizabeth II	0.004	0.020	328
Ealing	0.004	0.026	312
Northwick Park	0.003	0.019	399

In the table, higher GP access factor (GAF2) scores are apparent in HSAs closer to central London, such as St Mary's (STM2), Ealing (EALI) and Edgware (EDGW), and lower in relatively rural areas (such as Lister (LIST) Hemel Hempstead (HEME) and Queen Elizabeth II (QE2). Similar results were found in 1991/2 and 1992/3 (data not shown). For the hospital access factor (HAF2), the mean values were not obviously higher in HSAs closer to central London with the exception of St Mary's Paddington.

To examine the access factors further, the frequency distribution of the scores for GAF2 and HAF2 were plotted by HSA. Logged scores (using natural logarithms) were used to try to reduce the skewness of the distribution. The results are shown in appendix B. For simplicity data for one year, 1993/4 are shown – the results for the previous two years were similar. Appendix B shows that the scores for the logged GAF2 score (LGAF2) approximated roughly to a normal distribution, with skewing to the left in some HSAs. There was a more marked skew to the left for the logged HAF2 scores (LHAF2).

### 6.3.3 Bivariate analysis

### (a) Correlation between the main independent variables

Because of skewness in the distribution of the access factor scores, this bivariate analysis

was carried out using quintile scores rather than the raw values. The Pearson product moment correlation coefficients are shown in the table below. For simplicity one variable is used to represent each of the access factors.

There was a non-significant (p > 0.05) association between the Carstairs quintile and the SMR quintile, and a significant (p > =0.005) but weak *negative* association between the SIR quintile and the Carstairs quintile. The relationship between the SMR and SIR with the access quintiles was mostly not significant, although there was a positive and significant (p > 0.0001) relationship between Carstairs and both access scores. As to be expected there was a positive and significant (p < 0.0001) relationship between the access quintiles.

#### Variables Carsquin Smrquin Sirguin Gef2guin Haf2guin Carsquin 1.0000 -0.0242 Smrauin 1.0000 P a 0.1225 0.1740 1.0000 -0.0439 Sirquin 0.0000 0.0050 Gaf2quin 0.4383 -0.0338 -0.0264 1.0000 0.0308 0.0918 0.0000 Haf2quin 0.2908 -0.0190 -0.0385 0.3811 1.0000 0.0000 0.2254 0.0139 0.0000

### Table 34 Correlations between the main independent variables

Group B EDs

P= statistical significance

(b) Correlation between standardised admission rates and socio-economic deprivation

One of the main hypotheses in this thesis was that there would be a significant positive correlation between socio-economic deprivation with admissions for ACS conditions, and no significant correlation for marker conditions.

The analysis involved Group A EDs. Bivariate analyses were performed using age-sex standardised admission rates by ED (for all admission groups) and Carstairs scores (raw values). The Pearson product moment correlation was used to assess the closeness of the association between the two continuous variables. The results across the region for 1993/4 are shown in the table below.

Admission	Description	Permon correlation	Significance
Group	Detemption	with Constains	(2 tolled)
Group		With Carstairs	(2-(2))
1	Strong and weak ACS conditions, any	0.325	0.01
	urgency, all		
3	Strong and weak ACS conditions, any	0.225	0.01
	Strong and weak ACS conditions	0.265	0.01
5	urgent all admissions	0.205	0.01
7	Strong and weak ACS conditions.	0.172	0.01
	urgent, readmissions		
9	Strong ACS conditions, any urgency,	0.304	0.01
	all admissions		
10	Strong ACS conditions, any urgency,	0.168	0.01
	readmissions		
11	Strong ACS conditions, urgent, all	0.209	0.01
43		0.117	0.01
14	readmissions	0.117	0.01
13	Strong ACS conditions, very urgent, all	0 152	0.01
	admissions		
14	Strong ACS conditions, very urgent,	0.186	0.01
	readmissions		
15	Strong and weak markers, very urgent,	0.125	0.01
	all admissions		
16	Strong markers, very urgent, all	0.121	0.01
	admissions		
17	Admissions in surgical specialties	0.276	0.01
18	Admissions in medical specialties	0.206	0.01
19	Emergency admissions in surgical	0.018	not significant
	specialties		-
20	Emergency admissions in medical	0.065	0.01
	specialties		
21	Admissions for ACS conditions defined	0.104	0.01
	by Billings		
23	Admissions for marker conditions	0.034	0.05
	defined by Billings		

# Table 35 Correlation between Carstairs score and admission rates by admission group, 1993/4

Group A EDs

Table 35 shows that there was a significant positive correlation between admission rates for most groups and the Carstairs score. The correlations were higher in the ACS admission groups 1, 2, 3, 9 and 11 (ie the admission groups using *all admissions* rather than

*readmissions*) and lower in marker groups. The lower correlations in the readmissions groups may be because of the lower number of events. Correlations were higher in the all surgical and all medical admissions group compared to the emergency surgical and emergency medical admissions groups. Correlations were low with Billings' ACS and marker groups. The results were similar in the study years 1991/2 and 1992/3. The results are presented in such a way as to allow some comparison between groups, but it is important to note that interpretation is not straightforward because the numbers of admissions in each group is different, and some groups (eg group 3) are subsets of others (eg group 1).

As a further analysis to compare rates in deprived and non-deprived areas, the ratio of mean rates in the most deprived quintile of EDS (using Group A) and in the least deprived quintile of EDs were calculated for all 18 admission groups and for all three study years. The table in appendix C shows the results.

In each year the ratios varied mainly between 1.5 and 2.5; ratios for ACS conditions were generally nearer 2, and ratios for markers were nearer 1.5. Interestingly the ratios were generally higher in ACS groups in 1993/4 compared to other years, and for the ACS groups representing readmissions (in contrast to the results shown in table 35), suggesting that socioeconomic deprivation is associated with a higher rate of readmissions for ACS conditions particularly in 1993/4. The ratio was lowest for emergency surgical admissions (group 19), conditions for which the decision to admit is unlikely to be discretionary.

6.3.4 Multivariate analysis using Poisson regression

(a) Developing the model

The regression model used for the analysis was:

ADMISSIONS = f (POPN, AGE-GRP, SEX, SMR, SIR, CARST, HAF2, GAF2, HSA) (group)

Where:

ADMISSIONS (group)	-	number of admissions per ED in a particular
		admission group by age and sex group
POPN	=	total population count per ED, by age and sex group
AGE-GRP	=	age group (7 age bands were chosen - see above)
SEX	=	sex (male, female - see above)
SMR	-	standardised mortality ratio per ED
		(all causes aged under 75 years, 1988-92)
SIR	=	standardised illness ratio per ED
CARST	=	Carstairs' score per ED
GAF2 (GAF, GAF4)	=	GP access factor scores per ED
HAF2	=	hospital access factor scores per ED
HSA	=	hospital service area

The first step was try to simplify the dataset by calculating some derived variables. Using Group B EDs, the scores per ED for each of the five main variables (SMR, SIR, CARST, GAF2 and HAF2) were grouped into quintiles (eg SMRQUIN1 to SMRQUIN5). Using the exact scores would assume a linear relationship in the regression between these variables and the dependent variable, whereas grouping the scores by quintiles would allow for a non-linear (eg curvilinear) relationship, and would reduce problems in analysis with the skewed distribution of access factors. Using the quintiles in this way, the regression model (consisting of the variables shown above) was termed the 'full' model.

The results of the Poisson regression analysis are shown in appendix D. In the appendix, the 'IRR' shown represents the *incremental relative risk ratio*. The IRR is a value that represents the effect of the value of the variable in question, relative to a baseline value. For example, in the table, the value of CARSQ2 is 1.22 relative to the baseline variable CARSQ1 at 1.000. This means that, taking other variables into account, the mean number of admissions for EDs in the second quintile of the Carstairs score (CARSQ2) is 22% higher than that of the first quintile (CARSQ1).

The table in appendix D shows that, using the quintile scores in the full model (using group 1 ACS conditions in 1991/2 as the dependent variable and including GAF2 and HAF2 as access factors) the relationships with SMR, SIR and CARST were approximately linear. In other words, the IRRs for the quintiles increased successively from the first to the fifth quintile (eg from CARSQ1 to CARSQ5). These results were typical for other admission groups and other years. Therefore, for simplification, the quintile scores for each variable SMR, SIR and CARST were thus used as a single numerical variable (eg SMRQUIN, SIRQUIN) rather than using indicator variables (eg SMRq1, SMRq2 etc) as shown in appendix D for each quintile group.

However, as appendix D shows, it was not clear whether there was a linear relationship between admissions and HAF2 and GAF2 respectively. The IRRs did not suggest a clear linear pattern, the confidence intervals of the IRRs were wide, and the picture across the different ACS groups (data not shown) was mixed. Therefore to simplify the model, four of the quintiles for GAF2 and HAF2 were grouped in pairs (eg GAF2q1, GAF2q23, GAF2q45), rather than used as a single numerical variable (eg GAF2QUIN).

Using the single 'collapsed' variables – SMRQUIN, SIRQUIN, CARSQUIN – and the grouped quintiles for GAF2 and HAF2, the regression model was termed the 'simplified' model.

To test the effect on the fit using the simplified model, rather than the full model, Poisson regressions were run on both models and the 'pseudo' r-squared coefficients were compared. Pseudo r-squared statistics "provide a quick way to describe or compare the fit of different models for the same dependent variable" (295), although they "lack the straightforward explained-variance interpretation of the true r-squared coefficient as found in ordinary least squares regression". The values of pseudo r-squared statistics are always found in the range 0 to 1, 1 representing a perfect fit and 0 representing no fit. The results are shown appendix E. The IRR values were only slightly lower for the simplified model indicating little loss of explanatory power. As a result, the simplified model was used in all further analyses.

The pseudo r-squared scores in appendix E show that between approximately 10-20% of

variation in admissions for ACS conditions between EDs was 'explained' by either model, and around 5% of variation for marker conditions. The greatest explanatory power (22-24%) occurred for admission group 20 (emergency medical admissions) in contrast with group 19 (emergency surgical admissions) at 9-13%. The pseudo r-squared scores for the simplified model using Group C EDs (data not shown) were very similar to those shown using Group B EDs.

### (b) Running the Poisson regression model

The simplified model was first run using Group B EDs for each study year, and then Group C EDs (1991/2, 1992/3, 1993/4 and all years combined) to allow a comparison across all three study years. For simplicity, the results are presented here only for Group B EDs in 1993/4 (those using Group B EDs in earlier years were similar). Results for Group C EDs 1991-4 are shown in appendix F. The results are shown first for variables representing mortality (SMR), morbidity (SIR), and socio-economic deprivation (CARS), then for different age-groups, then for variables representing access to general practice and hospital facilities (GAF2 and HAF2). Results for the HSA variables are shown last.

### (i) SMR, SIR, and Carstairs index

The next table shows the results across all 18 admission groups for 1993/4.

The results show that there was a positive association between admissions and the three variables for all admission groups. For example, for group 1 the admission rate increases by an average of 11.6% for each quintile of Carstairs score. In most cases the association was significant. The positive relationship was very similar for SIR and SMR, and stronger for CARS, especially in admission groups 10, 7, 3, 12 and 14 (all representing readmissions).

		Carstairs				SMR				SIR			
Admission	Description	IRR	Ρ	95% ci		IRR	P	95% ci		IRR	ρ	95% ci	
Group			Value	lower	upper		value	lower	upper		value	lower	upper
1	Strong and weak ACS conditions, any urgency, all	1.1160	0.000	1.1087	1.1233	1.0446	0.000	1.0381	1.0511	1.0510	0.000	1.0445	1.0575
3	Strong and weak ACS conditions, any urgency,	1.1712	0.000	1.1554	1.1872	1.0551	0.000	1.0417	1.0686	1.0934	0.000	1.0796	1.1074
	readmissions												
5	Strong and weak ACS conditions, urgent, all admissions	1.1367	0.000	1.1249	1.1486	1.0339	0.000	1.0239	1.0440	1.0591	0.000	1.0488	1.0694
7	Strong and week ACS conditions, urgent, readmissions	1.1833	0.000	1.1595	1.2075	1.0235	0.015	1.0044	1.0430	1.1030	0.000	1.0823	1.1241
	Strong ACS conditions, any urgency, all admissions	1.1047	0.000	1.0929	1.1167	1.0314	0.000	1.0210	1.0419	1.0497	0.000	1.0391	1.0603
10	Strong ACS conditions, any urgency, readmissions	1.1909	0.000	1.1733	1.2088	1.0490	0.000	1.0345	1.0637	1.0814	0.000	1.0665	1.0964
11	Strong ACS conditions, urgent, all admissions	1.1101	0.000	1.0942	1.1261	1.0215	0.002	1.0079	1.0352	1.0605	0.000	1.0463	1.0748
12	Strong ACS conditions, urgent, readmissions	1.1760	0.000	1.1408	1.2123	1.0053	0.710	0.9775	1.0339	1.0987	0.000	1.0681	1.1302
13	Strong ACS conditions, very urgent, all admissions	1.1144	0.000	1.0941	1.1351	1.0083	0.342	0.9912	1.0256	1.0576	0.000	1.0396	1.0759
14	Strong ACS conditions, very urgent, readmissions	1.1334	0.000	1.1187	1.1483	1.0273	0.000	1.0149	1.0399	1.0504	0.000	1.0377	1.0633
15	Strong and week markers, very urgent, all admissions	1.1053	0.000	1.0879	1.1230	1.0595	0.000	1.0438	1.0754	1.0404	0.000	1.0251	1.0560
16	Strong markers, very urgent, all admissions	1.1036	0.000	1.0855	1.1219	1.0568	0.000	1.0405	1.0733	1.0351	0.000	1.0192	1.0512
17	Admissions in surgical specialties	1.0843	0.000	1.0794	1.0892	1.0518	0.000	1.0473	1.0563	1.0301	0.000	1.0257	1.0345
18	Admissions in medical specialties	1.0725	0.000	1.0676	1.0774	1.0489	0.000	1.0444	1.0535	1.0574	0.000	1.0529	1.0620
19	Emergency admissions in surgical specialties	1.0864	0.000	1.0713	1.1017	1.0604	0.000	1.0466	1.0744	1.0207	0.002	1.0074	1.0341
20	Emergency admissions in medical specialties	1.0960	0.000	1.0855	1.1066	1.0488	0.000	1.0394	1.0584	1.0548	0.000	1.0453	1.0644
					1								
21	Admissions for ACS conditions defined by Billings	1.1838	0.000	1.1505	1.2180	1.0237	0.086	0.9966	1.0515	1.0826	0.000	1.0541	1.1190
23	Admissions for marker conditions defined by Billings	1.0682	0.000	1.0440	1.0930	1.0592	0.000	1.0364	1.0825	1.0382	0.001	1.0160	1.0608

# Table 36 Incremental relative risk ratios for selected variables: Carstairs quintile; SMR quintile; and SIR quintile, 1993/4

Group B EDs
### (ii) Age-groups

Table 37 below shows the pattern of results across the seven different age groups for all 18 admission groups again for Group B EDs in 1993/4. The results are shown for age groups 2-7 relative to the baseline age group 1.

Admission	Description	Age	IRR	P	95% c.1	_
Group		Group		value	lower	upper
1	Strong and weak ACS conditions, any urgency,	1	1.00	-		•
	All admissions	2	0.59	0.000	0.56	0.62
		3	0.24	0.000	0.23	0.25
		4	0.12	0.000	0.12	0.13
		5	0.20	0.000	0.19	0.21
		6	0.46	0.000	0.44	0.48
			0.83	0.000	0.80	0.87
3	Strong and weak ACS conditions, any urgency,	1	1.00	-	•	•
	Readmissiona	2	0.39	0.000	0.37	0.44
		3	0.12	0.000	0.10	0.13
		1 4	0.06	0.000	0.05	0.06
		5	0.10	0.000	0.10	0.11
		2	0.37	0.000	0.34	0.40
			0.84	0.000	0.77	8.09
5	Strong and weak ACS conditions, urgent,		1.00	1	1	
	All admissions	2	0.59	0.000	0.50	0.63
			0.10	0.000	0.17	0.19
		1 2 1	0.08	0.000	0.00	0.08
		Ă	0.00	0.000	0.00	0.07
		ž	0.45	0.000	0.42	0.47
7	Strong and weak ACS conditions urgent		1.00	0.000	0.44	0.41
•	Beadmissions		0.42	0,000	0.38	0.49
		3	0.11	0.000	0.00	0.12
		Ă	0.03	0.000	0.03	0.04
		5	0.04	0.000	0.03	0.04
		6	0.16	0.000	0.15	0.18
		7	0.46	0.000	0.43	0.51
9	Strong ACS conditions, any urgency,	1	1.00	. · .	•	•
	All admissions	2	0.68	0.000	0.63	0.74
		3	0.34	0.000	0.31	0.37
			0.20	0.000	0.18	0.22
		2	0.24	0.000	0.22	0.26
		7	0.60	0.000	0.39	0.65
10	Strong ACS conditions, any urgency,	1	1.00		-	-
	Readmissions	2	0.50	0.000	0.45	0.57
		3	0.19	0.000	0.17	0.21
		4	0.09	0.000	0.08	0.10
		5	0.14	0.000	0.13	0.16
		6	0.71	0.000	0.64	0.86
		7	1.68	0.000	1.52	1.86
11	Strong ACS conditions, urgent,	1	1.00			
	All admissions	2	0.75	0.000	0.69	0.82
		3	0.34	0.000	0.31	0.37
			0.14	0.000	0.12	0.15
		2	0.11	0.000	0.10	0.12
		2	0.17	0.000	0.10	0.19
						1

Table 37	Incremental risk ratios for 'age group', by admission group, data for
	all years combined

Admission	Description	Age	IRR	p	95% c.l	
Group				value	lower	upper
12	Strong ACS conditions, urgent,	1	1.00	-	-	
	Readmissions	2	0.50	0.000	0.36	0.58
		3	0.16	0.000	0.14	0.20
		4	0.06	0.000	0.05	0.07
		5	0.05	0.000	0.04	0.06
		7	0.20	0.000	0.10	0.14
13	Strong ACS conditions, very urgent.	1	1.00	•	-	-
	All admissions	2	0.79	0.000	0.72	0.88
		3	0.35	0.000	0.32	0.39
		4	0.12	0.000	0.10	0.13
		5	0.08	0.000	0.08	0.09
		7	0.11	0.000	0.10	0.13
- 14	Strong ACS conditions year urgent		1.00	0.000	0.14	0.10
	Readmissions	2	0.82	0.000	0.76	0.85
		3	0.26	0.000	0.23	0.27
		4	0.09	0.000	0.08	0.10
		5	0.08	0.000	0.07	0.09
		6	0.20	0.000	0.19	0.22
		7	0.45	0.000	0.44	0.51
15	Strong and weak markers, very urgent,		1.00	-		
		2	1.93	0.320	0.75	1.15
		Å	1.10	0.399	0.00	1.31
		5	0.76	0.008	0.63	0.93
		6	1.44	0.000	1.18	1.74
		7	3.23	0.000	2.667	3.93
16	Strong markers, very urgent,	1	1.00	•	•	
	All admissions	2	1.01	0.945	0.80	1.25
		3	1.10	0.144	0.94	1.45
			0.75	0.138	0.85	0.93
		ĕ	1.40	0.001	1.66	1.74
		7	3.18	0.000	2.59	3.89
17	Admissions in surgical specialties	1	1.00	-	- 1	-
		2	1.35	0.000	1.26	1.45
		3	1.18	0.000	1.11	1.25
		1 4	0.95	0.113	0.88	1.01
		2	1.20	0.000	2.00	2.67
		7	3.81	0.000	3.57	4.05
18	Admissions in medical specialties	1 i	1.00			-
		2	0.24	0.000	0.23	0.25
		3	0.07	0.000	0.07	0.08
		4	0.05	0.000	0.05	0.05
		5	0.09	0.000	0.08	0.09
		7	0.28	0.000	0.25	0.26
40	Emergency adminutions in symplest encolotities		1.00	0.000	0.48	0.52
	Emergency somissions in surfice speciations	2	66	0.000	0.53	0.82
		3	1.13	0.187	0.93	1.40
		4	0.98	0.848	0.80	1.18
		5	0.99	0.991	0.83	1.20
		6	1.56	0.000	1.28	1.86
		7	2.93	0.000	2.42	3.54
20	Emergency admissions in medical specialties		1.00	1	0.00	-
		3	0.00	0.000	0.20	0.10
		Ă	0.04	0.000	0.03	0.04
		5	0.05	0.000	0.04	0.05
		6	0.13	0.000	0.13	0.14
		7	0.36	0.000	0.35	0.38

### Table 37 Continued

Admission	Description	Age group	IRR	P	95% c.l	
Group				value	lower	upper
21	Admissions for ACS conditions	1	1.00	•	-	
	Defined by Billings	2	0.75	0.213	0.48	1.17
		3	0.51	0.002	0.33	0.78
		4	0.36	0.000	0.23	0.55
		5	0.69	0.072	0.46	1.03
		6	3.06	0.000	2.07	4.52
		7	6.66	0.000	4.51	9.84
23	Admissions for marker conditions	1	1.00	•	-	-
	Defined by Billings	2	0.77	0.504	0.28	2.12
		3	9.31	0.000	3.85	22.49
		4	8.72	0.000	3.61	21.06
		5	5.29	0.000	2.19	12.76
		6	14.56	0.000	6.05	35.07
		7	38.82	0.000	16.12	93.47

#### Table 37 Continued

Group B EDs

The results show that the incremental relative risk ratios (IRRs) for ACS conditions generally showed a 'U' shaped pattern – that is higher values were seen in age groups 1 (under 1 year) and 7 (65-74) compared to the other groups. For all ACS admission groups (except admission group 10 – strong ACS conditions, readmissions) the IRRs were lower in most age groups (including age group 7) than age group 1 (aged under 1 year). For markers, the IRRs were approximately 3 times higher in age group 7 and age group 1. For medical and surgical admissions, the IRRs for age group 1 were higher than group 7 for medical admissions (groups 18 and 20), but lower for surgical admissions (groups 17 and 19). The strongest associations with age group 7 (65-74 years) were seen in admission groups 21 and 23 (Billings' ACS and markers) – in the latter group, the IRRs for age group 7 were 38 times higher than those for age group 1.

(iii) Sex

The next table shows the IRRs of sex group 2 (males) relative to group 1 (females).

Admission	Description	Sex	IRR	P	95% c.I	_
Group				value	lower	upper
1	Strong and weak ACS conditions, any urgency, all	1	1.00			
		2	0.89	0.000	0.88	0.91
3	Strong and weak ACS conditions, any	1	1.00			
	urgency, readmissions	2	0.73	0.000	0.71	0.76
5	Strong and weak ACS conditions, urgent,	1	1.00			-
	all admissions		0.75	0.000	0.72	
7	Strong and weak ACS conditions urgent	1	1.00	0.000	0.73	0.77
	readmissions	•	1.00			
		2	0.66	0.000	0.62	0.70
9	Strong ACS conditions, any urgency, all admissions	1	1.00		-	
		2	0.91	0.000	0.89	0.94
10	Strong ACS conditions, any urgency,	1	1.00			
	readmissions	2	0.72	0.000	0.70	0.75
11	Strong ACS conditions, urgent, all	1	1.00			-
	admissions	_	0.00	0.000		4.00
42	Strong ACE conditions urgent	4	0.90	0.000	0.93	1.00
	readmissions		1.00			
		2	0.96	0.101	0.84	0.99
13	Strong ACS conditions, very urgent, all	1	1.00			
		2	0.88	0.000	0.84	0.99
14	Strong ACS conditions, very urgent,	1	1.00			
	readmissions					
	Strong and weak markers were urgent all	2	1.00	0.000	0.67	0.72
13	admissions	•	1.00			
		2	0.61	0.000	0.59	0.64
16	Strong markers, very urgent, all	1	1.00			-
		2	0.60	0.000	0.58	0.63
17	Admissions in surgical specialties	1	1.00			
		2	0.77	0.000	0.76	0.78
18	Admissions in medical specialties	1	1.00			
		2	0.76	0.000	0.75	0.77
19	Emergency admissions in surgical apecialties	1	1.00	-		
		2	0.79	0.000	0.76	0.83
20	Emergency admissions in medical	1	1.00			
	sbacraurias	2	0.79	0.000	0.77	0.82
21	Admissions for ACS conditions defined	1	1.00	-		-
	by Billings	2	0.67	0.000	0.62	0.62
23	Admissions for marker conditions	4	1.00	0.000	0.03	0.02
	defined by Billings		1.00			
		2	0.62	0.000	0.58	0.67

## Table 38 Incremental relative risk ratios, by sex, 1993/4

Group B EDs

Table 38 shows that for all admission groups the IRR was lower for males (sex = 2) than females (sex = 1). The IRRs were lower to a greater extent for males compared to females for the ACS admission groups representing readmissions (groups 3, 7, 10, 12, 14) relative to admissions (groups 1, 5, 9, 11, 13) and, apart from the Billings' ACS conditions, lowest for the marker admission groups (15 and 16), with the exception of group 21 (Billings' ACS conditions).

#### (iv) Access factors

#### GP access factor

Table 39 below shows the results for the GP access factor GAF2 using Group B EDs for 1993/4. In the table the results for GAF2q23 and GAF2q45 are shown relative to the baseline (GAF2q1 which takes a value of 1.000).

#### Table 39 Incremental risk ratios for GP access factor (GAF2), by admission group, 1993/4

Admission Group	Description	Access	IRR	P value	95% c.l lower	95% c.l upper
1	Strong and weak ACS conditions, any urgency,	gaf2q1	1.0000	-	-	
	all admissions	gaf2q23 gaf2q45	1.0115	0.458	0.9812	1.0428
3	Strong and weak ACS conditions, any urgency,	gaf2q1	1.0000	•	•	-
	readmissions	gaf2q23 gaf2q45	1.0053 0.9954	0.872 0.905	0.9425	1.0723 1.0733
5	Strong and weak ACS conditions, urgent,	gaf2q1	1.0000			
	all admissions	gaf2q23 gaf2q45	0.9945	0.833	0.9447	1.0468
7	Strong and week ACS conditions, urgent,	gaf2q1	1.0000	-	•	•
	readmissions	gaf2q23 gaf2q45	0.9610	0.431 0.889	0.8703	1.0611 1.1138
9	Strong ACS conditions, any urgency, all admissions	gaf2q1 gaf2q23 gaf2q45	1.0000 1.0592 1.1032	- 0.028 0.001	1.0063 1.0391	1.1449 1.1713
10	Strong ACS conditions, any urgency, readmissions	gaf2q1 gaf2q23 gaf2q45	1.0000 1.0128 1.0302	0.687	- 0.9522 0.9565	- 1.0772 1.1096
11	Strong ACS conditions, urgent, all admissions	gaf2q1 gaf2q23 gaf2q45	1.0000 1.0343 1.0734	0.361 0.099	0.9620 0.9867	- 1.1120 1.1678
12	Strong ACS conditions, urgent, readmissions	gaf2q1 gaf2q23 gaf2q45	1.0000 0.9903 1.0345	0.901 0.707	0.8500 0.8666	1.1538 1.2350

Admission Group	Description	Access variable	IRR	p value	95% c.l lower	95% c.l upper
13	Strong ACS conditions, very urgent,	gaf2q1	1.0000		-	
	all admissions	gaf2q23	1.0224	0.634	0.9334	1.1198
		gaf2q45	1.0485	0.385	0.9423	1.1668
14	Strong ACS conditions, very urgent,	gaf2q1	1.0000			
	readmissions	gaf2q23	1.0195	0.555	0.9561	1.0872
		gaf2q45	1.0563	0.156	0.9794	1.1392
15	Strong and weak markers, very urgent,	gaf2q1	1.0000		-	
	all admissions	gaf2q23	0.9923	0.834	0.9233	1.0664
		gaf2q45	0.9702	0.496	0.8894	1.0584
16	Strong markers, very urgent,	gaf2q1	1.0000			
	all admissions	gaf2q23	0.9831	0.655	0.9125	1.0592
		gaf2q45	0.9465	0.233	0.8648	1.0359
17	Admissions in surgical specialties	gaf2q1	1.0000			
		gaf2q23	0.9932	0.524	0.9728	1.0141
		gaf2q45	0.9939	0.630	0.9695	1.0190
18	Admissions in medical apeciaities	gaf2g1	1.0000		-	-
		gaf2g23	1.1305	0.000	1,1072	1.1541
		gaf2q45	1.0613	0.000	1.0352	1.0880
19	Emergency admissions in surgical specialties	gaf2q1	1.0000			
		lasf2a23	0 9729	0.390	0.9139	1 0358
		gaf2g45	0.9619	0.332	0.8893	1.0404
20	Emergency admissions in medical apecialties	gaf2q1	1.0000	-	-	•
		gaf2g23	1.0332	0.115	0.9921	1.0761
		gaf2g45	1.0358	0.172	0.9849	1.0893
21	Admissions for ACS conditions	gaf2g1	1.0000			
	defined by Billings	gaf2g23	0.9875	0.854	0.8632	1.1298
		oaf2o45	1.0329	0.689	0.8815	1.2103
23	Admissions for marker conditions	aaf2a1	1,0000			
	defined by Billings	gaf2g23	0.8556	0.006	0.7651	0.9568
			0.0050	0.000	0.7000	0.0560

Table 39 Continued

Group B EDs

For all admission groups except groups 9, 18, and 23, the IRRs for the GP access factor (GAF2) were not significant (p < 0.05). For admission group 23, there was a negative significant (p < 0.01) relationship between admissions and GAF2 suggesting that *higher* access to GPs was associated with a *lower* number of admissions for these marker conditions. There was a positive and significant (p<0.001) relationship between GAF2 and admissions in group 18 (admissions in medical specialties) although a non-significant relationship for admissions in group 20 (emergency admissions in medical specialties). The difference between group 18 and 20 is that group 18 included emergency and non-emergency admissions, while group 20 included only emergencies. These results could suggest that *increased* access to GP facilities might result in a *higher* level of admissions for non-emergency medical conditions. To test this further a separate analysis was carried out – a new admissions *joinus* those in group 20 (emergency medical admissions) – ie non-

emergency medical admissions. The results for group 18, 20 and the new group are shown in the table below.

Admission	Description	Access	IRR	Р	95% c.l	95% c.l
Group		Variable		value	lower	upper
18	Admissions in medical specialties	Gaf2q1	1.0000	-	•	•
		Gaf2q23	1.1305	0.000	1.1072	1.1541
		Gaf2q45	1.0613	0.000	1.0352	1.0880
20	Emergency admissions in medical specialties	Gaf2q1	1.0000	•	•	•
		Gaf2q23	1.0332	0.115	0.9921	1.0761
		Gaf2q45	1.0358	0.172	0.9849	1.0893
New group	Non-emergency medical admissions	Gaf2q1	1.0000	-	•	
		Gaf2q23	1.1631	0.000	1.1353	1.1915
		Gaf2q45	1.0741	0.000	1.0437	1.1092

## Table 40 Incremental risk ratios (IRRs) of GP access factor (GAF2) for selected admission groups, 1993/4

Group B EDs

Table 40 shows that the results are slightly more pronounced using the new group (nonemergency medical admissions) – there is a significant (p < 0.001) positive relation with GP access factor. In the new group, the GP access factor does not behave linearly –that is, there is no evidence that GAF2q45 has a larger value than GAF2q23.

#### Hospital access factor (HAF2)

Table 41 below shows the results for the hospital access factor (HAF2) using Group B EDs for 1993/4. In the table the results for HAF2q23 and HAF2q45 are shown relative to the baseline (HAF2q1 which assumes a value of 1.000).

Admission	Description	Access	IRR	P	95% c.I	95% c.l
Group		variable		value	lower	upper
1	Strong and weak ACS conditions, any	haf2q1	1.0000	•	•	•
	all admissions	haf2q23	1.0928	0.000	1.0628	1.1237
		haf2q45	1.1221	0.000	1.0807	1.1443
3	Strong and weak ACS conditions, any	haf2q1	1.0000	-	•	-
	urgency,	b=12a22	1 1073	0.001	1.0441	4 4743
		haf2n45	1 1471	0.001	1.0802	1 2182
5	Strong and weak ACS conditions, urgent.	haf2q1	1.0000	0.000	1.0002	1.2102
	all admissions	haf2o23	1.0832	0.001	1.0338	1.1351
		haf2q45	1.1457	0.000	1.0922	1.2019
7	Strong and weak ACS conditions, urgent,	haf2g1	1.0000	-		-
	readmissions	haf2q23	1.0545	0.251	0.9631	1.1548
		haf2q45	1.1892	0.000	1.0843	1.3042
9	Strong ACS conditions, any urgency,	haf2q1	1.0000	-		-
	all admissions	haf2q23	1.1088	0.000	1.0579	1.1621
		haf2q45	1.1330	0.000	1.0798	1.1889
10	Strong ACS conditions, any urgency,	haf2q1	1.0000	-	•	-
	readmissions	haf2q23	1.2242	0.000	1.1537	1.2991
		haf2q45	1.2307	0.000	1.1590	1.3069
11	Strong ACS conditions, urgent,	haf2q1	1.0000	-	-	-
	all admissions	haf2q23	1.3010	0.000	1.0576	1.2075
		haf2q45	1.1775	0.000	1.1001	1.2603
12	Strong ACS conditions, urgent,	haf2q1	1.0000	-	-	-
	readmissions	haf2q23	1.2561	0.002	1.0886	1.4494
		haf2q45	1.4299	0.000	1.2372	1.6526
13	Strong ACS conditions, very urgent,	haf2q1	1.0000	-	•	-
	all admissions	haf2q23	1.1652	0.000	1.0711	1.2677
		haf2q45	1.2502	0.000	1.1472	1.3624
14	Strong ACS conditions, very urgent,	haf2q1	1.0000	-	•	-
	readmissions	haf2q23	1.0807	0.010	1.0191	1.1461
		haf2q45	1.1672	0.000	1.0992	1.2393
15	Strong and weak markers, very urgent,	haf2q1	1.0000	•	-	•
	ali admissions	haf2q23	1.1030	0.004	1.0319	1.1789
		haf2q45	1.1119	0.002	1.0380	1.1909
16	Strong markers, very urgent,	haf2q1	1.0000		•	-
	all admissions	haf2q23	1.0856	0.020	1.0133	1.1631
		haf2q45	1.0923	0.015	1.0174	1.1729
17	Admissions in surgical specialties	haf2q1	1.0000	-		
		nerząz3	1.0955	0.000	1.0749	1.1168
		nar2q45	1.0704	0.000	1.0495	1.0917
18	Admissions in medical specialties	haf2a4	1 0000			
		hemen a	1.0000	0.000	4.0704	
		hef2q23	1.0925	0.000	1.0/21	1.1133
		n812q45	0.9618	0.000	0.9430	0.9810

# Table 41Incremental risk ratios for hospital access factor (HAF2), by admission<br/>group, 1993/4

#### Table 41 Continued

Admission Group	Description	Access variable	IRR	p value	95% c.l lower	95% c.l upper
19	Emergency admissions in surgical specialties	haf2q1	1.0000	•		
		haf2q23	1.0749	0.017	1.0219	1.1408
		haf2q45	1.105	0.001	1.0392	1.1749
20	Emergency admissions in medical specialties	haf2q1	1.0000	*		
		haf2q23	1.0879	0.000	1.0457	1.1319
		haf2q45	1.1436	0.000	1.0982	1.1908
21	Admissions for ACS conditions	haf2q1	1.0000	-	•	•
	defined by Billings	haf2q23	1.0704	0.276	0.9472	1.2095
		haf2q45	1.0636	0.339	0.9373	1.2069
23	Admissions for marker conditions	haf2q1	1.0000	1.	•	-
	defined by Billings	haf2q23	1.0541	0.314	0.9514	1.1678
		haf2q45	1.1298	0.024	1.0161	1.2561

Group B EDs

Table 41 also shows that for almost all admission groups (except groups 21 and 23), there was a positive significant (p < or = 0.05) relationship between admissions and the hospital access factors (HAF2q23 and HAFq45 relative to HAF2q1). This suggests that admissions are greater with better access to hospital facilities. The IRR values suggest a dose-response relationship (with HAF2q45 generally being higher than HAF2q23) but since the confidence intervals for these variables overlapped in every case, the relationship may be spurious albeit suggestive. The IRR values for HAF2 were highest for admission groups 11 (strong ACS conditions, urgent, all admissions) and 12 (strong ACS conditions, urgent, readmissions).

There were similar results for the GP and hospital access factor for other years using Group B EDs (data not shown). The results for Group C EDs, which allow a better comparison across years, are shown in appendix F.

#### (v) Hospital service area.

The IRRs produced by the Poisson regression for HSAs are shown in the table below. Again Group B EDs are used. For simplicity, the results are shown only for 1993/4, and only for admission groups 1 (ACS) and 16 (markers). The results for other admission groups are shown in appendix G. The results for each HSA are shown relative to Barnet HSA, which assumes a value of 1.000.

Admission	Description	Hospital Service	IRR	Р	95% c.l	95% c.l
group		Area		value	lower	upper
1	Strong and weak ACS conditions, any urgency,	Barnet	1.0000	-	-	-
		Ealing	0.9665	0.000	0.9509	0.9823
	all admissions	Edgware	0.9925	0.233	0.9803	1.0048
		Hernel Hempstead	1.0097	0.036	1.0001	1.0189
		Hillingdon	0.9972	0.484	0.9897	1.0049
		Lister	0.9994	0.861	0.9924	1.0064
		Mt Vernon	1.0164	0.000	1.0099	1.0229
		Northwick Park	1.0067	0.007	1.0019	1.0116
		Queen Elizabeth II	0.9977	0.381	0.9926	1.0028
		St Mary's	0.9892	0.000	0.9849	0.9936
		Watford	1.0089	0.000	1.0051	1.0127
		West Middlesex	0.9982	0.342	0.9945	1.0019
16	Strong markers, very urgent,	Barnet	1.0000	-	-	•
	all admissions	Ealing	0.9546	0.028	0.9158	0.9951
		Edgware	0.9902	0.539	0.9596	1.0217
		Hernel Hempstead	1.0034	0.767	0.9809	1.0265
		Hillingdon	0.9837	0.099	0.9648	1.0031
		Lister	1.0200	0.023	1.0027	1.0375
		Mt Vernon	1.0014	0.868	0.9850	1.0180
		Northwick Park	0.9815	0.004	0.9692	0.9940
		Queen Elizabeth II	1.0024	0.708	0.9899	1.0150
		St Mary's	0.9664	0.000	0.9552	0.9777
		Watford	1.0017	0.722	0.9922	1.0113
		West Middlesex	0.9884	0.017	0.9789	0.9979

# Table 42 Incremental relative risk ratios (IRRs) for selected hospital service areas, and selected admission groups, 1993/4

Group B EDs

There are weak but significant relationships (positive and negative) in approximately half of the HSAs for each admission group, the other half having a non-significant relationship.

For ACS conditions (admission groups 1-14) there was no obvious pattern to the results – no HSA stood out as having a high or low IRR. The same was true for marker conditions (admission groups 15 and 16). However the IRR values were particularly low in Ealing and Hillingdon HSA (relative to Barnet) for emergency surgical (group 19) and emergency medical admissions (group 20).

A test for spatial correlation was performed (see chapter 2 on methodological implications), which was based on fitting a negative binomial model while allowing for extra-Poisson variation. The test indicated that there was significant spatial correlation. This means that, taking account of the variables in the model, a significant amount of so called 'extra-Poisson' variation existed between EDs which was not independent of spatial location. However, the test value of 1.2 indicated that the extent of spatial correlation was not big enough to be a problem in the analysis.

#### 6.4 Discussion

#### (a) Methodological issues

In any study such as this, there are methodological limitations. Chief among these must be the limitations of using HES data for small area analyses of admissions groups for specific diagnoses, because the accuracy and completeness of these data are known to be suspect. However this study has gone further than most small area analyses in investigating the accuracy and completeness of the admissions dataset used – chapters 4, 5 and section 6.3.1 of chapter 6 describe extensive investigation – and precautions were taken as a result. For example of all the possible enumeration districts under study between a fifth and one half were finally used in different parts of the analysis. It was not possible to take into account the detailed results of the data accuracy study (described in chapter 5), for example in a sensitivity analysis, partly because it would have been difficult, if not misleading, to extrapolate the findings of the necessarily few ACS and marker conditions investigated in this study to the whole set of ACS conditions, such as asthma and diabetes, methods might be developed so that coding accuracy can be taken into account more formally in the analysis.

If there were limitations with the numerator in this study, outlined above, there were also potential inaccuracies in the value of the denominator – the population size. The 1991 census was known to be incomplete in particular groups, for example young males in inner London. This was corrected as far as possible using adjustment factors recommended by the Office for National Statistics. It is not possible to know to what extent these adjustment factors are appropriate.

Another issue was how exactly to define hospital service areas. In this study EDs in which 40% or more admissions occurred at the hospital in question were included in the relevant hospital service area. Using a higher cut-off proportion (say 80%) might have produced different results, in particular possibly a more marked 'hospital effect' than that shown in table 42, but would have meant that large numbers of EDs were excluded from the analysis. Reviewing the literature on small area analyses, no consistent pattern is apparent in the definition of hospital service area; indeed most multivariate studies have not included HSA as a variable in the analysis.

The enumeration districts used in the study had an average population size of 429 – very small compared to that in an electoral ward of 10,000, or a district health authority of approximately 250,000. The most extensive small area analysis on the UK to date used small areas the size of electoral wards. The smaller enumeration districts were chosen in this study to reduce the potential for ecologic fallacy, and to be able to identify hospital service areas with more accuracy. But the drawback of a smaller area is that the variation in the number of relevant observations (admissions) is larger, thus increasing the potential for 'noise' to obscure relationships between variables. Again there is no right answer, except perhaps to repeat the analysis using electoral wards instead of EDs.

Assigning hospital admission records (coded by postcode) to EDs was not straightforward because not all postcodes (boundaries determined by the UK Post Office) uniquely map onto one ED (boundaries determined by the Office for National Statistics as part of the ten-yearly census). Therefore it might be that some admissions were assigned to EDs with inappropriate characteristics. However overall the number of postcodes this applied to was relatively small, and the problem is to a large extent intractable given that the way of geo-mapping the residence of census populations is different to that used for patients admitted to hospital.

Central to the investigation was the effect of the access factors on admissions for ACS and marker conditions. The formula used was complicated, drawing on routine sources of data. The basis of the formula to construct the GP access factor was the number of GPs in an area relative to the size of the population 'competing' for them, taking into account the distance of the population from the GPs. This is overly simplistic for at least four main reasons. First, it matters less to an individual how many GPs are available in the local area, than how accessible is *their own* GP, or the GPs in the practice in which they have registered. The availability of GPs in other practices may not affect this. Second, the number of full time equivalent GPs employed in a practice may be less important to access than, for example, the hours the GPs are available to patients. Third, access to a GP may be less of an issue than access to a GP with the appropriate knowledge of, and interest in, the ACS conditions concerned. Fourth, if a hypothesis is that access to primary care is relevant for the effective management of ACS conditions, then access to other relevant primary care staff and facilities may be more appropriate. However it would be very difficult to include these factors into a formula. For example, it is impossible to obtain from routine sources accurate data on opening hours and out-of-hours cover by GPs.

On the hospital access factor, there is arguably a better case for supposing that the supply of hospital resources (in this case beds) in different local hospitals is relevant to the accessibility of hospital care. Patients can in theory choose any hospital to visit for a consultation in the accident and emergency department, and the number of beds may be less important as a feature of access than the number of clinicians, the size of the accident and emergency unit or other on-site facilities, or the availability of public transport to the hospital. Some studies have demonstrated the importance of some of these factors, although the relative importance of each is not known. Again there should be further work on refining the quantification of access by developing the formula to include some of these variables.

In the multivariate analysis, Poisson regression was used but not more complicated techniques such as multi-level modelling (which probes more fully the effect of different 'layers' of characteristics - such as those relating to areas and institutions – on the dependent variable), or Bayesian analysis (which seeks to take account of the variation between areas due to proximity to each other), or two-stage least squares regression (which is purported to be more useful when examining the relationship between sets of variables when the direction of that relationship is two-way, eg high admissions resulting in a high supply of facilities, and vice versa (99)). All these were thought to beyond the immediate scope of this thesis, although investigation using these techniques could be a next stage of the analysis.

A large number of analyses were performed, for example using data referring to different years, to different admission groups, and different groups of EDs. If a large number of analyses are performed, then there is a greater possibility of finding significant results, even though there may be no significant relationships to be found. For example, using a 5% significance level, if 20 studies were performed based on random samples of data and in which there were no significant relationships, then one of the 20 studies would produce a significant result. In this study a balance was struck between conducting a thorough analysis, and not being at risk of finding a spurious result through conducting too many analyses. For example, in the multivariate analyses, results were presented mainly for one year only (1993/4) because the study in chapter 5 suggested that clinical coding of HES data was likely to be improving over time, and mainly for Group B EDs only (because Group B contained the largest number of EDs for which the GP access factors could be calculated).

#### (b) Substantive findings

#### (i) Univariate analyses and analysis by HSA

#### Need variables

The analysis showed that the area of North West Thames region under investigation had generally less socioeconomic deprivation compared to England and Wales as a whole apart from areas around St Mary's Hospital, Paddington, Ealing, Luton and West Middlesex Hospitals. More deprived enumeration districts also tended to be those with higher populations.

Much less variation across the region in values for SMR and SIR was found. Again, table 32 shows that the values were generally more favourable than for England & Wales as a whole. Low values for the SIR were found in EDs in St Mary's hospital service area in Paddington and also in Ealing - which are both relatively socially deprived areas. Why this should be so is not clear, but the finding might explain why the correlations between the need variables (shown in table 34) are also somewhat counter-intuitive. One might expect a positive and significant relationship between all three 'need' variables, but the table shows a non-significant relationship between the Carstairs and SMR scores and a weak negative relationship (p = 0.005) with SIR. The relationships may have been weakened by using quintiles rather than raw scores.

#### Access variables

There was a tendency for the mean values for the GP access factor to be higher in HSAs closer to central London, and lower in relatively rural areas, suggesting better access to the supply of general practitioners in and near central London. Similar results were found in 1991/2 and 1992/3 (data not shown).

For the hospital access factor (HAF2), the mean values were not obviously higher in HSAs closer to central London with the exception of St Mary's Paddington. This may due to the fact that competition for patient populations in London is more intense even though there are more hospitals in London.

#### Age-sex standardised rates

The ambulatory care sensitive (ACS) conditions and marker conditions selected for analysis represented a greater proportion of all admissions than those selected by Billings' *et al.* For

example admission group 1 represented approximately 26% of all admissions per study year, and markers 4% compared to Billings' ACS conditions (1.18% of all admissions) and marker conditions (1.83%)(see table 25).

Using the conditions identified by Billings *et al* as being ACS, the rate across the region (using Group A EDs) was found to be 1.56 per 1000 population (see table 28). This is much lower than the 7.38 per 1000 found in Toronto and 15.19 in New York City in 1993 (1)(2). There could be many reasons for this. For example in this study, many small areas from inner-city London were excluded (because of inadequate data), leaving EDs largely from suburban or rural areas, in which illness levels, deprivation, and the availability of hospital facilities for example may be relatively low. Table 30 does indeed suggest that rates of admissions for ACS conditions were generally higher for hospital service areas nearer London. It may also be that the practice style of clinicians varies across the three countries, with physicians in the UK less likely to admit patients at a certain level of severity compared to their Canadian and US counterparts. It will be important to investigate this further.

Age-standardised rates for ACS conditions (admission group 1) increased from 28.09 to 30.74 between 1991/2 and 1993/4 (see table 29) – a rise of just over 9% - although the rates for markers were stable. A similar trend was found in New York City, in which rates for ACS conditions defined by Billings rose 28.3% from 1981-93 (2).

The proportion of readmissions to admissions tended to be higher in admission groups of strong ACS conditions (see table 25). This might suggest that a lack of timely and effective ambulatory care has a greater effect in increasing the risk of readmission for patients with strong ACS conditions.

#### (ii) Bivariate analyses

#### The relationship between indicators of need and access factors

The relationship between the SMR and SIR and the access factors was mostly not significant (p>0.005). In a health system based on the principle of equal access to equal need, this was a surprising finding, presuming that the access factors used really did measure access. However given the non-significant relationship between SMR and SIR with the Carstairs score, which is

unexpected in small area analyses, it may be that the lack of association between the access indicators and SMR/SIR could be due to either measurement error in the calculation of the latter (especially the SIR, given the shortfall in the census data in inner London) or that the ED-based scores are unstable because of small populations. However there was a positive and significant (p>0.0001) relationship between Carstairs and both access scores (table 34), suggesting better access to GP and hospital facilities in deprived areas.

#### The relationship between indicators of need and admission rates

A significant (p=0.01) positive correlation was found between Carstairs scores and admission rates in all admission groups (except group 19 - emergency surgical admissions – for which the correlation was not significant). The effect was strongest in admission group 1 (ACS), but the correlation (0.325) was weaker than that found between the admission rates for ACS conditions (defined by Billings *et al*) and area income in the US (where between 60 and 70% of variation in ACS admission rates between areas could be explained by percentage of residents with an income below \$15,000). Using Billings' *et al* ACS conditions, the correlation with Carstairs score in North West Thames was much weaker at 0.104.

A weak positive but significant (p=0.01) relation was found in North West Thames region between Carstairs scores and markers (groups 15 and 16). This was in contrast to New York City where the relationship between admission rates for markers and area income was not found to be significant.

Comparing the ratio of admission rates for the fifth most deprived EDs with the fifth least deprived in North West Thames region, the ratios varied for UK-defined ACS admission groups from 1.5 to 2.5 (see appendix 3). In contrast, the ratio of admission rates for Billings' ACS conditions between low and high income zip code areas in New York City was found to be higher at 3.4 in 1993, although it is not clear exactly how this ratio was calculated. It is not clear which features of socioeconomic deprivation are more relevant to admissions for ACS conditions: those included in the Carstairs index, or income. It may be that in the US, where a lack of health insurance may be the single biggest factor reducing access to primary care for those with ACS conditions, income is the better indicator of socioeconomic deprivation whereas in the UK the composite measure might be more relevant. It may be possible to test this in the UK, although data on the average income of residents of small areas are difficult to obtain.

In North West Thames, higher ratios of admission rates for the upper to the lower socioeconomic quintiles were observed in admission groups for readmissions. People living in socioeconomically deprived areas were particularly susceptible to readmission for ACS conditions. Among the many possible reasons are that hospitals serving more deprived populations may experience greater pressures to discharge patients from hospital before they are ready, or the quality of primary care may be deficient in these areas.

#### (iii) The multivariate analysis

Overall, only 10-20% of the variation between EDs in admissions for ACS conditions could be 'explained' by the model, and only 5% for marker conditions – so the model has limited explanatory power. This is in contrast to the analysis by Billings *et al* in which 60-70% of the variation between areas could be explained by area income. A large part of the explanation will be that the populations of the small areas were of the order of 450, whereas for Billings they were approximately 20,000 –30,000. But the results are consistent with the bivariate analysis: socioeconomic differences between individuals, as measured, are not as closely related to hospital admission rate in the UK than the US.

The test for dispersion also indicated a significant amount of extra-Poisson variation. This suggests that either the right variables were identified but the indicators used in the study were not good measures of them, or that other variables (not measured) influenced the variation. These other variables may be very different from those included, for example ethnicity or physician practice style (as suggested in the literature). A next step in the analysis might be to experiment using different variables for socioeconomic deprivation and access, before moving onto potentially interesting variables suggested in the literature as discussed in chapter 2.

There was a positive and significant relationship between the three need variables and all admission groups, and the relationship was stronger for admission groups representing readmissions. If admissions for ACS conditions do suggest inadequate upstream primary care, then the effect of the latter on readmissions is apparently greater. Perhaps some general practices are particularly poor at providing support to patients after discharge from hospital, possibly because of a lack of clinical skill, time, primary care staff working in the community, inadequate organisational arrangements to identify when patients registered with the practice are discharged

from hospital, or inadequate support from social services locally.

The pattern of admissions by age was 'U'-shaped, as has been found elsewhere. The strongest age gradients were seen in Billings *et al's* ACS conditions (group 21) and markers (group 23) in age group 7 (65-74 years), in which the value of the IRR was six times higher (for group 21) and thirty eight times higher (for group 23). This may reflect that the conditions selected by Billings *et al* were more likely to be prevalent in this older age group. As expected, males had lower levels of admission than females, but this was especially true for readmissions. There may be a number of reasons for this. For example, older females are more likely to live alone without a live-in carer, than older males. All the admissions examined in this study however referred to those aged under 75 years.

The effect of the hospital access factor was positive and significant in almost all admission groups, indicating that the higher the availability of hospital beds to a population, the higher the admissions. In most admission groups there was a dose-response relationship, that is the higher the access to hospital facilities (as measured) the higher the admissions. The strength of the relationship was strongest for groups 11 (strong ACS, very urgent, admissions) and particularly group 12 (strong ACS, very urgent, readmissions). Access to hospital facilities appears to have a relatively strong positive effect on admissions for very urgent ACS conditions, whereas access to GPs appears to have a relatively strong positive effect on less urgent ACS and non-urgent medical conditions.

In almost all admission groups, the effect of the GP access factor was not significant. This access factor was found to have a significant if weak positive effect for admission group 9 (strong ACS conditions, any urgency), group 18 (all medical admissions) (weakly positive) and the new admission group (positive) – non-emergency medical admissions. In admission group 9 and 18 the effect of the GP access factor was linear – the higher the access to GPs the higher the admissions (in contrast to the hypothesis suggested in the introduction) – although in the non-emergency medical group the effect was not linear. The results suggest that people with non-urgent (and ACS) conditions are more likely to be admitted if they have greater access to general practitioners in primary care. Far from primary care preventing admission, it may be associated with higher levels of admission for non-urgent cases. The converse appears to be true for marker conditions, and admissions in surgical specialties, where a negative relation with access to general practitioner care was noted – that is lower access results in higher admissions – although only in one

admission group (23) was the relationship significant. Again this is in contrast to the hypothesis suggested at the outset in chapter 1, and suggested by Billings *et al.* It may be that patients with marker conditions are either managed at home, possibly inappropriately, in areas with lower access to GP care, or that the conditions are not spotted in primary care before a fatality occurs (reducing the potential for admission), or that the GP access factor used in the study does not accurately measure access to primary care: each of these possibilities should be investigated further.

Because the access factors might exert different effects according to the age of patients admitted, for example in older people, the model was run for individual age groups (data not presented). The results suggest, in contrast to the US, that there were no marked differences in the influence of GP access factor or hospital access factors by age. It may be that although the types of conditions, and the severity of the conditions, experienced by age groups are different, the effect of age *per se* is small or negligible on overall hospitalisation. It might be expected, given the financing and organisation of the NHS, that older people would have similar levels of access to health care to younger people, in contrast to the United States. Although evidence in the UK does point to the greater difficulty experienced by older people in accessing primary care, for example in some rural areas, overall in North West Thames if this occurs the effect on hospitalisation rates does not appear to be great.

Finally the 'hospital effect' did not seem to be an important factor influencing variations in admissions between areas (see table 42). As noted above this may be because the hospital effect was blurred because of the way that EDs were assigned to HSAs. Further analysis using higher cut-off levels may be useful to see if the hospital effect becomes stronger and more significant.

# Chapter 7 Summary of main findings, next steps, and implications for the NHS

Chap	ter outline
7.1	Context
7.2	Main findings
7.3	Next steps for analysis
7.4	Direct implications for the NHS

#### 7.1 Context

This is the first study in the UK to consider variations in admission rates for ambulatory care sensitive (ACS) and marker conditions across small areas.

The area under study is a relatively wealthy part of the south-east of England, with small pockets of deprivation. The area covers parts of London.

Almost all of the population studied received health care provided by the national health service (NHS) which is financed and organised on the principle of equal access for equal need. A comprehensive range of benefits are available (including primary and secondary care) and care is largely free at the point of use. However there may be significant personal costs to individuals in using health care, for example cost of time off work, and cost of transportation to facilities. Most small area studies of ACS conditions have been conducted in the USA, in which there are significant numbers of the population without health insurance cover for basic care, including primary and secondary care. One study, noted in chapter 2, examined variations in ACS conditions (as defined by Billings *et al*) in Canada compared to the USA (2). As in the NHS, the Canadian health system provides universal access to health care and comprehensive benefits, with no incentives for providers to limit services to people with low incomes. But unlike the NHS, the Canadian system does require more co-payments at the point of use.

The analysis was of data from 1991-1994 – a time when there were many mergers of hospitals as a result of the 1991 NHS reforms, and when 10-30% of the general practices had joined the GP fundholding scheme. The merging of hospitals created problems in the identification of hospital service areas, and possibly in the completeness and accuracy of the hospital episode statistics dataset. Also, the GP fundholding scheme may have had an effect on the quality of ambulatory care, since prescription drugs and diagnostic tests were included in GP fundholders' budgets whereas admissions to hospital for medical conditions (largely ACS conditions) were not. There were new potential incentives therefore for GPs to reduce expenditure on pharmaceuticals and diagnostic tests which could have affected the management of patients with ACS conditions. However the evidence on the latter, although weak, is not suggestive (296).

#### 7.2 Main findings

As others have shown (1)(22)(64)(65), it is possible for clinicians to achieve a consensus on which conditions may be ambulatory care sensitive and which are not. The study described in chapter 3 is the only one of its type published in the UK, and went further than previous studies in identifying not only ACS conditions but also the degree of sensitivity to ambulatory care and the likely urgency of admission. The study showed that general practitioners were more sanguine than their specialist colleagues as to the scope for preventing admissions through better management in ambulatory, particularly primary, care. A number of conditions were identified as being ACS, and there was similarity with those identified in American studies (see chapter 3 table 5) (1)(22)(64)(65).

The analysis of the completeness of the main dataset to be used in the small area analysis, described in chapter 4, was also unique to the small area variations literature (UK or US). Major shortcomings were found in the dataset and as a result a large number of areas, and years of data, were excluded from further analysis. Similarly the investigation of accuracy of the clinical coding of the admissions dataset was also unique to small area variations literature. Few good studies of this type have been conducted in the UK (284), yet the analysis showed major differences in coding between two sets of coders, including the coding for ACS conditions and markers. It was not possible to take account of most of the findings in the small area analysis.

Using Billings' list of ACS conditions, a much lower rate of admissions for ACS conditions was found in this study in the UK NHS (1.56 per 1000) than in New York City (15.16) or in Toronto (7.38). ACS admission rates increased over the three study years in this study, a trend also found by Billings in New York (1).

A significant positive relationship was found between admissions for ACS conditions and socioeconomic deprivation, independently of indicators of mortality and morbidity. However the relationship in the study between area *deprivation* and ACS admissions was much weaker than in the US or Canada between ACS admissions and area *income*. For example the ratio of admission rates for ACS conditions (as defined in the UK) of the most deprived compared to the least deprived areas in the UK was between 1.5 and 2.5 compared to 3.4 in the USA(using the ACS conditions defined by Billings). In the UK the relationship with deprivation was stronger for readmissions (for ACS conditions). In the USA, the main studies were not able to analyse readmissions separately. In contrast to findings in USA, a significant but weak relation was found between socioeconomic deprivation and admission rates for marker conditions.

The relationship between an indicator of access to primary care and admissions for ACS conditions was not significant, although weakly positive. The relationship reached statistical significance in a group representing 'strong' ACS conditions, and non-urgent medical conditions. In the USA, most small area studies of ACS conditions generally did not include an explicit measurement or analysis of access to primary care facilities. In three studies that did, two found a negative relationship between the supply of primary care physicians and admission rates (66)(123). However one of these ((123) found that, in a multivariate analysis, the negative association of supply of primary care with hospitalisation rates for ACS conditions was very weak. A third study found a negligible effect between ACS admission rates and the overall supply of physicians in an area (186), but this study was conducted in the Medicare population which, on the whole, enjoy significantly better access to health benefits than the rest of the US population, including primary care. The Medicare population, in this sense, is more similar than other population groups in the US to the population using NHS care in the UK. Bindman et al investigated a measure of perceived access to primary care (derived from self-rated responses to a community survey) and found that it was strongly and negatively associated with hospitalisation rates for the 5 ACS conditions studied (22).

Generally a positive relationship between admissions for ACS and marker conditions was found with the indicator of access to hospital care. This is in keeping with the overall findings elsewhere such as the small area analysis at the University of York (99).

While admission rates were found to be higher for the very young and those aged over 65 years, there was no obvious relationship with access to primary care and hospital care by age group. This is in contrast to the work by Billings *et al*, in which variations in admission rates across areas were less for these age groups compared to those aged 25-44 years (1).

The analysis suggested that there was a significant 'hospital effect' in some cases, but that this was not as strong as that found in the work by Tedeschi in the USA (173). This might be expected in a national health system in which most hospitals are owned, regulated, financed, and organised by the NHS.

The overall explanatory power of the multivariate model was weak at 10-20%, in contrast to the USA, where various investigators have found much stronger relationships between admission rates for ACS conditions and variables such as area income (1) and self-rated access to care (22). This will be largely due to the fact that the small areas used in this study were much smaller than those used in the US studies.

#### 7.3 Next steps for analysis

There could be many further avenues of enquiry to build upon the early findings reported here. In the paragraphs below, the most important next steps are outlined.

First, the same analysis could be repeated using electoral wards as small areas rather than the smaller enumeration districts. This would reduce the amount of 'noise' at the cost of some loss of resolution in defining the hospital service areas.

Second, the unit of analysis could be changed from a small geographical area to GP practice, or primary care group, in which case the population registered with a practice (or PCG), rather than the residents of a geographically defined area, would be investigated. This might help to illuminate the relevant differences in access to primary care between patients from different practices, and the relevant components to access (for example number of nursing staff, or the opening hours of the practice). However it would create difficulties because

indicators of population need would have to be calculated for each practice population. Some researchers have developed methods of doing this using census and other data (297), although such work is still fairly crude, and a number of demanding assumptions have to be made.

Third, as Billings *et al* have noted, access to care is still very much a 'black box' (1). It would be well worth conducting small scale qualitative studies to identify more fully the features which influence access to ambulatory care for patients living in areas of interest (perhaps those in which admission rates for ACS conditions are high). These features might include the *perceived* access of the population to ambulatory care, as investigated by Bindman *et al* (22).

Fourth, the GP access factor should be refined significantly, for example to take into consideration not just the supply of general practitioners in a practice and distance to them from the small area, but other factors such as the availability of other practice staff, opening hours, waiting time for a non-urgent appointment, or experience of the primary care staff in management of ACS conditions.

Fifth, conducting the analysis using a subset of common ACS conditions, for example asthma and diabetes, or single conditions may be fruitful. It may be that the factors influencing access to timely and effective ambulatory care for patients are condition specific. Similarly it may be worth focusing on readmissions for the full range of ACS conditions, since there may be factors affecting access that are specific to patients who have been discharged recently from hospital (for example access to care from social services departments).

Sixth, the list of ACS conditions developed in chapter 3 could be refined further, taking into account more recent evidence on the potential for ambulatory care to reduce the need for admission for various conditions.

Seventh, it might be useful to refine or add in other variables to the multivariate model, which have been suggested or examined by other investigators in the field. For example a priority might be to refine the variables used to indicate 'need'. This would not be easy, for the reasons outlined in chapter 2, but it might be possible to conduct a prospective study in which those cases admitted for ACS conditions might be investigated to assess their need (for example extent of ill health, or ability to benefit) for care, their socioeconomic status, or their ethnicity. Of all the possible new variables that could be included in a multivariate analysis, it would be worth exploring using GP consultation rates, or A&E attendances from the practice, for ACS conditions to investigate the relationship with admissions. At present this work

would have to be done on a case study basis since data on these variables are not available routinely for large areas.

Eighth, there are some important methodological aspects of small area analysis that should be addressed, as outlined in chapter 2. In particular it would be helpful for guidelines to be drawn up on how best researchers might assess the influence of 'need' for care on variations in admission rates observed. Similarly helpful would be guidance on how best to measure the supply of resources in hospital or primary care, and their availability or accessibility to the relevant populations. Also if the variation in admission rates for ACS conditions is to be developed as a potential indicator of equity of access to primary or ambulatory care across different countries, then it would be helpful if a set of ACS conditions could be agreed internationally and drawn up using well recognised and explicit techniques. On the other hand, it may be that the conditions that are defined as ambulatory care sensitive are too country-specific to be developed and agreed across countries. For example agreement on a list of ambulatory care sensitive conditions may be influenced by the expectations of care, availability of treatments, clinical practice style of physicians, and system of financing and organising health care in a country.

#### 7.4 Direct implications of the findings for the NHS

The implications of the findings from chapters 4 and 5 for the NHS are worrying. Information on hospital episode statistics is a potentially very valuable resource for all sorts of useful analyses, from economic evaluation to epidemiological investigation. Furthermore, unlike the case in Canada and the USA, NHS hospital episode statistics (HES) are reasonably comprehensive in that they cover the majority of hospital admissions for the population. While there is an increasing proportion of admissions to private providers (292), these are mostly for non-urgent surgery rather than for urgent or medical conditions. Although the NHS data investigated were from the early 1990's, when many hospitals were in the process of investing in new computerised information systems, it will be important to assess whether the situation is similar now. This is a matter of some urgency because HES are currently expensive to collect and grossly underused, given their significant potential to inform policy and practice. This is partly because of a lack of confidence in their accuracy which may not be completely justified.

The results of the main part of the study, at face value, suggest that in contrast to the USA, and to a much lesser extent Canada, access to ambulatory care is reasonably equitable in the

NHS. However, before policy-makers in the UK can relax, as the discussion above and in chapter 6 notes, the analysis represents a first small and tentative step into what could be a large area of investigation. Given the potential for admissions for ACS conditions to be an indicator of the quality of and access to ambulatory care (particularly primary care), it is important that further analysis along the lines suggested above be conducted.

Given the new developments in primary care, that is the formation of primary care groups and trusts which must now manage a budget for almost all the health care for their registered populations, scrutiny of the budget is likely to be on the agenda of all PCGs and PCTs in the future. Expenditures in the hospital sector will be the biggest call on these budgets, and so PCGs/PCTs have a bigger incentive than before to identify avoidable admissions and seek to reduce them where appropriate. Helping PCGs and PCTs to do this using a list of ACS conditions (and to investigate the reasons for variations) will become more relevant, and something that public health physicians in particular would be well-placed to lead and support. The fact that much information on activity in primary care (for example consultations by condition) is now available on computer, as is ambulatory care provided in accident in emergency departments, provides new opportunities for in-depth and creative local analysis.

#### References

(1) Billings J, Zeitel L, Lukomnik J, Carey TS, Blank AE, Newman L. Impact of socioeconomic status on hospital use in New York City. Health Affairs 1993;12:162-73

(2) Billings J, Anderson GM, Newman LS. Recent findings on preventable hospitalizations. Health Affairs 1996;15:239-49

(3) Casanova C, Starfield B. Hospitalizations of children and access to primary care: a cross national comparison. International Journal of Health Services 1995;25:283-94

(4) World Health Organisation, World Health Report 2000-Health Systems: Improving Performance <<u>www.who.int/whr/</u>>.

(5) Sanderson C, Dixon J. Conditions for which onset or hospital admission is potentially preventable by timely and effective ambulatory care. Journal of Health Services Research and Policy 2000;5:222-30

(6) Dixon J, Sanderson C, Elliott P, Walls P, Jones J, Petticrew M. Assessment of the reproducibility of clinical coding in routinely collected hospital activity data: a study in two hospitals. Journal of Public Health Medicine 1998;20:63-9

(7) Mooney G. Economics, medicine and health care. Harvester and Wheatsheaf, Hertfordshire, 1992.

(8) Le Grand J. Equity and choice. London, Harper Collins, 1991.

(9) Bradshaw J. A taxonomy of social need. In: Problems and progress in medical care: Essay on current research. 7<sup>th</sup> series. Ed: McLachan G. Published for the Nuffield Provincial Hospitals Trust by the Oxford University Press, London, 1972.

(10) Goddard M, Smith P. Equity of access to health care. The University of York, March 1998.

(11) Evans RG et al (Eds). Why are some people healthy and others not?: The determinants of health and population. De Gruyter, Berlin, New York, 1994

(12) Berk ML, Cunningham P, Beauregard K. The health care of poor persons living in wealthy areas. Social Science and Medicine 1991;32:1097-1103

(13) An introductory guide to the 1991 Census. Leventhal B, Moy C, Griffin J (Eds). Market Research Society: Census Interest Group. Henley-on-Thames: NTC publications; Market Research Society, 1993.

(14) MacIntyre S, MacIver S, Sooman A. Area, class and health: should we be focusing on places or people? Journal of Social Policy 1993;22:213-34

(15) Townsend P, Phillimore P, Beatti A. Health and deprivation: Inequality in the North. London: Croom Helm, 1988.

(16) Carstairs V, Morris R. Deprivation and mortality: an alternative to social class? Community Medicine 1989;11:210-19 (17) Jarman B. Identification of underprivileged areas. British Medical Journal 1983;286:1705-09

(18) Department of the Environment. Urban deprivation. Information note no 2. London. Inner Cities Directorate, Department of the Environment, 1983.

(19) Marsh P, Carlisle R, Avery AJ. How much does self-reported health status, measured by the SF-36, vary between electoral wards with different Jarman and Townsend scores? British Journal of General Practice 2000; 50:630-34

(20) Morris R, Carstairs V. Which deprivation? A comparison of selected deprivation indexes. Journal of Public Health Medicine 1991;13:318-326

(21) Mackenzie IF. My ward is more deprived than yours. Journal of Public Health Medicine 1998;20:186-90

(22) Bindman A, Grumbach K, Osmond D, Komaromy M, Vraniza K, Lurie N *et al.* Preventable hospitalization and access to health care. Journal of the American Medical Association 1995;274:346-7

(23) Payne N, Saul C. Variations in use of cardiology services in a health authority: comparison of coronary artery revascularisation rates with prevalence of angina and coronary mortality. British Medical Journal 1997;314:257-61

(24) Chaturvedi N, Ben Shlomo Y. From the surgery to the surgeon: does deprivation influence consultation and operation rates? British Journal of General Practice 1995;45:127-31

(25) Mooney G, Hall J, Donaldson C, Gerard K. Utilisation as a measure of equity: Weighing heat? Journal of Health Economics 1991; 10:475-80

(26) Mooney G. Reweighing heat. Journal of Health Economics 1992; 11:199-205

(27) Mooney G, Hall J, Donaldson C, Gerard K. Reweighing heat: Response to Culyer, van Doorslaer and Wagstaff. Journal of Health Economics 1992;11:199-205

(28) Culyer AJ, van Doorslaer E, Wagstaff A. Utilisation as a measure of equity by Mooney, Hall, Donaldson and Gerard. Journal of Health Economics 1992;11: 93-8

(29) Culyer AJ, van Doorslaer E, Wagstaff A. Access, utilisation and equity: a further comment. Journal of Health Economics 1992;11:207-10

(30) Le Grand J. The distribution of public expenditure: the case of health care. Economica 1978;45:125-142

(31) Le Grand J. Equity as an economic objective. Journal of Applied Philosophy 1984;1:39-51

(32) Le Grand J. Equity and choice. London, Harper Collins, 1991

(33) Donabedian A. Aspects of medical care administration. Cambridge, Massachusetts: Harvard University Press, 1973.

(34) Millman ML. Access to health care in America. Washington DC: Institute of Medicine, National Academy Press, 1993.

(35) Aday LA. The impact of health policy on access to medical care. Millbank Memorial Fund Quarterly 1976; Spring: 215-233

(36) Andersen RM, McCutcheon A, Aday LA, Chiu GY, Bell R. Exploring dimensions of access to medical care. Health Services Research 1983;18:49-74

(37) Weissman JS, Epstein AM. Falling through the safety net. Insurance status and access to health care. John's Hopkins University Press, Baltimore and London, 1995.

(38) Andersen R. Revisiting the behavioural model and access to medical care: does it matter? Journal of Health and Social Behavior 1995;36:1-10.

(39) Penchansky R, Thomas JW. The concept of access: definition and relationship to consumer satisfaction. Medical Care 1981;19;127-140

(40) Starfield B. Access – perceived or real, and to what? Journal of the American Medical Association 1995;274:346-7

(41) Donabedian A. Explorations in quality assessment and monitoring. Vol 1: The definitions of quality and approaches to its assessment. Ann Arbor MI: Health Administration Press

(42) Weinberger M, Oddone EZ, Henderson WG for the Veterans' Affairs Cooperative Study Group on Primary care and Hospital Admission. Does increased access to primary care reduce hospital readmissions? New England Journal of Medicine 1996;334:1441-47

(43) Henley DE. Trial of increased access to primary care [letter]. New England Journal of Medicine 1996;335:895-6

(44) Starfield B, Parrino TA. A trial of increased access to primary care. New England Journal of Medicine 1996;335:895-6

(45) Mohan G, Harrison BDW, Badminton RM, Mildenhall S, Wareham NJ. A confidential enquiry into deaths caused by asthma in an English health region: implications for general practice. British Journal of General Practice 1996;46;529-62

(46) Payne JN, Milner PC, Saul C, Browns IR, Hannay DR, Ramsay, L. Local confidential inquiry into avoidable factors in deaths from stroke and hypertensive disease. British Medical Journal 1993;307:1027-1030

(47) Buck C, Bull S. Preventable causes of death versus infant mortality as an indicator of the quality of health services. International Journal of Health Services 1986;16:553-62

(48) Rutstein DD, Berenberg W, Chalmers TC, Child CG, Fishman AP, Perrin EB. Measuring the quality of medical care. A clinical method. New England Journal of Medicine 1976;294:582-8

(49) Rutstein DD, Berenberg W, Chalmers TC, Fishman AP, Perrin EB, Zuidema GD. Measuring the quality of medical care: second revision of tables and indexes. New England Journal of Medicine 1980; (May 15):1146

(50) Heineken PA, Charles G, Stimson DH, Wenell C, Stimson R. The use of negative indexes of health to evaluate quality of care in a primary-care group practice. Medical Care 1985; 23:198-208

(51) Woolhandler S, Himmelstein DU, Silber R, Bader M, Harnly M, Jones AA. Medical care and mortality: racial differences in preventable deaths. International Journal of Health Services 1985;15:1-22

(52) Carr W, Szapiro N, Heisler T, Krasner MI. Sentinel health events as indicators of unmet needs. Social Science and Medicine 1989;29(6):705-14

(53) Charlton JRH, Hartley RM, Silver R, Holland WW. Geographical variations in mortality from conditions amenable to medical intervention in England and Wales. Lancet 1983; March 26 Vol (I):691-6

(54) Bauer RL, Charlton JRH. Area variation in mortality from diseases amenable to medical intervention: The contribution of differences in morbidity. International Journal of Epidemiology 1986;15(3):408-12

(55) Charlton J, Lakhani A, Aristidou M. How have 'avoidable death' indices for England and Wales changed? 1974-78 compared with 1979-83. Community Medicine 1986;8:304-14.

(56) Carr Hill RA, Hardman GF, Russell IT. Variations in avoidable mortality and variations in health care resources. Lancet 1987; April 4:789-92

(57) Martini CJM, Allan GJB, Davison J, Backett EM. Health indexes sensitive to medical care variation. International Journal of Health Services 1977;7(2):294-309

(58) Mackenbach JP, Bouvier-Colle MH, Jougla E. "Avoidable" mortality and health services : A review of aggregate data studies. Journal of Epidemiology and Community Health 1990;44:106-11

(59) Holland WW, Fitzgerald AP, Hildrey SJ, Phillips SJ. Heaven can wait. Journal of Public Health Medicine 1994;16:321-330

(60) Chambers M, Clarke A. Measuring readmission rates. British Medical Journal 1990;301:1134-6

(61) Milne R, Clarke A. Can readmission rates be used as an outcome indicator? British Medical Journal 1990;301:1139-40

(62) Cox A, Simpson WAC. The use of readmission rates as a measure of clinical outcome following cataract surgery. Health Bulletin 1998;56:799-802,

(63) Leng GC, Walsh D, Fowkes FGR, Swainson CP. Is the emergency readmission rate a valid outcome indicator? Quality in Health Care 1999;8:234-38

(64) Weissman JS, Gatsonis C, Epstein AM. rates of avoidable hospitalisation by insurance status in Massachusetts and Maryland. Journal of the American Medical Association 1992;268:2388-94

(65) Solberg LI, Peteron KE, Ellis RW, Romness K, Rohrenbach E, Thell T, Smith A, Routier A, Stillmank MW, Zak S. The Minnesota project: A focused approach to ambulatory quality assessment. Inquiry 1990;27:359-67

(66) Parchman ML, Culler S. Primary care physicians and avoidable hospitalizations. Journal of Family Practice 1994; 39 (2):123-128

(67) Begley CE, Slater CH, Engel M, Reynolds TF. Avoidable hospitalizations and socioeconomic status in Galveston County, Texas. Journal of Community Health 1994;19(5):377-87

(68) Ricketts TC, Randolph R, Howard HA, Pathman D, Carey T. Hospitalization rates as indicator of access to primary care. Health and Place 2001;7:27-38

(69) Guiffrida A, Gravelle H, Roland M. Performance indicators for primary care management in the NHS. University of York, York Health Economics Consortium. Discussion Paper 160, 1998.

(70) Williams DRR. Outcome indicators for diabetes services – what do we have and what do we need? Community Medicine 1989;11:57-64

(71) NHS Executive. Quality and performance in the NHS: High level performance indicators. NHS Executive, Quarry House, 2000

(72) Paul-Shaheen P, Clarke JD, Williams D. Small area analysis: A review of the North American literature. Journal of Health Politics, Policy and Law 1987;12:741-809

(73) Sanders D, Coulter A, McPherson K. Variations in hospital admission rates: A review of the literature. King's Fund Project Paper No 79. King's Fund, London 1989

(74) Mechanic D. Correlates of physician utilization: Why do major multivariate studies of physician utilization find trivial psychosocial and organizational effects? Journal of Health and Social Behavior 1979;20:387-96

(75) Living in Britain: Results from the 1998 General Household Survey: A survey carried out by the Social Survey Division. Social Survey Division., Office for National Statistics, Great Britain.

(76) Nazroo JY. The Health of Britain's Ethnic Minorities. Policy Studies Institute. London, 1997

(77) Roemer MI. Bed supply and hospitalization: a natural experiment. Journal of the American Hospital Association 1961;35:36-42

(78) Carr Hill RA, Hardman G, Martin S, Peacock S, Sheldon T, Smith P. A formula for distributing NHS revenues based on small area use of hospital beds. The University of York, Centre for Health Economics, 1994.

(79) Mechanic D. Social research in health and the American sociopolitical context: The changing fortunes of medical sociology. Social Science and Medicine 1993;36:95-102

(80) Scott A, Shiell A, King M. Is general practitioner decision-making associated with patient socio-economic status? Social Science and Medicine 1996;42:35-46

(81) Newton J. Factors influencing GPs' referral decisions. Family Practice 1991;8:308-13

(82) Carr Hill R, Rice T, Roland M. Socioeconomic determinants of rates of consultation in general practice based on fourth national morbidity survey of general practices. British Medical Journal 1996;312:1008-13

(83) Davis K. Inequality and access to health care. The Millbank Quarterly 1991;69:253-73

(84) Newacheck PW. Access to ambulatory care for poor persons. Health Services Research 1988;23:401-19

(85) Muller C. Review of 20 years of research on medical care utilization. Health Services Research 1986;21:129-44

(86) Stoddard JJ, St Peter, RF, Newacheck PW. Health insurance status and ambulatory care for children. New England Journal of Medicine 1994;330:1421-1425

(87) Gornick ME, Eggers PW, Reilly TW, Mentnech RM *et al.* Effects of race and income on mortality and use of services among Medicare beneficiaries. New England Journal of Medicine 1996;335:791-9

(88) Marquis MS, Long SH. Reconsidering the effect of Medicaid on health care services use. Health Services Research 1996;30 (6):791-808

(89) Lieu TA, Newacheck PW, McManus MA. Race, ethnicity and access to ambulatory care among US adolescents. American Journal of Public Health 1993;83:960-5

(90) Eddy DM. Clinical decision-making: anatomy of a decision. Journal of the American Medical Association 1990;263:441-43

(91) Weiner JP, Starfield BH, Powe NR, Stuart ME, Steinwachs DM. Ambulatory care practice variation within a Medicaid program. Health Services Research 1996;30:751-70.

(92) O'Donnell O, Propper C. Equity and the distribution of UK National Health Service resources Journal of Health Economics 1991;10:1-19

(93) O'Donnell O, Propper C. Equity and the distribution of UK National Health Service resources. Journal of Health Economics 1991;10:247-49

(94) Wagstaff A, van Doorslaer E, Paci P. On the measurement of horizontal equity in the delivery of health care. Journal of Health Economics 1991;10:169-205

(95) Wagstaff A, van Doorslaer E, Paci P. Horizontal equity in the delivery of health care. Journal of Health Economics 1991;10:251-6

(96) Le Grand J. The distribution of health care revisited: A commentary on Wagstaff, van Doorslaer and Paci, and O'Donnell and Propper. Journal of Health Economics 1991;10:239-45

(97) Collins E, Klein R. Equity and the NHS: self-reported morbidity, access and primary care. British Medical Journal 1980;281:1111-1115

(98) Evandrou M, Falkingham J, Le Grand J, Winter D. Equity in health and social care. Journal of Social Policy 1992;21:489-23

(99) Carr Hill RA, Hardman G, Martin S, Peacock S, Sheldon T, Smith P. A formula for distributing NHS revenues based on small area use of hospital beds. The University of York, Centre for Health Economics, 1994.

(100) Reid FDA, Cook DG, Majeed A. Explaining variation in hospital admission rates between general practices: cross sectional study. British Medical Journal 1999;319:98-103 (101) Carr Hill RA, Place M, Posnett J. Access and the utilisation of healthcare services. In: Ferguson B, Sheldon T, Posnett J, editors. Concentration and choice in healthcare. London: FT Healthcare, 1997 pp 37-49.

(102) The Black Report. Ed: Townsend P, Davidson N. In: Inequalities in Health. Penguin, London 1988

(103) Whitehead M. the health divide. In: Inequalities in health. Penguin, London, 1988

(104) Ben Shlomo Y, Chaturvedi N. Assessing equity in access to health care provision in the UK; does where you live affect your chances of getting a coronary artery bypass graft? Journal of Epidemiology and Community Health 1995;49:200-4

(105) Balarajan R, Yuen P, Machin D. Socioeconomic differentials in the uptake of medical care in Great Britain. Journal of Epidemiology and Community Health 1987;41:196-9

(106) Carlisle RD, Johnstone SP, Pearson JCG. Relation between night visit rates and deprivation measures in one general practice. British Medical Journal 1993;306:1383-5

(107) Baker D, Hann M. General practitioner services in primary care groups in England: is there inequity between service availability and population need? Health and Place 2001; 7: 67-74

(108) Health and lifestyles. Black and ethnic minority ethnic groups in England. Health Education Authority, London, 1994.

(109) Saxena S, Majeed A, Jones M. Socioeconomic differences in childhood consultation rates in general practice in England and Wales: Prospective cohort study. British Medical Journal 1999;318:642-6

(110) Worrall A, Rea JN, Ben Shlomo Y. Counting the cost of social disadvantage in primary care: Retrospective analysis of patient data. British Medical Journal 1997;314:38-42

(111) Price CE, Paul EA, Bevan RG, Holland WW. Equity and medical practice variation: relationships between standardised discharge ratios in total and for selected conditions in English districts. Journal of Epidemiology and Community Health 1992;46:58-62

(112) Wennberg JE, Barnes BA, Zubkoff M. Professional uncertainty and the problem of supplier-induced demand. Social Science and Medicine 1982;16:811-24

(113) Mort EA, Weissman JS, Epstein AM. Physician discretion and racial variation in the use of surgical procedures. Archives of Internal Medicine 1994; 154 (7): 761-767

(114) Peterson ED, Shaw LK, DeLong ER, Pryor DB *et al.* Racial variation in the use of coronary revascularisation procedures. New England Journal of Medicine 1997;336:480-6

(115) Gittlesohn AM, Halpern J, Sanchez RL. Income, race and surgery in Maryland. American Journal of Public Health 1991;81:1435-1441

(116) Gittlesohn A, Powe N. Small area variations in health care delivery in Maryland. Health Services Research 1995;30:295-317

(117) Durkin MS, Davidson LL, Kuhn L, O'Connor P, Barlow B. Low-income neighborhoods and the risk of severe pediatric injury: A small-area analysis in Northern Manhattan. American Journal of Public Health 1994;.84(4):587-592 (118) Perrin JM, Homer CJ, Berwick DM, Woolf AD, Freeman JL, Wennberg JE. Variations in rates of hospitalization of children in three urban communities. New England Journal of Medicine 1989;320:1183-7

(119) Gottlieb DJ, Beiser AS, O'Connor GT. Poverty, race and medication use are correlates of asthma hospitalization rates. A small area analysis in Boston. Chest 1995;108:28-35

(120) McConnochie KM, Roghmann KJ, Liptak GS. Hospitalization for lower respiratory tract illness in infants: Variation in rates among counties in New York State and areas in Monroe County. Journal of Pediatrics 1995;126:220-9

(121) Blumberg MS. Regional differences in hospital use standardised by reported morbidity. Medical Care 1982;20:931-44

(122) Carr W, Szapiro N, Heisler T, Krasner MI. Sentinel health events as indicators of unmet needs. Social Science and Medicine 1989;29(6):705-14

(123) Komaromy M, Lurie N, Osmond D, Vranzian K, Keane D, Bindman AB. Physician practice style and rates of hospitalization for chronic medical conditions, Medical Care 1996; 34(6):594-609

(124) Coulter A, McPherson K. Socioeconomic variations in the use of common surgical operations British Medical Journal 1985;291: 183-7

(125) McCormick A, Fleming D, Charlton J. Morbidity statistics from general practice: Fourth National Study 1991-2. London: HMSO, 1995.

(126) Skobeloff PM, Spivey WH, St Clair SS, Schoffstall JM. The influence of age and sex on asthma admissions. Journal of the American Medical Association 1992;268:3437-3440

(127) Strachan DP, Butland B, Anderson HR. Incidence and prognosis of asthma and wheezing illness from early childhood to age 33 in a national British cohort. British Medical Journal 1996;312:1195-9

(128) McNiece R, Majeed A. Socioeconomic differences in general practice consultation rates in patients aged 65 and over: prospective cohort study. British Medical Journal 1999;319:26-8

(129) Raine R. Does gender bias exist in the use of specialist health care? Journal of Health Services Research and Policy 2000;5:237-49

(130) Propper C. Who pays for and who gets health care: equity in the finance and delivery of health care in the UK. Nuffield Provincial hospitals Trust, Occasional Paper, 1997.

(131) Hippisley-Cox J, Pringle M. Inequalities in access to coronary angiography and revascularisation: The association of deprivation and location of primary care services. British Journal of General Practice 2000;50:449-54.

(132) Johnstone F, Whitehead M.. Deprivation and health in North Cheshire : An equity audit of health services. University of Liverpool, Department of Public Health: Liverpool Public Health Observatory, 1996.

(133) Manson-Siddle CJ, Robinson MB. Super Profile analysis of socioeconommic variations in coronary investigation and revascularisation rates. Journal of Epidemiology and Community Health 1998;52:507-12 (134) Kee F, Gaffney B, Canavan C, Little J et al. Expanding access to coronary artery bypass surgery: who stands to gain? British Heart Journal 1995;73:129-33

(135) Buckingham K, Freeman P. Socioeconomic and morbidity indicators of need in relation to the use of community health services: An observational study. British Medical Journal 1997;315: 994-6

(136) Milner PC, Nicholl JP, Williams BT. Variation in demand for accident and emergency departments in England from 1974 to 1985. Journal of Epidemiology and Community Health 1988;42:274-8

(137) Lynch M. Effect of practice and patient population characteristics on the uptake of childhood immunisations. British Journal of General Practice 1995;45:205-08

(138) Majeed A, Cook DG, Anderson H, Hilton S, *et al.* Using patient and general practice characteristics to explain variations in cervical smear uptake rates. British Medical Journal 1994;308:1272-76

(139) Fleming DM, Charlton JRH. Morbidity and healthcare utilisation of children in households with one adult: comparative observational study. British Medical Journal 1998;316:1572-6

(140) Tackling inequalities in health: An agenda for action. Benzeval M, Judge K, Whitehead M (Eds). Kings Fund. London, 1995

(141) Which patients consult?: A multivariate analysis of socio-economic factors. MSGP-4 1991-92 Office for National Statistics, London, 1996

(142) Melnyk KAM. Barriers: A critical review of the literature. Nursing Research 1988;37:196-201

(143) Billings J, Teicholz N. Uninsured patients in District of Columbia hospitals. Health Affairs 1990;9:158-165

(144) The Medicaid Access Study Group. Access of Medicaid recipients to outpatient care. New England Journal of Medicine 1994;330:1426-30

(145) Davidson AE, Klein DE, Settipane GA, Alario AJ. Access to care among children visiting the emergency room with acute exacerbations of asthma. Annals of Allergy 1994; 72 (5):469-473

(146) Grumbach K, Keane D, Bindman A. Primary care and public emergency department overcrowding. American Journal of Public Health 1993;83:372-8

(147) Stern RS, Weissman JS, Epstein AM. The emergency department as a pathway to admission for poor and high-cost patients. Journal of the American Medical Association 1991;266:2238-43

(148) Lambrew JM, DeFriese GH, Carey TS, Ricketts TC, Biddle AK. The effects of having a regular doctor in access to primary care. Medical Care 1996;34(2):138-51

(149) Kuder JM, Vilmain JA, Demlo LK. Exploring physician responses to patients' extramedical characteristics. Medical Care 1987;25:882:93

(150) Mushlin AI, Appel FA. Extramedical factors in the decision to hospitalise medical patients. American Journal of Public Health 1976;66:170-2

(151) Gardner KG, Chapple A. barriers to referral for patients with angina: qualitative study. British Medical Journal 1999;319:418-21

(152) Burkey Y. Barriers to cervical screening amongst women who have never had a smear. Grimsby & Scunthorpe Health Authority, Kirton-in-Lindsey, United Health, 1993

(153) Bolden KJ. Inner cities. Occasional Paper 19. Royal College of General Practitioners, London, 1981

(154) Lorentzon M, Jarman B, Bajekal M. Report of the Inner City Task Force of the Royal College of General Practitioners. London: RCGP, 1994.

(155) National Survey of NHS Patients: General Practice 1998. NHS Executive, October 1999.

(156) Drummond N, McConnachie A, O'Donnell C, Moffat KJ et al. Social variation in reasons for contacting general practice out-of-hours: implications for daytime service provision? British Journal of General Practice 2000;50:460-64

(157) Martin E, Russell D, Goodwin S, Chapman R et al. Why patients consult and what happens when they do. British Medical Journal 1991;303:289-92

(158) Sweeney KG, Gray DP. Patients who do not receive continuity of care from their general practice – are they a vulnerable group? British Journal of General Practice 1995;45:133-5

(159) Coulter A, Peto V, Doll H. Patients' preferences and general practitioners' decisions in the treatment of menstrual disorders. Family Practice 1994;11:67-74.

(160) Armstrong D, Fry J, Armstrong P. Doctors' perceptions of pressure from patients for referral. British Medical Journal 1991;302:1186-8

(161) Lissauer T, Richman S, Tempia M, Jenkins S et al. Influence of homelessness on acute admissions to hospital. Archives of Diseases of Childhood 1993;69:423-9

(162) Department of Health. Public Information Strategy: Improving Information on the Health of Ethnic Groups. Department of Health: London, 1993.

(163) Modhood T, Berthoud R, Lakey J, Nasroo J, Smith P, Virdee S, Beishon S. Diversity and disadvantage: ethnic minorities in Britain. The 4<sup>th</sup> National Survey of Ethnic Minorities. Policy Studies Institute, London 1997.

(164) Smaje C, Le Grand J. Ethnicity, equity and the use of health services in the British NHS. Social Science and Medicine 1997;45:485-96

(165) Access to health care for ethnic minorities: A systematic review of the literature. Community Health Sciences Group 2001. Submitted to NHS Executive London Regional Office R&D Directorate. Unpublished

(166) Shaukat N, de Bono DP, Cruickshank JK. Clinical features, risk factors, and referral delay in British patients of Indian and European origin with angina matched for age and extent of coronary atheroma. British Medical Journal 1993;307:717-8
(167) Lear JT, Lawrence IG, Burden AC Pohl JEF. A comparison of stress test referral rates and outcome between Asians and Europeans. Journal of the Royal Society of Medicine 1994; 87: 661-2

(168) Smaje C. Health Race and ethnicity: making sense of the evidence. King's Fund Institute, London 1995

(169) Mason WB, Bedwell CL, Vander Zwaag R, Runyan JW. Why poor people are hospitalized. A description of preventable factors leading to admission for medical illness. Medical Care 1980;18(2):147-63

(170) Bigby J, Dunn J, Barclay Adams J, Jen P, Landefeld CS, Komaroff AL. Assessing the preventability of emergency hospital admissions. A method for evaluating the quality of medical care in a primary care facility. The American Journal of Medicine 1987;83:1031

(171) Frankl SE, Breeling JL, Goldman L. Preventability of emergent hospital readmission. American Journal of Medicine 1991;90:667-74

(172) McCaul KD, Glasgow RE, Schafer LC. Diabetes regimen behaviors. Predicting adherence. Medical Care 1987;25:868-91

(173) Tedeschi PJ, Wolfe RA, Griffith JR. Micro-area variation in hospital use. Health Services Research 1990;24:729-39

(174) Wennberg J, Gittlesohn A. Variations in Medical Care among small areas. Scientific American 1982;246:100-11

(175) Slack R, Fergusson B, Ryder S. Analysis of hospitalisation rates by electoral ward; Relationship to accessibility and deprivation data. Health Services Management 1997;10:24-31

(176) Wennberg JE, Freeman JL, Culp WJ. Are hospital services rationed in New Haven or over-utilised in Boston? Lancet 1987;ii:1185-88

(177) Black N, Langham S, Pettigrew M. Coronary revascularisation: why do rates vary geographically in the UK? Journal of Epidemiology and Community Health 1995;49:408-12

(178) Roos NP, Flowerdew G, Wadja A, tate RB. Variations in hospitalization practices: A population-based study in Manitoba, Canada. American Journal of Public Health 1986;76:45-51

(179) Fisher ES, Wennberg JE, Stukel TA, Sharp SM. Hospital readmission rates for cohorts of Medicare beneficiaries in Boston and New Haven. New England Journal of Medicine 1994;331:989-95

(180) Wissow LS, Gittlesohn A, Szklo M, Starfield B, Mussman M. Poverty, race and hospitalization for childhood asthma. American Journal of Public Health 1988;78:777-82

(181) Brown LJ, Barnett JR. Influence of bed supply and health care organization on regional and local patterns of diabetes related hospitalization. Social Science and Medicine 1992;35:1157-90

(182) Dalziel M, Garrett C. Intraregional variations in the treatment of end stage renal failure. British Medical Journal 1987:294:382-3 (183) Deprez RD, Pennell B, Libby MA. The substitutability of outpatient primary care in rural community health centers for inpatient hospital care. Health Services Research 1987;2(2):207-33

(184) Welch WP, Miller ME, Welch HG, Fisher ES, Wennberg JE. Geographic variation in expenditures for physician's services in the United States. New England Journal of Medicine 1993;328:621-7

(185) Rosenberg MW, Hanlon NT. Access and utilization: a continuum of health service environments. Social Science and Medicine 1996;43:975-83

(186) Krakauer H, Jacoby I, Millman M, Lukomnik JE. Physician impact on hospital admission and on mortality rates in the Medicare population. Health Services Research 1996;31:191-211.

(187) Jarman B, Gault S, Alves B, Hider A et al. Explaining differences in English hospital death rates using routinely collected data. British Medical Journal 1999;318:1515-20

(188) Jones Elwyn G, Stott NCH. Avoidable referrals? Analysis of 170 consecutive referrals to secondary care. British Medical Journal 1994;309:576-578

(189) Coast J, Inglis A, Morgan K, Gray S *et al.* The hospital admissions study in England: Are there alternatives to emergency hospital admission? Journal of Epidemiology and Community Health 1995;49:194-99.

(190) Watson JP, Cowen P, Lewis RA. The relationship between asthma admission rates, routes of admission, and socio-economic deprivation. European Respiratory Journal 1996;6:2087-93

(191) Gribben B. Do access factors affect utilisation of general practice services in South Auckland? New Zealand Medical Journal 1992;105:453-5

(192) Parkin D. Distance as an influence on demand in general practice. Journal of Epidemiology and Community Health 1979;33:96-99

(193) Haynes RM, Bentham CG. The effects of accessibility on general practitioner consultations, outpatient attendances and in-patient admissions in Norfolk, England. Social Science and Medicine 1982;16:561-9

(194) Watt IS, Franks AJ, Sheldon TA. Rural health and health care. British Medical Journal 1993;306:1358-9

(195) Piette JD, Moos RH. The influence of distance on ambulatory care use, death and readmission following an acute myocardial infarction. Health Services Research 1996;31:573-91

(196) Tudor Hart J. The inverse care law. The Lancet 1971:i:405-12

(197) Benzeval M, Judge K. Access to health care in England: Continuing inequalities in the distribution of GPs. Journal of Public Health Medicine 1996;18:33-40

(198) Gravelle H, Sutton M. Inequality in the geographical distribution of general practitioners in E&W 1974-95. Journal of Health Services Research and Policy 2001;6:6-13

(199) Manning WG, Leibowitz A, Goldberg GA, Rogers WH, Newhouse JP. A controlled trial on the effect of a prepaid group practice on use of services. New England Journal of Medicine 1984;310:1505-10.

(200) Hadley J, Steinberg EP, Feder J. Comparison of uninsured and privately insured patients: Condition on admission, resource use and outcome. Journal of the American Medical Association 1991;265:374-79

(201) Legnini MW, Thompson JD, DerSimonian R, Lynch JT. Changes in Connecticut hospital use rates: have small area variations been affected? Inquiry 1990;27:51-60

(202) Dyck FJ, Murphy FA, Murphy JK, Road D et al. Effect or surveillance on the number of hysterectomies in the province of Saskatchewan. New England Journal of Medicine 1977;296:1326-8

(203) Zeitel LV, Goodwin EJ, Fox KS. Access to medical care. Monitoring changes in hospitalization among elderly New York city residents. Medical Care 1991;29:287-96

(204) Anderson GM, Grombach K, Luft HS, Roos LL, Mustard C, Brook R. Use of coronary artery bypass surgery in the United States and Canada. Influence of age and income. Journal of the American Medical Association 1993;269:1661-66

(205) Briggs LW, Rohrer JE, Ludke RL, Hilsenrath PE, Phillips KT. Geographic variation in primary care visits in Iowa. Health Services Research 1995;30:657-71

(206) Newhouse JP, Manning WG, Morris CN, Orr LL *et al.* Some interim results from a controlled trial of cost sharing in health insurance. New England Journal of Medicine 1981;305:1501-7

(207) Selby JV, Fireman BH, Swain BE. Effect of co-payment on use of the emergency department in a health maintenance organization. New England Journal of Medicine 1996;334:657-8

(208) Ware JE Jr, Bayliss MS, Rogers WH, Kosinski M, Tarlov AR. Differences in 4-year health outcomes for elderly and poor and chronically ill patients treated in HMO and fee-forservice systems. Results from the Medical Outcomes Study. Journal of the American Medical Association 1996;276;1952-3

(209) Le Grand, Mays N, Mulligan J. Learning from the Internal Market: a review of the evidence. King's Fund, 1998.

(210) Dixon J, Glennerster H. What do we know about fundholding in general practice? British Medical Journal:1995;311;727-30

(211) Gillam SJ. Provision of health promotion clinics in relation to population need: another example of the inverse care law? British Journal of General Practice 1992;42:54-6

(212) Lovejoy FH, Carper JM, Janeway CA, Kosa J. Unnecessary and preventable hospitalizations. Report on an internal audit. Medical Care 1971;79(5):868-72

(213) Conway SP, Newport MJ. Are all hospital admissions for acute gastroenteritis necessary? Journal of Infection 1994;29:5-8

(214) Kreger BE, Restuccia JD. Assessing the need to hospitalize children: Pediatric appropriateness evaluation protocol. Pediatrics 1989;84:242-7

(215) Soulen JL, Duggan, AK, DeAngelis CD. Identification of potentially avoidable pediatric hospital use: Admitting physician judgment as a complement to utilization review. Pediatrics 1994;94:421-424

(216) Campbell SM, Roland MO, Quayle JA, Buetow SA, Shekelle PG. Quality indicators for general practice: which ones can general practitioners and health authority managers agree are important and how useful are they? Journal of Public Health Medicine 1998;20:414-21

(217) Campbell SM, Roland MO, Buetow SA. Defining quality of care. Social Science and Medicine 2000;51:1611-1625

(218) Padgett DK, Brodsky B. Psychosocial factors influencing non-urgent use of the emergency room: A review of the literature and recommendations for research and improved service delivery. Social Science and Medicine 1992;35(9):1189-97

(219) Young GP, Wagner MB, Kellermann AL, Ellis JE, Bouley D. Ambulatory visits to hospital emergency departments: Patterns and reasons for use. Journal of the American Medical Association 1996;276:460-65

(220) Baker DW, Stevens CD, Brook, RH. Regular source of ambulatory care and medical care utilization by patients presenting to a public hospital emergency department. Journal of the American Medical Association 1994; 271 (24):1909-1912

(221) Bedford HE, Jenkins SM, Shore C, Kenny PA. Use of an East End children's accident and emergency department for infants: a failure of primary health care? Quality in Health Care 1992;1:29-33

(222) Bowling A, Isaacs D, Arnston J, Roberts JE *et al.* Patient use of a paediatric hospital casualty department in the East End of London. Family Practice 1987;4:85-90

(223) Davis T. Accident department or general practice? British Medical Journal 1988;292:241-3

(224) Singh S. Self referral to accident and emergency department: Patients' perceptions. British Medical Journal 1988; 297: 1179-1180

(225) Davison AG, Hildrey AC, Floyer MA. Use and misuse of an accident and emergency department in the East End of London. Journal of the Royal Society of Medicine 1983;76:37-40

(226) McKee CM, Gleadhill DNS, Watson JD. A&E attendance rates: Variation among patients from different general practitioners. British Journal of General Practice 1990;40:150-3

(227) Campbell JL. General practitioner appointment systems. patient satisfaction, and use of accident and emergency services - a study in one geographical area. Family Practice 1994;11:438-45

(228) Hershey JC, Luft HS, Glanvis JM. Making sense of utilization data. Medical Care 1975;XIII:838-54

(229) Freeman GK, Richards SC. How much personal care in four group practices? British Medical Journal 1990;301:1028-30

(230) Roland M, Mayor V, Morris R. Factors associated with achieving continuity of care in general practice. Journal of the Royal College of General Practitioners 1986;36:102-4

(231) Ettlinger PRA, Freeman GK. GP compliance study: is it worth being a personal doctor? British Medical Journal 1981;282:1192-95

(232) Hjortdahl P. Continuity of care: general practitioners' knowledge about, and sense of responsibility toward their patients. Family Practice 1992;9:3-8

(233) Lahdensuo A, Haahtela T, Herrala J, Kava T, Kiviranta K et al. Randomised comparison of guided self management and treatment of asthma over one year. British Medical Journal 1996;312:748-52

(234) Connett GJ, Warde C, Wooler E, Lenney W. Audit strategies to reduce hospital admissions for childhood asthma. Archives of Diseases in Childhood 1993;69(2):202-5

(235) Osman LM, Abdalla MI, Beattie J, Ross SJ, Russell IT, Friend JA, Legge JS, Douglas JG (on behalf of the Grampian Asthma Study of Integrated care). Reducing hospital admissions through computer supported education for asthma patients. British Medical Journal 1994; 308:568-71

(236) Bendefy IM. Home nebulisers in childhood asthma: survey of hospital supervised use. British Medical Journal 1991; 302:1180-1

(237) Bryce FP, Neville RG, Crombie IK, Clarke RA, McKenzie P. Controlled trial of an audit facilitator in diagnosis and treatment of childhood asthma in general practice. British Medical Journal 1995;310:838-42

(238) Hughes DM, McLeod M, Garner B, Goldblum RM. Controlled trial of a home and ambulatory program for asthmatic children. Pediatrics 1991;87:54-61

(239) Griffiths C, Sturdy P, Naish J, Omar R, Dolan S, Feder G. Hospital admissions for asthma in east London: associations with characteristics of local general practices, prescribing, and population. British Medical Journal 1997;314:481-2

(240) Grampian Asthma Study of Integrated Care. Integrated care for asthma: a clinical, social, and economic evaluation. British Medical Journal 1994;308:559-64

(241) Mayo PH, Richman J, Harris HW. Results of a programme to reduce admissions for adult asthma. Annals of Internal Medicine 1990;112:864-71

(242) Farmer A, Coulter A. Organisation of care for diabetic patients in general practice: influence on hospital admissions. British Journal of General Practice 1990;40:56-8

(243) Swift PGF, Hearnshaw JR, Botha JL, Wright G, Raymond NT, Jamieson KF. A decade of diabetes: Keeping children out of hospital. British Medical Journal 1993;307:96-8

(244) Neville RG, Hoskins G, Smith B Clark RC. Observations on the structure, process and clinical outcomes of asthma care in general practice. British Journal of General Practice 1996;46:96-8

(245) Blainey D, Lomas D, Beale A, partridge M. The cost of acute ashtma – how much is preventable? Health Trends 1990/1;4:151-3

(246) Pringle M, Stewart Evans C, Coupland C, Williams I *et al.* Influences on control in diabetes: patient, doctor, practice or delivery of care? British Medical Journal 1993;306:630-4

(247) Eastwood AJ, Sheldon TS. Organisation of asthma care: what difference does it make? A systematic review of the literature. Quality in Health Care 1996;5:134-43.

(248) Connell FA, Day RW, LoGerfo JP. Hospitalization of Medicaid children: analysis of small-area variations in admission rates. American Journal of Public Health 1981;71:606-13

(249) Stano M. Further issues in small area variations' analysis. Journal of Health Politics, Policy and Law 1991;16:573-88

(250) Eisenberg JM. Sociologic influences on decision-making by clinicians. Annals of Internal Medicine 1979;90:957-64

(251) Roos NP. Hospitalization style of physicians in Manitoba: The disturbing lack of logic in medical practice. Health Services Research 1992;27:361-384

(252) Roos NP. Predicting hospital utilization by the elderly: the importance of patient, physician and hospital characteristics. Medical Care 1989;27:905-19

(253) Connell FA, Blide LA, Hanken MA. Clinical correlates of small area variation in population-based admission rates for diabetes. Medical Care 1984;24:939

(254) Coulter A, Peto V, Doll H. Influence of sex of general practitioner on the management of menorrhagia. British Journal of General Practice 1995;45:471-5

(255) Majeed FA, Cook DG, Anderson HR, Hilton S *et al.* Using patient and general practitioner characteristics to explain variations in cervical smear uptake rates. British Medical Journal 1994;308:1272-6

(256) Mort EA, Edwards JN, Emmons DW, Convery K at al. Physician response to patient insurance status in ambulatory care clinical decision-making. Implications for quality of care. Medical Care 1996;34:783-97

(257) Coulter A, Seagroatt V, McPherson K. Relation between general practices' outpatient referral rates and rates of elective admission to hospital. British Medical Journal 1990;301:273-75

(258) Diehr P, Cain P, Connell F, Volinn E. What is too much variation? The null hypothesis in small area analysis. Health Services Research 1990;24:741-71

(259) Diehr P, Cain KC, Kreuter W, Rosenkrantz S. Can small-area analysis detect variation in surgery rates? The power of small-area variation analysis. Medical Care 1992;30:484-502

(260) Elliott P, Martuzzi M, Shaddick G. Spatial statistical methods in environmental epidemiology: A critique. Statistical Methods in Medical Research 1995;4:137-159

(261) Kirkwood BR. Essentials of medical statistics. Blackwell Scientific Publications. Oxford 1990.

(262) The Dartmouth Atlas of Health Care. The Center for Evaluative Clinical Sciences. Dartmouth Medical School. American Hospital Publishing, Inc. Chicago, Illinois, 1996.

(263) Stocking B. First consensus development conference in United Kingdom: On coronary artery bypass grafting. British Medical Journal 1985;291:713-5.

(264) McGlynn EA, Kosekoff J, Brook RH. Format and conduct of consensus conferences. International Journal of Technology and Access to Health Care 1990:6;450-69.

(265) Ziglio E. The Delphi method and its contribution to decision-making. In: Gazing into the Oracle. The Delphi Method and its Application to Social Policy and Public Health. Adler M, Ziglio E (Eds). Jessica Kingsley Publishers, London 1996.

(266) Pill J. The Delphi method: Substance, content, a critique and an annotated bibliography. Socioeconomic Planning Science 1971;5:57-71

(267) Delbecq AL et al. Group techniques for program planning: a guide to nominal group and Delhi processes. Scott, Foresman and Co., Glennview, Illinois, 1975.

(268) Manual of the international classification of diseases, injuries and causes of death. Based on the recommendations of the ninth revision conference, 1975. 1975 (ninth) revision. Volume 1. Geneva: World Health Organisation 1977.

(269) Armitage P, Berry G. Statistical methods for medical research, 3<sup>rd</sup> edition. Oxford: Blackwell Scientific, 1971.

(270) Lomas J, Anderson G, Enkin M, Vayda E, Roberts R, MacKinnon B. The role of evidence in the consensus process. Journal of the American Medical Association 1988;259:3001-3005.

(271) Jacoby I. Evidence and consensus. Journal of the American Medical Association 1988;259:3039.

(272) McKee M, Priest M, Ginzler M, Black N. How representative are members of expert panels? Quality Assurance in Health Care 1991;3:89-94.

(273) Rettig BA. A randomised study of the effects of a home diabetes education program. Diabetes Care 1986;9:173-8

(274) Fishbein HA, Faich GA, Ellis SE. Incidence and hospitalisation patterns of insulin dependent diabetes mellitus. Diabetes Care 1982; 5:630-633

(275) Shekelle PG, Kahan JP, Bernstein SJ, Leape LL, Kamberg CJ, Parke RE. the reproducibility of a method to identify the overuse and underuse of medical procedures. New England Journal of Medicine 1998;338:1888-95

(276) Burnand B, Vader JP, Froehlich F, Dupriez K, Larequi-Lauber T, Pache I et al. Reliability of panel-based guidelines for colonoscopy: an international comparison. Gastrointestinal Endoscopy 1998;47:162-66

(277) Hospital Episode Statistics 1992/3. Department of Health 1994. London, HMSO.

(278) Barrie JL, Marsh DR. Quality of data in the Manchester Orthopaedic Database. British Medical Journal 1992;304:159-62

(279) George AM, Maddocks GB. Accuracy of diagnostic content of hospital activity analysis in infectious diseases. British Medical Journal 1979;1:1332-34

(280) Whates PP, Birzgalis AR, Irving M. Accuracy of hospital activity analysis operation codes. British Medical Journal 1982;284:1855-58

(281) Cleary R, Beard RW, Coles J, Devlin HB, Hopkins A Roberts S, Schumacher D, Wickings HI. Comparative hospital databases: Value for management and quality. Quality in Health Care 1994;3:3-10

(282) Yeoh C, Davis H. Clinical coding: Accuracy and completeness when doctors take it on. British Medical Journal 1993;306:972

(283) Walshe K, Harrison N, Renshaw M. Comparison of the quality of patient data collected by hospital and department computer systems. Health Trends 1993;25:105-8

(284) Campbell SE, Campbell MK, Grimshaw JM, Walker AE. A systematic review of discharge coding accuracy. Journal of Public Health Medicine 2001;23:205-11

(285) Cleary R. Beard RW, Coles J, Devlin HB, Hopkins A, Schumacher D, Wickings HI. The quality of routinely collected maternity data. British Journal of Obstetrics and Gynaecology 1994;101(12):1042-7

(286) Martini CJM, Hughes AO, Patton VA. A study of the validity of the Hospital Activity Analysis information. British Journal of Preventive Social Medicine 1976;30:180-6

(287) Office for Population Censuses and Surveys. Tabular list of the classification of surgical operations and procedures. Fourth revision. London: HMSO.

(288) Clinical Coding Toolbox. Information Management Group, NHS Executive, Loughborough, September 1991.

(289) National Clinical Coding Standards. Information Management Group, NHS Executive, Loughborough, June 1995

(290) Audit Commission. Setting the records straight: A study of hospital medical records. London: Audit Commission, 1995.

(291) Wyatt JC. Hospital information management: the need for clinical leadership. British Medical Journal 1995;311:175-80

(292) Williams B, Whatmough P, McGill N, Rushton L. Private funding of elective hospital treatment in England and Wales 1997-8: National survey. British Medical Journal 2000;320:904-5

(293) OPCS/GRO(s). 1991 Census General Report, Great Britain. HMSO, London, 1995.

(294) Population Statistics Division in collaboration with the Census Division at OPCS. How complete was the 1991 census? Population Trends 1993;71:22-25

(295) Hamilton LC. Statistics with STATA 5.Duxbury Press, Brooks/Cole Publishing Company, Pacific Grove, California 93950

(296) Goodwin N. The evidence on fundholding practices. In: Le Grand, Mays N, Mulligan J (Eds). Learning from the Internal Market: a review of the evidence. King's Fund, 1998.

(297) Majeed FA, Cook DG, Poloniecki J, Griffiths J, Stones C. Sociodemographic variables for general practices: Use of census data. British Medical Journal 1995;310:1373-4

# Original In Colour





#### Appendix B 1/8

### BARN HSA Frequency Std. Dev = .90 lean = -2.31 256 00 0 0 -3.00 -2.00 -1.00 -3.80 -2.60 -1.50 -50 0.00 60 LGAF2 EALI HSA 50 30 Frequency Std. Dev = .69 Mean = -1.28 N = 313.00 -2 28 -176 -126 -76 -28 28 .76 126 -2.00 -160 -1.00 -80 0.00 50 1.00 160 LGAF2 EDGW HSA . Frequency Std. Dev = .78 Mean = -1.48 N = 272.00 -4.50 10 -3.50 -2.50 -1.50 -.50 .50 1.50 -4.00 -3.00 -2.00 -1.00 0.00 1.00 LGAF2

Frequency distribution of values for LGAF2 by HSA, 1993/4 (Group B EDs)

#### Appendix B 2/8

Frequency distribution of values for LGAF2 by HSA, 1993/4 (Group B EDs)



## Frequency distribution of values for LGAF2 by HSA, 1993/4

#### (Group B EDs)





# Frequency distribution of values for LGAF2 by HSA, 1993/4 (Group B EDs)



Appendix B 5/8 Frequency distribution of values for LHAF2 by HSA, 1993/4 (Group B EDs)







Frequency distribution of values for LHAF2 by HSA, 1993/4 (Group B EDs)



Frequency distribution of values for LHAF2 by HSA, 1993/4

#### (Group B EDs)







#### Appendix B 8/8

Frequency distribution of values for LHAF2 by HSA, 1993/4 (Group B EDs)



Ratio of means\* of age-sex standardised admission rates by admission group, by year (\* means = mean of the rates in the fifth most deprived EDs/ mean of the rates in the fifth least deprived EDs)

Admission	Description	Ratio of means		
group		912	923	934
1	Strong and weak ACS conditions, any urgency, all	1.66	1.66	1.79
3	Strong and weak ACS conditions, any urgency, readmissions	2.00	1.88	2.16
5	Strong and weak ACS conditions, urgent, all admissions	1.92	1.95	2.02
7	Strong and weak ACS conditions, urgent, readmissions	2.44	2.10	2.44
9	Strong ACS conditions, any urgency, all admissions	1.64	1.74	2.16
10	Strong ACS conditions, any urgency, readmissions	2.34	2.53	2.17
11	Strong ACS conditions, urgent, all admissions	1.76	1.88	2.17
12	Strong ACS conditions, urgent, readmissions	2.28	2.07	2.84
13	Strong ACS conditions, very urgent, all admissions	1.69	1.81	2.13
14	Strong ACS conditions, very urgent, readmissions	1.89	1.95	1.92
15	Strong and weak markers, very	1.60	1.71	1.46
16	Strong markers, very urgent, all admissions	1.58	1.67	1.47
17	Admissions in surgical specialties	1.39	1.37	1.45
18	Admissions in medical specialties	1.68	1.64	1.95
19	Emergency admissions in surgical specialties	0.95	0.77	1.12
20	Emergency admissions in medical specialties	1.14	0.80	1.27
21	Admissions for ACS conditions defined by Billings	2.04	2.09	1.80
23	Admissions for marker conditions defined by Billings	1.47	1.41	1.19

Group A EDs

#### Appendix D 1/1

The incremental relative risk ratio (IRR)\* for key variables for admission group 1 (ACS), 1991/2

Variable	IRR*	P value	95% confidence interval
Carsq1	1.00		-
Carsq2	1.22	0.000	1.1674 - 1.2696
Carsq3	1.31	0.000	1.2535 - 1.3629
Carsq4	1.42	0.000	1.3598 - 1.4774
Carsq5	1.62	0.000	1.5523 - 1.6858
Smrq1	1.00		
Smrq2	1.04	0.030	1.0041 - 1.0843
Smrq3	1.07	0.000	1.0335 - 1.1151
Smrq4	1.12	0.000	1.0793 - 1.1645
Smrq5	1.22	0.000	1.1773 - 1.2706
Sirq1	1.00		-
Sirq2	1.05	0.011	1.0114 - 1.0935
Sirq3	1.08	0.000	1.0414 - 1.1252
Sirq4	1.13	0.000	1.0873 - 1.1746
Sirq5	1.2	0.000	1.1546 - 1.2481
Gaf2q2	1.00		-
Gaf2q2	1.04	0.030	1.0041 - 1.0843
Gaf2q3	1.10	0.000	1.0528 - 1.1419
Gaf2q4	1.06	0.004	1.0196 - 1.1089
Gaf2q5	1.08	0.001	1.0323 - 1.1252
Gaf2q2	1.00		-
Haf2q2	1.14	0.000	1.0891 - 1.1952
Haf2q3	1.18	0.000	1.1191 - 1.2320
Haf2q4	1.16	0.000	1.1129 - 1.2183
Haf2q5	1.14	0.000	1.0921 - 1.1902

Group B EDs

\* The IRRs are unadjusted for other variables in the model.

#### Appendix E 1/1

The value of 'pseudo r-squared' coefficients\* using full and simplified models

Admission	Model	Pseudo R-s	quared	
group		912	923	934
1	Full	0.1271	0.1282	0.1366
	Simplified	0.1254	0.1277	0.1354
3	Fuil	0.1216	0.126	0.1464
	Simplified	0.1194	0.1274	0.1454
5	Full	0.1560	0.1531	0.1599
	Simplified	0.1541	0.1522	0.1586
7	Full	0.1326	0.1366	0.1501
	Simplified	0.1304	0.1366	0.1489
9	Full	0.0548	0.0557	0.0573
	Simplified	0.0543	0.0553	0.0566
10	Full	0.1218	0.1425	0.1594
	Simplified	0.1199	0.1419	0.1588
11	Full	0.1003	0.0954	0.0948
	Simplified	0.0991	0.0942	0.0939
12	Full	0.1046	0.1016	0.1015
	Simplified	0.1014	0.0987	0.0999
13	Full	0.1327	0.1166	0.1144
	Simplified	0.1304	0.1162	0.1132
14	Full	0.1375	0.1377	0.135
	Simplified	0.1353	0.1374	0.1336
15	Full	0.0557	0.0557	0.0549
	Simplified	0.0550	0.0547	0.0547
16	Full	0.0531	0.0537	0.0518
	Simplified	0.0526	0.0526	0.0516
17	Full	0.1054	0.1196	0.1243
	Simplified	0.1035	0.1177	0.1232
18	Full	0.2150	0.2073	0.2334
	Simplified	0.2117	0.2056	0.232
19	Full	0.0920	0.1016	0.0654
	Simplified	0.0874	0.0986	0.0646
20	Full	0.2427	0.2317	0.2256
	Simplified	0.2382	0.2273	0.2236
21	Full	0.0980	0.1339	0.1379
	Simplified	0.0974	0.1331	0.1372
23	Full	0.1131	0.1116	0.0914
	Simplified	0.1102	0.1084	0.0911

• The pseudo r-squared values are adjusted using the variables representing: SMR; SIR; CARST; GAF2 and HAF2.

Appendix F 1/5

<b>\$12</b>							923						934					
Group	Access	In		Ρ	95% c.l	95% c.l	Group	Access	Irr	P	95% c.l	95% c.l	Group	Access	in	P	95% c.l	95% c.l
	variable				lower	upper		variable			Lower	upper		variable			lower	upper
1	2012023		1.0819	0.001	1.0335	1.1327	1	gəf2q23	1.0663	0.000	1.0329	1.1008	1	gar2q23	1.0424	0.078	0.9953	1.0917
	gaf2q45		1.0916	0.005	1.0264	1.1610		gaf2q45	1.0719	0.000	1.0308	1.1147		gaf2q45	1.0573	0.048	1.0004	1.1174
	haf2q23		1.1437	0.000	1.0866	1.2038		haf2q23	1.0559	0.001	1.0235	1.0894		haf2q23	1.0287	0.190	0.986	1.0732
	haf2q45		1.1343	0.000	1.0765	1.1952		haf2q45	1.0873	0.000	1.0526	1.1232		haf2q45	1.0764	0.001	1.0291	1.126
3	gaf2q23		1.0804	0.109	0.9830	1.1874	3	gaf2q23	1.0399	0.251	0.9725	1.1220	3	gaf2q23	1.0560	0.264	0.9597	1.162
	gaf2q45		1.0572	0.390	0.9312	1.2002		gaf2q45	1.0733	0.087	0.9898	1.1639		gaf2q45	1.1079	0.076	0.9894	1.2405
	haf2q23		1.2419	0.000	1.1157	1.3823		haf2q23	1.0434	0.202	0.9774	1.1138		haf2q23	1.1061	0.033	1.0083	1.2134
	haf2q45		1.2078	0.001	1.0825	1.3477		haf2q45	1.0821	0.022	1.0113	1.1579		haf2q45	1.1061	0.033	1.0083	1.2134
5	gaf2q23		1.1458	0.000	1.0648	1.2329	5	gaf2q23	1.0513	0.054	0.9991	1.1063	5	gaf2q23	1.0145	0.706	0.9413	1.0933
	gaf2q45		1.1916	0.000	1.0810	1.3134		gaf2q45	1.1074	0.001	1.0405	1.1786		gaf2q45	1.0835	0.078	0.9911	1.1846
	haf2q23		1.0893	0.040	1.0039	1.1819		haf2g23	1.0814	0.002	1.0285	1.137		haf2g23	1.0200	0.570	0.9525	1.0924
	haf2q45		1.0658	0.053	0.9989	1.1802		haf2q45	1.1533	0.000	1.0951	1.2146		haf2q45	1.1194	0.002	1.0415	1.2032
7	gaf2q23		1.1586	0.042	1.0056	1.3349	7	gaf2g23	1.0275	0.595	0.9295	1.1359	7	gaf2g23	0.9773	0.758	0.8448	1.1307
	gaf2q45		1.1623	0.118	0.9626	1.4036		gaf2g45	1.1160	0.076	0.9884	1.2602		gaf2g45	1.1115	0.228	0.9358	1.3202
	haf2g23		1.1476	0.093	0.9773	1.3475		haf2g23	1.0320	0.529	0.9354	1.1386		haf2o23	0.9622	0.569	0.8427	1.0986
	haf2q45		1.1827	0.044	1.0044	1.3928		haf2q45	1.1711	0.002	1.0587	1.2955		haf2q45	1.1254	0.094	0.9799	1.2927
9	gaf2q23		1.0677	0.098	0.9879	1.1541	9	gaf2q23	1.0987	0.001	1.0419	1.1585	9	gaf2q23	1.0611	0.131	0.9824	1.1461
	gaf2q45	1	1.0578	0.284	0.9544	1.1725		gaf2q45	1.1599	0.000	1.0875	1.2371		gaf2q45	1.1165	0.018	1.0189	1.2235
	hef2q23		1.0542	0.226	0.9678	1.1483		haf2q23	1.0947	0.001	1.0391	1.1532		haf2q23	1.1008	0.008	1.0256	1.1816
	haf2q45		1.0238	0.598	0.9380	1.1174		haf2q45	1.1405	0.000	1.0806	1.2039		haf2q45	1.1588	0.000	1.0752	1.2489
10	2-20-22		1.0285	0.502	0.9214	1.1480	10	gaf2q23	1.0736	0.046	1.0012	1.1513	10	gaf2q23	1.0687	0.173	0.9713	1.1759
	gaf2q45		1.2192	0.007	1.0568	1.4065		gaf2q45	1.1800	0.000	1.0839	1.2846		gaf2q45	1.1021	0.099	0.9819	1.2371
	haf2q23		1.2113	0.002	1.0731	1.3673		haf2q23	1.0573	0.109	0.9876	1.1319		haf2q23	1.0123	0.786	0.9264	1.1062
	haf2045		1.1329	0.049	1.0070	1.2826		haf2o45	1.1083	0.004	1.0324	1.1898		haf2045	1.1201	0.018	1 0199	1.2302

Incremental risk ratios for GP (GAF2) and hospital (HAF2) access factors, Group C EDs, by admission group, 1991/2, 1992/3, 1993/4

Appendix F 2/5

Incremental risk ratios for GP (GAF2) and hospital (HAF2) access factors, Group C EDs, by admission group, 1991/2, 1992/3, 1993/4

912						923						934					
Group	Access	lm	P	95% c.l	95% c.l	Group	Access	lπ	P	95% c.l	95% c.l	Group	Access	Irr	D	95% c.l	95% c.l
	variable			lower	upper		variable			lower	upper		variable		•	lower	Upper
- 11	a-12+22	1.0787	0.146	0.9739	1.1949	11	gaf2q23	1.0767	0.042	1.0025	1.1563	11	gaf2q23	1.0588	0.290	0.9523	1.1773
	gaf2045	1.1289	0.083	0.9844	1.2945		gaf2q45	1.1191	0.002	1.0428	1.2009		gaf2q45	1.1229	0.070	0.9904	1.2731
	hai2q23	1.0228	0.697	0.9129	1.1459		haf2q23	1.211	0.000	1.1355	1.3129		haf2q23	1.0573	0.262	0.9591	1.1656
	haf2q45	1.0540	0.373	0.9387	1.1834		haf2q45	1.2210	0.000	1.1355	1.3129		haf2q45	1.1409	0.012	1.0297	1.264
12	<b>gal</b> 2q23	1.1638	0.176	0.9343	1.4498	12	ga12q23	0.9856	0.862	0.8374	1.16	12	gaf2q23	1.0141	0.909	0.7977	1.2892
	gaf2q45	1.1198	0.441	0.8395	1.4936		gaf2q45	1.1822	0.085	0.9769	1.4306		gaf2q45	1.1600	0.292	0.9429	1.4606
	hef2q23	1.0474	0.712	0.8193	1.3390		haf2q23	1.2626	0.004	1.0752	1.4827		haf2q23	1.1735	0.152	0.9429	1.4606
	halzq45	1.1004	0.451	0.8578	1.4115		haf2q45	1.4440	0.000	1.226	1.7009		haf2q45	1.3442	0.011	1.0713	1.6867
13	gaf2q23	1.0364	0.579	0.9131	1.1765	13	g <b>af2</b> q23	1.0705	0.153	0.9749	1.1755	13	gaf2q23	1.0625	0.392	0.9246	1.2209
	caf2q45	1.1679	0.071	0.9868	1.3821		gaf2q45	1.2017	0.001	1.0729	1.346		gaf2q45	1.1341	0.137	0.9608	1.3387
	haf2q23	1.0340	0.642	0.8977	1.1910		haf2q23	1.1775	0.001	1.0728	1.2923		haf2q23	1.0820	0.228	0.9518	1.2302
	haf2q45	1.0807	0.290	0.9360	1.2479		haf2q45	1.3604	0.000	1.2371	1.4961		haf2q45	1.2625	0.001	1.1048	1.4429
14	gal2q23	1.1280	0.009	1.0310	1.2342	14	gaf2q23	1.0380	0.257	0.9731	1.1073	14	gaf2q23	1.0295	0.549	0.9361	1.1323
	gal2q45	1.2533	0.000	1.1224	1.4121		gaf2q45	1.1321	0.002	1.046	1.2253		gaf2q45	1.1305	0.035	1.0085	1.2672
	haf2q23	1.0501	0.338	0.9501	1.1607		h <b>af2</b> q23	1.0829	0.014	1.016	1.1541		haf2q23	1.0101	0.821	0.9253	1.1027
	hal2q45	1.0569	0.290	0.9540	1.1709		haf2q45	1.1886	0.000	1.1131	1.2693		haf2q45	1.1444	0.004	1.0438	1.2548
15	g <b>a 2</b> g23	1.0857	0.141	0.9730	1.2114	15	g <b>af2</b> q23	1.0317	0.396	0.9598	1.1091	15	gaf2q23	1.0203	0.708	0.9183	1.1335
	gaf2q45	1.1711	0.034	1.0123	1.3548		gaf2q45	0.9899	0.831	0.9021	1.0863		gaf2q45	1.0263	0.699	0.8997	1.1707
	haf2q23	1.0254	0.683	0.9086	1.1572		h <b>af2</b> q23	1.0456	0.220	0.9737	1.1228		haf2q23	0.9964	0.944	0.9039	1.0984
	hal2q45	1.0600	0.354	0.9369	1.1993		haf2q45	1.0045	0.906	0.9316	1.0832		haf2q45	1.0323	0.552	0.9295	1.1464
16	gal2q23	1.0788	0.193	0.9624	1.2092	16	gaf2q23	1.0394	0.313	0.9641	1.1206	16	gaf2q23	1.0184	0.743	0.9131	1.1359
	gaf2q45	1.1656	0.047	1.0019	1.3559		gaf2q45	0.9934	0.894	0.9018	1.0942		gaf2q45	1.0253	0.720	0.8944	1.1752
	h <b>af</b> 2q23	1.0330	0.610	0.9109	1.1722		haf2q23	1.0421	0.275	0.9676	1.1223		haf2q23	0.9804	0.702	0.886	1.0847
	ha/2q45	1.0802	0.239	0.9499	1.2284		haf2q45	1.0019	0.961	0.9263	1.0837		haf2q45	1.0105	0.850	0.9062	1.1268

Appendix F 3/5

Incremental risk ratios for GP (GAF2) and hospital (HAF2) access factors, Group C EDs, by admission group, 1991/2, 1992/3, 1993/4

\$12						923						934					
Group	Access	lin p	)	95% c.l	95% c.l	Group	Access	In	p	95% c.l	95% c.l	Group	Access	Irr	p	95% c.l	95% c.l
	variable			lower	upper		variable			lower	upper		variable			lower	upper
17	5-13-33	1.0568	0.000	1.0256	1.0889	17	gaf2q23	1.0392	0.000	1.0178	1.061	17	gaf2q23	1.0078	0.615	0.9776	1.0389
	gaf2q45	1.0770	0.000	1.0333	1.1226		gaf2q45	1.0135	0.319	0.987	1.0408		gaf2q45	0.9928	0.710	0.956	1.031
	haf2q23	1.0732	0.000	1.0380	1.1095		haf2q23	1.0667	0.000	1.0451	1.0888		haf2q23	1.0656	0.000	1.0364	1.0957
	haf2q45	1.0548	0.022	1.0193	1.0916		haf2q45	1.0499	0.000	1.0274	1.0728		haf2q45	1.0326	0.037	1.0019	1.0644
18	<del>6-13</del> 433	1.1651	0.000	1.1589	1.2684	18	gaf2q23	1.2360	0.000	1.2118	1.2623	18	gaf2q23	1.2729	0.000	1.2378	1.3091
	gal2q45	1.2124	0.000	1.1589	1.2684		gaf2q45	1.2379	0.000	1.2062	1.2705		gaf2q45	1.2816	0.000	1.2374	1.3274
	haf2q23	0.9752	0.171	0.9407	1.0109		h <b>af</b> 2q23	0.9360	0.000	0.9173	0.9551		haf2q23	0.9568	0.001	0.9317	0.9826
	hal2q45	1.0283	0.138	0.9910	1.0670		haf2q45	0.9477	0.000	0.9277	0.9682		haf2q45	0.9762	0.103	0.9484	1.0048
19	gaf2q23	1.0267	0.529	0.9456	1.1147	19	gaf2q23	0.9973	0.926	0.9425	1.0552	19	gaf2q23	0.9948	0.907	0.9127	1.0844
	gal2q45	1.0832	0.253	0.9445	1.2422		gaf2q45	0.8817	0.002	0.8143	0.9545		gaf2q45	0.9626	0.516	0.8581	1.0798
	hal2q23	1.1080	0.032	1.0087	1.217		haf2q23	1.0380	0.195	0.9807	1.0999		haf2q23	1.0309	0.465	0.95	1.1188
	narzęłs	1.1466	0.006	1.041	1.263		har2q45	0.9912	0.779	0.9323	1.0539		hal2q45	1.0073	0.873	0.9208	1.102
20	gerząza	1.1536	0.000	1.0871	1.2242	20	08/2023	1.0510	0.018	1.0087	1.0951	20	garzq23	1.0704	0.023	1.0095	1.1348
	ganzq45	1.0790	0.129	0.978	1.1034		garzq45	1.0329	0.245	0.978	1.0908		gar2q45	1.0725	0.071	0.9941	1.157
	narzązs	1.0309	0.378	0.9633	1.1034		narząz3	1.0082	0.698	0.967	1.0513		haf2q23	0.9745	0.372	0.9208	1.0313
	hal2q45	1.1788	0.000	1.1004	1.2627		haf2q45	1.0968	0.000	1.0501	1.1454		haf2q45	1.0626	0.049	1.0002	1.129
21	227073	1.2130	0.061	0.9912	1.4844	21	gaf2q23	1.0072	0.914	0.8839	1.1477	21	gaf2q23	0.8535	0.114	0.7014	1.0385
	g812045	1.1261	0.384	0.8619	1.4713		ga12q45	1.0619	0.475	0.9005	1.252		gaf2q45	1.0127	0.917	0.7975	1.2859
		0.9245	0.402	0.7507	1.1008		natząza	1.1341	0.054	0.9978	1.2891		natzg23	1.0512	0.5//	0.8819	1.253
		0.3397	0.000	4.0500	1.1/03		nanz (945	1.0412	0.362	0.9082	1.1935		nat2045	0.9514	0.616	0.7835	1.1554
23	ganzqz3	1,2165	0.008	1.0528	1.4102	23		0.9784	0.682	0.8816	1.0858	23	garzq23	0.7919	0.005	0.6719	0.9335
	ganzq43	1.3508	0.002	1.1141	1.0525		gm2q45	0.9083	0.170	0.7916	1.0421		gar2q45	0.7983	0.032	0.6501	0.9803
	haf2od5	0.9623	0.027	0.0309	1 1569			1.0085	0.207	0.9039	1.1845		hat2q23	1.0050	0.520	0.8695	1.1616
	ILLER CH3	C006.V	0.010	0.031	1.1300		[nan2043	1.0320	0.300	0.9234	1.1522		jnaizq45	1.0511	0.532	0.9988	1.2291

Group C EDs

#### Appendix F 4/5

Incremental risk ratios for GP (GAF2) and hospital (HAF2) access factors, by admission group, Group C EDs, all three years combined

912-934					
Group	Access	lm P		95% c.l	95% c.l
	l gaf2g23	1.0687	0.000	1 0412	1 0067
	gaf2g45	1 0406	0.015	1 0076	1 0746
	haf2g23	1 0714	0.000	1 0435	1 0000
	haf2g45	1 0763	0.000	1 0474	1 1061
	3 gaf2g23	1.0533	0.060	0 9978	1 1 1 2
	gaf2g45	1 0259	0.450	0.9599	1 0964
	haf2g23	1 0895	0.002	1 0313	1 151
	haf2g45	1 0893	0.003	1 029	1 1531
	S gaf2g23	1.0753	0.001	1.0318	1 1208
	gaf2q45	1,1094	0.000	1.0543	1.1675
	haf2q23	1.0751	0.001	1.0307	1,1214
	haf2q45	1.1115	0.000	1.0641	1,1609
	gaf2g23	1.0689	0.106	0.9858	1.1591
	gaf2g45	1,1090	0.014	1.0042	1.2247
	haf2q23	1.0558	0.197	0.9721	1.1467
	haf2q45	1.1498	0.001	1.0563	1.2517
	gaf2q23	1.0799	0.001	1.0341	1.1278
	gaf2q45	1.0994	0.000	1.0424	1.1595
	haf2q23	1.0726	0.002	1.0264	1.1208
	haf2q45	1.0817	0.001	1.0336	1.1322
10	gaf2q23	1.0497	0.102	0.9903	1.1127
	gaf2q45	1.1079	0.005	1.0316	1.19
	haf2q23	1.0619	0.045	1.0013	1.1262
	haf2q45	1.0526	0.100	0.9902	1.1189
11	gaf2q23	1.0708	0.021	1.0104	1.1347
	gaf2q45	1.1340	0.001	1.0561	1.2176
	haf2q23	1.0841	0.008	1.0216	1.1503
	haf2q45	1.1520	0.000	1.0838	1.2244
12	2 gaf2q23	1.0441	0.512	0.9175	1.882
	gaf2q45	1.1733	0.042	1.0058	1.3686
	haf2q23	1.1979	0.008	1.0484	1.3687
	haf2q45	1.3265	0.000	1.158	1.5196
13	gaf2q23	1.0577	0.140	0.9816	1.1397
	gaf2q45	1.2007	0.000	1.096	1.3154
	haf2q23	1.1350	0.001	1.0509	1.2259
	haf2q45	1.2675	0.000	1.1716	1.3712
14	gef2q23	1.0635	0.020	1.0096	1.1202
	gaf2q45	1.1663	0.000	1.094	1.2434
	haf2q23	1,1386	0.000	1.0779	1.2027
	Inarzq45	1.1386	0.000	1.0779	1.2027
18	garzq23	1.0330	0.285	0.9732	1.0965
	gar2q45	1.0406	0.304	0.9645	1.1226
	harząz3	1.0120	0.699	0.9524	1.0753
	na12q45	1.0040	0.990	0.9389	1.0659
16	gerząza	1.0354	0.271	0.9731	1.1017
	gar2q45	1.0471	0.253	0.9676	1.133
	na12q23	1.0107	0.741	0.9488	1.0765
	jnarzg45	1.0050	0.880	0.9409	1.0736

#### Appendix F 5/5

Incremental risk ratios for GP (GAF2) and hospital (HAF2) access factors, by admission group, Group C EDs, all three years combined

912-934					
Group	Access variable	lrr ş	)	95% c.l lower	95% c.l upper
17	gaf2q23	1.0399	0.000	1.0224	1.0576
	gaf2q45	1.0060	0.589	0.9843	1.0281
	haf2q23	1.0603	0.000	1.0422	1.0788
	haf2q45	1.0356	0.000	1.017	1.0545
11	gaf2q23	1.2093	0.000	1.1888	1.2302
	gaf2q45	1.1501	0.000	1.1251	1.1757
	haf2q23	0.9233	0.000	0.9074	0.9395
	haf2q45	0.9309	0.000	0.9141	0.9481
19	gaf2q23	1.0120	0.610	0.9666	1.0596
	gaf2q45	0.8654	0.000	0.8095	0.9252
	haf2q23	1.0497	0.048	1.003	1.1015
	haf2q45	1.0215	0.633	0.962	1.0656
20	) gaf2q23	1.0682	0.000	1.0331	1.1045
	gaf2q45	1.0237	0.497	0.9773	1.0483
	haf2q23	1.0122	0.497	0.9773	1.0483
	haf2q45	1.1125	0.000	1.0729	1.1535
21	gaf2q23	1.0316	0.573	0.9257	1.1498
	gaf2q45	1.0767	0.286	0.94	1.2333
	haf2q23	1.0364	0.523	0.9287	1.1566
	haf2q45	0.9962	0.949	0.8878	1.1179
2:	3 gaf2q23	1.0457	0.294	0.9618	1.1369
	gaf2q45	1.0682	0.234	0.9582	1.909
	haf2q23	1.0407	0.361	0.9552	1.1337
	haf2q45	1.0251	0.588	0.9368	1.1218

Group C EDs

#### Appendix G 1/4

Incremental relative risk ratios (IRRs) for selected hospital service areas, and for selected admission groups, 1993/4

3	Strong and weak ACS conditions,	Barnet	1.0000	-	•	•
	Postmissions	Feller	0.0444	0.004	0.0405	0.0707
	Readmissions	Edensee	1 0490	0.001	0.9125	0.9/6/
		Edgware	1.0180	0.102	0.9929	1.0438
			0.9881	0.223	0.9693	1.0073
		Hillingaon	1.0016	0.838	0.9861	1.01/4
		Lister	1.0033	0.656	0.9888	1.0180
		Mt Vernon	1.0173	0.012	1.0038	1.0309
		Northwick Park	1.0183	0.000	1.0083	1.0284
		Queen Elizabeth II	0.9939	0.264	0.9833	1.0046
		St Mary's	0.9957	0.343	0.9867	1.0046
		Watford	1.0117	0.003	1.0039	1.0195
		West Middlesex	0.9952	0.223	0.9876	1.0029
5	Strong and weak ACS conditions, urgent,	Barnet	1.0000		•	
	all admissions	Ealing	0.9876	0.346	0.9623	1.0136
		Edgware	0.9872	0.208	0.9676	1.0072
		Hemel Hempstead	0.9699	0.000	0.9551	0.9848
		Hillingdon	0.9890	0.082	0.9768	1.0014
		Lister	1.0077	0.188	0.9963	1.0192
		Mt Vernon	1.0016	0.772	0.9908	1.0125
		Northwick Park	1.0111	0.005	1.0033	1.0191
		Queen Elizabeth II	0.9925	0.079	0.9841	1.0008
		St Mary's	0.9857	0.000	0.9787	0.9928
		Watford	1.0044	0.162	0.9982	1.0105
		West Middlesex	0.9814	0.000	0.9752	0.9876
7	Strong and weak ACS conditions, urgent,	Barnet	1.0000	•	•	•
	Readmissions	Falloo	0 9480	0.038	0 9014	0 9972
		Edgware	0 9723	0 155	0.9353	1 0107
		Hemel Hemostead	0.9613	0.009	0.9334	0.9901
		Hillingdon	0 9873	0.293	0 9641	1 0111
		Lister	1.0090	0419	0.9873	1 0312
		Mt Vernon	1 0136	0 195	0.9931	1 0344
		Northwick Park	1 0103	0 181	0.9952	1 0255
		Queen Elizabeth II	0 9808	0.022	0.9648	0.9972
		St Mary's	0.9838	0.019	0 9704	0 9973
		Wattord	1 0081	0.178	0.0064	1 0100
		West Middlesey	0.9716	0.000	0.9597	0.9836
9	Strong ACS conditions, any unserver	Remet	L 1 000	0.000	0.0007	0.0030
-	all admissions	Sallag	0.077	0.087	0 0512	1 0024
		Edaware	0.075	0.007	0.0559	0.0064
		Hemel Hempsterd	0.007	7 0 789	0.0834	1 0124
		Hillingdon	0.007	0.700	0.9024	1.0131
		Lister	1 000	5 0 274	0.8014	1.0000
		Mt Versen	1.000	0.274	4 04940	1.0402
		Northwick Bart	1.028		1.0100	1.0402
		Oueen Ellenbeth	1.0112		0.0002	1.0183
		Queen Elizabeth II	0.9994	0.005	0.9908	1.0080
		ot mary's	1.001	5 0.014	0.9947	1.0090
		DIOLOG	1.017	0.000	1.0115	1.0240
		West Middlesex	0.990	0.004	0.9847	0.9972

Incremental relative risk ratios	(IRRs)	for	selected	hospital	service	areas,	and for	r
selected admission groups, 1993/4				-				

10	Strong ACS conditions any urgency	Remet	Т	1.000	0			
	Readmissions	Eellog		1.000	ă	0 756	0.0602	1 0440
		Edaware		1 031	2	0.730	1 0025	1 0800
		Hemel Hemneteed	.	1 1 2 9	3	0.033	1 1057	1 1 5 1 1
		Hillingdon	'	1.120	7	0.000	1.1037	1.1011
		Lister		1 047	÷.	0.000	1.0200	1.0040
		Mt Vernon		1.047	3	0.000	1.0309	1.0030
		Northwick Back		1.042	7	0.000	1.0275	1.0000
		Queen Elizabeth I	.	1.023		0.000	1.0122	1.0303
		St Mand's	'	1.027	2	0.000	0.0007	1.0390
		St Mary B		1.000		0.902	0.9907	1.0107
		West Middlesov		1.038	6	0.000	0.0917	0.0005
44		West Millulesex	+	0.990	0	0.038	0.9017	0.9995
	Strong ACS conditions, urgent,	Eellee		1.000		•	-	4 0450
		Edourono	1	1.008	4	0.003	0.9/44	1.0400
		Eugware	.	0.978	0	0.145	0.9520	1.0072
		Hemel Hempstead	•	0.904	0	0.001	0.9435	0.9849
		Lister		1.000	5	0.003	0.9004	0.9900
				1.000	2	0.320	0.9921	1.0242
		Nextbudek Deck		1.994	1	0.490	0.9790	1.0100
		Oueen Elizabeth I	, I	0.097		0.000	0.0755	0.0001
		St Maru's	1	0.907	3	0.030	0.9755	0.9991
		Wetford		1 011	3	0.000	1 0020	1 0109
		West Middlesey	-1	0.073	3	0.000	0.0640	0.0822
49	Rivers ACR conditions turnent	Remet	-+	1.000	0	0.000	0.9049	0.8023
14	Boodmissions	Seling		0.000		0 004	0 9394	-
	Keadmissions	Edaware	- 1	0.000		0.004	0.0304	0.9070
		Hamel Hampeteed	. I	0.813	2	0.002	0.0037	0.8003
		Hillingdon	'	0.837	5	0.043	0.9111	0.0570
		Lister		0.020	0	0.000	0.0357	0.0001
		Mt Vernon		0.000	ă	0.593	0.8557	1 0220
		Northwick Park		1 001	ň	0.000	0.0020	1 0221
		Queen Elizabeth I	ı I	0.951	Ř	0.000	0 9278	0.9765
		St Mary's	1	0.966	Ř	0.001	0 9478	0.9857
		Watford	1	0.998		0.850	0.9821	1 1050
		West Middlesex		0.941	5	0.000	0 9240	0 8594
	Strong ACS conditions year urgent	Barnet	1	0000	-		-	
	all admissions	Ealing	0	9829	n	449	0 9401	1 0278
		Edoware	ň	9599	ň	023	0.0269	0 0043
		Hemel	n.	9822	ň	182	0.9566	1.0085
		Hemostead	•	UULL			0.0000	1.0000
		Hillingdon	0	9545	n	000	0.9335	0.9760
		Lister	Ō.	9948	ō	.013	0.9750	1.0150
		Mt Vernon	0	9960	0	678	0.9771	1.0151
		Northwick Park	0	9922	0	266	0.9787	1.0060
		Queen Elizabeth	0	9754	0	.002	0.9606	0 9905
		1	Ψ.					
		St Mary's	0.	9711	0	.000	0.9588	0.9835
		Watford	1.	0104	0	.050	1.0000	1.0210
		West Middlesex	0.	9695	0	.000	0.9588	0.9803

Incremental relative risk ratios (IRRs) for selected hospital service areas, and for selected admission groups, 1993/4

14	Strong ACS conditions, very urgent,	Barnet	1.0000	-	-	-
	Readmissions	Ealing	0.9759	0.140	0.9448	1.0080
		Edgware	0.9763	0.060	0.9522	1.0010
		Hemel	0.9768	0.015	0.9586	0.9954
		Hempstead				
		Hillingdon	0.9709	0.000	0.9557	0.9863
		Lister	1.0054	0.455	0.9913	1.0197
		Mt Vernon	0.9957	0.540	0.9823	1.0094
		Northwick Park	0.9948	0.303	0.9851	1.0047
		Queen Elizabeth	0.9837	0.003	0.9733	0.9942
		111				0.00.1
		St Mary's	0.9721	0.000	0.9633	0.9810
		Watford	1 0007	0.854	0 9931	1 0084
		West Middlesex	0.9801	0.000	0.9725	0 9877
15	Strong and weak markers very	Barnet	1 0000		0.0120	0.0077
	urgent.		1.0000			
	all admissions	Falino	0 9530	0.018	0.0158	0 0018
		Edowara	0.9887	0.450	0.0100	1 0180
		Hemel	1 0010	0.408	0.0000	1.0241
		Hempsterd	1.0013	0.001	0.9002	1.0241
		Hillingdon	0 0830	0.000	0.0857	1 0025
		Lieter	1 0218	0.050	1 0052	1.0025
		Mal	1.0210	0.010	0.0944	1.0307
		Northwick Book	0.0704	0.900	0.0070	0.0014
		Ronnwick Park	4.0024	0.001	0.9070	4.0440
		Queen Elizabeth	1.0021	0.737	0.9900	1.0142
		los Manuda	0.0040	0.000	0.0540	0.0750
		St Mary s	0.9046	0.000	0.9540	0.9730
			1.0000	0.995	0.9909	1.0092
		TABEL WIGGIESEX	0.9871	0.005	0.9781	0.9963
17	Admissions in surgical specialties	Bernet	1.0000			-
		Ealing	0.9652	0.000	0.9544	0.9761
		Edgware	0.9790	0.000	0.9712	0.9881
		Hemel Hempstead	0.9896	0.001	0.9834	0.9958
		Hillingdon	0.9924	0.005	0.9872	0.9977
		Lister	1.0132	0.000	1.0084	1.0180
		Mt Vernon	0.9979	0.365	0.9934	1.0024
		Northwick Park	0.9811	0.000	0.9777	0.9845
		Queen Elizabeth II	1.0037	0.031	1.0003	1.0071
		St Mary's	0.9837	0.000	0.9806	0.9867
		Watford	0.9966	0.011	0.9940	0.9992
		West Middlesex	0.9925	0.000	0.9899	0.9951
18	Admissions in medical specialties	Barnet	1.0000	-	-	-
		Ealing	1.0494	0.000	1.0368	1.0621
		Edgware	1.0472	0.000	1.0376	1.0569
		Hemel Hempstead	1.0225	0.000	1.0156	1.0296
		Hillingdon	1.0518	0.000	1.0460	1.0577
		Lister	1.1038	0.000	1.0985	1.1091
		Mt Vernon	1.0389	0.000	1.0341	1.0438
		Northwick Park	1.0282	0.000	1.0245	1.0319
		Queen Elizabeth II	1.0124	0.000	1.0086	1.0162
		St Mary's	1.0796	0.000	1.0763	1.0829
		Watford	1.0232	0.000	1.0203	1.0260
		West Middlesex	1.0063	0.000	1.0035	1.0092

Incremental relative risk ratios (IRRs) for selected hospital service areas, and for selected admission groups, 1993/4

19     Emergency admissions in surgical specialties     Barnet     1.0000     -     -     -       specialties     Ealing     0.7429     0.0000     0.7086     0.7786       Edgware     1.0543     0.001     1.0130     0.6560     0.000     0.8527       Hamel Hempstaad     1.0343     0.001     1.0130     0.6560     9.7786       Northwick Park     1.0128     0.2029     1.0013     1.0244     0.000     1.0366     9.9777       Northwick Park     1.0128     0.029     1.0016     1.0306     1.0326       20     Emergency admissions in modical specialties     Barnet     1.0000     -     -       20     Emergency admissions in modical specialties     Barnet     1.0000     -     -     -       20     Emergency admissions in modical specialties     Barnet     1.0000     -     -     -       20     Emergency admissions in modical specialties     Barnet     1.0000     -     -     -       20     Ealing     0.6917     0.000     0.6813							
20     Emergency admissions in modical specialties     Ealing Edgware Hempstead 1.0343     0.001     1.0366     0.9901       20     Emergency admissions in modical specialties     0.6917     0.000     0.8120     0.000     0.9117     0.8523       20     Emergency admissions in modical specialties     0.6917     0.000     0.9165     0.000     0.9165     0.000     1.0163     1.0244       20     Emergency admissions in modical specialties     Barnet     1.0019     0.000     1.0111     1.0287       20     Emergency admissions in modical specialties     Barnet     1.0000     -     -     -       20     Emergency admissions in modical specialties     Barnet     1.0027     0.000     1.0816     0.000     1.0817     0.000     1.0801     1.0806       20     Ealing     0.6917     0.000     1.6663     0.7159     1.0866     0.001     1.0170     0.755     0.006     1.0801     1.0376     0.000     1.0801     1.0376     0.000     1.0817     1.0475     0.000     1.0817     1.0475     0.000	19	Emergency admissions in surgical specialties	Barnet	1.0000	•	•	-
20     Edgware Hemai Hempstead Hillingdon     1.0599     0.000     1.0306     1.0901       111     Lister     1.0811     0.0021     1.0851     0.002     1.0851       111     Lister     1.0815     0.000     0.8317     0.8527       Lister     1.0815     0.000     0.8117     0.8527       Northwick Park     1.0128     0.002     1.0306     1.0304       10128     0.000     0.9856     0.9778     1.0319     0.9851     1.0030       20     Emorgency admissions in modical spocialties     Barnet     1.0000     -     -     -       Edgware     1.1425     0.000     1.0814     1.0030     1.0814     1.0030       1111     0.0225     0.000     1.0814     1.0326     0.000     1.0814     1.0326       20     Emorgency admissions for ACS conditions     Barnet     1.0472     0.000     1.0814     1.0326       1111     0.0275     0.401     0.907     0.004     0.9866     0.9874       11111     0.0277<			Eeling	0.7429	0.000	0.7086	0.7788
20     Emorgency admissions in medical specialties     Hermai Hempstead Hillingdon     1.0343     0.001     1.0168     1.1016       20     Emorgency admissions in medical specialties     Barnet     1.0103     1.0240     0.000     0.9568     0.9778       20     Emorgency admissions in medical specialties     Barnet     1.0010     1.0240     0.000     1.0213     1.0241       20     Emorgency admissions in medical specialties     Barnet     1.0000     -     -     -     -       20     Emorgency admissions in medical specialties     Barnet     1.0000     -			Edgware	1.0599	0.000	1.0306	1.0901
20     Emergency admissions in modical spocialties     Hillingdon (1.0395)     0.000     0.8110     0.0029     1.0013     1.0244       20     Emergency admissions in modical spocialties     Barnet     1.0010     0.000     0.9236     0.9577       20     Emergency admissions in modical spocialties     Barnet     1.0019     0.000     0.9851     1.0030       20     Emergency admissions in modical spocialties     Barnet     1.0000     -     -     -       20     Emergency admissions in modical spocialties     Barnet     1.0000     -     -     -       20     Emergency admissions in modical spocialties     Barnet     1.0000     -     -     -       20     Emergency admissions in modical spocialties     Barnet     1.0270     0.000     1.0830     0.1181     0.000     1.0830     0.000     1.0830     0.000     1.0830     0.000     1.0830     0.000     1.0830     0.000     1.0836     0.000     1.0811     1.0475     0.000     1.0814     1.0355     0.000     1.0814     1.0355     1.0379			Hemel Hempstead	1.0343	0.001	1.0130	1.0560
Lister     1.0851     0.000     1.0689     1.1018       Mt Vernon     0.9405     0.000     0.9236     0.9577       Northwick Park     1.0124     0.029     1.0013     1.0244       Queen Elizabeth II     1.0419     0.000     0.9566     0.9778       Watford     1.0194     0.000     0.9566     0.9778       Watford     1.0194     0.000     0.9586     0.7159       Poclaities     Barnet     1.000     -     -       Ealing     0.6917     0.000     1.6683     0.7159       Edgware     1.1025     0.000     1.0214     1.0300       Hemel Hempstead     1.0749     0.000     1.0814     1.0306       Hillingdon     0.7932     0.000     1.0814     1.0390       Lister     1.0927     0.441     0.9957     1.0046       Mt Vernon     0.9778     0.000     1.0612     1.0729       Met Middlesex     0.9807     0.004     0.9846     0.9869       Met Sons for ACS conditions     Bar			Hillingdon	0.8320	0.000	0.8117	0.8527
Mt Vernon     0.9402     0.0236     0.9361     1.0230       20     Emergency admissions in medical specialties     Barnet     1.0270     0.900     1.6314     1.1038     1.1632     1.1632     1.1632     1.1632     1.1632     1.1632     1.1632     1.1632     1.1032     1.217     1.0270     0.200     0.917     0.200     0.916			Lister	1.0851	0.000	1.0689	1.1016
Admissions for ACS conditions Defined by Billings     Barnet Ealing Edgware Hemel Hempstead Hillingdon Uset Middlesex     1.0128 0.000     0.001 0.000     0.0306 0.0307 0.000 0.0366 0.0307 0.000 0.0366 0.0307 0.000 0.0366 0.0307 0.000 0.0366 0.0306 0.0307 0.000 0.0306 0.0306 0.0307 0.000 0.0346 0.0380 0.0306 0.0340 0.0390 0.0340 0.0340 0.0390 0.0340 0.0340 0.0340 0.0390 0.0340 0.0340 0.0390 0.0340 0.0340 0.0390 0.0340 0.0340 0.0340 0.0390 0.0340 0.0340 0.0390 0.0340 0.0340 0.0390 0.0340 0.0340 0.0390 0.0340 0.			Mt Vernon	0.9405	0.000	0.9236	0.9577
Queen Elizabeth II     1.0419     0.000     1.0534       20     Emergency admissions in modical spocialties     Barnet     0.9942     0.000     1.0111     1.0230       20     Emergency admissions in modical spocialties     Barnet     1.0000     -     -     -       spocialties     Ealing     0.6917     0.000     1.0610     1.0390       20     Energency admissions in modical spocialties     Barnet     1.0000     -     -       spocialties     Ealing     0.6917     0.000     1.6833     0.7159       Edgware     Hernel Hempstead     1.0749     0.000     1.0814     1.0390       Lister     1.0925     0.000     1.0814     1.0380     0.000     1.0817     1.0475       Mt Vernon     0.9778     0.000     1.0817     1.0472     1.0840     0.000     1.0812     1.0729       21     Admissions for ACS conditions     Barnet     1.0047     0.1865     0.014     1.0128     1.1217       Edgware     Hemel Hempstead     1.0970     0.441     0			Northwick Park	1.0128	0.029	1.0013	1.0244
St Mary's     0.9672     0.000     0.9566     0.9778       20     Emergency admissions in medical specialities     Barnet     1.0198     0.000     1.0111     1.0287       20     Emergency admissions in medical specialities     Barnet     1.0000     -     -     -       20     Emergency admissions in medical specialities     Barnet     1.0000     -     -     -     -       21     Admissions for ACS conditions     Lister     1.0749     0.000     1.0800     1.0376     0.000     1.0806     0.000     1.0866     0.9897     1.0027     0.441     0.9957     1.0048     0.9957     1.0048     0.9957     1.0027     1.0445     0.000     1.0846     0.9967     1.0046     0.9967     1.0048     0.9957     1.0028     0.9957     1.0027     1.0445     0.9957     1.0027     1.0217     1.1217     1.0227     1.0445     0.9957     1.0028     0.9977     1.1217     1.0257     1.0024     0.9248     0.9956     1.0170     1.1217     1.0257     1.0121     1.221			Queen Elizabeth II	1.0419	0.000	1.0306	1.0534
Wattord     1.0198     0.000     1.0111     1.0287       20     Emergency admissions in medical specialties     Barnet     0.0940     0.193     0.9851     1.0030       20     Emergency admissions in medical specialties     Barnet     1.0000     -     -     -       20     Ealing     0.6917     0.000     0.6683     0.7159       20     Edgware     1.1425     0.000     1.0900     1.0902     1.1636       21     Admissions for ACS conditions     Defined by Billings     1.0378     0.000     1.0418     1.0575       21     Admissions for ACS conditions     Barnet     1.0670     0.000     1.0418     1.0575       21     Admissions for ACS conditions     Barnet     1.0670     -			St Mary's	0.9672	0.000	0.9566	0.9778
West Middlesex     0.9940     0.1930     0.9851     1.0030       20     Emergency admissions in medical specialties     Barnet     1.0000     -     -       21     Specialties     Ealing     0.6917     0.000     0.6683     0.7159       22     Ealing     0.6917     0.000     1.6200     1.1220     1.1634       23     Admissions for ACS conditions     Barnet     1.0749     0.000     1.0614     1.0360       21     Admissions for ACS conditions     Barnet     1.0925     0.000     1.0814     1.0375       21     Admissions for ACS conditions     Barnet     1.0027     0.441     0.9957     1.0084       21     Admissions for ACS conditions     Barnet     1.0027     0.441     0.9846     0.9849       21     Admissions for ACS conditions     Barnet     1.00472     0.188     0.9777     1.1217       23     Admissions for marker conditions     Barnet     1.0072     0.441     0.9846     0.9847     1.0428       23     Admissions for marker conditions			Watford	1.0198	0.000	1.0111	1.0287
20     Emergency admissions in medical specialties     Barnet     1.0000     -     -       21     Ealing     0.6917     0.000     1.1425     0.000     1.1425       20     Edgware     1.1425     0.000     1.1425     0.000     1.1420       20     Edgware     1.1425     0.000     1.1420     1.1634       20     Millingdon     0.7932     0.000     1.786     0.8083       21     Admissions for ACS conditions     Mit Vernon     0.9778     0.000     1.0475       21     Admissions for ACS conditions     Barnet     1.0000     -     -     -       21     Admissions for ACS conditions     Barnet     1.0496     0.000     1.0412     1.0287       21     Admissions for ACS conditions     Barnet     1.0000     -     -     -       221     Admissions for ACS conditions     Barnet     1.0472     0.188     0.9861     1.0472       223     Admissions for ACS conditions     Barnet     1.0472     0.188     0.9769     1.0200			West Middlesex	0.9940	0.193	0.9851	1.0030
21     Admissions for ACS conditions Defined by Billings     Barnet     1.0420     0.000     1.1220     1.1634       4     Hemai Hempstead     1.0749     0.000     1.0800     1.0900       Hillingdon     0.7785     0.000     1.0814     1.1038       Lister     1.0925     0.000     1.0814     1.1038       Northwick Park     1.0927     0.441     0.9957     1.0028       Northwick Park     1.0027     0.441     0.9957     1.0028       21     Admissions for ACS conditions     Barnet     1.0000     -     -       Defined by Billings     Ealing     1.0472     0.188     0.9777     1.1217       Edgware     1.0476     0.247     0.9366     1.0170       Hillingdon     1.0476     0.387     0.9818     1.0472       Northwick Park     0.9697     0.247     0.9366     1.0170       Hillingdon     1.0476     0.247     0.9366     1.0170       Hillingdon     1.0476     0.387     0.9818     1.0428       No	20	Emergency admissions in medical specialties	Barnet	1.0000	•	•	-
21     Admissions for ACS conditions     Barnet     1.0425     0.000     1.1620     1.0900       21     Admissions for ACS conditions     Barnet     1.0027     0.441     0.9987     1.0092       21     Admissions for ACS conditions     Barnet     1.0020     0.000     0.9866     0.9989       21     Admissions for ACS conditions     Barnet     1.0027     0.441     0.9957     1.0029       21     Admissions for ACS conditions     Barnet     1.0000     -     -     -       Uset Middlesex     0.9907     0.004     0.9846     0.9969       21     Admissions for ACS conditions     Barnet     1.0000     -     -     -       Edgware     1.0665     0.014     1.0128     1.1231     1.0428     1.0212     1.0428       Mt Vernon     0.9760     0.247     0.188     0.9777     1.0475       Mt Vernon     0.9749     0.186     0.9497     1.0151     0.4579     0.9818     1.0429       Mt Vernon     0.9740     0.3861     0.000			Ealing	0.6917	0.000	0.6683	0.7159
21     Admissions for ACS conditions Defined by Billings     Barnet Hillingdon     1.0749     0.000     1.0600     1.0900       21     Admissions for ACS conditions Defined by Billings     Barnet Hillingdon     1.0475     0.000     1.0418     0.0577       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -     -       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -     -       23     Admissions for marker conditions Defined by Billings     Barnet     1.0015     0.247     0.9366     0.917     1.1217       23     Admissions for marker conditions Defined by Billings     Barnet     1.0017     0.399     0.9919     1.0359       24     Admissiona for marker conditions Defined by Billings     Barnet     1.0017     0.399     0.9919     1.026			Edgware	1.1425	0.000	1.1220	1.1634
21     Admissions for ACS conditions Defined by Billings     Hillingdon Lister     0.7932     0.000     0.7785     0.8083       21     Admissions for ACS conditions Defined by Billings     St Mary's Horthwick Park Queen Elizabeth II     1.0396     0.000     1.0317     1.0475       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0027     0.441     0.9957     1.0098       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0027     0.441     0.9957     1.0286       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0007     -			Hemel Hempstead	1.0749	0.000	1.0600	1.0900
21     Admissions for ACS conditions Defined by Billings     Barnet     1.0925     0.000     1.0814     1.1038       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0027     0.441     0.9957     1.0028       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0027     0.441     0.9957     1.0028       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       Edgware     1.0665     0.014     1.028     1.231       Hemel Hempstead     0.9760     0.247     0.9366     1.0170       Hillingdon     1.0146     0.387     0.9818     1.0428       1.002     0.9760     0.247     0.9366     1.0170       Hillingdon     1.0146     0.387     0.9812     1.0428       1.0115     0.459     0.9812     1.0428     1.0428       1.022     0.9366     0.000     0.9427     0.9816     1.0426       1.0115     0.459     0.9819     1.0350     1.0247			Hillingdon	0.7932	0.000	0.7785	0.8083
Nt Vernon Northwick Park Queen Elizabeth II     0.9778     0.000     0.9666     0.9891       21     Admissions for ACS conditions Defined by Billings     St Mary's Barnet     1.0470     0.000     1.0418     1.0575       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0070     0.000     1.0612     1.0729       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0070     0.000     1.0612     1.1217       Edgware     1.0665     0.014     1.0128     1.1231       Hemel Hempstead     0.9760     0.247     0.9366     1.0479       Northwick Park     0.9561     0.000     0.9481     1.0429       Mt Vernon     0.9774     0.186     0.9497     1.0429       Mt Vernon     0.9760     0.247     0.9366     1.0470       Queen Elizabeth II     1.0173     0.2487     0.9818     1.0429       Mt Vernon     0.9764     0.186     0.9497     1.0160       St Mary's     0.9619     0.000     0.9427     0.9815       Defined by Billi			Lister	1.0925	0.000	1.0814	1.1038
Northwick Park Queen Elizabeth II     1.0396     0.000     1.0317     1.0475       Admissions for ACS conditions Defined by Billings     St Mary's Watford     1.0027     0.441     0.9957     1.00812     1.0729       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -			Mt Vernon	0.9778	0.000	0.9666	0.9891
Admissions for ACS conditions Defined by Billings     Barnet Ealing     1.0007     0.441     0.9957     1.0098       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -			Northwick Park	1.0396	0.000	1.0317	1.0475
St Mary's Watford     1.0027     0.441     0.9957     1.0029       21     Admissions for ACS conditions Defined by Billings     Barnet Ealing     1.0072     0.188     0.9969       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0072     0.188     0.9977     1.1217       Edgware     1.0665     0.014     1.0128     1.1231       Hemel Hempstead     0.9760     0.247     0.9868     1.0472       Hillingdon     1.0115     0.459     0.9818     1.0429       Mt Vernon     0.9764     0.186     0.9477     1.0148       Lister     1.0115     0.459     0.9818     1.0429       Mt Vernon     0.9744     0.186     0.9497     1.0100       Northwick Park     0.9619     0.000     0.9427     0.9818       Queen Elizabeth II     1.0000     -     -     -       23     Admissions for marker conditions     Barnet     1.0000     -     -     -       24     West Middlesex     0.9359     0.041     0.8784			Queen Elizabeth II	1.0496	0.000	1.0418	1.0575
Watford West Middlesex     1.0670     0.000     1.0612     1.0729       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       21     Defined by Billings     Barnet     1.0472     0.188     0.9777     1.1217       Edgware     1.0665     0.014     1.0128     1.1231       Hemel Hempstead     0.9760     0.247     0.9366     1.0479       Hillingdon     1.0145     0.387     0.9818     1.0489       Lister     1.0115     0.459     0.9919     1.0359       Mt Vernon     0.9744     0.9861     0.9977     1.0249       Wastford     1.0071     0.399     0.9906     1.0240       Wastford     0.9025     0.551     0.9473     1.0359       Z3     Admissions for marker conditions     Barnet     1.0000     -     -       Ealing     0.9359     0.041     0.8784			St Mary's	1.0027	0.441	0.9957	1.0098
West Middlesex     0.9907     0.004     0.9846     0.9969       21     Admissions for ACS conditions Defined by Billings     Barnet Ealing     1.0070     -     -     -       Ealing     1.0472     0.188     0.9777     1.1217       Edgware     1.0665     0.014     1.0128     1.1231       Hemel Hempstead     0.9760     0.247     0.9366     1.0170       Hemel Hempstead     0.9760     0.247     0.9366     1.0170       Mt Vernon     0.9794     0.186     0.9497     1.0408       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9815       Watford     1.0071     0.399     0.9906     1.0240       VestMiddlesex     0.9222     0.354     0.9759     1.0087       23     Admissions for marker conditions     Barnet     1.0000     -     -       Ealing     0.9359     0.041     0.8784     0.9972       24     Admissions for marker conditions			Watford	1.0670	0.000	1.0612	1.0729
21     Admissions for ACS conditions Defined by Billings     Barnet     1.0000     -     -     -       Ealing     1.0472     0.188     0.9777     1.1217       Edgware     1.0665     0.014     1.0128     1.1231       Hemel Hempstead     0.9760     0.247     0.9366     1.0170       Hamel Hempstead     0.9760     0.247     0.9368     1.0170       Hillingdon     1.0146     0.387     0.9818     1.0486       Lister     1.0115     0.459     0.9812     1.0426       Mt Vernon     0.9794     0.186     0.9497     1.0100       Northwick Park     0.9061     0.000     0.9340     0.9768       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9818       Defined by Billings     Barnet     1.0071     0.399     0.9906     1.0240       West Middlesex     0.9252     0.751     0.9473     1.0397       Eding     0.93269     0.041			West Middlesex	0.9907	0.004	0.9846	0.9969
Defined by Billings     Ealing Edgware     1.0472     0.188     0.9777     1.1217       Edgware     1.0665     0.014     1.0128     1.1231       Hemel Hempstead     0.9760     0.247     0.9366     1.0170       Hillingdon     1.0115     0.459     0.9818     1.0486       Lister     1.0115     0.459     0.9812     1.0429       Mt Vernon     0.9764     0.186     0.9947     1.0109       Northwick Park     0.9561     0.000     0.9340     0.9769       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9818       Vartford     1.0071     0.399     0.9906     1.0240       West Middlesex     0.9225     0.354     0.9759     1.0087       Eding     0.9359     0.041     0.8784     0.9972       Defined by Billings     Ealing     0.9358     0.000     0.9014     0.9673       Hemel Hempstead     0.9861     0.348     0.9578	21	Admissions for ACS conditions	Barnet	1.0000	-	-	-
Edgware     1.0665     0.014     1.0128     1.1231       Hemel Hempstad     0.9760     0.247     0.9366     1.0170       Hillingdon     1.0146     0.387     0.9818     1.0486       Lister     1.0115     0.459     0.9812     1.0429       Mt Vernon     0.9764     0.186     0.9477     1.0109       Northwick Park     0.9561     0.000     0.9340     0.9788       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9815       Wastford     1.0071     0.399     0.9906     1.0240       West Middlesex     0.9925     0.354     0.9759     1.0087       Z3     Admissions for marker conditions     Barnet     1.0000     -     -       Ealing     0.9359     0.041     0.8784     0.9972       Edgware     0.9925     0.751     0.9473     1.0379       Hemel Hempstead     0.9388     0.000     0.9014     0.9673		Defined by Billings	Ealing	1.0472	0.188	0.9777	1.1217
23     Admissions for marker conditions Defined by Billings     Barret Hemei Hempstead Hillingdon     1.0146     0.387     0.9818     1.0486       23     Admissions for marker conditions Defined by Billings     Barret Hemei Hempstead     0.9760     0.247     0.9818     1.0486       24     Mt Vernon     0.9794     0.186     0.9497     1.0100       Northwick Park     0.9561     0.000     0.9340     0.9818     1.0359       25     Matry's     0.9619     0.000     0.9427     0.9815       Watford     1.0071     0.399     0.9906     1.0240       West Middlesex     0.922     0.354     0.9759     1.0087       Hemel Hempstead     0.9359     0.041     0.8784     0.9972       26     Admissions for marker conditions     Barnet     1.0000     -     -       Hemei Hempstead     0.9359     0.041     0.8784     0.9972     0.9815     1.0387       Lister     1.0107     0.9925     0.751     0.9473     1.0397     1.0387       Lister     1.0167 <td< th=""><th></th><th></th><th>Edgware</th><th>1.0665</th><th>0.014</th><th>1.0128</th><th>1.1231</th></td<>			Edgware	1.0665	0.014	1.0128	1.1231
Hillingdon     1.0146     0.387     0.9818     1.0448       Lister     1.0115     0.459     0.9818     1.0429       Mt     Uernon     0.9784     0.186     0.9497     1.0100       Northwick Park     0.9561     0.000     0.9340     0.9788       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9815       West Middlesex     0.9922     0.354     0.9759     1.0027       Defined by Billings     Barnet     1.0000     -     -     -       Hemei Hempstead     0.9925     0.751     0.9473     1.0397       Hemei Hempstead     0.9338     0.000     0.9473     1.0397       Hatter     1.0167     0.1943     0.9975     1.0167       Hemei Hempstead     0.9338     0.000     0.9014     0.9673       Hillingdon     0.9861     0.348     0.9578     1.0153       Lister     1.0167     0.194     0.9916     1.0426			Hemel Hempstead	0.9760	0.247	0.9366	1.0170
Lister     1.0115     0.459     0.9812     1.0429       Mt Vernon     0.9794     0.186     0.9947     1.0100       Northwick Park     0.9561     0.000     0.9340     0.9788       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9815       Watford     1.0071     0.399     0.9906     1.0240       Wast Middlesex     0.9225     0.354     0.9759     1.0087       Defined by Billings     Ealing     0.9359     0.041     0.8784     0.9975       Hemel Hempstead     0.9025     0.751     0.9473     1.0397       Hemel Hempstead     0.9861     0.348     0.9578     1.0153       Lister     1.0167     0.1944     0.9916     1.0426       Mt Vernon     1.014     0.386     0.9869     1.0344       Morthwick Park     0.9825     0.065     0.9847     1.0212       Mt Vernon     1.0104     0.386     0.9869     1.0344			Hillingdon	1.0146	0.387	0.9818	1.0486
Mt Vernon     0.9794     0.186     0.9497     1.0100       Northwick Park     0.9561     0.000     0.9340     0.9788       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9815       Watford     1.0071     0.399     0.9906     1.0240       West Middlesex     0.922     0.354     0.9759     1.0087       Defined by Billings     Barnet     1.000     -     -     -       Hemsi Hempstead     0.9359     0.041     0.8784     0.9972     0.9815       Lister     1.0167     0.9473     1.0359     1.0367     1.0359       Hillingdon     0.98359     0.041     0.8784     0.9972     0.9816     1.0426       Kt Vernon     1.0167     0.9473     1.0359     1.0147     0.9868     1.0144       Lister     1.0167     0.9842     1.0014     0.9864     1.0212       Mt Vernon     1.0104     0.9864     1.0212     0.9763			Lister	1.0115	0.459	0.9812	1.0429
Northwick Park     0.9561     0.000     0.9340     0.9788       Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9815       Watford     1.0071     0.399     0.9906     1.0240       West Middlesex     0.9922     0.354     0.9759     1.0087       23     Admissions for marker conditions     Barnet     1.0000     -     -       Edgware     0.9925     0.751     0.8473     0.9972       Edgware     0.9925     0.751     0.9473     1.0397       Hemel Hempstead     0.9338     0.000     0.9014     0.9673       Lister     1.0167     0.194     0.9673     1.0153       Lister     1.0167     0.9861     0.348     0.9578     1.0153       Northwick Park     0.9825     0.065     0.9642     1.0212       Mt Vernon     1.01028     0.763     0.9847     1.0212       Mary's     0.9829     0.000     0.9421     1.0212 </th <th></th> <th></th> <th>Mt Vernon</th> <th>0.9794</th> <th>0.186</th> <th>0.9497</th> <th>1.0100</th>			Mt Vernon	0.9794	0.186	0.9497	1.0100
Queen Elizabeth II     1.0137     0.218     0.9919     1.0359       St Mary's     0.9619     0.000     0.9427     0.9815       Watford     1.0071     0.399     0.9906     1.0240       West Middlesex     0.9922     0.354     0.9759     1.0087       23     Admissions for marker conditions     Barnet     1.0000     -     -       Defined by Billings     Ealing     0.9359     0.041     0.8784     0.9972       Edgware     0.9925     0.751     0.9473     1.0397       Hemei Hempstead     0.9255     0.751     0.9473     1.0397       Hamei Hempstead     0.9255     0.751     0.9473     1.0397       Hemei Hempstead     0.9861     0.348     0.9578     1.0153       Lister     1.0167     0.194     0.9616     1.0426       Morthwick Park     0.9825     0.065     0.9642     1.0011       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.98589     0.000     0.9421 </th <th></th> <th></th> <th>Northwick Park</th> <th>0.9561</th> <th>0.000</th> <th>0.9340</th> <th>0.9788</th>			Northwick Park	0.9561	0.000	0.9340	0.9788
St Mary's     0.9619     0.000     0.9427     0.9815       Watford     1.0071     0.399     0.9906     1.0240       West Middlesex     0.922     0.354     0.9759     1.0087       23     Admissions for marker conditions     Barnet     1.0000     -     -       Defined by Billings     Ealing     0.9359     0.041     0.8784     0.9972       Edgware     0.9925     0.751     0.9473     1.0397       Hemel Hempstead     0.9358     0.000     0.9014     0.9673       Hillingdon     0.9861     0.348     0.9578     1.0153       Lister     1.0107     0.1944     0.9916     1.0426       Mt Vernon     1.0104     0.386     0.9869     1.0344       Northwick Park     0.9825     0.065     0.9847     1.0212       St Mary's     0.9859     0.000     0.9421     0.9761       West Middlesex     0.9859     0.000     0.9421     0.9761       Usen Elizabeth II     1.0023     0.748     0.9864     1			Queen Elizabeth II	1.0137	0.218	0.9919	1.0359
Watford     1.0071     0.399     0.9906     1.0240       23     Admissions for marker conditions     Barnet     1.000     -     -       Defined by Billings     Barnet     1.000     -     -     -       Hemel Hempstead     0.9359     0.041     0.8784     0.9972       Edgware     0.9925     0.751     0.9473     1.0397       Hemel Hempstead     0.9388     0.000     0.9014     0.9673       Hillingdon     0.9861     0.348     0.9578     1.0153       Lister     1.0167     0.194     0.9916     1.0426       Mt Vernon     1.0167     0.194     0.9868     1.0041       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.9589     0.000     0.9421     0.9761       West Middlesex     0.9589     0.000     0.9421     0.9761       Mary's     0.9589     0.000     0.9421     0.9761       Mary's     0.9589     0.000     0.9421     0.9761			St Mary's	0.9619	0.000	0.9427	0.9815
West Middlesex     0.9922     0.354     0.9759     1.0087       23     Admissions for marker conditions Defined by Billings     Barnet     1.0000     -     -     -       Ealing     0.9359     0.041     0.8784     0.9972       Defined by Billings     Edgware     0.9925     0.751     0.9473     1.0397       Hemel Hempstead     0.9338     0.000     0.9014     0.9673       Lister     1.0167     0.194     0.9978     1.0153       Northwick Park     0.9861     0.348     0.9578     1.0153       Lister     1.0107     0.194     0.9916     1.0426       Mt Vernon     1.0107     0.194     0.9816     1.0347       Northwick Park     0.9825     0.065     0.9642     1.0011       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.9589     0.000     0.9421     0.9761       West Middlesex     0.9956     0.542     0.9816     1.0486			Watford	1.0071	0.399	0.9906	1.0240
23     Admissions for marker conditions Defined by Billings     Barnet Ealing     1.0000     -     -       Ealing     0.9359     0.041     0.8784     0.9972       Edingere     0.9925     0.751     0.9473     1.0397       Hemei Hempstead     0.9338     0.000     0.9014     0.9673       Hillingdon     0.9861     0.348     0.9578     1.0153       Lister     1.0167     0.194     0.9616     1.0426       Mt Vernon     1.0104     0.386     0.9869     1.0344       Northwick Park     0.9825     0.065     0.9642     1.0212       St Mary's     0.9859     0.000     0.9421     0.9761       West Middlesex     0.9956     0.542     0.9864     1.0164	_		West Middlesex	0.9922	0.354	0.9759	1.0087
Defined by Billings     Ealing Edgware     0.9359     0.041     0.8784     0.9972       Edgware     0.9925     0.751     0.9473     1.0397       Hemel Hempstead     0.9338     0.000     0.9014     0.9673       Hillingdon     0.9386     0.09014     0.9673     1.0153       Lister     1.0167     0.194     0.9916     1.0426       Mt Vernon     1.0104     0.386     0.9869     1.0344       Northwick Park     0.9825     0.065     0.9842     1.0011       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.9859     0.000     0.9421     0.9761       West Middlesex     0.9956     0.542     0.9864     1.0048	23	Admissions for marker conditions	Barnet	1.0000	•	-	•
Edgware   0.9925   0.751   0.9473   1.0397     Hemel Hempstead   0.9338   0.000   0.9014   0.9873     Hillingdon   0.9861   0.348   0.9578   1.0153     Lister   1.0167   0.194   0.9869   1.0426     Mt Vernon   1.0104   0.388   0.9869   1.0344     Northwick Park   0.9825   0.065   0.9642   1.0011     Queen Elizabeth II   1.0028   0.763   0.9847   1.0212     St Mary's   0.99589   0.000   0.9421   0.9761     West Middlesex   0.9958   0.542   0.9816   1.0048		Defined by Billings	Ealing	0.9359	0.041	0.8784	0.9972
Hemel Hempstead     0.9338     0.000     0.9014     0.9673       Hillingdon     0.9861     0.348     0.9578     1.0153       Lister     1.0167     0.194     0.9916     1.0426       Mt Vernon     1.0167     0.386     0.9809     1.0344       Northwick Park     0.9825     0.065     0.9642     1.0011       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.99589     0.000     0.9421     0.9761       Watford     1.0023     0.748     0.9864     1.0164       West Middlessx     0.9956     0.542     0.9916     1.0098			Edgware	0.9925	0.751	0.9473	1.0397
Hillingdon     0.9861     0.348     0.9578     1.0153       Lister     1.0167     0.194     0.9916     1.0426       Mt Vernon     1.0167     0.194     0.9869     1.0344       Northwick Park     0.9825     0.065     0.9642     1.0011       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.9559     0.000     0.9421     0.9761       West Middlessx     0.9956     0.542     0.9864     1.0164			Hemel Hempstead	0.9338	0.000	0.9014	0.9673
Lister 1.0167 0.194 0.9916 1.0426 Mt Vernon 1.0104 0.388 0.9869 1.0344 Northwick Park 0.9825 0.065 0.9642 1.0011 Queen Elizabeth II 1.0028 0.763 0.9647 1.0212 St Mary's 0.9559 0.000 0.9421 0.9761 Wast Middlessx 0.9956 0.542 0.9816 1.0098			Hillingdon	0.9861	0.348	0.9578	1.0153
Mt Vernon     1.0104     0.386     0.9869     1.0344       Northwick Park     0.9825     0.065     0.9642     1.0011       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.9589     0.000     0.9421     0.9761       Watford     1.0023     0.748     0.9844     1.0164       West Middlesex     0.9956     0.542     0.9816     1.0098			Lister	1.0167	0.194	0.9916	1.0426
Northwick Park     0.9825     0.065     0.9642     1.0011       Queen Elizabeth II     1.0028     0.763     0.9847     1.0212       St Mary's     0.9589     0.000     0.9421     0.9761       Watford     1.0023     0.748     0.9884     1.0164       West Middlesex     0.9956     0.542     0.9816     1.0098			Mt Vernon	1.0104	0.386	0.9869	1.0344
Queen Elizabeth II     1.0028     0.763     0.9647     1.0212       St Mary's     0.9589     0.000     0.9421     0.9761       Watford     1.0023     0.748     0.9884     1.0164       West Middlessx     0.9956     0.542     0.9816     1.0098			Northwick Park	0.9825	0.065	0.9642	1.0011
St Mary's     0.9589     0.000     0.9421     0.9761       Watford     1.0023     0.748     0.9884     1.0164       West Middlesex     0.9956     0.542     0.9816     1.0098			Queen Elizabeth II	1.0028	0.763	0.9847	1.0212
Watford 1.0023 0.748 0.9884 1.0164 West Middlesex 0.9956 0.542 0.9816 1.0098			St Mary's	0.9589	0.000	0.9421	0.9761
West Middlesex 0.9956 0.542 0.9816 1.0098			Watford	1.0023	0.748	0.9884	1.0164
			West Middlesex	0.9956	0.542	0.9816	1.0098

Group B EDs

.

